

**Exhibit B - DRAFT** 



Medicaid Care Management

**EXHIBIT B** 

SCOPE OF SERVICES



# **Exhibit B - DRAFT**

# **Table of Contents**

1	INTRO	ODUCTION	5
	1.1	Purpose	5
	1.2	Term	5
2	DEFI	NITIONS AND ACRONYMS	5
	2.1	Definitions	5
	2.2	Acronym List	30
3	GENE	ERAL TERMS AND CONDITIONS	35
	3.1	Program Management and Planning	35
	3.2	Agreement Elements	37
	3.3	Delegation of Authority	
	3.5	Time of the Essence	38
	3.6	CMS Approval of Agreement and Any Amendments	38
	3.7	Cooperation with Other Vendors and Prospective Vendors	39
	3.8	Renegotiation and Re-Procurement Rights	39
	3.9	Organization Requirements	40
	3.10	Subcontractors	52
	3.11	Staffing	60
4	PROC	GRAM REQUIREMENTS	71
	4.1	Covered Populations and Services	71
	4.2	Pharmacy Management	91
	4.3	Member Enrollment and Disenrollment	103
	4.4	Member Services	109
	4.5.	Member Grievances and Appeals	133
	4.6.	Provider Appeals	148
	4.7.	Access	152
	4.8.	Utilization Management	172
	4.9.	Member Education and Incentives	184
	4.10	Primary Care and Prevention Focused Care Model	188
	4.11	Care Coordination and Care Management	196
	4.12	Behavioral Health	211



# **Exhibit B - DRAFT**

4.14       Network Management       2         4.15       Alternative Payment Models       2         4.16       Provider Payments       2         4.17       Readiness Requirements Prior to Operations       3         4.18       Managed Care Information System       3         4.19       Claims Quality Assurance Standards       3         5       OVERSIGHT AND ACCOUNTABILITY       3         5.1       Reporting       3         5.2       Contract Oversight Program       3         5.3       Program Integrity       3         5.4       MCM Withhold and Incentive Program       3         5.5       Remedies       3         5.6       State Audit Rights       3         5.7       Dispute Resolution Process       3         6       FINANCIAL MANAGEMENT       3         6.1       Financial Standards       3         6.2       Capitation Payments       3         6.3       Medical Loss Ratio (MLR) Reporting and Settlement       3         6.5       Medical Cost Accruals       3         6.5       Medical Cost Accruals       3         6.7       Member Liability       3         6.8       Deni		4.13	Quality Management	255
4.16       Provider Payments       2         4.17       Readiness Requirements Prior to Operations       3         4.18       Managed Care Information System       3         4.19       Claims Quality Assurance Standards       3         5       OVERSIGHT AND ACCOUNTABILITY       3         5.1       Reporting       3         5.2       Contract Oversight Program       3         5.3       Program Integrity       3         5.4       MCM Withhold and Incentive Program       3         5.5       Remedies       3         5.6       State Audit Rights       3         5.7       Dispute Resolution Process       3         6       FINANCIAL MANAGEMENT       3         6.1       Financial Standards       3         6.2       Capitation Payments       3         6.3       Medical Loss Ratio (MLR) Reporting and Settlement       3         6.5       Medical Cost Accruals       3         6.5       Medical Cost Accruals       3         6.7       Member Liability       3         6.8       Denial of Payment       3         6.9       Federal Matching Funds       3         6.10       Third Part		4.14	Network Management	267
4.17       Readiness Requirements Prior to Operations       3         4.18       Managed Care Information System       3         4.19       Claims Quality Assurance Standards       3         5       OVERSIGHT AND ACCOUNTABILITY       3         5.1       Reporting       3         5.2       Contract Oversight Program       3         5.3       Program Integrity       3         5.4       MCM Withhold and Incentive Program       3         5.5       Remedies       3         5.6       State Audit Rights       3         5.7       Dispute Resolution Process       3         6       FINANCIAL MANAGEMENT       3         6.1       Financial Standards       3         6.2       Capitation Payments       3         6.3       Medical Loss Ratio (MLR) Reporting and Settlement       3         6.4       Financial Responsibility for Dual-Eligible Members       3         6.5       Medical Cost Accruals       3         6.6       Audits       3         6.7       Member Liability       3         6.8       Denial of Payment       3         6.9       Federal Matching Funds       3         6.10		4.15	Alternative Payment Models	285
4.18       Managed Care Information System       3         4.19       Claims Quality Assurance Standards       3         5       OVERSIGHT AND ACCOUNTABILITY       3         5.1       Reporting       3         5.2       Contract Oversight Program       3         5.3       Program Integrity       3         5.4       MCM Withhold and Incentive Program       3         5.5       Remedies       3         5.6       State Audit Rights       3         5.7       Dispute Resolution Process       3         6.1       Financial Standards       3         6.2       Capitation Payments       3         6.3       Medical Loss Ratio (MLR) Reporting and Settlement       3         6.4       Financial Responsibility for Dual-Eligible Members       3         6.5       Medical Cost Accruals       3         6.6       Audits       3         6.7       Member Liability       3         6.8       Denial of Payment       3         6.9       Federal Matching Funds       3         6.10       Third Party Liability       3         7       TERMINATION OF AGREEMENT       3		4.16	Provider Payments	297
4.19 Claims Quality Assurance Standards       3         5 OVERSIGHT AND ACCOUNTABILITY       3         5.1 Reporting       3         5.2 Contract Oversight Program       3         5.3 Program Integrity       3         5.4 MCM Withhold and Incentive Program       3         5.5 Remedies       3         5.6 State Audit Rights       3         5.7 Dispute Resolution Process       3         6 FINANCIAL MANAGEMENT       3         6.1 Financial Standards       3         6.2 Capitation Payments       3         6.3 Medical Loss Ratio (MLR) Reporting and Settlement       3         6.4 Financial Responsibility for Dual-Eligible Members       3         6.5 Medical Cost Accruals       3         6.6 Audits       3         6.7 Member Liability       3         6.8 Denial of Payment       3         6.9 Federal Matching Funds       3         6.10 Third Party Liability       3         7 TERMINATION OF AGREEMENT       3		4.17	Readiness Requirements Prior to Operations	305
5         OVERSIGHT AND ACCOUNTABILITY         3           5.1         Reporting         3           5.2         Contract Oversight Program         3           5.3         Program Integrity         3           5.4         MCM Withhold and Incentive Program         3           5.5         Remedies         3           5.6         State Audit Rights         3           5.7         Dispute Resolution Process         3           6         FINANCIAL MANAGEMENT         3           6.1         Financial Standards         3           6.2         Capitation Payments         3           6.3         Medical Loss Ratio (MLR) Reporting and Settlement         3           6.4         Financial Responsibility for Dual-Eligible Members         3           6.5         Medical Cost Accruals         3           6.6         Audits         3           6.7         Member Liability         3           6.8         Denial of Payment         3           6.9         Federal Matching Funds         3           6.10         Third Party Liability         3           7         TERMINATION OF AGREEMENT         3		4.18	Managed Care Information System	306
5.1       Reporting       3         5.2       Contract Oversight Program       3         5.3       Program Integrity       3         5.4       MCM Withhold and Incentive Program       3         5.5       Remedies       3         5.6       State Audit Rights       3         5.7       Dispute Resolution Process       3         6       FINANCIAL MANAGEMENT       3         6.1       Financial Standards       3         6.2       Capitation Payments       3         6.3       Medical Loss Ratio (MLR) Reporting and Settlement       3         6.4       Financial Responsibility for Dual-Eligible Members       3         6.5       Medical Cost Accruals       3         6.6       Audits       3         6.7       Member Liability       3         6.8       Denial of Payment       3         6.9       Federal Matching Funds       3         6.10       Third Party Liability       3         7       TERMINATION OF AGREEMENT       3		4.19	Claims Quality Assurance Standards	321
5.2       Contract Oversight Program       3         5.3       Program Integrity       3         5.4       MCM Withhold and Incentive Program       3         5.5       Remedies       3         5.6       State Audit Rights       3         5.7       Dispute Resolution Process       3         6       FINANCIAL MANAGEMENT       3         6.1       Financial Standards       3         6.2       Capitation Payments       3         6.3       Medical Loss Ratio (MLR) Reporting and Settlement       3         6.4       Financial Responsibility for Dual-Eligible Members       3         6.5       Medical Cost Accruals       3         6.6       Audits       3         6.7       Member Liability       3         6.8       Denial of Payment       3         6.9       Federal Matching Funds       3         6.10       Third Party Liability       3         7       TERMINATION OF AGREEMENT       3	5	OVER	SIGHT AND ACCOUNTABILITY	323
5.3       Program Integrity       3         5.4       MCM Withhold and Incentive Program       3         5.5       Remedies       3         5.6       State Audit Rights       3         5.7       Dispute Resolution Process       3         6       FINANCIAL MANAGEMENT       3         6.1       Financial Standards       3         6.2       Capitation Payments       3         6.3       Medical Loss Ratio (MLR) Reporting and Settlement       3         6.4       Financial Responsibility for Dual-Eligible Members       3         6.5       Medical Cost Accruals       3         6.6       Audits       3         6.7       Member Liability       3         6.8       Denial of Payment       3         6.9       Federal Matching Funds       3         6.10       Third Party Liability       3         7       TERMINATION OF AGREEMENT       3		5.1	Reporting	323
5.4       MCM Withhold and Incentive Program		5.2	Contract Oversight Program	335
5.5       Remedies       3         5.6       State Audit Rights       3         5.7       Dispute Resolution Process       3         6       FINANCIAL MANAGEMENT       3         6.1       Financial Standards       3         6.2       Capitation Payments       3         6.3       Medical Loss Ratio (MLR) Reporting and Settlement       3         6.4       Financial Responsibility for Dual-Eligible Members       3         6.5       Medical Cost Accruals       3         6.6       Audits       3         6.7       Member Liability       3         6.8       Denial of Payment       3         6.9       Federal Matching Funds       3         6.10       Third Party Liability       3         7       TERMINATION OF AGREEMENT       3		5.3	Program Integrity	.335
5.6       State Audit Rights       3         5.7       Dispute Resolution Process       3         6       FINANCIAL MANAGEMENT       3         6.1       Financial Standards       3         6.2       Capitation Payments       3         6.3       Medical Loss Ratio (MLR) Reporting and Settlement       3         6.4       Financial Responsibility for Dual-Eligible Members       3         6.5       Medical Cost Accruals       3         6.6       Audits       3         6.7       Member Liability       3         6.8       Denial of Payment       3         6.9       Federal Matching Funds       3         6.10       Third Party Liability       3         7       TERMINATION OF AGREEMENT       3		5.4	MCM Withhold and Incentive Program	352
5.7       Dispute Resolution Process       3         6       FINANCIAL MANAGEMENT       3         6.1       Financial Standards       3         6.2       Capitation Payments       3         6.3       Medical Loss Ratio (MLR) Reporting and Settlement       3         6.4       Financial Responsibility for Dual-Eligible Members       3         6.5       Medical Cost Accruals       3         6.6       Audits       3         6.7       Member Liability       3         6.8       Denial of Payment       3         6.9       Federal Matching Funds       3         6.10       Third Party Liability       3         7       TERMINATION OF AGREEMENT       3		5.5	Remedies	353
6       FINANCIAL MANAGEMENT       3         6.1       Financial Standards       3         6.2       Capitation Payments       3         6.3       Medical Loss Ratio (MLR) Reporting and Settlement       3         6.4       Financial Responsibility for Dual-Eligible Members       3         6.5       Medical Cost Accruals       3         6.6       Audits       3         6.7       Member Liability       3         6.8       Denial of Payment       3         6.9       Federal Matching Funds       3         6.10       Third Party Liability       3         7       TERMINATION OF AGREEMENT       3		5.6	State Audit Rights	362
6.1       Financial Standards       3         6.2       Capitation Payments       3         6.3       Medical Loss Ratio (MLR) Reporting and Settlement       3         6.4       Financial Responsibility for Dual-Eligible Members       3         6.5       Medical Cost Accruals       3         6.6       Audits       3         6.7       Member Liability       3         6.8       Denial of Payment       3         6.9       Federal Matching Funds       3         6.10       Third Party Liability       3         7       TERMINATION OF AGREEMENT       3		5.7	Dispute Resolution Process	364
6.2 Capitation Payments	6	FINAN	CIAL MANAGEMENT	366
6.3 Medical Loss Ratio (MLR) Reporting and Settlement		6.1	Financial Standards	366
6.4 Financial Responsibility for Dual-Eligible Members 3 6.5 Medical Cost Accruals 3 6.6 Audits 3 6.7 Member Liability 3 6.8 Denial of Payment 3 6.9 Federal Matching Funds 3 6.10 Third Party Liability 3 7 TERMINATION OF AGREEMENT 3		6.2	Capitation Payments	367
6.5       Medical Cost Accruals       3         6.6       Audits       3         6.7       Member Liability       3         6.8       Denial of Payment       3         6.9       Federal Matching Funds       3         6.10       Third Party Liability       3         7       TERMINATION OF AGREEMENT       3		6.3	Medical Loss Ratio (MLR) Reporting and Settlement	372
6.6       Audits       3         6.7       Member Liability       3         6.8       Denial of Payment       3         6.9       Federal Matching Funds       3         6.10       Third Party Liability       3         7       TERMINATION OF AGREEMENT       3		6.4	Financial Responsibility for Dual-Eligible Members	378
6.7 Member Liability		6.5	Medical Cost Accruals	.378
6.8 Denial of Payment		6.6	Audits	.378
6.9 Federal Matching Funds		6.7	Member Liability	.379
6.10 Third Party Liability		6.8	Denial of Payment	.380
7 TERMINATION OF AGREEMENT		6.9	Federal Matching Funds	.380
		6.10	Third Party Liability	.380
7.1 Termination for Cause 3	7	TERMI	INATION OF AGREEMENT	391
7.1 Terrilliation for Gade		7.1	Termination for Cause	391
7.2 Termination for Other Reasons3		7.2		392
7.3 Claims Responsibilities3		7.3	Claims Responsibilities	.393
·			•	393
5			G	393



# **Exhibit B - DRAFT**

7.6	State Owned Devices, Systems and Network Usage	394
7.7	Website And Social Media	396
7.8	Privacy Impact Assessment	397



**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

#### 1 INTRODUCTION

#### 1.1 Purpose

1.1.1. This Medicaid Care Management Agreement is a comprehensive full risk prepaid capitated Agreement that sets forth the terms and conditions for the Managed Care Organization's (MCO's) participation in the New Hampshire (NH) Medicaid Care Management (MCM) program.

## 1.2 **Term**

- 1.2.1. The Agreement and all contractual obligations, including Readiness Review, shall become effective on the date the Governor and Executive Council approves the executed MCM Agreement or, if the MCO does not have health maintenance organization (HMO) licensure in the State of New Hampshire by the New Hampshire Insurance Department on the date of Governor and Executive Council approval, the date the MCO obtains HMO licensure in the State of New Hampshire, whichever is later.
  - 1.2.1.1 If the MCO fails to obtain HMO licensure within thirty (30) calendar days of Governor and Executive Council approval, this Agreement shall become null and void without further recourse to the MCO.
- 1.2.2. The Program Start Date shall begin September 1, 2024, and the Agreement term shall continue through August 31, 2029.
- 1.2.3. The MCO's participation in the MCM program is contingent upon approval by the Governor and Executive Council, the MCO's successful completion of the Readiness Review process as determined by the Department, and obtaining HMO licensure in the State of New Hampshire as set forth above.
- 1.2.4. The MCO is solely responsible for the cost of all work during the Readiness Review and undertakes the work at its sole risk.
- 1.2.5. If at any time the Department determines that any MCO will not be ready to begin providing services on the MCM Program Start Date, at its sole discretion, the Department may withhold enrollment and require corrective action or terminate the Agreement without further recourse to the MCO.

#### 2 DEFINITIONS AND ACRONYMS

#### 2.1 **Definitions**

#### 2.1.1 **Abuse**

2.1.1.1 Practices that are inconsistent with sound fiscal, business, or medical practices that result in an unnecessary cost to the Medicaid program or in reimbursement for goods services that are not medically necessary or that fail to meet professionally recognized standards for care; or recipient practices that result in unnecessary cost to the Medicaid program.

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

# 2.1.2 Adults with Special Health Care Needs

- 2.1.2.1 Members who have or are at increased risk of having a chronic illness and/or a physical, developmental, behavioral, acquired brain disorder, or emotional condition and who also require health and related services of a type or amount beyond that usually expected for Members of similar age.
- 2.1.2.2 This includes, but is not limited to Members diagnosed with Human Immunodeficiency Virus (HIV)/Acquired Immune Deficiency Syndrome (AIDS), a Severe Mental Illness (SMI), Serious Emotional Disturbance (SED), Intellectual and/or Developmental Disability (I/DD), Substance Use Disorder diagnosis, or chronic pain.

#### 2.1.3 Advance Directive

2.1.3.1 As applicable, written instruction, such as a living will or durable power of attorney for health care, recognized under the laws of the State of New Hampshire, relating to the provision of health care when a Member is incapacitated. [42 CFR 489.100]

#### 2.1.4 Adverse Action

2.1.4.1 The denial or limited authorization of a requested service, including the type or level of service, pursuant to 42 CFR 438.400(b). The reduction, suspension, or termination of a previously authorized service. The denial, in whole or in part, of payment for a service. The failure to provide services in a timely manner, as defined by the State. The failure of the MCO to act on a grievance or an appeal within the time limits defined in this Agreement.

#### 2.1.5 Affordable Care Act

2.1.5.1 The Patient Protection and Affordable Care Act, P.L. 111-148, enacted on March 23, 2010 and the Health Care and Education Reconciliation Act of 2010, P.L. 111-152, enacted on March 30, 2010.

# 2.1.6 Agreement

2.1.6.1 This entire written Agreement between the Department and the MCO duly executed and legally binding.

### 2.1.7 Alternative Payment Model (APM)

2.1.7.1 A payment approach that gives added incentive payments to provide high-quality cost-efficient care.

## 2.1.8 Alternative Payment Model Implementation Plan

2.1.8.1 A MCO's plan for meeting the APM requirements described in this Agreement.

Page 6 of 397 Date
--------------------

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

## 2.1.9 American Society of Addiction Medicine (ASAM) Criteria

2.1.9.1 The National set of criteria for providing outcome-oriented and results-based care in the treatment of addiction. The Criteria provides guidelines for placement, continued stay and transfer/discharge of patients with addiction and co-occurring conditions.

## 2.1.10 Americans with Disabilities Act (ADA)

2.1.10.1 The civil rights law that prohibits discrimination against Members with disabilities in all areas of public life, including jobs, schools, transportation, and all public and private places that are open to the general public.

#### 2.1.11 Area Agency

2.1.11.1 An entity established as a nonprofit corporation in the State of New Hampshire which is established by rules adopted by the Commissioner to provide services to developmentally disabled persons in the area as defined in RSA 171-A:2.

#### 2.1.12 **ASAM Level of Care**

2.1.12.1 The standard nomenclature for describing the continuum of recovery-oriented addiction services. With the continuum, clinicians are able to conduct multidimensional assessments that explore individual risks and needs, and recommended ASAM Level of Care that matches intensity of treatment services to identified patient needs.

#### 2.1.13 Assertive Community Treatment (ACT)

2.1.13.1 The evidence-based practice of delivering comprehensive and effective services to Members with SMI by a multidisciplinary team primarily in Member homes, communities, and other natural environments.

#### 2.1.14 Automatic Assignment (or Auto-Assign)

2.1.14.1 The enrollment of an eligible Medicaid recipient, for whom enrollment is mandatory, in a MCO chosen by the Agency or its agent, and/or the assignment of a new enrollee to a PCP chosen by the MCO. In addition, Auto-Assignment may occur based on MCO performance as described in Section 4.3.4 (Auto-Assignment).

## 2.1.15 **Auxiliary Aids**

2.1.15.1 Services or devices that enable persons with impaired sensory, manual, or speaking skills to have an equal opportunity to participate in, and enjoy, the benefits of programs or activities conducted by the MCO.

Page 7 of 397	Date

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

2.1.15.2 Such aids include readers, Braille materials, audio recordings, telephone handset amplifiers, telephones compatible with hearing aids, telecommunication devices for deaf persons (TDDs), interpreters, note takers, written materials, and other similar services and devices.

#### 2.1.16 Behavioral Health Services

2.1.16.1 Mental health and Substance Use Disorder services that are Covered Services under this Agreement.

## 2.1.17 **Bright Futures**

2.1.17.1 A National health promotion and prevention initiative, led by the American Academy of Pediatrics (AAP) that provides theory-based and evidence-driven guidance for all preventive care screenings and well-child visits.

#### 2.1.18 Capitation Payment

- 2.1.18.1 The monthly payment by the Department to the MCO for each Member enrolled in the MCO's plan for which the MCO provides Covered Services under this Agreement.
- 2.1.18.2 Capitation payments are made only for Medicaid-eligible Members and retained by the MCO for those Members. The Department makes the payment regardless of whether the Member receives services during the period covered by the payment. [42 CFR 438.2]

## 2.1.19 Care Coordination

2.1.19.1 A process that assesses, plans, implements, coordinates, monitors, and evaluates the options and services required to meet a Member's physical, behavioral health and psychosocial needs using communication, closed-loop referral processes, and all available resources to promote quality cost-effective outcomes.

## 2.1.20 Care Management

2.1.20.1 Direct contact with a Member focused on the provision of various aspects of the Member's physical, behavioral health and needed supports that will enable the Member to achieve the best health outcomes.

## 2.1.21 Care Manager

2.1.21.1 A qualified and trained individual who is primarily responsible for providing Member supportive services as defined by this Agreement.

#### 2.1.22 **Care Plan**

2.1.22.1 A document prepared and updated by a Member's Provider and interdisciplinary Care Team with input from the Member which

Page 8 of 397	Date
---------------	------

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

summarizes the Member's health conditions, specific care needs, and current treatments. The Care Plan outlines what is needed to manage the Member's care needs and helps organize and prioritize care and treatment, including referrals relative to health-related social needs as defined in this Agreement.

#### 2.1.23 **Care Team**

2.1.23.1 Chosen and/or approved by the Member, or their parent(s) or guardian(s) if a minor, or their guardian(s) if an adult and applicable, whose composition best meets the unique care needs to be addressed and with whom the Member has already established relationships. The care team shall include the PCP.

### 2.1.24 Case Management

2.1.24.1 Service provided for supervising or coordinating care on behalf of Members, including gaining access to needed waivers and other Medicaid State Plan services, as well as monitoring the continuity of patient care services. Proper case management occurs across a continuum of care, addressing the ongoing individual needs of a Member rather than being restricted to a single practice setting.

### 2.1.25 Centers for Medicare & Medicaid Services (CMS)

2.1.25.1 The federal agency within the United States Department of Health and Human Services (HHS) with primary responsibility for the Medicaid and Medicare programs.

# 2.1.26 Certified Community Behavioral Health Clinic (CCBHC)

2.1.26.1 A state certified clinic that is responsible for providing all required CCBHC services in a manner that meets or exceeds CCBHC criteria. CCBHCs must either directly or through its Designated Collaborating Organizations (DCOs) provide, in a manner reflecting person-centered and family-centered care: crisis services; screening, assessment, and diagnosis; person-centered and family-centered treatment planning; outpatient mental health and substance use services; primary care screening and monitoring; targeted case management services; psychiatric rehabilitation services; peer supports and family/caregiver supports; and community care for uniformed service members and veterans.

#### 2.1.27 Children with Special Health Care Needs

2.1.27.1 Members under age twenty-one (21) who have or are at increased risk of having a serious or chronic physical, developmental, behavioral, or emotional condition and who also require health and related services of a type or amount beyond that usually expected for the child's age.

Page 9 of 397 Date \_\_\_\_\_

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

2.1.27.2 This includes, but is not limited to, children or infants: in foster care; requiring care in the Neonatal Intensive Care Units; with Neonatal Abstinence Syndrome (NAS); in high stress social environments/caregiver stress; receiving Family Centered Early Supports and Services, or participating in Special Medical Services or Partners in Health Services with a SED, I/DD or Substance Use Disorder diagnosis.

# 2.1.28 Children's Health Insurance Program (CHIP)

2.1.28.1 A program to provide health coverage to eligible children under Title XXI of the Social Security Act.

# 2.1.29 Choices for Independence (CFI)

- 2.1.29.1 Home and Community-Based Services (HCBS) 1915(c) waiver program that provides a system of Long Term Services and Supports (LTSS) to seniors and adults who are financially eligible for Medicaid and medically qualify for institutional level of care provided in nursing facilities.
- 2.1.29.2 The CFI waiver is also known as HCBS for the Elderly and Chronically III (HCBS-ECI). Long term care definitions are identified in RSA 151 E and He-E 801, and Covered Services are identified in He-E 801.

#### 2.1.30 Chronic Condition

2.1.30.1 A physical or mental impairment or ailment of indefinite duration or frequent recurrence such as heart disease, stroke, cancer, diabetes, obesity, arthritis, mental illness or a Substance Use Disorder.

## 2.1.31 Clean Claim

2.1.31.1 A claim that can be processed without obtaining additional information from the provider of the service or from a third party. It includes a claim with errors originating in a health plan's claims system. It does not include a claim from a provider who is under investigation for Fraud or Abuse, or a claim under review for medical necessity pursuant to 42 CFR 447.45(b).

## 2.1.32 Cold Call Marketing

2.1.32.1 Any unsolicited personal contact by the MCO or its designee, with a potential Member or a Member with another contracted MCO for the purposes of Marketing. [42 CFR 438.104(a)]

#### 2.1.33 Community Mental Health Services

2.1.33.1 The mental health services provided by a Community Mental Health Program ("CMH Program") or Community Mental Health Provider ("CMH Provider") to eligible Members as defined under He-M 426.

Page 10 of 397	Date	

New Hampshire Department of Health and Human Services Medicaid Care Management Services



#### **Exhibit B - DRAFT**

## 2.1.34 Community Mental Health Program ("CMH Program")

2.1.34.1 Synonymous with Community Mental Health Center, means a program established and administered by the State of New Hampshire, city, town, or county, or a nonprofit corporation for the purpose of providing mental health services to the residents of the area and which minimally provides emergency, medical or psychiatric screening and evaluation, Case Management, and psychotherapy services, [RSA 135-C:2, IV] A CMH Program is authorized to deliver the comprehensive array of services described in He-M 426 and is designated to cover a region as described in He-M 425.

## 2.1.35 Community Mental Health Provider ("CMH Provider")

2.1.35.1 The Medicaid Provider of Community Mental Health Services that has been previously approved by the DHHS Commissioner to provide specific mental health services pursuant to He-M 426 [He-M 426.02: (g)]. The distinction between a CMH Program and a CMH Provider is that a CMH Provider offers a more limited range of services.

## 2.1.36 Comprehensive Assessment

2.1.36.1 A person-centered assessment to help identify a Member's health condition, functional status, accessibility needs, strengths and supports, health care goals and other characteristics to inform whether a Member requires Care Management services and the level of services that should be provided.

#### 2.1.37 Comprehensive Medication Review (CMR)

- 2.1.37.1 A systematic process of collecting patient-specific information, assessing medication therapies to identify medication-related problems, developing a prioritized list of medication-related problems, and creating a plan to resolve them with the patient, caregiver and/or prescriber.
- 2.1.37.2 The related CMR counseling is an interactive person-to-person, telephonic, or telehealth consultation conducted in real-time between the patient and/or other qualified individual, such as a prescriber or caregiver, and the pharmacist or other qualified provider and is designed to improve a patient's knowledge of their prescriptions, over-the-counter medications, herbal therapies and dietary supplements; identify, and address problems or concerns the patient may have, and empower them to self-manage their medications and health conditions.

#### 2.1.38 Confidential Information and Confidential Data

2.1.38.1 The definition for this term is located in Exhibit K: DHHS Information Security Requirements.

Page 11 of 397	Date

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

## 2.1.39 Consumer Assessment of Health Care Providers and Systems (CAHPS®)

2.1.39.1 Family of standardized survey instruments, including a Medicaid survey, used to measure Member experience of health care.

# 2.1.40 Continuity of Care

2.1.40.1 Provision of continuous care for chronic or acute medical conditions through Member transitions between: facilities and home; facilities; Providers; service areas; managed care contractors; Marketplace, Medicaid fee-for-service (FFS) or private insurance and managed care arrangements. Continuity of Care occurs in a manner that prevents unplanned or unnecessary readmissions, ED visits, or adverse health outcomes.

#### 2.1.41 Continuous Quality Improvement (CQI)

2.1.41.1 Systematic process of identifying, describing, and analyzing strengths and weaknesses and then testing, implementing, learning from, and revising solutions.

## 2.1.42 Copayment

2.1.42.1 Monetary amount that a Member pays directly to a Provider at the time a Covered Service is rendered.

## 2.1.43 Corrective Action Plan (CAP)

2.1.43.1 Plan that the MCO completes and submits to the Department to identify and respond to any issues and/or errors in instances where it fails to comply with Department requirements.

## 2.1.44 Cost Sharing

2.1.44.1 A monetary amount that a Member pays directly to a Provider at the time a Covered Service is rendered.

#### 2.1.45 Covered Services

2.1.45.1 Health care services as defined by the Department and State and federal regulations and includes Medicaid State Plan services specified in this Agreement, including authorized In Lieu of Services and Value-Added Services and services required to meet Mental Health Parity and Addiction Equity Act.

#### 2.1.46 Cultural Competence

2.1.46.1 The level of knowledge-based skills required to provide effective clinical care to members of particular ethnic or racial groups.

#### 2.1.47 **Data**

2.1.47.1 Department records, files, forms, electronic information and other documents or information, in either electronic or paper form, that will be used /converted by the Vendor during the contract term, that

Page 12 of 397	Date
----------------	------

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

may be defined as "Confidential Data" within Exhibit K: DHHS Information Security Requirements.

#### 2.1.48 Data Breach

2.1.48.1 The definition for this term is located in Attachment 2 – Exhibit K: DHHS Information Security Requirements.

# 2.1.49 **Designated Receiving Facility (DRF)**

2.1.49.1 A hospital-based psychiatric unit or a non-hospital-based residential treatment program designated by the Commissioner to provide care, custody, and treatment to persons involuntarily admitted to the state mental health services system as defined in He-M 405. A DRF may also provide care for persons admitted to the facility voluntarily.

#### 2.1.50 Determinants of Health/Health-related Social Needs

2.1.50.1 A wide range of factors known to have an impact on healthcare, ranging from socioeconomic status, education and employment, to one's physical environment and access to healthcare.

#### 2.1.51 **Disenrollment**

2.1.51.1 The discontinuation of a Member's entitlement to receive Covered Services under the terms of this Agreement, and deletion from the approved list of members furnished by the Department.

#### 2.1.52 Data Certification

2.1.52.1 Encounter Data submitted to the Department, which must be certified by one of the following: MCO's CEO, CFO, or an individual who has delegated authority to sign for, and who reports directly to, the MCO's CEO or CFO [42 CFR 438.604; 42 CFR 438.606(a)].

#### 2.1.53 Quality Assessment and Performance Improvement (QAPI) Program

2.1.53.1 An ongoing and comprehensive program for the Covered Services the MCO furnishes to Members consistent with the requirements of this Agreement.

#### 2.1.54 **Dual-Eligible Members**

2.1.54.1 Members who are eligible for both Medicare and Medicaid.

#### 2.1.55 Emergency Medical Condition

2.1.55.1 A medical condition manifesting itself by acute symptoms of sufficient severity (including severe pain) that a prudent layperson, who possesses an average knowledge of health and medicine, could reasonably expect the absence of immediate medical attention to result in: placing the health of the Member (or, for a pregnant woman, the health of the woman or her unborn child) in

Page 13 of 397 Date \_\_\_\_\_

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

- serious jeopardy; serious impairment to bodily functions; or serious dysfunction of any bodily organ or part. [42 CFR 438.114(a)]
- 2.1.55.2 With respect to a pregnant woman, an emergency medical condition exists when:
  - 2.1.55.2.1 There is inadequate time to effect safe transfer to another Provider prior to delivery;
  - 2.1.55.2.2 Transfer may pose a threat to the health and safety of the patient or fetus; or
  - 2.1.55.2.3 There is evidence of onset of uterine contractions or rupture of the membranes.

## 2.1.56 **Emergency Services**

2.1.56.1 Covered inpatient and outpatient services that are furnished by a Provider that is qualified to furnish the services needed to evaluate or stabilize an Emergency Medical Condition. [42 CFR 438.114(a)]

#### 2.1.57 Encounter Data

2.1.57.1 A record of Covered Services provided to a MCO Member. An "encounter" is an interaction between a patient and a provider (MCO, rendering dentist, pharmacy, lab, etc.) who delivers services or is professionally responsible for services delivered to a patient. Encounter Data is considered to be Confidential Data as defined in Exhibit K: DHHS Information Security Requirements.

### 2.1.58 Enrollment

2.1.58.1 The process by which a person becomes a Member of the MCO's plan through the Department.

#### 2.1.59 Equal Access

2.1.59.1 All Members have the same access to all Providers and Covered Services.

#### 2.1.60 Evidence-Based Supported Employment (EBSE)

2.1.60.1 The provision of vocational supports to Members following the Supported Employment Implementation Resource Kit developed by Dartmouth Medical School to promote successful competitive employment in the community.

#### 2.1.61 Exclusion Lists

2.1.61.1 The HHS Office of the Inspector General's (OIG) List of Excluded Individuals/Entities; the System of Award Management; the Social Security Administration Death Master File; the list maintained by the Office of Foreign Assets Controls; and to the extent applicable, National Plan and Provider Enumeration System (NPPES).

Page 14 of 397	Date
----------------	------

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

## 2.1.62 External Quality Review (EQR)

2.1.62.1 The analysis and evaluation described in 42 CFR 438.350 by an External Quality Review Organization (EQRO) detailed in 42 CFR 438.364 of aggregated information on quality, timeliness, and access to Covered Services that the MCO or its Subcontractors furnish to Medicaid recipients.

## 2.1.63 **Facility**

2.1.63.1 Any premises (a) owned, leased, used, or operated directly or indirectly by or for the MCO or its affiliates for purposes related to this Agreement; or (b) maintained by a Subcontractor to provide Covered Services on behalf of the MCO.

#### 2.1.64 Family Planning Services

- 2.1.64.1 Services available to Members by Participating or Non-Participating Providers without the need for a referral or Prior Authorization that include: Consultation with trained personnel regarding family planning, contraceptive procedures, immunizations, and sexually transmitted diseases;
- 2.1.64.2 Distribution of literature relating to family planning, contraceptive procedures, and sexually transmitted diseases;
- 2.1.64.3 Provision of contraceptive procedures and contraceptive supplies by those qualified to do so under the laws of the State in which services are provided;
- 2.1.64.4 Referral of Members to physicians or health agencies for consultation, examination, tests, medical treatment and prescription for the purposes of family-planning, contraceptive procedures, and treatment of sexually transmitted diseases, as indicated; and
- 2.1.64.5 Immunization services where medically indicated and linked to sexually transmitted diseases, including but not limited to Hepatitis B and Human papillomaviruses vaccine.

#### 2.1.65 Federally Qualified Health Centers (FQHCs)

2.1.65.1 A public or private non-profit health care organization that has been identified by the Health Resources and Services Administration (HRSA) and certified by CMS as meeting criteria under Sections 1861(aa)(4) and 1905(l)(2)(B) of the Social Security Act.

#### 2.1.66 Fraud

2.1.66.1 An intentional deception or misrepresentation made by a person with the knowledge that the deception results in unauthorized benefit to that person or another person. The term includes any act that constitutes Fraud under applicable federal or State law.

#### 2.1.67 **Granite Advantage Members**

Page 15 of 397	Date
1 ago 10 01 001	

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

2.1.67.1 Members who are covered under the NH Granite Advantage waiver, which includes individuals in the Medicaid new adult eligibility group, covered under Title XIX of the Social Security Act who are adults, aged nineteen (19) up to and including sixty-four (64) years, with incomes up to and including one hundred and thirty-eight percent (138%) of the federal poverty level (FPL) who are not pregnant, not eligible for Medicare and not enrolled in NH's Health Insurance Premium Payment (HIPP) program.

#### 2.1.68 **Grievance Process**

2.1.68.1 The procedure for addressing Member grievances and which is in compliance with 42 CFR 438 Subpart F and this Agreement.

### 2.1.69 Home and Community Based Services (HCBS)

2.1.69.1 The waiver of Sections 1902(a)(10) and 1915(c) of the Social Security Act, which permits the federal Medicaid funding of LTSS in non-institutional settings for Members who reside in the community or in certain community alternative residential settings, as an alternative to long term institutional services in a nursing facility or Intermediate Care Facility (ICF). This includes services provided under the HCBS-CFI waiver program, Developmental Disabilities (HCBS-DD) waiver program, Acquired Brain Disorders (HCBS-ABD) waiver program, and In Home Supports (IHS) waiver program.

#### 2.1.70 Hospital-Acquired Conditions and Provider Preventable Conditions

2.1.70.1 A condition that meets the following criteria: Is identified in the Medicaid State Plan; has been found by NH Medicaid, based upon a review of medical literature by qualified professionals, to be reasonably preventable through the application of procedures supported by evidence-based guidelines; has a negative consequence for the Member; is auditable; and includes, at a minimum, wrong surgical or other invasive procedure performed on a Member, surgical or other invasive procedure performed on the wrong body part, or surgical or other invasive procedure performed on the wrong Member.

### 2.1.71 Implementation

2.1.71.1 The process for making the System fully operational for processing the Data.

#### 2.1.72 In Lieu of Services

2.1.72.1 An alternative medically appropriate and cost-effective substitute for a Covered Service or setting under the Medicaid State Plan. The utilization and actual cost of In Lieu Of Services shall be taken into account in developing the component of the capitation rates that represents the Medicaid State Plan Covered Services, unless a

Page 16 of 397	Date
----------------	------

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

statute or regulation explicitly requires otherwise. A Member cannot be required by the MCO to use the alternative service or setting.

## 2.1.73 Incomplete Claim

2.1.73.1 A claim that is denied for the purpose of obtaining additional information from the Provider.

# 2.1.74 Indian Health Care Provider (IHCP)

2.1.74.1 A health care program operated by the Indian Health Service (IHS) or by an Indian Tribe, Tribal Organization, or Urban Indian Organization (I/T/U) as those terms are defined in the Indian Health Care Improvement Act (25 U.S.C. 1603). [42 CFR 438.14(a)]

## 2.1.75 Integrated Care

2.1.75.1 The systematic coordination of mental health, Substance Use Disorder, and primary care services to effectively care for people with multiple health care needs.<sup>1</sup>

#### 2.1.76 **Licensed**

2.1.76.1 A facility, equipment, or an individual that has formally met State, county, and local requirements, and has been granted a license by a local, State, or federal government entity.

## 2.1.77 Limited English Proficiency (LEP)

2.1.77.1 Member's primary language is not English and the Member may have limited ability to read, write, speak or understand English.

## 2.1.78 List of Excluded Individuals and Entities (LEIE)

2.1.78.1 A database maintained by the Department of Health & Human Services, Office of the Inspector General. The LEIE provides information to the public, medical health care providers, patients, and others relating to parties excluded from participation in Medicare, Medicaid, and all other federal medical health care programs.

## 2.1.79 Long Term Services and Supports (LTSS)

2.1.79.1 Nursing facility services, all four of NH's Home and Community Based Care waivers, and services provided to children and families through the Division for Children, Youth and Families (DCYF).

## 2.1.80 Managed Care Information System (MCIS)

2.1.80.1 A comprehensive, automated and integrated system that: collects, analyzes, integrates, and reports data [42 CFR 438.242(a)]; provides information on areas, including but not limited to utilization, claims, grievances and appeals, and disenrollment for reasons

Page 17 of 397 Date \_\_\_\_\_

<sup>&</sup>lt;sup>1</sup> SAMHSA-HRSA Center for Integrated Solutions, "What is Integrated Care?"

New Hampshire Department of Health and Human Services Medicaid Care Management Services



## **Exhibit B - DRAFT**

other than loss of Medicaid eligibility [42 CFR 438.242(a)]; collects and maintains data on Members and Providers, as specified in this Agreement and on all services furnished to Members, through an encounter data system [42 CFR 438.242(b)(2)]; is capable of meeting the requirements listed throughout this Agreement; and is capable of providing all of the data and information necessary for the Department to meet State and federal Medicaid reporting and information regulations.

## 2.1.81 Managed Care Organization (MCO)

2.1.81.1 An entity that has a certificate of authority from the NH Insurance Department (NHID) and who contracts with the Department under a comprehensive risk Agreement to provide health care services to eligible Members under the MCM program.

## 2.1.82 Marketing

2.1.82.1 Any communication from the MCO to a potential Member, or Member who is not enrolled in that MCO, that can reasonably be interpreted as intended to influence the Member to enroll with the MCO or to either not enroll, or disenroll from another the Department contracted MCO. [42 CFR 438.104(a)]

## 2.1.83 Marketing Materials

2.1.83.1 Materials that are produced in any medium, by or on behalf of the MCO that can be reasonably interpreted as intended as Marketing to potential Members.

#### 2.1.84 MCO Formulary or Prescription Drug List (PDL)

2.1.84.1 List of prescription drugs covered by the MCO and the tier on which each medication is placed, in compliance with the Department-developed Preferred Drug List (PDL) and 42 CFR 438.10(i).

# 2.1.85 MCO Quality Assessment and Performance Improvement (QAPI) Program

2.1.85.1 An ongoing and comprehensive program for the services the MCO furnishes to Members consistent with the requirements of this Agreement and federal requirements for the QAPI program. [42 CFR 438.330(a)(1); 42 CFR 438.330(a)(3)]

## 2.1.86 MCO Utilization Management Program

2.1.86.1 "MCO Utilization Management Program" means a program developed, operated, and maintained by the MCO that meets the criteria contained in this Agreement related to Utilization Management. The MCO Utilization Management Program shall include defined structures, policies, and procedures for Utilization Management.

#### 2.1.87 Medicaid Director

Page 18 of 397	Date
. age .e c. cc.	24.0

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

2.1.87.1 The State Medicaid Director of NH DHHS.

## 2.1.88 Medicaid Management Information System (MMIS)

2.1.88.1 A system defined by the CMS.gov glossary as: a CMS approved system that supports the operation of the Medicaid program. The MMIS includes the following types of sub-systems or files: recipient eligibility, Medicaid provider, claims processing, pricing, Surveillance and Utilization Review Subsystem (SURS), Management and Administrative Reporting System (MARS), and potentially encounter processing.

#### 2.1.89 Medicaid State Plan

2.1.89.1 An agreement between a State and the Federal government describing how that State administers its Medicaid and CHIP programs. It gives an assurance that a State will abide by Federal rules and may claim Federal matching funds for its program activities. The State Plan establishes groups of individuals to be covered, services to be provided, methodologies for providers to be reimbursed and the administrative activities that are underway in the State.

#### 2.1.90 Medical Loss Ratio (MLR)

2.1.90.1 The proportion of premium revenues spent on clinical services and quality improvement, calculated in compliance with the terms of this Agreement and with all federal standards, including 42 CFR 438.8(b) for the application of the minimum federal loss ratio provision.

## 2.1.91 Medically Necessary

- 2.1.91.1 For Members twenty-one (21) years of age and older, services that a licensed Provider, exercising prudent clinical judgment, would provide, in accordance with generally accepted standards of medical practice, to a recipient for the purpose of evaluating, diagnosing, preventing, or treating an acute or chronic illness, injury, disease, or its symptoms, and that are:
- 2.1.91.2 Clinically appropriate in terms of type, frequency of use, extent, site, and duration, and consistent with the established diagnosis or treatment of the Member's illness, injury, disease, or its symptoms;
  - 2.1.91.2.1 Not primarily for the convenience of the Member or the Member's family, caregiver, or health care Provider:
  - 2.1.91.2.2 No more costly than other items or services which would produce equivalent diagnostic, therapeutic, or treatment results as related to the Member's illness, injury, disease, or its symptoms; and

Page 19 of 397	Date
----------------	------

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

2.1.91.2.3 Not experimental, investigative, cosmetic, or duplicative in nature [He-W 530.01(e)].

## 2.1.92 Medication Assisted Treatment (MAT)

2.1.92.1 The use of medications in combination with treatment planning, counseling and behavioral therapies or referral thereto for the treatment of Substance Use Disorder.

#### 2.1.93 **Member**

2.1.93.1 An individual who is enrolled in managed care through an MCO having an Agreement with the Department. [42 CFR 438.2]

## 2.1.94 Member Advisory Board

2.1.94.1 A group of Members that represents the Member population, established and facilitated by the MCO. The Member Advisory Board shall adhere to the requirements set forth in this Agreement.

#### 2.1.95 Member Appeal Process

2.1.95.1 The procedure for handling, processing, collecting and tracking Member requests for a review of an adverse benefit determination which is in compliance with 42 CFR 438 Subpart F and this Agreement.

## 2.1.96 Member Encounter Confidential Data (Encounter Data)

2.1.96.1 The information relating to the receipt of any item(s) or service(s) by a Member, under this Agreement, between the Department and an MCO that is subject to the requirements of 42 CFR 438.242 and 42 CFR 438.818.

#### 2.1.97 Member Handbook

2.1.97.1 A handbook based upon the Model Member Handbook developed by the Department and published by the MCO that enables the Member to understand how to effectively use the MCM program in accordance with this Agreement and 42 CFR 438.10(g).

## 2.1.98 National Committee for Quality Assurance (NCQA)

2.1.98.1 The organization responsible for developing and managing health care measures that assess the quality of care and services that managed care clients receive.

#### 2.1.99 NCQA Health Plan Accreditation

2.1.99.1 MCO accreditation, including the Medicaid module obtained from the NCQA, based on an assessment of clinical performance and consumer experience.

## 2.1.100 Neonatal Abstinence Syndrome (NAS)

Page 20 of 397	Date

New Hampshire Department of Health and Human Services Medicaid Care Management Services



## **Exhibit B - DRAFT**

2.1.100.1 A constellation of symptoms in newborn infants exposed to any of a variety of substances in utero, including opioids.

#### 2.1.101 Non-Covered Service

2.1.101.1 A service that is not a benefit under either the Medicaid State Plan or the MCO.

# 2.1.102 Non-Emergency Medical Transportation (NEMT)

2.1.102.1 Transportation services arranged by the MCO and provided free of charge to Members who are unable to pay for the cost of transportation to Provider offices and facilities for Medically Necessary care and services covered by the Medicaid State Plan, regardless of whether those Medically Necessary services are covered by the MCO.

## 2.1.103 Non-Participating Provider

2.1.103.1 A person, health care Provider, practitioner, facility or entity acting within their scope of practice or licensure, that does not have a written Agreement with the MCO to participate in the MCO's Provider network, but provides health care services to Members under appropriate scenarios (e.g., a referral approved by the MCO).

## 2.1.104 Non-Symptomatic Office Visits

2.1.104.1 Office visits available from the Member's Primary Care Provider (PCP) or another Provider within forty-five (45) calendar days of a request for the visit. Non-Symptomatic Office Visits may include, but are not limited to, well/preventive care such as physical examinations, annual gynecological examinations, or child and adult immunizations.

## 2.1.105 Non-Urgent, Symptomatic Office Visits

2.1.105.1 Routine care office visits available from the Member's PCP or another Provider within ten (10) calendar days of a request for the visit. Non-Urgent, Symptomatic Office Visits are associated with the presentation of medical signs or symptoms not requiring immediate attention, but that require monitoring.

## 2.1.106 Ongoing Special Condition

2.1.106.1 In the case of an acute illness, a condition that is serious enough to require medical care or treatment to avoid a reasonable possibility of death or permanent harm; in the case of a chronic illness or condition, a disease or condition that is life threatening, degenerative, or disabling, and requires medical care or treatment over a prolonged period of time; in the case of pregnancy, pregnancy from the start of the second trimester; in the case of a terminal illness, a Member has a medical prognosis that the Member's life expectancy is six (6) months or less.

Page 21 of 397 Date \_\_\_\_\_

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

## 2.1.107 Overpayments

2.1.107.1 Any amount received to which the Provider is not entitled. An overpayment includes payment that should not have been made and payments made in excess of the appropriate amount.

# 2.1.108 Participating Provider

2.1.108.1 A person, health care Provider, practitioner, facility, or entity, acting within the scope of practice and licensure, and who is under a written contract with the MCO to provide services to Members under the terms of this Agreement.

## 2.1.109 **Pay and Chase**

2.1.109.1 Recovery of claims paid in which the Standard Medicare, Medicare Advantage plan or private insurance was not known at the time the claim was adjudicated.

# 2.1.110 Peer Recovery Program

2.1.110.1 "Peer Recovery Program" means a program that is accredited by the Council on Accreditation of Peer Recovery Support Services (CAPRSS) or another accrediting body approved by the Department, is under contract with the Department's contracted facilitating organization, or is under contract with the Department's Bureau of Drug and Alcohol Services to provide Peer Recovery Support Services (PRSS).

## 2.1.111 Performance Improvement Project (PIP)

2.1.111.1 An initiative included in the QAPI program that focuses on clinical and non-clinical areas. A PIP shall be developed in consultation with the EQRO. [42 CFR 438.330(b)(1); 42 CFR 438.330(a)(2)].

## 2.1.112 Physician Group

2.1.112.1 A partnership, association, corporation, individual practice association, or other group that distributes income from the practice among its Members. An individual practice association is a Physician Group only if it is composed of individual physicians and has no Subcontracts with Physician Groups.

## 2.1.113 Physician Incentive Plan

2.1.113.1 Any compensation arrangement between the MCO and Providers that apply to federal regulations found at 42 CFR 422.208 and 42 CFR 422.210, as applicable to Medicaid managed care on the basis of 42 CFR 438.3(i).

## 2.1.114 Post-Stabilization Services

Page 22 of 397	Date
----------------	------

New Hampshire Department of Health and Human Services Medicaid Care Management Services



#### **Exhibit B - DRAFT**

2.1.114.1 Covered Services related to an Emergency Medical Condition that are provided after a Member is stabilized in order to maintain the stabilized condition or to improve or resolve the Member's condition. [42 CFR 438.114; 422.113]

#### 2.1.115 Practice Guidelines

2.1.115.1 Evidence-based clinical guidelines adopted by the MCO that are in compliance with 42 CFR 438.236 and with NCQA's requirements for health plan accreditation. The Practice Guidelines shall be based on valid and reasonable clinical evidence or a consensus of Providers in the particular field, shall consider the needs of Members, be adopted in consultation with Participating Providers, and be reviewed and updated periodically as appropriate.

## 2.1.116 Prescription Drug Monitoring Program (PDMP)

2.1.116.1 The program operated by the NH Office of Professional Licensure and Certification that facilitates the collection, analysis, and reporting of information on the prescribing, dispensing, and use of controlled substances in NH.

#### 2.1.117 Primary Care

2.1.117.1 All health services and laboratory services, including periodic examinations, preventive health care services and counseling, immunizations, diagnosis and treatment of illness or injury, coordination of overall medical care by the PCP, record maintenance, and initiation and coordination of closed loop referrals to specialty providers, including but not limited to Behavioral Health Service providers, and collaboration with such providers, for maintaining continuity of the Member's care and to collaboratively support achievement of the Member's whole-person health care goals.

# 2.1.118 Primary Care and Prevention Focused Model of Care

2.1.118.1 Model of Care as described in Section 4.10 of this Agreement.

#### 2.1.119 Primary Care Provider (PCP)

2.1.119.1 A Participating Provider who has the responsibility for supervising, coordinating, and providing primary health care to Members, initiating referrals for specialist care, and maintaining the Continuity of Member Care. PCPs include, but are not limited to Pediatricians, Family Practitioners, General Practitioners, Internists, Obstetricians/Gynecologists (OB/GYNs), Physician Assistants (under the supervision of a physician), or Advanced Registered Nurse Practitioners (ARNP), as designated by the MCO. The definition of PCP is inclusive of primary care physician as it is used in 42 CFR 438. All federal requirements applicable to primary care

Page 23 of 397 Date \_\_\_\_\_

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

physicians shall also be applicable to PCPs as the term is used in this Agreement.

#### 2.1.120 Prior Authorization

2.1.120.1 The process by which the Department, the MCO, or another MCO participating in the MCM program, whichever is applicable, authorizes, in advance, the delivery of Covered Services based on factors, including but not limited to medical necessity, cost-effectiveness, and compliance with this Agreement.

# 2.1.121 Program Start Date

2.1.121.1 The date when the MCO is responsible for coverage of Covered Services to its Members in the MCM program, contingent upon Agreement approval by the Governor and Executive Council and the Department's determination of successful completion of the Readiness Review period.

# 2.1.122 Post Payment Recovery

2.1.122.1 The process of seeking reimbursement from third parties whenever claims have been paid for which there is Third Party Liability (TPL). Also known as "Cost Recovery" or "pay and chase".

# 2.1.123 Provider Appeal Process

2.1.123.1 The procedure for handling, processing, collecting and tracking Provider appeal requests in accordance with Section 4.6 (Provider Appeals) of this Agreement.

## 2.1.124 Provider Directory

2.1.124.1 Information on the MCO's Participating Providers for each of the Provider types covered under this Agreement, available in electronic form and paper form upon request to the Member in accordance with 42 CFR 438.10 and the terms of this Agreement.

## 2.1.125 Psychiatric Boarding

2.1.125.1 The continued presence of a Member experiencing a mental health crisis in a hospital emergency room while waiting for admission in a designated receiving facility.

## 2.1.126 Qualified Bilingual/Multilingual Staff

2.1.126.1 An employee of the MCO who is designated by the MCO to provide oral language assistance as part of the individual's current, assigned job responsibilities and who has demonstrated to the MCO that they are proficient in speaking and understanding spoken English and at least one (1) other spoken language, including any necessary specialized vocabulary, terminology and phraseology; and is able to effectively, accurately, and impartially communicate directly with Members with LEP in their primary languages.

Page 24 of 397 Date \_\_\_\_\_

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

# 2.1.127 Qualified Interpreter for a Member with a Disability

- 2.1.127.1 An interpreter who, via a remote interpreting service or an on-site appearance, adheres to generally accepted interpreter ethics principles, including Member confidentiality; and is able to interpret effectively, accurately, and impartially, both receptively and expressively, using any necessary specialized vocabulary, terminology and phraseology.
- 2.1.127.2 Qualified interpreters can include, for example, sign language interpreters, oral transliterators (employees who represent or spell in the characters of another alphabet), and cued language transliterators (employees who represent or spell by using a small number of handshapes).

## 2.1.128 Qualified Interpreter for a Member with LEP

2.1.128.1 An interpreter who, via a remote interpreting service or an on-site appearance adheres to generally accepted interpreter ethics principles, including Member confidentiality; has demonstrated proficiency in speaking and understanding spoken English and at least one (1) other spoken language; and is able to interpret effectively, accurately, and impartially, both receptively and expressly, to and from such language(s) and English, using any necessary specialized vocabulary, terminology and phraseology.

#### 2.1.129 Qualified Translator

2.1.129.1 A translator who adheres to generally accepted translator ethics principles, including Member confidentiality; has demonstrated proficiency in writing and understanding written English and at least one (1) other written language; and is able to translate effectively, accurately, and impartially to and from such language(s) and English, using any necessary specialized vocabulary, terminology and phraseology. [45 CFR 92.4, 45 CFR 92.101]

# 2.1.130 Qualifying APM

2.1.130.1 An APM approved by the Department as consistent with the standards specified in this Agreement and in any subsequent Department guidance, including the Department Medicaid APM Strategy.

#### 2.1.131 **Quality**

2.1.131.1 The degree to which a MCO increases the likelihood of desired health outcomes of its Members through its structural and operational characteristics and through the provision of health services that are consistent with current professional knowledge.

#### 2.1.132 Quality Improvement (QI)

Page 25 of 397	Date

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

2.1.132.1 The process of monitoring that the delivery of oral, behavioral, and physical health care services are available, accessible, timely, and medically necessary. The MCO must have a quality improvement program (QI program) that includes standards of excellence. It also must have a written quality improvement plan (QI plan) that draws on its quality monitoring to improve health care outcomes for Members.

## 2.1.133 Readiness Review

2.1.133.1 The review process through which the MCO demonstrates, to the Department's satisfaction, the MCO's operational readiness and its ability to provide Covered Services to Members at the start of this Agreement in accordance with 42 CFR 438.66(d)(2), (d)(3), and (d)(4). [42 CFR 437.66(d)(1)(i) and the terms and conditions of this Agreement.

## 2.1.134 **Recovery**

2.1.134.1 A process of change through which Members improve their health and wellness, live self-directed lives, and strive to reach their full potential. Recovery is built on access to evidence-based clinical treatment and Recovery support services for all populations.<sup>2</sup>

## 2.1.135 Referral Provider

2.1.135.1 A Provider, who is not the Member's PCP, to whom a Member is referred for Covered Services.

# 2.1.136 Required Priority Population

2.1.136.1 The population mandated by the Department for MCO-Delivered Care Management services as described in this Agreement (Section 4.11.2). The MCO may provide Care Management services for other Members or populations at the plan's option.

## 2.1.137 Rural Health Clinic (RHC)

2.1.137.1 A clinic located in an area designated by the Department as rural, located in a federally designated medically underserved area, or has an insufficient number of physicians, which meets the requirements under 42 CFR 491.

# 2.1.138 Second Opinion

2.1.138.1 The opinion of a qualified health care professional within the Provider network, or the opinion of a Non-Participating Provider with whom the MCO has permitted the Member to consult, at no cost to the Member. [42 CFR 438.206(b)(3)]

## 2.1.139 Health-related Social Needs

Page 26 of 397 Date \_\_\_\_\_

<sup>&</sup>lt;sup>2</sup> SAMHSA, "Recovery and Recovery Support".

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

2.1.139.1 A wide range of factors known to have an impact on healthcare, ranging from socioeconomic status, education and employment, to one's physical environment and access to healthcare.

#### 2.1.140 **Software**

2.1.140.1 All Custom, Open Source, IaaS, SaaS and/or COTS Software and/or applications provided by the Contractor under the Agreement.

# 2.1.141 Specifications

2.1.141.1 Refer to Contract Exhibit P-37: General Provisions Section 12 – Assignment, Delegation, Subcontracts.

#### 2.1.142 State

2.1.142.1 The State of New Hampshire and any of its agencies.

#### 2.1.143 **State Data**

2.1.143.1 All Data created or in any way originating with the State, and all Data that is the output of computer processing of or other electronic manipulation of any Data that was created by or in any way originated with the State, whether such Data or output is stored on the State's hardware, the Contractor's hardware or exists in any system owned, maintained or otherwise controlled by the State or by the Contractor not defined as "Confidential Data" within Exhibit K: DHHS Information Security Requirements

# 2.1.144 Subcontract

2.1.144.1 Any separate contract or written arrangement between the Contractor and an individual or entity ("Subcontractor") to perform all or a portion of the duties and obligations that the Contractor is obligated to perform pursuant to this Agreement.

#### 2.1.145 Subcontractor

2.1.145.1 A person or entity that is delegated by the Contractor to perform an administrative function or service on behalf of the Contractor that directly or indirectly relates to the performance of all or a portion of the duties or obligations under this Agreement. A Subcontractor does not include a Participating Provider.

# 2.1.146 Substance Use Disorder

2.1.146.1 A cluster of symptoms meeting the criteria for Substance Use Disorder as set forth in the Diagnostic and Statistical Manual of Mental Disorders (DSM), 5th edition (2013), as described in He-W 513.02.

## 2.1.147 Substance Use Disorder Provider

Page 27 of 397	Date

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

2.1.147.1 All Substance Use Disorder treatment and Recovery support service Providers as described in He-W 513.04.

## 2.1.148 **System**

2.1.148.1 All Software, specified hardware, and interfaces and extensions, integrated and functioning together in accordance with the Specifications.

#### 2.1.149 Term

2.1.149.1 The duration of this Agreement.

# 2.1.150 Third Party Liability (TPL)

- 2.1.150.1 The legal obligation of third parties (e.g., certain individuals, entities, insurers, or programs) to pay part or all of the expenditures for medical assistance furnished under a Medicaid State Plan.
- 2.1.150.2 By law, all other available third party resources shall meet their legal obligation to pay claims before the Medicaid program pays for the care of an individual eligible for Medicaid.
- 2.1.150.3 States are required to take all reasonable measures to ascertain the legal liability of third parties to pay for care and services that are available under the Medicaid State Plan.

## 2.1.151 Transitional Care Management

- 2.1.151.1 The responsibility of the MCO to manage Covered Services care transitions for all Members moving from one clinical setting to another or from a clinical setting to home, to prevent unplanned or unnecessary ED visits or adverse health outcomes.
- 2.1.151.2 The MCO shall maintain and operate a formalized hospital and/or institutional discharge planning program that includes effective post-discharge Transitional Care Management, including appropriate discharge planning for short-term and long-term hospital and institutional stays. [42 CFR 438.208(b)(2)(i)]

#### 2.1.152 Transportation

2.1.152.1 An appropriate means of conveyance furnished to a Member to obtain Covered Services.

#### 2.1.153 Transitional Health Care

2.1.153.1 Care that is available from a primary or specialty Provider for clinical assessment and care planning within two (2) business days of discharge from inpatient or institutional care for physical or mental health disorders or discharge from a Substance Use Disorder treatment program.

#### 2.1.154 Transitional Home Care

Page 28 of 397	Date	

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

2.1.154.1 Care that is available with a home care nurse, a licensed counselor, and/or therapist (physical therapist or occupational therapist) within two (2) calendar days of discharge from inpatient or institutional care for physical or mental health disorders, if ordered by the Member's PCP or specialty care Provider or as part of the discharge plan.

#### 2.1.155 Trauma Informed Care

2.1.155.1 A program, organization, or system that realizes the widespread impact of trauma and understands potential paths for Recovery; recognizes the signs and symptoms of trauma in Members, families, staff, and others involved with the system; responds by fully integrating knowledge about trauma into policies, procedures, and practices; and seeks to actively resist re-traumatization

# 2.1.156 Urgent, Symptomatic Office Visits

2.1.156.1 Office visits available from the Member's PCP or another Provider within forty-eight (48) hours, for the presentation of medical signs or symptoms that require immediate attention, but are not life threatening and do not meet the definition of Emergency Medical Condition.

# 2.1.157 Utilization Management

2.1.157.1 The criteria of evaluating the necessity, appropriateness, and efficiency of Covered Services against established guidelines and procedures.

#### 2.1.158 Value-Added Services

2.1.158.1 Services not included in the Medicaid State Plan that the MCO elects to purchase and provide to Members at the MCO's discretion and expense to improve health and reduce costs. Value-Added Services are not included in capitation rate calculations.

## 2.1.159 Verification

2.1.159.1 Supports the confirmation of authority to enter a computer system application or network.

#### 2.1.160 Waste

2.1.160.1 The thoughtless or careless expenditure, mismanagement, or abuse of resources to the detriment (or potential detriment) of the U.S. government. Waste also includes incurring unnecessary costs resulting from inefficient or ineffective practices, systems, or controls.

## 2.1.161 Wellness Visit

2.1.161.1 A PCP visit that includes health risk and social determinant of health needs assessments, evaluation of the Member's physical and

Page 29 of 397	Date

New Hampshire Department of Health and Human Services Medicaid Care Management Services



## **Exhibit B - DRAFT**

behavioral health, including screening for depression, mood, suicidality, and Substance Use Disorder.

## 2.1.162 Willing Provider

2.1.162.1 A Provider credentialed as a qualified treatment provider according to the requirements of the Department and the MCO, who agrees to render Covered Services as authorized by the MCO and in compliance with terms of the MCO's Provider Agreement, including reimbursement rates and policy manual.

#### 2.1.163 Withhold

2.1.163.1 The actuarially sound amount retained as a percent of the MCO's risk adjusted total Capitation for a rating period which is withheld annually and may be available for distribution to the MCO in future contract years upon meeting specific performance criteria.

#### 2.1.164 Work Plan

2.1.164.1 Documentation that details the activities for the Project created in accordance with the Agreement. The plan and delineation of tasks, activities and events to be performed and Deliverables to be produced under the Project as specified in Appendix B: Business/Technical Requirements and Deliverables. The Work Plan must include a detailed description of the Schedule, tasks/activities, Deliverables, critical events, task dependencies, and the resources that would lead and/or participate on each task.

## 2.2 Acronym List

- 2.2.1 AAP means American Academy of Pediatrics.
- 2.2.2 ABD means Acquired Brain Disorder.
- 2.2.3 ACT means Assertive Community Treatment.
- 2.2.4 ADA means Americans with Disabilities Act.
- 2.2.5 ADL means Activities of Daily Living.
- 2.2.6 ADT means Admission, Discharge and Transfer.
- 2.2.7 AIDS means Acquired Immune Deficiency Syndrome.
- 2.2.8 ANSA means Adult Needs and Strengths Assessment.
- 2.2.9 APM means Alternative Payment Model.
- 2.2.10 ARNP means Advanced Registered Nurse Practitioner.
- 2.2.11 ASAM means American Society of Addiction Medicine.
- 2.2.12 ASC means Accredited Standards Committee.
- 2.2.13 ASFRA means Assisted Suicide Funding Restriction Act.
- 2.2.14 ASL means American Sign Language.

Page 30 of 397 Date \_\_\_\_\_

New Hampshire Department of Health and Human Services Medicaid Care Management Services



## **Exhibit B - DRAFT**

- 2.2.15 BCCP means Breast and Cervical Cancer Program.
- 2.2.16 CAHPS means Consumer Assessment of Healthcare Providers and Systems.
- 2.2.17 CANS means Child and Adolescent Needs and Strengths Assessment.
- 2.2.18 CAP means Corrective Action Plan.
- 2.2.19 CAPRSS means Council on Accreditation of Peer Recovery Support Services.
- 2.2.20 CARC means Claim Adjustment Reason Code.
- 2.2.21 CCBHC means Certified Community Behavioral Health Clinic
- 2.2.22 CEO means Chief Executive Officer.
- 2.2.23 CFI means Choices for Independence.
- 2.2.24 CFO means Chief Financial Officer.
- 2.2.25 CHIP means Children's Health Insurance Program.
- 2.2.26 CHIS means Comprehensive Health Care Information System.
- 2.2.27 CMH means Community Mental Health.
- 2.2.28 CMO means Chief Medical Officer.
- 2.2.29 CMR means Comprehensive Medication Review.
- 2.2.30 CMS means Centers for Medicare & Medicaid Services.
- 2.2.31 COB means Coordination of Benefits.
- 2.2.32 COBA means Coordination of Benefits Agreement.
- 2.2.33 CPT means Current Procedural Terminology.
- 2.2.34 CQI means Continuous Quality Improvement.
- 2.2.35 DBT means Dialectical Behavioral Therapy.
- 2.2.36 DCO means Designated Collaborating Organization.
- 2.2.37 DCYF means New Hampshire Division for Children, Youth and Families.
- 2.2.38 DD means Developmental Disability.
- 2.2.39 DEA means Drug Enforcement Administration.
- 2.2.40 DHHS means New Hampshire Department of Health and Human Services.
- 2.2.41 DME means Durable Medical Equipment.
- 2.2.42 DOB means Date of Birth.
- 2.2.43 DOD means Date of Death.
- 2.2.44 DOJ means (New Hampshire or United States) Department of Justice.
- 2.2.45 DRA means Deficit Reduction Act.
- 2.2.46 DSM means Diagnostic and Statistical Manual of Mental Disorders.
- 2.2.47 DSRIP means The New Hampshire Delivery System Reform Incentive Payment Program.

Page 31 of 397 Date \_\_\_\_\_

New Hampshire Department of Health and Human Services Medicaid Care Management Services



## **Exhibit B - DRAFT**

2.2.48 DUR means Drug Utilization Review. 2.2.49 EBSE means Evidence-Based Supported Employment. 2.2.50 ECI means Elderly and Chronically III. 2.2.51 ED means Emergency Department. 2.2.52 EDI means Electronic Data Interchange. 2.2.53 EFT means Electronic Funds Transfer. 2.2.54 EOB means Explanation of Benefits. 2.2.55 EPSDT means Early and Periodic Screening, Diagnostic and Treatment. 2.2.56 EQR means External Quality Review. 2.2.57 EQRO means External Quality Review Organization. 2.2.58 ERISA means Employees Retirement Income Security Act of 1974. 2.2.59 EST means Eastern Standard Time. 2.2.60 ETL means Extract, Transformation and Load. 2.2.61 FAR means Federal Acquisition Regulation. 2.2.62 FCA means False Claims Act. 2.2.63 FDA means Food and Drug Administration for the United States Department of Health and Human Services. 2.2.64 FFATA means Federal Funding Accountability & Transparency Act. 2.2.65 FFS means Fee-for-Service. 2.2.66 FPL means Federal Poverty Level. 2.2.67 FQHC means Federally Qualified Health Center. 2.2.68 HEDIS means Healthcare Effectiveness Data and Information Set. 2.2.69 HCBS means Home and Community Based Services. 2.2.70 HCBS-I means Home and Community Based Services In Home Supports. 2.2.71 HCPCS means Health Care Common Procedure Coding System. 2.2.72 HCQI means Health Care Quality Improvement. 2.2.73 HHS means United States Department of Health and Human Services. 2.2.74 HIPAA means Health Insurance Portability and Accountability Act. 2.2.75 HIPP means Health Insurance Premium Payment. 2.2.76 HITECH means Health Information Technology for Economic and Clinical Health Act of 2009.

2.2.77 HIV means Human Immunodeficiency Virus.2.2.78 HMO means Health Maintenance Organization.

Page 32 of 397 Date \_\_\_\_\_

2.2.107 NH means New Hampshire.

2.2.109 NPI means National Provider Identifier.



## **Exhibit B - DRAFT**

2.2.79	HRSA means Health Resources and Services Administration for the United States Department of Health and Human Services.
2.2.80	I/T/U means Indian Tribe, Tribal Organization, or Urban Indian Organization.
2.2.81	IADL means Instrumental Activities of Daily Living.
2.2.82	IBNR means Incurred But Not Reported.
2.2.83	ICF means Intermediate Care Facility.
2.2.84	ID means Intellectual Disabilities.
2.2.85	IEA means Involuntary Emergency Admission.
2.2.86	IHCP means Indian Health Care Provider.
2.2.87	IHS means Indian Health Service.
2.2.88	IMD means Institution for Mental Disease.
2.2.89	IVR means Interactive Voice Response.
2.2.90	LEIE means List of Excluded Individuals & Entities.
2.2.91	LEP means Limited English Proficiency.
2.2.92	LTSS means Long-Term Services and Supports.
2.2.93	MACRA means Medicare Access and CHIP Reauthorization Act of 2015.
2.2.94	MAT means Medication Assisted Treatment.
2.2.95	MCIS means Managed Care Information System.
2.2.96	MCM means Medicaid Care Management.
2.2.97	MCO means Managed Care Organization.
2.2.98	MED means Morphine Equivalent Dosing.
2.2.99	MFCU means Medicaid Fraud Control Unit, Office of Attorney General.
2.2.100	MLADCs means Masters Licensed Alcohol and Drug Counselors.
2.2.101	MLR means Medical Loss Ratio.
2.2.102	MMIS means Medicaid Management Information System.
2.2.103	NAS means Neonatal Abstinence Syndrome.
2.2.104	NCPDP means National Council for Prescription Drug Programs.
2.2.105	NCQA means National Committee for Quality Assurance.
2.2.106	NEMT means Non-Emergency Medical Transportation.
0 0 407	All I I I I

Page 33 of 397

2.2.111 OB/GYN means Obstetrics/Gynecology or Obstetricians/ Gynecologists.

2.2.110 NPPES means National Plan and Provider Enumeration System.

2.2.108 NHID means New Hampshire Insurance Department.

Date \_\_\_\_\_

New Hampshire Department of Health and Human Services Medicaid Care Management Services



## **Exhibit B - DRAFT**

- 2.2.112 OIG means Office of the Inspector General for the United States Department of Health and Human Services.
- 2.2.113 OTP means Opioid Treatment Program.
- 2.2.114 PBM means Pharmacy Benefits Manager.
- 2.2.115 PCA means Personal Care Attendant.
- 2.2.116 PCP means Primary Care Provider.
- 2.2.117 PDL means Preferred Drug List.
- 2.2.118 PDMP means Prescription Drug Monitoring Program.
- 2.2.119 PHI means Protected Health Information.
- 2.2.120 PI means Personal Information.
- 2.2.121 PIP means Performance Improvement Plan.
- 2.2.122 POS means Point of Service.
- 2.2.123 PRSS means Peer Recovery Support Services.
- 2.2.124 QAPI means Quality Assessment and Performance Improvement.
- 2.2.125 QI means Quality Improvement.
- 2.2.126 QM means Quality Management.
- 2.2.127 QOS means Quality of Service.
- 2.2.128 RARC means Reason and Remark Codes.
- 2.2.129 RFP means Request for Proposal.
- 2.2.130 RHC means Rural Health Clinic.
- 2.2.131 SAMHSA means Substance Abuse and Mental Health Services Administration for the United States Department of Health and Human Services.
- 2.2.132 SBIRT means Screening, Brief Intervention, and Referral to Treatment.
- 2.2.133 SED means Serious Emotional Disturbance.
- 2.2.134 SFY means State Fiscal Year.
- 2.2.135 SHIP means State's Health Insurance Assistance Program.
- 2.2.136 SIU means Special Investigations Unit.
- 2.2.137 SMART means Specific, Measurable, Attainable, Realistic, and Time Relevant.
- 2.2.138 SMDL means State Medicaid Director Letter.
- 2.2.139 SMI means Severe Mental Illness.
- 2.2.140 SNF means Skilled Nursing Facility.
- 2.2.141 SPMI means Severe or Persistent Mental Illness.
- 2.2.142 SSADMF means Social Security Administration Death Master File.
- 2.2.143 SSAE means Statement on Standards for Attestation Engagements.

Page 34 of 397 Date \_\_\_\_\_

New Hampshire Department of Health and Human Services Medicaid Care Management Services



## **Exhibit B - DRAFT**

- 2.2.144 SSI means Supplemental Security Income.
- 2.2.145 SSN means Social Security Number.
- 2.2.146 TAP means Technical Assistance Publication.
- 2.2.147 TDD means Telecommunication Device for Deaf Persons.
- 2.2.148 TPL means Third Party Liability.
- 2.2.149 TTY means Teletypewriter.
- 2.2.150 UAT means User Acceptance Testing.
- 2.2.151 Utilization Management means Utilization Management.
- 2.2.152 UDS means Urine Drug Screenings.
- 2.2.153 VA means United States Department of Veterans Affairs.

#### 3 GENERAL TERMS AND CONDITIONS

## 3.1 **Program Management and Planning**

#### 3.1.1 **General**

- 3.1.1.1 The MCO shall provide a comprehensive risk-based, capitated program for providing health care services to Members enrolled in the MCM program and who are enrolled in the MCO.
- 3.1.1.2 The MCO shall provide for all aspects of administrating and managing such program and shall meet all service and delivery timelines and milestones specified by this Agreement, applicable law or regulation incorporated directly or indirectly herein, or the MCM program.

#### 3.1.2 Representation and Warranties

- 3.1.2.1 The MCO represents and warrants that it shall fulfill all obligations under this Agreement and meet the specifications as described in the Agreement during the Term, including any subsequently negotiated, and mutually agreed upon, specifications.
- 3.1.2.2 The MCO acknowledges that, in being awarded this Agreement, the Department has relied upon all representations and warrants made by the MCO in its response to the Department's Request for Proposal (RFP) as referenced in Exhibit M: The MCM Proposal by Reference including any addenda, with respect to delivery of Medicaid managed care services and affirms all representations made therein.
- 3.1.2.3 The MCO represents and warrants that it shall comply with all of the material submitted to, and approved by the Department as part of its Readiness Review. Any material changes to such approved materials or newly developed materials require prior written approval by the Department before implementation.

Page 35 of 397 Date \_\_\_\_\_

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

- 3.1.2.4 The MCO shall not take advantage of any errors and/or omissions in the RFP or the resulting Agreement and amendments.
- 3.1.2.5 The MCO shall promptly notify the Department of any such errors and/or omissions that are discovered.
- 3.1.2.6 This Agreement shall be signed and dated by all parties, and is contingent upon approval by Governor and Executive Council.

# 3.1.3 **Program Management Plan**

- 3.1.3.1 The MCO shall develop and submit a Program Management Plan for the Department's review and approval.
- 3.1.3.2 The MCO shall provide the initial Program Management Plan to the Department for review and approval at the beginning of the Readiness Review period; in future years, any modifications to the Program Management Plan shall be presented for prior approval to the Department at least sixty (60) calendar days prior to the coverage year.
- 3.1.3.3 The Program Management Plan shall:
  - 3.1.3.3.1 Elaborate on the general concepts outlined in the MCO's Proposal and the section headings of the Agreement;
  - Describe how the MCO shall operate in NH by outlining management processes such as workflow, overall systems as detailed in the section headings of Agreement, evaluation of performance, and key operating premises for delivering efficiencies and satisfaction as they relate to Member and Provider experiences;
  - 3.1.3.3.3 Describe how the MCO shall ensure timely notification to the Department regarding:
    - 3.1.3.3.3.1. Expected or unexpected interruptions or changes that impact MCO policy, practice, operations, Members or Providers,
    - 3.1.3.3.3.2. Correspondence received from the Department on emergent issues and non-emergent issues; and
    - 3.1.3.3.3. Outline the MCO integrated organizational structure including NH-based resources and its support from its parent company, affiliates, or Subcontractors.

Page 36 of 397	Date

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

3.1.3.4 On an annual basis, the MCO shall submit to the Department either a certification of "no change" to the Program Management Plan or a revised Program Management Plan together with a redline that reflects the changes made to the Program Management Plan since the last submission.

# 3.1.4 **Key Personnel Contact List**

- 3.1.4.1 The MCO shall submit a Key Personnel Contact List to the Department that includes the positions and associated information indicated in Section 3.11.1. (Key Personnel) of this Agreement at least sixty (60) calendar days prior to the scheduled start date of the MCM program.
- 3.1.4.2 Thereafter, the MCO shall submit an updated Contact List immediately upon any Key Personnel staff changes.

## 3.2 Agreement Elements

- 3.2.1 The Agreement between the parties shall consist of the following:
  - 3.2.1.1 General Provisions, Form Number P-37
  - 3.2.1.2 Exhibit A: Revisions to Standard Agreement Provisions
  - 3.2.1.3 Exhibit B: Scope of Services
  - 3.2.1.4 Exhibit C: Payment Terms
  - 3.2.1.5 Exhibit D: Certification Regarding Drug Free Workplace Requirements
  - 3.2.1.6 Exhibit E: Certification Regarding Lobbying
  - 3.2.1.7 Exhibit F: Certification Regarding Debarment, Suspension, and Other Responsibility Matters
  - 3.2.1.8 Exhibit G: Certification of Compliance with Requirements Pertaining to Federal Nondiscrimination, Equal Treatment of Faith-Based Organizations and Whistleblower Protections
  - 3.2.1.9 Exhibit H: Certification Regarding Environmental Tobacco Smoke
  - 3.2.1.10 Exhibit I: Health Insurance Portability Act Business Associate Agreement
  - 3.2.1.11 Exhibit J: Certification Regarding Federal Funding Accountability & Transparency Act (FFATA) Compliance
  - 3.2.1.12 Exhibit K: DHHS Information Security Requirements
  - 3.2.1.13 Exhibit L: MCO Implementation Plan
  - 3.2.1.14 Exhibit M: MCO Proposal submitted in response to RFP, by reference
  - 3.2.1.15 Exhibit N: Liquidated Damages Matrix

Page 37 of 397	Date
. 490 0. 0. 00.	

New Hampshire Department of Health and Human Services Medicaid Care Management Services



## **Exhibit B - DRAFT**

- 3.2.1.16 Exhibit O: Quality and Oversight Reporting Requirements
- 3.2.1.17 Exhibit P: MCO Program Oversight Plan
- 3.2.1.18 Exhibit Q: DoIT Technical Requirements Workbook

#### 3.3 **Delegation of Authority**

3.3.1 Whenever, by any provision of this Agreement, any right, power, or duty is imposed or conferred on the Department, the right, power, or duty so imposed or conferred is possessed and exercised by the Commissioner unless any such right, power, or duty is specifically delegated to the duly appointed agents or employees of the Department and NHID.

#### 3.4 Authority of the New Hampshire Insurance Department

- 3.4.1 Pursuant to this Agreement and under the laws and rules of the State, the NHID shall have authority to regulate and oversee the licensing requirements of the MCO to operate as a health maintenance organization (HMO) in the State of New Hampshire.
- 3.4.2 The MCO is subject to all applicable laws and rules (and as subsequently amended) including but not limited to RSA 420-B; Managed Care Law and Rules RSA. 420-J; RSA 420-F and N.H. Administrative Rules Chapter Ins 2700; compliance with Bulletin INSNO. 12-015-AB, and further updates made by the New Hampshire Insurance Department (NHID); and the NH Comprehensive Health Care Information System (CHIS) Confidential Data reporting submission under NHID rules and/or bulletins.

#### 3.5 Time of the Essence

3.5.1 In consideration of the need to ensure uninterrupted and continuous services under the MCM program, time is of the essence in the performance of the MCO's obligations under the Agreement.

#### 3.6 CMS Approval of Agreement and Any Amendments

- 3.6.1 This Agreement and the implementation of amendments, modifications, and changes to this Agreement are subject to and contingent upon the approval of CMS.
- 3.6.2 This Agreement submission shall be considered complete for CMS's approval if:
  - 3.6.2.1 All pages, appendices, attachments, etc. were submitted to CMS; and
  - 3.6.2.2 Any documents incorporated by reference (including but not limited to State statute, regulation, or other binding document, such as a Member Handbook) to comply with federal regulations and the requirements of this review tool were submitted to CMS.

Page 38 of 397 Date \_\_\_\_\_

New Hampshire Department of Health and Human Services Medicaid Care Management Services



## **Exhibit B - DRAFT**

- 3.6.3 As part of this Agreement, the Department shall submit to CMS for review and approval the MCO rate certifications concurrent with the review and approval process for this Agreement. [42 CFR 438.7(a)]
- 3.6.4 The Department shall also submit to CMS for review and approval any Alternative Payment arrangements or other Provider payment arrangement initiatives based on the Department's description of the initiatives submitted and approved outside of the Agreement. [42 CFR 438.6(c)]

#### 3.7 Cooperation with Other Vendors and Prospective Vendors

- 3.7.1 This is not an exclusive Agreement and the Department may award simultaneous and/or supplemental contracts for work related to the Agreement, or any portion thereof. The MCO shall reasonably cooperate with such other vendors, and shall not knowingly or negligently commit or permit any act that may interfere with the performance of work by any other vendor, or act in any way that may place Members at risk.
- 3.7.2 The MCO is required to notify the Department within twelve (12) hours of a report by a Member, Member's relative, guardian or authorized representative of an allegation of a serious criminal offense against the Member by any employee of the MCO, its subcontractor or a Provider.
- 3.7.3 For the purpose of this Agreement, a serious criminal offense should be defined to include murder, arson, rape, sexual assault, assault, burglary, kidnapping, criminal trespass, or attempt thereof.
- 3.7.4 The MCO's notification shall be to a member of senior management of the Department such as the Commissioner, Deputy Commissioner, Associate Commissioner, Medicaid Director, or Deputy Medicaid Director.

#### 3.8 Renegotiation and Re-Procurement Rights

## 3.8.1 Renegotiation of Agreement

- 3.8.1.1 Notwithstanding anything in the Agreement to the contrary, the Department may at any time during the Term exercise the option to notify the MCO that the Department has elected to renegotiate certain terms of the Agreement.
- 3.8.1.2 Upon the MCO's receipt of any Department notice pursuant to this section to renegotiate this Agreement, the MCO and the Department shall undertake good faith negotiations of the subject terms of the Agreement, and may execute an amendment to the Agreement subject to approval by Governor and Executive Council.

#### 3.8.2 Re-Procurement of the Services or Procurement of Additional Services

3.8.2.1 Notwithstanding anything in the Agreement to the contrary, whether or not the Department has accepted or rejected MCO's services and/or deliverables provided during any period of the Agreement, the Department may at any time issue requests for proposals or

Page 39 of 397	Date
. age co c. co.	

New Hampshire Department of Health and Human Services Medicaid Care Management Services



## **Exhibit B - DRAFT**

- offers to other potential contractors for performance of any portion of the scope of work covered by the Agreement or scope of work similar or comparable to the scope of work performed by the MCO under the Agreement.
- 3.8.2.2 The Department shall give the MCO ninety (90) calendar days' notice of intent to replace another MCO participating in the MCM program or to add an additional MCO or other contractors to the MCM program.
- 3.8.2.3 If, upon procuring the services or deliverables or any portion of the services or deliverables from a Subcontractor in accordance with this section, the Department, in its sole discretion, elects to terminate this Agreement, the MCO shall have the rights and responsibilities set forth in Section 7 (Termination of Agreement) and Section 5.7 (Dispute Resolution Process).

#### 3.9 Organization Requirements

#### 3.9.1 **General Organization Requirements**

- 3.9.1.1 As a condition to entering into this Agreement, the MCO shall be licensed by the NHID to operate as an HMO in the State as required by RSA 420-B, and shall have all necessary registrations and licensures as required by the NHID and any relevant State and federal laws and regulations.
- 3.9.1.2 As a condition to entering into this Agreement, and during the entire Agreement Term, the MCO shall ensure that its articles of incorporation and bylaws do not prohibit it from operating as an HMO or performing any obligation required under this Agreement.
- 3.9.1.3 The MCO shall not be located outside of the United States. [42 CFR 438.602(i)] The MCO is prohibited from making payments or deposits for Medicaid-covered items or services to financial institutions located outside of the United States or its territories.
- 3.9.1.4 At the Department's discretion and at a Member effective date to be determined, the MCO shall initiate a Centers for Medicare and Medicaid Services defined application process to implement a highly integrated dual eligible special needs plan (HIDE SNP) or an alternate dual-eligible special needs plan (D-SNP) as defined at 42 CFR 422.2.

#### 3.9.2 Articles

3.9.2.1 The MCO shall provide, by the beginning of each Agreement year and at the time of any substantive changes, written assurance from MCO's legal counsel that the MCO is not prohibited by its articles of incorporation from performing the services required under this Agreement.

Page 40 of 397	Date
rage recor	

**Ownership and Control Disclosures** 

New Hampshire Department of Health and Human Services Medicaid Care Management Services

393



## **Exhibit B - DRAFT**

0.0.0		
	3.9.3.1	The MCO shall submit to the Department, in compliance with
		Exhibit K: Information Security Requirements, the name of any

persons or entities with an ownership or control interest in the MCO that:

3.9.3.1.1 Has direct, indirect, or combined direct/indirect ownership interest of five percent (5%) or more of the MCO's equity:

3.9.3.1.2 Owns five percent (5%) or more of any mortgage, deed of trust, note, or other obligation secured by the MCO if that interest equals at least five percent (5%) of the value of the MCO's assets: or

3.9.3.1.3 Is an officer or director of an MCO organized as a corporation or is a partner in an MCO organized as a partnership. [Section 1124(a)(2)(A) of the Social Security Act; section 1903(m)(2)(A)(viii) of the Social Security Act; 42 CFR 438.608(c)(2); 42 CFR 455.100 - 1041

3.9.3.2 The submission shall include for each person or entity, as applicable:

3.9.3.2.1 The address, including the primary business address, every business location, and P.O. Box address, for every entity;

3.9.3.2.2 The date of birth (DOB) and social security number (SSN) of any individual;

3.9.3.2.3 Tax identification number(s) of any corporation;

3.9.3.2.4 Information on whether an individual or entity with an ownership or control interest in the MCO is related to another person with ownership or control interest in the MCO as a spouse, parent, child, or sibling:

3.9.3.2.5 Information on whether a person or corporation with an ownership or control interest in any Subcontractor in which the MCO has a five percent (5%) or more interest is related to another person with ownership or control interest in the MCO as a spouse, parent, child, or sibling;

3.9.3.2.6 The name of any other disclosing entity, as such term is defined in 42 CFR 455.101, in which an owner of the MCO has an ownership or control interest;

	Page 41 of 397	Date
--	----------------	------

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

3.9.3.2.7	The name, address, DOB, and SSN of any managing
	employee of the MCO, as such term is defined by 42
	CFR 455.101; and

- 3.9.3.2.7.1. Certification by the MCO's CEO that the information provided in this Section 3.9.3 (Ownership and Control Disclosures) to the Department is accurate to the best of his or her information, knowledge, and belief.
- 3.9.3.2.8 The MCO shall disclose the information set forth in this Section 3.9.3 (Ownership and Control Disclosures) on individuals or entities with an ownership or control interest in the MCO to the Department at the following times:
  - 3.9.3.2.8.1. At the time of Agreement execution;
  - 3.9.3.2.8.2. When the Provider or disclosing entity submits a Provider application;
  - 3.9.3.2.8.3. When the Provider or disclosing entity executes a Provider agreement with the Department;
  - 3.9.3.2.8.4. Upon request of the Department during the revalidation of the Provider enrollment; and
  - 3.9.3.2.8.5. Within thirty-five (35) calendar days after any change in ownership of the disclosing entity. [Section 1124(a)(2)(A) of the Social Security Act; section 1903(m)(2)(A)(viii) of the Social Security Act; 42 CFR 438.608(c)(2); 42 CFR 455.100 103; 42 CFR 455.104(c)(1) and (4)]
- 3.9.3.3 The Department shall review the ownership and control disclosures submitted by the MCO and any Subcontractors. [42 CFR 438.602(c); 42 CFR 438.608(c)]
- 3.9.3.4 The MCO shall be fined in accordance with Exhibit N: Liquidated Damages Matrix for any failure to comply with ownership disclosure requirements detailed in this Section.

#### 3.9.4 Change in Ownership or Proposed Transaction

3.9.4.1 The MCO shall inform the Department and the NHID of its intent to merge with or be acquired, in whole or in part, by another entity or another MCO or of any change in control within seven (7) calendar

Page 42 of 397 Da	ie
-------------------	----

New Hampshire Department of Health and Human Services Medicaid Care Management Services



#### **Exhibit B - DRAFT**

days of a management employee learning of such intent. The MCO shall receive prior written approval from the Department and the NHID prior to taking such action.

## 3.9.5 **Prohibited Relationships**

- 3.9.5.1 Pursuant to Section 1932(d)(1)(A) of the Social Security Act (42 USC 1396u-2(d)(1)(A)), the MCO shall not knowingly have a director, officer, partner, or person with beneficial ownership of more than five percent (5%) of the MCO's equity who has been, or is affiliated with another person who has been debarred or suspended from participating in procurement activities under the Federal Acquisition Regulation (FAR) or from participating in non-procurement activities under regulations issued pursuant to Executive Order No. 12549 or under guidelines implementing such order. [Section 1932(d)(1) of the Social Security Act; 42 CFR 438.610(a)(1)-(2); 42 CFR 438.610(c)(2); Exec. Order No. 12549]
- 3.9.5.2 The MCO shall not have an individual:

3.9.5.2.3

- 3.9.5.2.1 With a direct or indirect ownership or control interest of 5 percent (5%) or more in the entity or with an ownership or control interest, as defined in Section 1124(a)(3) of the Social Security Act, in that entity; or
- 3.9.5.2.2 Who is an officer, director, agent, or managing employee as defined in section 1126(b) of the Social Security Act. The term "agent" shall include non-officer, non-director, non-managing employees as defined in section 1126(b) and Subcontractors for the purposes of this section to the extent required by CMS or other federal authority: or
  - Who no longer has a direct or indirect ownership or control interest of 5 percent (5%) or more in the entity or with an ownership or control interest in that entity as defined in section 1124(a)(3) of the Social Security Act due to a transfer of such ownership or control to an immediate family member or member of the household as defined in 1128(j) of the Social Security Act who continues to maintain a direct or indirect ownership or control interest of 5% or more in the entity; and
- 3.9.5.2.4 Has been convicted of any offense in Sections 1128(a) or 1128(b)(1)-(3) of the Social Security Act, to the extent required by CMS or other federal authority; or

Page 43 of 397 Date \_\_\_\_\_

Medicaid Care Management Services Contract
New Hampshire Department of Health and Human Services
Medicaid Care Management Services



# **Exhibit B - DRAFT**

	3.9.5.2.5	Has been excluded from participation under a program under title XVIII or under a State health care program; or
	3.9.5.2.6	Has been assessed a civil monetary penalty under Section 1128A or 1129 of the Social Security Act.
3.9.5.3		nall retain any data, information, and documentation above described relationships for a period of no less years.
3.9.5.4	written discled provide written to the Departing individual or excluded from or from particular issued under implementing under 42 CF Act; 42 CFF 438.610(b);	color calendar days of discovery, the MCO shall provide obsure to the Department, and Subcontractors shall en disclosure to the MCO, which shall provide the same timent, of any individual or entity (or affiliation of the entity) who/that is debarred, suspended, or otherwise in participating in procurement activities under the FAR dipating in non-procurement activities under regulations or Executive Order No. 12549 or under guidelines of Executive Order No. 12549, or prohibited affiliation of R 438.610. [Section 1932(d)(1) of the Social Security R 438.608(c)(1); 42 CFR 438.610(a)(1-2); 42 CFR 42 CFR 438.610(c)(1-4); SMDL 6/12/08; SMDL c. Order No. 12549]
3.9.5.5	with an individual; (ii	ment learns that the MCO has a prohibited relationship vidual or entity that (i) is debarred, suspended, or coluded from participating in procurement activities AR or from participating in non-procurement activities tions issued under Executive Order No. 12549 or under aplementing Executive Order No. 12549, or if the MCO whip with an individual who is an affiliate of such an individual who is an affiliate
	3.9.5.5.1	Terminate the existing Agreement with the MCO;
	3.9.5.5.2	Continue an existing Agreement with the MCO unless the HHS Secretary directs otherwise;
	3.9.5.5.3	Not renew or extend the existing Agreement with the MCO unless the HHS Secretary provides to the State and to Congress a written statement describing compelling reasons that exist for renewing or extending the Agreement despite the prohibited affiliation. [42 CFR 438.610(d)(2)-(3); 42 CFR 438.610(a); 42 CFR 438.610(b); Exec. Order No.

12549]

Date \_\_\_\_\_ Page 44 of 397

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

## 3.9.6 **Background Checks**

- 3.9.6.1 The MCO shall perform criminal history record checks on its owners, directors, and managing employees, as such terms are defined in 42 CFR 455.101 and clarified in applicable subregulatory guidance such as the Medicaid Provider Enrollment Compendium.
- 3.9.6.2 The MCO or its Subcontractors shall conduct background checks upon hire and monthly exclusion checks on all employees (or contractors and their employees) to ensure that the MCO and Subcontractors do not employ or contract with any individual or entity, in accordance with Prohibited Relationship provisions in Section 3.9.5 of this Agreement, on an Exclusion List who are:
  - 3.9.6.2.1 Convicted of crimes described in Section 1128(b)(8) of the Social Security Act;
  - 3.9.6.2.2 Debarred, suspended, or excluded from participating in procurement activities under the FAR or from participating in non-procurement activities under regulation issued under Executive Order No. 12549 or under guidelines implementing Executive Order No. 12549; and/or
  - 3.9.6.2.3 Is excluded from participation in any federal health care program under Section 1128 or 1128A of the Social Security Act. [[42 CFR 438.808(a); 42 CFR 438.808(b)(1); 42 CFR 431.55(h); section 1903(i)(2) of the Social Security Act; 42 CFR 1001.1901(c); 42 CFR 1002.3(b); SMDL 6/12/08; SMDL 1/16/09; 76 Fed. Reg. 5862, 5897 (February 2, 2011)]
- 3.9.6.3 In addition, the MCO or its Subcontractor shall conduct screenings upon hire and monthly of its employees (except its directors and officers), and contractors and MCO Subcontractors' contractor employees (except its directors and officers) to ensure that none of them appear on:
  - 3.9.6.3.1 HHS-OIG's List of Excluded Individuals/Entities;
  - 3.9.6.3.2 The System of Award Management;
  - 3.9.6.3.3 The list maintained by the Office of Foreign Assets Control; and
  - 3.9.6.3.4 To the extent applicable, NPPES (collectively, these lists are referred to as the "Exclusion Lists").
- 3.9.6.4 The MCO shall certify to the Department annually that it or its Subcontractors performs screenings upon hire and monthly thereafter against the Exclusion Lists and that neither the MCO nor

Page 45 of 397 Date	
---------------------	--

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

its Subcontractors,	including	contractor	employees	of	MCO
Subcontractors, have	e any emplo	oyees, direc	tly or indirect	ly, w	/ith:

- 3.9.6.4.1 Any individual or entity excluded from participation in the federal health care program;
- 3.9.6.4.2 Any entity for the provision of such health care, utilization review, medical social work, or administrative services through an excluded individual or entity or who could be excluded under Section 1128(b)(8) of the Social Security Act as being controlled by a sanctioned individual;
- 3.9.6.4.3 Any individual or entity excluded from Medicare, Medicaid or NH participation by the Department per the Department system of record;
- 3.9.6.4.4 Any entity that has a contractual relationship (direct or indirect) with an individual convicted of certain crimes as described in Section 1128(b)(8) of the Social Security Act; and/or
- 3.9.6.4.5 Any individual entity appearing on any of the Exclusion Lists.
- 3.9.6.5 In the event that the MCO or its Subcontractor identifies that it has employed or contracted with a person or entity which would make the MCO unable to certify as required under this Section 3.9.6 (Background Checks) or Section 3.9.3 (Ownership and Control Disclosures) above, then the MCO should notify the Department in writing and shall begin termination proceedings within forty-eight (48) hours unless the individual is part of a federally-approved waiver program.
- 3.9.6.6 The MCO shall maintain documentation to ensure screenings have been completed by Subcontractors and reviewed by the MCO monthly.

#### 3.9.7 Conflict of Interest

- 3.9.7.1 The MCO shall ensure that safeguards, at a minimum equal to federal safeguards (41 USC 423), are in place to guard against conflict of interest. [Section 1932(d)(3) of the Social Security Act; SMDL 12/30/97]. The MCO shall report transactions between the MCO and parties in interest to the Department and any other agency as required, and make it available to MCO Members upon reasonable request. [Section 1903(m)(4)(B) of the Social Security Act]
- 3.9.7.2 The MCO shall report to the Department and, upon request, to the HHS Secretary, the HHS Inspector General, and the Comptroller General a description of transactions between the MCO and a party

Page 46 of 397	Date

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

in interest (as defined in Section 1318(b) of the Social Security Act), including the following transactions:

- 3.9.7.2.1 Any sale or exchange, or leasing of any property between the MCO and such a party;
- 3.9.7.2.2 Any furnishing for consideration of goods, services (including management services), or facilities between the MCO and such a party, but not including salaries paid to employees for services provided in the normal course of their employment; and
- 3.9.7.2.3 Any lending of money or other extension of credit between the MCO and such a party. [Section 1903(m)(4)(A) of the Social Security Act; Section 1318(b) of the Social Security Act]

## 3.9.8 Compliance with State and Federal Laws

#### 3.9.8.1 General Requirements

- 3.9.8.1.1 The MCO, its Subcontractors, and Participating Providers, shall adhere to all applicable State and federal laws and applicable regulations and subregulatory guidance which provides further interpretation of law, including subsequent revisions whether or not listed in this Section 3.9.8 (Compliance with State and Federal Laws), and any laws, regulations or administrative rules effective after the execution of this Agreement.
- 3.9.8.1.2 The MCO shall comply with any applicable federal and State laws that pertain to Member rights and ensure that its employees and Participating Providers observe and protect those rights. [42 CFR 438.100(a)(2)]
- 3.9.8.1.3 The MCO shall comply, at a minimum, with the following:
  - 3.9.8.1.3.1. Medicare: Title XVIII of the Social Security Act, as amended; 42 U.S.C.A. Section 1395 et seq.; Related rules: Title 42 Chapter IV of the Code of Federal Regulations;
  - 3.9.8.1.3.2. Medicaid: Title XIX of the Social Security Act, as amended; 42 U.S.C.A. Section 1396 et seq. (specific to managed care: Section 1902(a)(4), 1903(m), 1905(t), and 1932 of the SSA); Related rules: Title 42 Chapter IV of the Code of

Page 47 of 397	Date	
----------------	------	--

Medicaid Care Management Services Contract
New Hampshire Department of Health and Human Services
Medicaid Care Management Services



# **Exhibit B - DRAFT**

		Federal Regulations (specific to managed care: 42 CFR Section 438; see also 431 and 435);
	3.9.8.1.3.3.	CHIP: Title XXI of the Social Security Act, as amended; 42 U.S.C. 1397aa; Regulations promulgated thereunder: 42 CFR 457;
	3.9.8.1.3.4.	Regulations related to the operation of a waiver program under Section 1915c of the Social Security Act, including: 42 CFR 430.25, 431.10, 431.200, 435.217, 435.726, 435.735, 440.180, 441.300-310, and 447.50-57;
	3.9.8.1.3.5.	State administrative rules and laws pertaining to transfers and discharges, such as RSA 151:26;
	3.9.8.1.3.6.	State administrative rules and laws pertaining to confidentiality;
	3.9.8.1.3.7.	American Recovery and Reinvestment Act;
	3.9.8.1.3.8.	Title VI of the Civil Rights Act of 1964;
	3.9.8.1.3.9.	The Age Discrimination Act of 1975;
	3.9.8.1.3.10.	The Rehabilitation Act of 1973;
	3.9.8.1.3.11.	Title IX of the Education Amendments of 1972 (regarding education programs and activities);
	3.9.8.1.3.12.	The ADA;
	3.9.8.1.3.13.	42 CFR Part 2; and
	3.9.8.1.3.14.	Section 1557 of the Affordable Care Act. [42 CFR438.3(f)(1); 42 CFR 438.100(d)]
3.9.8.1.4	Agreement changes, vectors that of incorpor	shall provide, by the beginning of each year and at the time of any substantive written assurance from MCO's legal to the MCO is not prohibited by its articles ration from performing the services der this Agreement.
3.9.8.1.5	Department	shall comply with all aspects of the Sentinel Event Reporting and Review 003, and any subsequent versions and/or s;

Date \_\_\_\_ Page 48 of 397

3.9.8.1.6

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

prevail.

	sentinel event, and provide any additional reporting information requested by the Department, and participate in a Department sentinel event review, if requested;
3.9.8.1.7	The MCO shall report to the Department within twenty-four (24) hours any time a sentinel event occurs with one of its Members. This does not replace the MCO's responsibility to notify the appropriate authority if the MCO suspects a crime has occurred;
3.9.8.1.8	The MCO shall comply with all statutorily mandated reporting requirements, including but not limited to, RSA 161-F:42-54 and RSA 169-C:29;
3.9.8.1.9	In instances where the time frames detailed in the Agreement conflict with those in the Department

The MCO shall cooperate with review of any reported

Sentinel Event Policy, the policy requirements will

#### 3.9.9 **Non-Discrimination**

3.9.9.1 The MCO shall require Participating Providers and Subcontractors to comply with the laws listed in Section 3.9.8 (Compliance with State and Federal Laws) and the provisions of Executive Order 11246, Equal Opportunity, dated September 24, 1965, and all rules and regulations issued thereunder, and any other laws, regulations, or orders which prohibit discrimination on grounds of age, race, ethnicity, mental or physical disability, sexual or affection orientation or preference, marital status, genetic information, source of payment, sex, color, creed, religion, or national origin or ancestry. [42 CFR 438.3(d)(4)]

#### 3.9.10 Reporting Discrimination Grievances

- 3.9.10.1 The MCO shall forward to the Department copies of all grievances alleging discrimination against Members because of race, color, creed, sex, religion, age, national origin, ancestry, marital status, sexual or affectional orientation, physical or mental disability or gender identity for review and appropriate action within three (3) business days of receipt by the MCO.
- 3.9.10.2 Failure to submit any such grievance within three (3) business days may result in the imposition of liquidated damages as outlined in Section 5.5.2. (Liquidated Damages).

#### 3.9.11 Americans with Disabilities Act

3.9.11.1 The MCO shall have written policies and procedures that ensure compliance with requirements of the ADA, and a written plan to

Page 49 of 397	Date

New Hampshire Department of Health and Human Services Medicaid Care Management Services



## **Exhibit B - DRAFT**

- monitor compliance to determine the ADA requirements are being met.
- 3.9.11.2 The ADA compliance plan shall be sufficient to determine the specific actions that shall be taken to remove existing barriers and/or to accommodate the needs of Members who are qualified individuals with a disability.
- 3.9.11.3 The ADA compliance plan shall include the assurance of appropriate physical access to obtain included benefits for all Members who are qualified individuals with a disability, including but not limited to street level access or accessible ramp into facilities; access to lavatory; and access to examination rooms.
- 3.9.11.4 A "Qualified Individual with a Disability," defined pursuant to 42 U.S.C. Section 12131(2), is an individual with a disability who, with or without reasonable modifications to rules, policies, or practices, the removal of architectural, communication, or transportation barriers, or the provision of Auxiliary Aids and services, meets the essential eligibility requirements for the receipt of services or the participation in programs or activities provided by a public entity.
- 3.9.11.5 The MCO shall require Participating Providers and Subcontractors to comply with the requirements of the ADA. In providing Covered Services, the MCO shall not directly or indirectly, through contractual, licensing, or other arrangements, discriminate against Medicaid Members who are qualified individuals with disabilities covered by the provisions of the ADA.
- 3.9.11.6 The MCO shall survey Participating Providers of their compliance with the ADA using a standard survey document that shall be provided by the Department. Completed survey documents shall be kept on file by the MCO and shall be available for inspection by the Department.
- 3.9.11.7 The MCO shall, in accordance with Exhibit G (Certification Regarding ADA Compliance), annually submit to the Department a written certification that it is conversant with the requirements of the ADA, that it is in compliance with the ADA, that it has complied with this Section 3.9.11 (Americans with Disabilities Act) of the Agreement, and that it has assessed its Participating Provider network and certifies that Participating Providers meet ADA requirements to the best of the MCO's knowledge.
- 3.9.11.8 The MCO warrants that it shall hold the State harmless and indemnify the State from any liability which may be imposed upon the State as a result of any failure of the MCO to be in compliance with the ADA.
- 3.9.11.9 Where applicable, the MCO shall abide by the provisions of Section 504 of the Federal Rehabilitation Act of 1973, as amended, 29

Page 50 of 397 Date \_\_\_\_\_

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

U.S.C. Section 794, regarding access to programs and facilities by people with disabilities.

#### 3.9.12 Non-Discrimination in Employment

- 3.9.12.1 The MCO shall not discriminate against any employee or applicant for employment because of age, sex, gender identity, race, color, sexual orientation, marital status, familial status, or physical or mental disability, religious creed or national origin.
- 3.9.12.2 The MCO shall take affirmative action to ensure that applicants are employed, and that employees are treated during employment, without regard to their age, sex, gender identity, race, color, sexual orientation, marital status, familial status, or physical or mental disability, religious creed or national origin.
- 3.9.12.3 Such action shall include, but not be limited to the following: employment, upgrading, demotion, or transfer; recruitment or recruitment advertising; layoff or termination; rates of pay or other forms of compensation; and selection for training, including apprenticeship.
- 3.9.12.4 The MCO agrees to post in conspicuous places, available to employees and applicants for employment, notices to be provided by the contracting officer setting forth the provisions of this nondiscrimination clause.
- 3.9.12.5 The MCO shall, in all solicitations or advertisements for employees placed by or on behalf of the MCO, state that all qualified applicants shall receive consideration for employment without regard to age, sex, gender identity, race, color, sexual orientation, marital status, familial status, or physical or mental disability, religious creed or national origin.
- 3.9.12.6 The MCO shall send to each labor union or representative of workers with which it has a collective bargaining agreement or other agreement or understanding, a notice, to be provided by the agency contracting officer, advising the labor union or workers' representative of the MCO's commitments under Section 202 of Executive Order No. 11246 of September 24, 1965, and shall post copies of the notice in conspicuous places available to employees and applicants for employment.
- 3.9.12.7 The MCO shall comply with all provisions of Executive Order No. 11246 of Sept. 24, 1965, and of the rules, regulations, and relevant orders of the Secretary of Labor.
- 3.9.12.8 The MCO shall furnish all information and reports required by Executive Order No. 11246 of September 24, 1965, and by the rules, regulations, and orders of the Secretary of Labor, or pursuant thereto, and shall permit access to its books, records, and accounts

Page 51 of 397	Date

New Hampshire Department of Health and Human Services Medicaid Care Management Services



## **Exhibit B - DRAFT**

- by the Department and the Secretary of Labor for purposes of investigation to ascertain compliance with such rules, regulations, and orders.
- 3.9.12.9 The MCO shall include the provisions described in this Section 3.9.12 (Non-Discrimination in Employment) in every contract with a Subcontractor or purchase order unless exempted by rules, regulations, or orders of the Secretary of Labor issued pursuant to Section 204 of Executive Order No. 11246 of September 24, 1965, so that such provisions shall be binding upon each Subcontractor or vendor.
- 3.9.12.10 The MCO shall take such action with respect to any contract with a Subcontractor or purchase order as may be directed by the Secretary of Labor as a means of enforcing such provisions including sanctions for noncompliance, provided, however, that in the event the MCO becomes involved in, or is threatened with, litigation with a Subcontractor or vendor as a result of such direction, the MCO may request the United States to enter into such litigation to protect the interests of the United States.

#### 3.9.13 **Non-Compliance**

3.9.13.1 In the event of the MCO's noncompliance with the non-discrimination clauses of this Agreement or with any of such rules, regulations, or orders, this Agreement may be cancelled, terminated or suspended in whole or in part and the MCO may be declared ineligible for further government contracts in accordance with procedures authorized in Executive Order No. 11246 of Sept. 24, 1965, and such other sanctions may be imposed and remedies invoked as provided in Executive Order No. 11246 of September 24, 1965, or by rule, regulation, or order of the Secretary of Labor, or as otherwise provided by law.

## 3.9.14 Changes in Law

3.9.14.1 The MCO shall implement appropriate program, policy or system changes, as required by changes to State and federal laws or regulations or interpretations thereof.

#### 3.10 Subcontractors

#### 3.10.1 MCO Obligations

3.10.1.1 The MCO shall maintain ultimate responsibility for adhering to, and otherwise fully complying with the terms and conditions of this Agreement, notwithstanding any relationship the MCO may have with the Subcontractor, including being subject to any remedies contained in this Agreement, to the same extent as if such obligations, services and functions were performed by the MCO.

Page 52 of 397	Date

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

- 3.10.1.2 For the purposes of this Agreement, such work performed by any Subcontractor shall be deemed performed by the MCO. [42 CFR 438.230(b)]
- 3.10.1.3 The Department reserves the right to require the replacement of any Subcontractor or other contractor found by the Department to be unacceptable or unable to meet the requirements of this Agreement, and to object to the selection or use of a Subcontractor or contract.
- 3.10.1.4 The MCO, regardless of its written agreements with any Subcontractors, maintains ultimate responsibility for complying with this Agreement.
- 3.10.1.5 The MCO shall have oversight of all Subcontractors' policies and procedures for compliance with the False Claims Act (FCA) and other State and federal laws described in Section 1902(a)(68) of the Social Security Act, including information about rights of employees to be protected as whistleblowers.

#### 3.10.2 Contracts with Subcontractors

- 3.10.2.1 The MCO shall have a written agreement between the MCO and each Subcontractor which includes, but shall not be limited to:
  - 3.10.2.1.1 Full disclosure of the method and amount of compensation or other consideration received by the Subcontractor;
  - 3.10.2.1.2 Amount, duration, and scope of services to be provided by the Subcontractor;
  - 3.10.2.1.3 Term of the agreement, methods of extension, and termination rights;
  - 3.10.2.1.4 Information about the grievance and appeal system and the rights of the Member as described in 42 CFR 438.414 and 42 CFR 438.10(g);
  - 3.10.2.1.5 Requirements to comply with all applicable Medicaid laws, regulations, including applicable subregulatory guidance and applicable provisions of this Agreement; and
  - 3.10.2.1.6 In accordance with Prohibited Relationship provisions in Section 3.9.5.
  - 3.10.2.1.7 Requirements for the Subcontractor
    - 3.10.2.1.7.1. Provided that the Department makes timely payments to the MCO under this Agreement to hold harmless the Department and its employees, and all

Date _	
	Date _

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

Members served under the terms of this Agreement in the event of non-payment by the MCO;

3.10.2.1.7.2. To indemnify and hold harmless the Department and its employees against all injuries, deaths, losses, damages, claims, suits, liabilities, judgments, reasonable costs and expenses which may in any manner accrue against the Department or its employees through intentional misconduct, negligence, or omission of the Subcontractor, its agents, officers, employees or contractors.

# 3.10.2.1.8 Requirements that provide that:

Department, 3.10.2.1.8.1. The MCO, the Medicaid Fraud Control Unit (MFCU), NH Department of Justice (DOJ), U.S. DOJ, the OIG, and the Comptroller General or their respective designees shall have the right to audit, evaluate, and inspect, and that it shall make available for the purpose of audit, evaluation or inspection, any premises, physical facilities, equipment, books, records, contracts, computer or other electronic systems of the Subcontractor, or of the Subcontractor's contractor, that pertain to any aspect of the services and/or activities performed determination of amounts payable under Agreement; [42 438.230(c)(3)(i) & (ii); 42 CFR 438.3(k)]

3.10.2.1.8.2. The Subcontractor shall further agree that it can be audited for ten (10) years from the final date of the Term or from the date of any completed audit, whichever is later; and [42 CFR 438.230(c)(3)(iii); 42 CFR 438.3(k)]

3.10.2.1.8.3. The MCO, the Department, MFCU, NH DOJ, U.S. DOJ, OIG, and the Comptroller General or their respective designees may conduct an audit at any time if the Department, MFCU, NH DOJ.

Page 54 of 397 Date \_\_\_\_\_

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

U.S. DOJ, the OIG, and the Comptroller General or their respective designee determines that there is a reasonable possibility of Fraud, potential Member harm or similar risk. [42 CFR 438.230(c)(3)(iv); 42 CFR 438.3(k)]

- 3.10.2.1.8.4. Subcontractor's agreement to notify the MCO within one (1) business day of being cited by any State or federal regulatory authority;
- 3.10.2.1.8.5. Require Subcontractor to submit ownership and controlling interest information as required by Section 3.9.3 (Ownership and Control Disclosures);
- 3.10.2.1.8.6. Require Subcontractors to investigate and disclose to the MCO, at contract execution or renewal, and upon request by the MCO of the identified person who has been convicted of a criminal offense related to that person's involvement in any program under Medicare or Medicaid since the inception of those programs and who is [42 CFR 455.106(a)]:
  - 3.10.2.1.8.6.1A person who has an ownership or control interest in the Subcontractor or Participating Provider; [42 CFR 455.106(a)(1)];
  - 3.10.2.1.8.6.2An agent or person who has been delegated the authority to obligate or act on behalf of the Subcontractor or Participating Provider; or [42 CFR 455.101; 42 CFR 455.106(a)(1)];
  - 3.10.2.1.8.6.3An agent, managing employee, general manager, business manager, administrator, director, or other individual who exercises operational or managerial control over, or who directly or indirectly conducts the day-to-day operation of, the Subcontractor or Participating

Page 55 of 397 Date \_\_\_\_\_

New Hampshire Department of Health and Human Services Medicaid Care Management Services



## **Exhibit B - DRAFT**

Provider [42 CFR 455.101; 42 CFR 455.106(a)(2)];

- 3.10.2.1.8.6.4 Require Subcontractor to screen its directors, officers, employees, contractors and Subcontractors against each of the Exclusion Lists on a monthly basis and report to the MCO any person or entity appearing on any of the Exclusion Lists and begin termination proceedings within forty-eight (48) hours unless the individual is part of a federally-approved waiver program;
- 3.10.2.1.8.6.5 Require Subcontractor to have a compliance plan that meets the requirements of 42 CFR 438.608 and policies and procedures that meet the Deficit Reduction Act (DRA) of 2005 requirements;
- 3.10.2.1.8.6.6 Prohibit Subcontractor from making payments or deposits for Medicaid-covered items or services to financial institutions located outside of the United States or its territories;
- 3.10.2.1.8.6.7A provision for revoking delegation of activities or obligations, or imposing other sanctions if the Subcontractor's performance is determined to be unsatisfactory by the MCO or the Department;
- 3.10.2.1.8.6.8 Subcontractor's agreement to comply with the ADA, as required by Section 3.9.11 (Americans with Disabilities Act) above;
- 3.10.2.1.8.6.9 Include provisions of this Section 3.10.2 (Contracts with Subcontractors) in every Subcontract or purchase order unless exempted by rules, regulations, or orders of the

Page 56 of 397 Date \_\_\_\_\_

New Hampshire Department of Health and Human Services Medicaid Care Management Services



#### **Exhibit B - DRAFT**

Secretary of Labor issued pursuant to Section 204 of Executive Order No. 11246 of September 24, 1965;

- 3.10.2.1.8.6.10 Require Subcontractor, to the extent that the Subcontractor is delegated responsibility by the MCO for coverage of services and payment of claims under this Agreement, to implement policies and procedures, as reviewed by the Department, for reporting of all Overpayments identified, embezzlement including receipt of Capitation Payments to which it was not entitled or recovered, specifying Overpayments due to potential Fraud, to the State;
- 3.10.2.1.8.6.11 Require any Subcontractor to comply with all applicable Medicaid laws, regulations, including applicable subregulatory guidance and Agreement provisions. [42 CFR 438.230(c)(2); 42 CFR 438.3(k)]; and
- 3.10.2.1.8.6.12 Require any Subcontractor to comply with any other provisions specifically required under this Agreement or the applicable requirements of 42 CFR 438. [42 CFR 438.230]
- 3.10.2.2 The MCO shall notify the Department in writing within one (1) business day of becoming aware that its Subcontractor is cited as non-compliant or deficient by any State or federal regulatory authority.
- 3.10.2.3 If any of the MCO's activities or obligations under this Agreement are delegated to a Subcontractor:
  - 3.10.2.3.1 The activities and obligations, and related reporting responsibilities, are specified in the contract or written agreement between the MCO and the Subcontractor; and

Page 57 of 397	Date

New Hampshire Department of Health and Human Services Medicaid Care Management Services



#### **Exhibit B - DRAFT**

- 3.10.2.3.2 The contract or written arrangement between the MCO and the Subcontractor shall either provide for revocation of the delegation of activities or obligations, or specify other remedies in instances where the State or the MCO determines that the Subcontractor has not performed satisfactorily. [42 CFR 438.230(c)(1)(i)-(iii); 42 CFR 438.3(k)]
- 3.10.2.4 Subcontractors or any other party performing utilization review are required to be licensed in New Hampshire.

#### 3.10.3 **Notice and Approval**

- 3.10.3.1 The MCO shall submit all Subcontractor agreements and Subcontractor Provider agreements to the Department, for review at least sixty (60) calendar days prior to the anticipated implementation date of that Subcontractor agreement, any time there is a renewal or extension amendment to a Subcontractor agreement already reviewed by the Department or there is a substantial change in scope or terms of the Subcontractor agreement.
- 3.10.3.2 The MCO remains responsible for ensuring that all Agreement requirements are met, including requirements requiring the integration of physical and behavioral health, and that the Subcontractor adheres to all State and federal laws, regulations and related guidance and guidelines.
- 3.10.3.3 The MCO shall notify the Department of any change in Subcontractors and shall submit a new Subcontractor agreement for review sixty (60) calendar days prior to the start date of the new Subcontractor agreement.
- 3.10.3.4 Review by the Department of a Subcontractor agreement does not relieve the MCO from any obligation or responsibility regarding the Subcontractor and does not imply any obligation by the Department regarding the Subcontractor or Subcontractor agreement.
- 3.10.3.5 The Department may grant a written exception to the notice requirements of this Section 3.10.3 (Notice and Approval) if, in the Department's reasonable determination, the MCO has shown good cause for a shorter notice period.
- 3.10.3.6 The MCO shall notify the Department within five (5) business days of receiving notice from a Subcontractor of its intent to terminate a Subcontractor agreement.
- 3.10.3.7 The MCO shall notify the Department of any material breach by Subcontractor of an agreement between the MCO and the Subcontractor that may result in the MCO being non-compliant with

Page 58 of 397	Date

New Hampshire Department of Health and Human Services Medicaid Care Management Services



#### **Exhibit B - DRAFT**

- or violating this Agreement within one (1) business day of validation that such breach has occurred.
- 3.10.3.8 The MCO shall take any actions directed by the Department to cure or remediate said breach by the Subcontractor.
- 3.10.3.9 In the event of material change, breach or termination of a Subcontractor agreement between the MCO and a Subcontractor, the MCO's notice to the Department shall include a transition plan for the Department's review and approval.

## 3.10.4 MCO Oversight of Subcontractors

- 3.10.4.1 The MCO shall provide its Subcontractors with training materials regarding preventing Fraud, waste and abuse and shall require the MCO's hotline to be publicized to Subcontractors' staff who provide services to the MCO.
- 3.10.4.2 The MCO shall oversee and be held accountable for any functions and responsibilities that it delegates to any Subcontractor in accordance with 42 CFR 438.230 and 42 CFR Section 438.3, including:
  - 3.10.4.2.1 Prior to any delegation, the MCO shall evaluate the prospective Subcontractor's ability to perform the Social Security activities to be delegated;
  - 3.10.4.2.2 The MCO shall audit the Subcontractor's compliance with its agreement with the MCO and the applicable terms of this Agreement, at least annually and when there is a substantial change in the scope or terms of the Subcontractor agreement; and
  - 3.10.4.2.3 The MCO shall identify deficiencies or areas for improvement, if any. The MCO shall prompt the Subcontractor to take corrective action.
- 3.10.4.3 The MCO shall develop and maintain a system for regular and periodic monitoring of each Subcontractor's compliance with the terms of its agreement and this Agreement.
- 3.10.4.4 If the MCO identifies deficiencies or areas for improvement in the Subcontractor's performance that affect compliance with this Agreement, the MCO shall notify the Department within seven (7) calendar days and require the Subcontractor to develop a CAP. The MCO shall provide the Department with a copy of the Subcontractor's CAP within thirty (30) calendar days upon the Department request, which is subject to the Department approval [42 CFR 438.230 and 42 CFR Section 438.3]

Page 59 of 397 Date \_\_\_\_\_

New Hampshire Department of Health and Human Services Medicaid Care Management Services



## **Exhibit B - DRAFT**

# 3.11 Staffing

## 3.11.1 **Key Personnel**

- 3.11.1.1 The MCO shall commit key personnel to the MCM program on a full-time basis. Positions considered to be key personnel, along with any specific requirements for each position, include:
  - 3.11.1.1.1 CEO/Executive Director: Individual shall have clear authority over the general administration and day-to-day business activities of this Agreement.
  - 3.11.1.2 Finance Officer: Individual shall be responsible for accounting and finance operations, including all audit activities.
  - 3.11.1.3 Medical Director: Individual shall be a physician licensed by the NH Board of Medicine, shall oversee and be responsible for all clinical activities, including but not limited to, the proper provision of Covered Services to Members, developing clinical practice standards and clinical policies and procedures.
  - 3.11.1.4 The Medical Director shall have substantial involvement in QAPI Program activities and shall attend monthly, or as otherwise requested, in-person meetings with the Department's Medical Director.
  - 3.11.1.1.5 The Medical Director shall have a minimum of five (5) years of experience in government programs (e.g. Medicaid, Medicare, and Public Health).
  - 3.11.1.1.6 The Medical Director shall have oversight of all utilization review techniques and methods and their administration and implementation.
- 3.11.1.2 Quality Improvement Director: Individual shall be responsible for all QAPI program activities.
  - 3.11.1.2.1 Individual shall have relevant experience in quality management for physical and/or behavioral health care and shall participate in regular Quality Improvement meetings with the Department and the other MCOs to review quality related initiatives and how those initiatives can be coordinated across the MCOs.
- 3.11.1.3 Compliance Officer: Individual shall be responsible for developing and implementing policies, procedures, and practices designed to ensure compliance with the requirements of the Agreement.

Page 60 of 397	Date
1 age 00 01 007	

New Hampshire Department of Health and Human Services Medicaid Care Management Services



#### **Exhibit B - DRAFT**

- 3.11.1.3.1 The Compliance Officer shall report directly to the NH-based CEO or the executive director thereof.
- 3.11.1.4 Network Management Director: Individual shall be responsible for development and maintenance of the MCO's Participating Provider network.
- 3.11.1.5 Provider Relations Manager: Individual shall be responsible for provision of all MCO Provider services activities.
  - 3.11.1.5.1 The manager shall have prior experience with individual physicians, Provider groups and facilities.
- 3.11.1.6 Member Services Manager: Individual shall be responsible for provision of all MCO Member Services activities.
  - 3.11.1.6.1 The manager shall have prior experience with Medicaid populations.
- 3.11.1.7 Utilization Management (UM) Director: Individual shall be responsible for all UM activities.
  - 3.11.1.7.1 This person shall be under the direct supervision of the Medical Director and shall ensure that UM staff has appropriate clinical backgrounds in order to make appropriate UM decisions regarding Medically Necessary Services.
  - 3.11.1.7.2 The MCO shall also ensure that the UM program assigns responsibility to appropriately licensed clinicians, including a behavioral health and a LTSS professional for those respective services.
- 3.11.1.8 Systems Director/Manager: Individual shall be responsible for all MCO information systems supporting this Agreement, including but not limited to continuity and integrity of operations, continuity flow of records with the Department's information systems and providing necessary and timely reports to the Department.
- 3.11.1.9 Encounter Manager: Individual shall be responsible for and qualified by training and experience to oversee encounter submittal and processing to ensure the accuracy, timeliness, and completeness of encounter reporting.
- 3.11.1.10 Claims Manager: Individual shall be responsible for and qualified by training and experience to oversee claims processing and to ensure the accuracy, timeliness, and completeness of processing payment and reporting.
  - 3.11.1.10.1 Pharmacy Manager: Individual shall be a pharmacist licensed by the NH Board of Pharmacy and shall have a minimum of five (5) years pharmacy experience as a practicing pharmacist.

age 61 of 397	Date

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

- 3.11.1.10.1.1. The individual shall be responsible for all pharmacy activities, including but not limited to the Lock-In Program, coordinating clinical criteria for Prior Authorizations, compliance with the opioid prescribing requirements outlined in Section 4.13.23 (Substance Use Disorder) and overseeing the Drug Utilization Review (DUR) Board or the Pharmacy and Therapeutics Committee.
- 3.11.1.10.2 Substance Use Disorder Physician: Individual shall be an Addiction Medicine Physician licensed by the NH Board of Medicine and participate under the terms of this Agreement.
  - 3.11.1.10.2.1. The SUD Physician's responsibilities shall include, but are not limited to:
    - 3.11.1.10.2.1.1 In-person and in-state presence for greater than .50 FTE to meet with SUD Providers and PCPs to help expand SUD services. Discussion subjects shall include, but are not limited to, appropriate prescribing of medications for the treatment of opioid use disorder (MOUD);
    - 3.11.1.10.2.1.2 In person and in-state to educate SUD Providers regarding appropriate treatment plans, and documentation, and billing practices;
    - 3.11.1.10.2.1.3 Responsibility for providing clinical oversight and guidance for the MCO on Substance Use Disorder issues, including issues such as the use of ASAM or other evidence-based assessments and treatment protocols, the use of MAT, engagements with PRSS, and discharge planning for Members who visit an ED or are hospitalized for an overdose;

Page 62 of 397 Date \_\_\_\_\_

New Hampshire Department of Health and Human Services Medicaid Care Management Services



#### **Exhibit B - DRAFT**

- 3.11.1.10.2.1.4 Active meeting participation, and at least yearly, meetings with organizations that support persons with a substance use disorder, including OTPs, reduction hospitals, harm organizations, The Doorway program sites, CMHCs, sober living homes, and other non-profit for-profit organizations assisting persons with substance use disorder: and
- 3.11.1.10.2.1.5 Provide consultative support for the MCM program on a routine basis, including but not limited to, clinical policy related to Substance Use Disorders and individual Member cases, as needed.
- 3.11.1.11 MCO coordinators shall be responsible for overseeing Care Coordination and Care Management activities, and also serve as liaisons to Department staff for their respective functional areas. The MCO shall assign coordinators to each of the following areas on a full-time basis:
  - 3.11.1.11.1 Special Needs Coordinator: Individual shall have a minimum of a Master's Degree from a recognized college or university with major study in Social Work, Psychology, Education, Public Health or a related field.
    - 3.11.1.11.1. Individual shall have a minimum of eight (8) years demonstrated experience both in the provision of direct care services as well as progressively increasing levels of management responsibilities with a particular focus on special needs populations.
    - 3.11.1.11.2. The Special Needs Coordinator shall be responsible for ensuring compliance with and implementation of requirements for Adults and Children with Special Care Needs related to Care Management, Network Adequacy, access to Benefits, and Utilization Management.

age 63 of 397	Date

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

- 3.11.1.11.1.3. The Developmental Disability and Special Needs Coordinator positions may be either consolidated or established as individual part-time positions.
- 3.11.1.12 Developmental Disability Coordinator: Individual shall have a minimum of a Master's Degree from a recognized college or university with major study in Social Work, Psychology, Education, Public Health or a related field.
  - 3.11.1.12.1. Individual shall have a minimum of eight (8) years demonstrated experience both in the provision of direct care services as well as progressively increasing levels of management responsibilities, with a particular focus on direct care and administrative responsibilities related to services provided for developmentally disabled individuals.
  - 3.11.1.11.2.2. The Developmental Disability
    Coordinator shall be responsible for
    ensuring coordination with LTSS Case
    Managers for Members enrolled in the
    MCO but who have services covered
    outside of the MCO's Covered Services.
  - 3.11.1.11.2.3. The Developmental Disability and Special Needs Coordinator positions may be either consolidated or established as individual part-time positions.
- 3.11.1.13 Mental Health Coordinator: Individual shall oversee the delivery of Mental Health Services to ensure that there is a single point of oversight and accountability.
  - 3.11.1.13.1. Individual shall have a minimum of a Master's Degree from a recognized college or university with major study in Social Work, Psychology, Education, Public Health or a related field.
  - 3.11.1.13.2. Individual shall have a minimum of eight (8) years demonstrated experience both in the provision of direct care services as well as progressively increasing levels of management responsibilities, with a

Page 64 of 397	Date

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

particular focus on direct care and administrative responsibilities within Community Mental Health Services.

- 3.11.1.13.3. Other key functions shall include coordinating Mental Health Services across all functional areas including: quality management; oversight of the behavioral health Subcontract, as applicable: Care Management; Management: Utilization network development and management; Provider relations: implementation and interpretation of clinical policies and procedures; and Health-related social needs Health-related social needs and community-based resources.
- 3.11.1.14 Substance Use Disorder Coordinator: Individual shall be an addiction medicine specialist on staff or under contract who works with the Substance Use Disorder Physician to provide clinical oversight and guidance to the MCO on Substance Use Disorder issues.
  - 3.11.1.14.1. The Substance Use Disorder Coordinator shall be a Masters Licensed Alcohol and Drug Counselor (MLADC) or Licensed Mental Health Professional who is able to demonstrate experience in the treatment of Substance Use Disorder.
  - 3.11.1.14.2. The individual shall have expertise in screening, assessments, treatment, and Recovery strategies; use of MAT; strategies for working with child welfare agencies, correctional institutions and other health and social service agencies that serve individuals with Substance Use Disorders
  - 3.11.1.11.4.3. The individual shall be available to the MCM program on a routine basis for consultations on clinical, policy and operational issues, as well as the disposition of individual cases.
  - 3.11.1.11.4.4. Other key functions shall include coordinating Substance Use Disorder services and treatment across all

Page 65 of 397	Date

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

functional areas including: quality management; oversight of the behavioral health Subcontract, as applicable; Care Management; Utilization Management; network and development management; Provider relations; and Health-related social needs health-related social needs and community-based resources.

- 3.11.1.15 Long Term Care Coordinator: Individual shall be responsible for coordinating managed care Covered Services with FFS and waiver programs.
  - 3.11.1.15.1. The individual shall have a minimum of a Master's Degree in a Social Work, Psychology, Education, Public Health or a related field and have a minimum of eight (8) years of demonstrated experience both in the provision of direct care services at progressively increasing levels of management responsibilities, with a particular focus on direct care and administrative responsibilities related to long term care services.
- 3.11.1.11.6 Grievance Coordinator: Individual shall be responsible for overseeing the MCO's Grievance System.
- 3.11.1.11.7 Fraud, Waste, and Abuse Coordinator: Individual shall be responsible for tracking, reviewing, monitoring, and reducing Fraud, waste and abuse.
- 3.11.1.11.8 Transportation Coordinator: Individual shall oversee the delivery of NEMT services to Members to ensure that there is a single point of oversight and accountability for all transportation and NEMT services.
  - 3.11.1.11.8.1. The Transportation Coordinator shall be responsible for helping to identify, secure, and maintain transportation for Members, including but not limited to overseeing the MCO's NEMT Subcontractor.
  - 3.11.1.11.8.2. The Transportation Coordinator is responsible for ensuring the integration

Page 66 of 397 Date	3
---------------------	---

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

of transportation services into Member Care Plans.

- 3.11.1.18.3. The Transportation Coordinator shall ensure that the NEMT Subcontractor meets all NEMT requirements, including requirements as described in Section 4.1.9 (Non-Emergency Medical Transportation (NEMT)) and Exhibit O: Quality and Oversight Reporting Requirements of this Agreement as well as all other requirements in guidance provided by the Department.
- 3.11.1.11.8.4. The Transportation Coordinator shall be responsible for providing resolution to issues requiring immediate attention, including:
  - 3.11.1.11.8.4.1 Resolution of complaints made by Members and transportation Providers.
  - 3.11.1.11.8.4.2 Service delivery failures, including real-time assistance with rescheduling service appointments and/or transportation
- 3.11.1.18.5. The Transportation Coordinator shall have a minimum of four (4) years' experience relevant to the oversight of transportation services for vulnerable populations.
- 3.11.1.19 Housing Coordinator: The individual shall be responsible for helping to identify, secure, and maintain community based housing for Members and developing, articulating, and implementing a broader housing strategy within the MCO to expand housing availability/options. The Housing Coordinator shall:
  - 3.11.1.11.9.1. Act as the MCO's central housing expert/resource, providing education and assistance to all MCO's relevant staff (care managers and others) regarding supportive housing services and related issues.

Page 67 of 397	Date

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

- 3.11.1.11.9.2. Be a dedicated staff person whose primary responsibility is housing-related work.
- 3.11.1.11.9.3. Be a staff person to whom housingrelated work has been added to their existing responsibilities and function within the MCO.
- 3.11.1.11.9.4. At as a liaison with the Department's Bureau of Housing and Homeless Services to receive training and work in collaboration on capacity requirements/building.
- 3.11.1.11.9.5. Have at least two (2) year's full-time experience is assisting vulnerable populations to secure accessible, affordable housing.
- 3.11.1.19.6. Be familiar with the relevant public and private housing resources and stakeholders.
- 3.11.1.11.10 Prior Authorization Coordinator: Individual shall be responsible for all MCO Utilization Management activities and shall work under the direct supervision of the Medical Director.
  - 3.11.1.11.10.1. The Prior Authorization Coordinator shall ensure that all staff performing prior authorization functions have the necessary clinical backgrounds needed to apply established coverage criteria and make appropriate decisions based on medical necessary.
  - 3.11.1.11.10.2. The individual shall be licensed by the NH Board of Nursing and have a minimum of eight (8) years of demonstrated experience in both the provision of direct clinical services as well as progressively increasing levels of management responsibilities with a particular focus on performance of a variety of utilization functions including conducting inter-rater reliability quality audits.
- 3.11.1.11 Third Party Liability (TPL) Coordinator: Individual shall be responsible for ensuring the MCO and its

Page 68 of 397 Date \_\_\_\_\_

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

subcontractors are performing all required TPL functions when processing claims, that MCOs are properly identifying and recovering on claims not cost avoided, that the MCO has a system in place to manage subrogation cases and comply with contract requirements, and act as liaison between the Department's TPL unit and the MCO. This person shall have claims experience and a financial background.

## 3.11.2 Other MCO Required Staff

- 3.11.2.1 Fraud, Waste, and Abuse Staff: The MCO shall establish a Special Investigations Unit (SIU), which shall be comprised of experienced Fraud, waste and abuse investigators who have the appropriate training, education, experience, and job knowledge to perform and carry out all of the functions, requirements, roles and duties contained herein.
  - 3.11.2.1.1 At a minimum, the SIU shall have at least two (2) Fraud, waste and abuse investigators and one (1) Fraud, Waste and Abuse Coordinator.
  - 3.11.2.1.2 The MCO shall adequately staff the SIU to ensure that the MCO meets Agreement provisions of Section 5.3.2 (Fraud, Waste and Abuse).
- 3.11.2.2 Behavioral Health Staff: The MCO shall designate one (1) or more staff who have behavioral health specific managed care experience to provide assistance to Members who are homeless and oversee:
  - 3.11.2.2.1 Behavioral health Care Management;
  - 3.11.2.2.2 Behavioral health Utilization Management;
  - 3.11.2.2.3 Behavioral health network development; and
  - 3.11.2.2.4 The behavioral health Subcontract, as applicable.
- 3.11.2.3 Any subcontracted personnel or entity engaged in decision-making for the MCO regarding clinical policies related to Substance Use Disorder or mental health shall have demonstrated experience working in direct care for Members with Substance Use Disorder or mental health.
- 3.11.2.4 Emergency Services teams shall employ clinicians and certified Peer Support Specialists who are trained to manage crisis intervention and who have access to a clinician available to evaluate the Member on a face-to-face basis in the community to address the crisis and evaluate the need for hospitalization

#### 3.11.3 **On-Site Presence**

Page 69 of 397	Date

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

3.11.3.1	The MCO shall have an on-site presence in New Hampshire. On-
	site presence for the purposes of this section of the Agreement
	means that the MCO's full-time equivalent (1.0 FTE) personnel for
	each position identified below regularly reports to work in the State
	of New Hampshire unless otherwise specified:

- 3.11.3.1.1 CEO/Executive Director: 3.11.3.1.2 Medical Director: 3.11.3.1.3 **Network Management Director**; 3.11.3.1.4 Provider Relations Manager; 3.11.3.1.5 Pharmacy Manager; 3.11.3.1.6 Substance Use Disorder Physician; 3.11.3.1.7 Special Needs Coordinator; Mental Health Coordinator: 3.11.3.1.8 3.11.3.1.9 Substance Use Disorder Coordinator; 3.11.3.1.10 Developmental Disabilities Coordinator; 3.11.3.1.11 Long Term Care Coordinator; 3.11.3.1.12 Transportation Coordinator; 3.11.3.1.13 Housing Coordinator; 3.11.3.1.14 Grievance Coordinator; and
- 3.11.3.2 Upon the Department's request, MCO required staff who are not located in New Hampshire shall travel to New Hampshire for inperson meetings.

3.11.3.1.15 Fraud, Waste, and Abuse Coordinator

- 3.11.3.3 The MCO shall provide to the Department for review and approval key personnel and qualifications no later than sixty (60) calendar days prior to the start of the program.
- 3.11.3.4 The MCO shall staff the program with the key personnel as specified in this Agreement, or shall propose alternate staffing subject to review and approval by the Department, which approval shall not be unreasonably withheld.
- 3.11.3.5 The Department may grant a written exception to the notice requirements of this section if, in the Department's reasonable determination, the MCO has shown good cause for a shorter notice period.

#### 3.11.4 **General Staffing Provisions**

3.11.4.1 The MCO shall provide sufficient staff to perform all tasks specified in this Agreement. The MCO shall maintain a level of staffing

Page 70 of 397	Date

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

necessary to perform and carry out all of the functions, requirements, roles, and duties in a timely manner as contained herein. In the event that the MCO does not maintain a level of staffing sufficient to fully perform the functions, requirements, roles, and duties, the Department may impose liquidated damages, in accordance with Section 5.5.2 (Liquidated Damages).

- 3.11.4.2 The MCO shall ensure that all staff receive appropriate training, education, experience, and orientation to fulfill the requirements of the positions they hold and shall verify and document that it has met this requirement.
- 3.11.4.3 This includes keeping up-to-date records and documentation of all individuals requiring licenses and/or certifications and such records shall be available for the Department inspection.
- 3.11.4.4 All key personnel shall be generally available during Department hours of operation and available for in-person or video conferencing meetings as requested by the Department.
- 3.11.4.5 The MCO key personnel, and others as required by the Department, shall, at a minimum, be available for monthly in-person meetings in NH with the Department.
- 3.11.4.6 The MCO shall make best efforts to notify the Department at least thirty (30) calendar days in advance of any plans to change, hire, or reassign designated key personnel.
- 3.11.4.7 If a member of the MCO's key personnel is to be replaced for any reason while the MCO is under Agreement, the MCO shall inform the Department within seven (7) calendar days, and submit a transition plan with proposed alternate staff to the Department for review and approval, for which approval shall not be unreasonably withheld.
- 3.11.4.8 The Staffing Transition Plan shall include, but is not limited to:
  - 3.11.4.8.1 The allocation of resources to the Agreement during key personnel vacancy;
  - 3.11.4.8.2 The timeframe for obtaining key personnel replacements within ninety (90) calendar days; and
  - 3.11.4.8.3 The method for onboarding staff and bringing key personnel replacements/additions up-to-date regarding this Agreement.

## 4 PROGRAM REQUIREMENTS

- 4.1 Covered Populations and Services
  - 4.1.1 Overview of Covered Populations

Page 71 of 397	Date

New Hampshire Department of Health and Human Services Medicaid Care Management Services



## **Exhibit B - DRAFT**

- 4.1.1.1 The MCO shall provide and be responsible for the cost of managed care services to population groups deemed by the Department to be eligible for managed care and to be covered under the terms of this Agreement, as indicated in the table below.
- 4.1.1.2 Members enrolled with the MCO who subsequently become ineligible for managed care during MCO enrollment shall be excluded from MCO participation. The Department shall, based on State or federal statute, regulation, or policy, exclude other Members as appropriate.

Member Category	Eligible for Managed Care	Not Eligible for Managed Care (DHHS Covered)
Aid to the Needy Blind Non-Dual	X	
Aid to the Permanently and Totally Disabled Non-Dual	X	
American Indians and Alaskan Natives	×	
Auto Eligible and Assigned Newborns	х	
Breast and Cervical Cancer Program	Х	
Children Enrolled in Special Medical Services/Partners in Health	Х	
Children with Supplemental Security Income	Х	
Family Planning Only Benefit		Х
Foster Care/Adoption Subsidy	Х	
Granite Advantage (Medicaid Expansion Adults, Frail/Non-Frail)	Х	
Health Insurance Premium Payment		Х
Home Care for Children with Severe Disabilities (Katie Beckett)	Х	
In and Out Spend-Down		Х

Page 72 of 397 Date \_\_\_\_\_

New Hampshire Department of Health and Human Services Medicaid Care Management Services



### **Exhibit B - DRAFT**

Member Category	Eligible for Managed Care	Not Eligible for Managed Care (DHHS Covered)
Incarcerated individuals in the State's prison system eligible for participation in the Department's Community Reentry demonstration waiver	X	
Medicaid Children Funded through the Children's Health Insurance Program	X	
Medicaid for Employed Adults with Disabilities Non-Dual	X	
Medicaid for Employed Older Adults with Disabilities	X	
Medicare Duals with full Medicaid Benefits	X	
Medicare Savings Program Only (no Medicaid services)		Х
Members with Veterans Affairs Benefits		Х
Non-Expansion Poverty Level Adults (Including Pregnant Women) and Children Non-Dual	X	
Old Age Assistance Non-Dual	X	
Retroactive/Presumptive Eligibility Segments (excluding Auto Eligible Newborns)		Х
Third Party Coverage Non-Medicare, Except Members with Veterans Affairs Benefits	X	

#### 4.1.2 Overview of Covered Services

- 4.1.2.1 The MCO shall cover the physical health, behavioral health, pharmacy, and other benefits for all MCO Members, as indicated in the summary table below and described in this Agreement. Additional requirements for Behavioral Health Services are included in Section 4.13 (Behavioral Health), and additional requirements for pharmacy are included in Section 4.2 (Pharmacy Management).
- 4.1.2.2 The MCO shall provide, at a minimum, all Covered Services identified in the following matrix, and all Covered Services in accordance with the CMS-approved Medicaid State Plan and

Page 73 of 397	Date	

New Hampshire Department of Health and Human Services Medicaid Care Management Services



### **Exhibit B - DRAFT**

Alternative Benefit Plan State Plan. The MCO shall cover services consistent with 45 CFR 92.207(b).

4.1.2.3 While the MCO may provide a higher level of service and cover more services than required by the Department (as described in Section 4.1.3 (Covered Services Additional Provisions), the MCO shall, at a minimum, cover the services identified at least up to the limits described in NH Code of Administrative Rules, chapter He-E 801, He-E 802, He-W 530, and He-M 426. The Department reserves the right to alter this list at any time by providing reasonable notice to the MCO. [42 CFR 438.210(a)(1)-(3), (4)(i), (5) (i)-(ii)(A)-(C) and (b).

## 4.1.2.4 Summary of Covered Services

Services	MCO Covered	Not Included in Managed Care (DHHS Covered)
Acquired Brain Disorder Waiver Services		Х
Adult Medical Day Care	X	
Advanced Practice Registered Nurse	×	
Ambulance Service	X	
Ambulatory Surgical Center	X	
Audiology Services	X	
Certified Non-Nurse Midwife	X	
Choices for Independence Waiver Services		X
Child Health Support Service – Division for Children, Youth & Families, except for services eligible under EPSDT		Х
Community Mental Health Services	X	
Crisis Intervention–Division for Children, Youth & Families		Х
Developmental Disability Waiver Services		X

Page 74 of 397 Date \_\_\_\_\_



### **Exhibit B - DRAFT**

Services	MCO Covered	Not Included in Managed Care (DHHS Covered)
Dental Benefit Services <sup>3</sup>		X
Designated Receiving Facilities	Х	
Developmental Services Early Supports and Services		Х
Early and Periodic Screening, Diagnostic and Treatment Services including Applied Behavioral Analysis Coverage	X	
Family Planning Services	X	
Freestanding Birth Centers	Х	
Furnished Medical Supplies & Durable Medical Equipment	Х	
Glencliff Home		Х
Home Based Therapy–Division for Children, Youth & Families		Х
Home Health Services	X	
Home Visiting Services	Х	
Hospice	X	
Home and Community-Based In Home Support Services		Х
Inpatient Hospital	Х	
Inpatient Hospital Swing Beds, Intermediate Care		X
Inpatient Hospital Swing Beds, Skilled Nursing		Х
Inpatient Psychiatric Facility Services Under Age Twenty- One (21) <sup>4</sup>	Х	

<sup>&</sup>lt;sup>3</sup> Certain preventive, restorative, denture and other oral health services are carved-out of the MCM program and covered under the State's contract with Delta Dental of New Hampshire, Inc. for eligible adults ages 21 years and over. Dental and oral health emergency services for Medicaid enrolled children and adults of all ages are Covered Services under the MCM program.

Page 75 of 397

<sup>4</sup> Under age 22 if individual admitted prior to age 21.

New Hampshire Department of Health and Human Services Medicaid Care Management Services



### **Exhibit B - DRAFT**

Services	MCO Covered	Not Included in Managed Care (DHHS Covered)
Inpatient Psychiatric Treatment in State-owned New Hampshire Hospital and Hampstead Hospital, and Other State Determined IMD for Mental Illness <sup>5</sup>	Х	
Intensive Home and Community-Based Services–Division for Children, Youth & Families		X
Intermediate Care Facility Atypical Care		X
Intermediate Care Facility for Members with Intellectual Disabilities (e.g., Cedarcrest)		Х
Intermediate Care Facility Nursing Home		X
Laboratory (Pathology)	Х	
Medicaid to Schools Services		X
Medical Services Clinic (e.g., Opioid Treatment Program)	X	
Mental health services (e.g., psychology, psychotherapy, psychological and neurological testing)	Х	
Mobile Crisis Services	X	
Non-Emergency Medical Transportation <sup>6</sup>	Х	
Occupational Therapy <sup>7</sup>	Х	
Optometric Services Eyeglasses	X	
Outpatient Hospital <sup>8</sup>	Х	
Pediatric Residential Treatment Facility Services		Х
Personal Care Services	Х	

<sup>&</sup>lt;sup>5</sup> Medicaid managed care inpatient psychiatric treatment at State-owned New Hampshire Hospital and Hampstead Hospital, and other State determined IMD for mental illness are covered up to sixty (60) days for adults age 21-64 due to a primary diagnosis of mental illness.

<sup>&</sup>lt;sup>6</sup> Also includes mileage reimbursement for Medically Necessary travel.

<sup>&</sup>lt;sup>7</sup> Services are limited to twenty (20) visits per benefit year for each type of therapy including combined habilitation services and outpatient rehabilitation services.

<sup>&</sup>lt;sup>8</sup> Including facility and ancillary services for dental procedures.

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



### **Exhibit B - DRAFT**

Services	MCO Covered	Not Included in Managed Care (DHHS Covered)
Physical Therapy <sup>9</sup>	X	
Physicians Services	X	
Placement Services-Division for Children, Youth & Families		Х
Podiatrist Services	Х	
Prescribed Drugs	Х	
Preventative Services (e.g., nicotine cessation, SBIRT, transitional care management, chronic care management) <sup>10</sup>	X	
Private Duty Nursing	X	
Private Non-Medical Institutional For Children–Division for Children, Youth & Families		Х
Psychology	X	
Qualified Residential Treatment Program Services		Х
Rehabilitative Services Post Hospital Discharge	Х	
Rural Health Clinic & Federally Qualified Health Centers	Х	
Non-Swing Bed Skilled Nursing Facilities		Х
Skilled Nursing Facilities Skilled Nursing Facilities Atypical Care		Х
Speech Therapy <sup>11</sup>	X	
Substance Use Disorder Services (Per He-W 513), including services provided in Institutions for Mental Diseases pursuant to an approved 1115(a) research and demonstration waiver	х	

<sup>&</sup>lt;sup>9</sup> Outpatient Physical Therapy, Occupational Therapy and Speech Therapy services are limited to twenty (20) visits per benefit year for each type of therapy. Benefit limits are shared between habilitation services and outpatient rehabilitation services.

<sup>&</sup>lt;sup>10</sup> See Law of the State of New Hampshire 2023, Chapter79:203 (HB2) (authorizing preventative services which may include, but is not necessarily limited to those listed).

<sup>&</sup>lt;sup>11</sup>Outpatient Physical Therapy, Occupational Therapy and Speech Therapy services are limited to twenty (20) visits per benefit year for each type of therapy. Benefit limits are shared between habilitation services and outpatient rehabilitation services.

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



### **Exhibit B - DRAFT**

Services	MCO Covered	Not Included in Managed Care (DHHS Covered)
Transitional Housing Program Services and Community Residential Services With Wrap-Around Services and Supports	X	
Wheelchair Van	х	
X-Ray Services	x	

#### 4.1.3 Covered Services Additional Provisions

- 4.1.3.1 Nothing in this Section 4.1.3 shall be construed to limit the MCO's ability to otherwise voluntarily provide any other services in addition to the Covered Services required to be provided under this Agreement.
- 4.1.3.2 The MCO shall seek written approval from the Department, bear the entire cost of the service, and the utilization and cost of such voluntary services shall not be included in determining capitation rates.
- 4.1.3.3 All Covered Services shall be provided in accordance with 42 CFR 438.210 and 42 CFR 438.207(b). The MCO shall ensure there is no disruption in service delivery to Members or Providers as the MCO transitions these services into Medicaid managed care from FFS.
- 4.1.3.4 The MCO shall adopt written policies and procedures to verify that Covered Services are actually provided. [42 CFR 455.1(a)(2)]
- 4.1.3.5 In Lieu of Services
  - 4.1.3.5.1 The MCO may provide Members with services or settings that are "In Lieu of" Services or settings with prior approval and in accordance with federal regulations.
  - 4.1.3.5.2 The MCO may cover In Lieu of Services if:
    - 4.1.3.5.2.1. The alternative service or setting is a medically appropriate and cost-effective substitute:
    - 4.1.3.5.2.2. The Member is not required to use the alternative service or setting;
    - 4.1.3.5.2.3. The In Lieu of Service has been authorized by the Department and/or CMS, as appropriate; and

Page 78 of 397 Date \_\_\_\_\_

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

- 4.1.3.5.2.4. The in Lieu of Service has been offered to Members at the option of the MCO. [42 CFR 438.3(e)(2)(i-iii)]
- 4.1.3.5.3 For the MCO to obtain approval for In Lieu of Services not previously authorized by the Department, the MCO shall submit an In Lieu of Service request to the Department for each proposed In Lieu of Service not yet authorized.
- 4.1.3.5.4 The Department has authorized critical time intervention (CTI) services, diabetes self-management, and assistance in finding and keeping housing (not including rent), as In Lieu of Services (subject to CMS approval, as appropriate). This list may be expanded upon or otherwise modified by the Department through amendments of this Agreement, and CMS approval, as appropriate.
- 4.1.3.5.5 The MCO shall monitor the cost-effectiveness of each approved In Lieu of Service in accordance with Exhibit O: Quality and Oversight Reporting Requirements.

#### 4.1.3.6 Telemedicine

- 4.1.3.6.1 The MCO shall comply with provisions of RSA 167:4(d) by providing access to telemedicine services to Members in certain circumstances.
- 4.1.3.6.2 The MCO shall develop a telemedicine clinical coverage policy and submit the policy to the Department during Readiness Review for review. Covered telemedicine modalities shall comply with all local, State and federal laws including the HIPAA and record retention requirements, and Exhibit K: Information Security Requirements and the Exhibit Q: IT Requirements Workbook.
- 4.1.3.6.3 The clinical policy shall include security requirements which demonstrate how each covered telemedicine modality complies with Exhibit K, Information Security Requirements.

#### 4.1.3.7 Non-Participating Indian Health Care Providers

4.1.3.7.1 American Indian/Alaska Native Members are permitted to obtain Covered Services from Non-Participating Indian Health Care Providers (IHCP) from whom the Member is otherwise eligible to receive such services. [42 CFR 438.14(b)(4)]

Page 79 of 397 Date \_\_\_\_\_

New Hampshire Department of Health and Human Services Medicaid Care Management Services



#### **Exhibit B - DRAFT**

4.1.3.7.2 The MCO shall permit any American Indian/Alaska Native Member who is eligible to receive services from an IHCP PCP that is a Participating Provider, to choose that IHCP as their PCP, as long as that Provider has capacity to provide the services. [American Reinvestment and Recovery Act 5006(d); SMDL 10-001; 42 CFR 438.14(b)(3)]

## 4.1.3.8 Moral and Religious Grounds

- 4.1.3.8.1 An MCO that would otherwise be required to provide, reimburse for, or provide coverage of a counseling or referral service is not required to do so if the MCO objects to the service on moral or religious grounds. [Section 1932(b)(3)(B)(i) of the Social Security Act; 42 CFR 438.102(a)(2)]
- 4.1.3.8.2 If the MCO elects not to provide, reimburse for, or provide coverage of, a counseling or referral service because of an objection on moral or religious grounds, the MCO shall furnish information about the services it does not cover to the Department with its application for a Medicaid contract and any time thereafter when it adopts such a policy during the Term of this Agreement. [Section 1932(b)(3)(B)(i) of the Social Security Act: 42 CFR 438.102(b)(1)(i)(A)(1-2)]
- 4.1.3.8.3 If the MCO does not cover counseling or referral services because of moral or religious objections and chooses not to furnish information on how and where to obtain such services, the Department shall provide that information to potential Members upon request. [42 CFR 438.10(e)(2)(v)(C)]

## 4.1.4 Cost Sharing

- 4.1.4.1 Any cost sharing imposed on Medicaid Members shall be in accordance with NH's Medicaid Cost Sharing State Plan Amendment and Medicaid FFS requirements pursuant to 42 CFR 447.50 through 42 CFR 447.57. [Sections 1916(a)(2)(D) and 1916(b)(2)(D) of the Social Security Act; 42 CFR 438.108; 42 CFR 447.50-57.
- 4.1.4.2 With the exception of Members who are exempt from cost sharing as described in the Medicaid Cost Sharing State Plan Amendment, the MCO shall require point of service (POS) Cost Sharing for Covered Services for Members deemed by the Department to have annual incomes at or above one hundred percent (100%) of the FPL, as follows:

Page 80 of 397	Date	



# **Exhibit B - DRAFT**

	4.1.4.2.1	A Copayment of one dollar (\$1.00) shall be required for each preferred prescription drug and each refill of a preferred prescription drug;
	4.1.4.2.2	A Copayment of two dollars (\$2.00) shall be required for each non-preferred prescription drug and each refill of a non-preferred prescription drug, unless the prescribing Provider determines that a preferred drug will be less effective for the recipient and/or will have adverse effects for the recipient, in which case the Copay for the non-preferred drug shall be one dollar (\$1.00);
	4.1.4.2.3	A Copayment of one dollar (\$1.00) shall be required for a prescription drug that is not identified as either a preferred or non-preferred prescription drug; and
4.1.4.3	The following	services are exempt from cost-sharing:
	4.1.4.3.1	Emergency services,
	4.1.4.3.2	Family planning services,
	4.1.4.3.3	Preventive services provided to children,
	4.1.4.3.4	Pregnancy-related services,
	4.1.4.3.5	Services resulting from potentially preventable events, and,
	4.1.4.3.6	Cloraril (Clozapine) prescriptions. [42 CFR 447.56(a)]
4.1.4.4	Members are	exempt from Copayments when:
	4.1.4.4.1	The Member falls under the designated income threshold (one hundred percent (100%) or below the FPL);
	4.1.4.4.2	The Member is under eighteen (18) years of age;
	4.1.4.4.3	The Member is in a nursing facility or in an ICF for Members with IDs;
	4.1.4.4.4	The Member participates in one (1) of the HCBS waiver programs;
	4.1.4.4.5	The Member is pregnant and receiving services related to their pregnancy or any other medical condition that might complicate the pregnancy;
	4.1.4.4.6	The Member is receiving services for conditions related to their pregnancy and the prescription is filled or refilled within sixty (60) calendar days after the month the pregnancy ended;

Date \_\_\_\_\_ Page 81 of 397

New Hampshire Department of Health and Human Services Medicaid Care Management Services



#### **Exhibit B - DRAFT**

4.1.4.4.6.1.	The Member is in the Breast and Cervical Cancer Treatment Program;
4.1.4.4.6.2.	The Member is receiving hospice care; or
4.1.4.4.6.3.	The Member is an American Indian/Alaska Native.

4.1.4.5 Any American Indian/Alaskan Native who has ever received or is currently receiving an item or service furnished by an IHCP or through referral under contract health services shall be exempt from all cost sharing including Copayments and Premiums. [42 CFR 447.52(h); 42 CFR 447.56(a)(1)(x); ARRA 5006(a); 42 CFR 447.51; SMDL 10-001]

#### 4.1.5 **Emergency Services**

- 4.1.5.1 The MCO shall cover and pay for Emergency Services at rates that are no less than the equivalent Department FFS rates if the Provider that furnishes the services has an agreement with the MCO. [Section 1932(b)(2)(A) of the Social Security Act; 42 CFR 438.114(b)]
- 4.1.5.2 If the Provider that furnishes the Emergency Services does not have an agreement with the MCO, the MCO shall cover and pay for the Emergency Services in compliance with Section 1932(b)(2)(D) of the Social Security Act, 42 CFR 438.114(c)(1)(i), and the SMDL 3/20/98.
- 4.1.5.3 The MCO shall cover and pay for Emergency Services regardless of whether the Provider that furnishes the services is a Participating Provider.
- 4.1.5.4 The MCO shall pay Non-Participating Providers of Emergency and Post-Stabilization Services an amount no more than the amount that would have been paid under the Department FFS system in place at the time the service was provided. [SMDL 3/31/06; Section 1932(b)(2)(D) of the Social Security Act]
- 4.1.5.5 The MCO shall not deny treatment obtained when a Member had an Emergency Medical Condition, including cases in which the absence of immediate medical attention would not have had the outcomes specified in 42 CFR 438.114(a) of the definition of Emergency Medical Condition.
- 4.1.5.6 The MCO shall not deny payment for treatment obtained when a representative, such as a Participating Provider, or the MCO instructs the Member to seek Emergency Services [Section 1932(b)(2) of the Social Security Act; 42 CFR 438.114(c)(1)(i); 42 CFR 438.114(c)(1)(ii)(A-B)].

Page 82 of 397	Date

New Hampshire Department of Health and Human Services **Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

- 4.1.5.7 The MCO shall not limit what constitutes an Emergency Medical Condition on the basis of lists of diagnoses or symptoms.
- 4.1.5.8 The MCO shall not refuse to cover Emergency Services based on the emergency room Provider, hospital, or fiscal agent not notifying the Member's PCP. MCO. or the Department of the Member's screening and treatment within ten (10) calendar days of presentation for Emergency Services. [42 CFR 438.114(d)(1)(i-ii)]
- 4.1.5.9 The MCO may not hold a Member who has an Emergency Medical Condition liable for payment of subsequent screening and treatment needed to diagnose the specific condition or stabilize the patient. [42 CFR 438.114(d)(2)]
- 4.1.5.10 The attending emergency physician, or the Provider actually treating the Member, is responsible for determining when the Member is sufficiently stabilized for transfer or discharge, and that determination is binding on the entities identified in 42 CFR 438.114(b) as responsible for coverage and payment. [42 CFR 438.114(d)(3)]

#### 4.1.6 **Post-Stabilization Services**

- Post-Stabilization Services shall be covered and paid for in 4.1.6.1 accordance with provisions set forth at 42 CFR 422.113(c). The MCO shall be financially responsible for medically necessary Post-Stabilization Services:
  - 4.1.6.1.1 Obtained within or outside the MCO that are preapproved by a Participating Provider or other MCO representative;
  - 4.1.6.1.2 Obtained within or outside the MCO that are not preapproved by a Participating Provider or other MCO representative, but administered to maintain the Member's stabilized condition within one (1) hour of a request to the MCO for pre-approval of further poststabilization care services; and/or
  - 4.1.6.1.3 Administered to maintain, improve or resolve the stabilized condition without Member's authorization, and regardless of whether the Member obtains the services within the MCO network if:
    - 4.1.6.1.3.1. The MCO does not respond to a request for pre-approval within one (1) hour:
    - 4.1.6.1.3.2. The MCO cannot be contacted; or
    - е n

4.1.6.1.3.3.	The MCO representative treating physician cannot agreement concerning the	reach	ar
Page 83 of 397	Dat	te	

New Hampshire Department of Health and Human Services Medicaid Care Management Services



### **Exhibit B - DRAFT**

care and an MCO physician is not available for consultation. In this situation, the MCO shall give the treating physician the opportunity to consult with an MCO physician, and the treating physician may continue with care of the patient until an MCO physician is reached or one (1) of the criteria of 42 CFR 422.133(c)(3) is met. [42 CFR 438.114(e); 42 CFR 422.113(c)(2)(i)-(ii); 422.113(c)(2)(iii)(A)-(C)]

- 4.1.6.2 The MCO shall limit charges to Members for Post-Stabilization Services to an amount no greater than what the organization would charge the Member if the Member had obtained the services through the MCO. [[42 CFR 438.114(e); 42 CFR 422.113(c)(2)(iv)]
- 4.1.6.3 The MCO's financial responsibility for Post-Stabilization Services, if not pre-approved, ends when:
  - 4.1.6.3.1 The MCO physician with privileges at the treating hospital assumes responsibility for the Member's care;
  - 4.1.6.3.2 The MCO physician assumes responsibility for the Member's care through transfer;
  - 4.1.6.3.3 The MCO representative and the treating physician reach an agreement concerning the Member's care; or
  - 4.1.6.3.4 The Member is discharged. [42 CFR 438.114(e); 42 CFR 422.113(c)(3)(i-iv)]

#### 4.1.7 Value-Added Services

- 4.1.7.1 The MCO may elect to offer Value-Added Services that are not covered in the Medicaid State Plan or under this Agreement in order to improve health outcomes, the quality of care, or reduce costs, in compliance with 42 CFR 438.3(e)(i).
- 4.1.7.2 Value-Added Services are services that are not currently provided under the Medicaid State Plan. The MCO may elect to add Value-Added Services not specified in the Agreement at the MCO's discretion, but the cost of these Value-Added Services shall not be included in Capitation Payment calculations. The MCO shall submit to the Department an annual list of the Value-Added Services being provided.

#### 4.1.8 Early and Periodic Screening, Diagnostic, and Treatment

Page 84 of 397	Date

New Hampshire Department of Health and Human Services Medicaid Care Management Services



## **Exhibit B - DRAFT**

- 4.1.8.1 The MCO shall provide the full range of preventive, screening, diagnostic and treatment services including all medically necessary 1905(a) services that correct or ameliorate physical and mental illnesses and conditions for EPSDT eligible beneficiaries ages birth to twenty-one in accordance with 1905(r) of the Social Security Act. [42 CFR 438.210(a)(5)]
- 4.1.8.2 The MCO shall determine whether a service is Medically Necessary on a case by case basis, taking into account the medical necessity criteria specific to EPSDT defined in 42 U.S.C. Section 1396d(r), 42 CFR 438.210, and 42 CFR Subpart B—Early and Periodic Screening, Diagnosis, and Treatment (EPSDT) of Individuals Under Age 21, and the particular needs of the child and consistent with the definition for Medical Necessity included in this Agreement.
- 4.1.8.3 Upon conclusion of an individualized review of medical necessity, the MCO shall cover all Medically Necessary services that are included within the categories of mandatory and optional services listed in 42 U.S.C. Section 1396d(a), regardless of whether such services are covered under the Medicaid State Plan and regardless of whether the request is labeled as such, with the exception of all services excluded from the MCO.
- 4.1.8.4 The MCO may provide Medically Necessary services in the most economic mode possible, as long as:
  - 4.1.8.4.1 The treatment made available is similarly efficacious to the service requested by the Member's physician, therapist, or other licensed practitioner;
  - 4.1.8.4.2 The determination process does not delay the delivery of the needed service; and
  - 4.1.8.4.3 The determination does not limit the Member's right to a free choice of Participating Providers within the MCO's network.
- 4.1.8.5 Specific limits (number of hours, number of visits, or other limitations on scope, amount or frequency, multiple services same day, or location of service) in the MCO clinical coverage policies, service definitions, or billing codes do not apply to Medicaid Members less than twenty-one (21) years of age, when those services are determined to be Medically Necessary per federal EPSDT criteria.
- 4.1.8.6 If a service is requested in quantities, frequencies, or at locations or times exceeding policy limits and the request is reviewed and approved per EPSDT criteria as Medically Necessary to correct or ameliorate a defect, physical or mental illness, it shall be provided. This includes limits on visits to physicians, therapists, dentists, or other licensed, enrolled clinicians.

Page 85 of 397	Date

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

- 4.1.8.7 The MCO shall not require Prior Authorization for Non-Symptomatic Office Visits (early and periodic screenings/Wellness Visits) for Members less than twenty-one (21) years of age. The MCO may require Prior Authorization for other diagnostic and treatment products and services provided under EPSDT.
- 4.1.8.8 The MCO shall conduct Prior Authorization reviews using current clinical documentation, and shall consider the individual clinical condition and health needs of the child Member. The MCO shall not make an adverse benefit determination on a service authorization request for a Member less than twenty-one (21) years of age until the request is reviewed per EPSDT criteria.
- 4.1.8.9 While an EPSDT request is under review, the MCO may suggest alternative services that may be better suited to meet the Member's needs, engage in clinical or educational discussions with Members or Providers, or engage in informal attempts to resolve Member concerns as long as the MCO makes clear that the Member has the right to request authorization of the services he or she wants to request.
- 4.1.8.10 The MCO shall develop effective methods to ensure that Members less than twenty-one (21) years of age receive all elements of preventive health screenings recommended by the AAP in the Academy's most currently published Bright Futures preventive pediatric health care periodicity schedule using a validated screening tool. The MCO shall be responsible for requiring in contracts that all Participating Providers that are PCPs perform such screenings.
- 4.1.8.11 The MCO shall require that PCPs that are Participating Providers include all the following components in each medical screening:
  - 4.1.8.11.1 Comprehensive health and developmental history that assesses for both physical and mental health, as well as for Substance Use Disorders;
  - 4.1.8.11.2 Screening for developmental delay at each visit through the fifth (5th) year using a validated screening tool;
  - 4.1.8.11.3 Screening for Autism Spectrum Disorders per AAP guidelines;
  - 4.1.8.11.4 Comprehensive, unclothed physical examination:
  - 4.1.8.11.5 All appropriate immunizations, in accordance with the schedule for pediatric vaccines, laboratory testing (including blood lead screening appropriate for age and risk factors); and

Page 86 of 397 Date \_\_\_\_\_

New Hampshire Department of Health and Human Services Medicaid Care Management Services



## **Exhibit B - DRAFT**

- 4.1.8.11.6 Health education and anticipatory guidance for both the child and caregiver.
- 4.1.8.12 The MCO shall include the following information related to EPSDT in the Member Handbook:
  - 4.1.8.12.1 The benefits of preventive health care;
  - 4.1.8.12.2 Services available under the EPSDT program and where and how to obtain those services;
  - 4.1.8.12.3 That EPSDT services are not subject to cost-sharing; and
  - 4.1.8.12.4 That the MCO shall provide scheduling and transportation assistance for EPSDT services upon request by the Member.
- 4.1.8.13 The MCO shall perform outreach to Members who are due or overdue for an EPSDT screening service on a monthly basis.
  - 4.1.8.13.1 The MCO shall provide referral assistance for non-medical treatment not covered by the plan but found to be needed as a result of conditions disclosed during screenings and diagnosis.
- 4.1.8.14 The MCO shall submit its EPSDT plan for the Department's review and approval as part of its Readiness Review and in accordance with Exhibit O: Quality and Oversight Reporting Requirements.
- 4.1.9 Non-Emergency Medical Transportation (NEMT)
  - 4.1.9.1 The MCO shall arrange for the NEMT of its Members to ensure Members receive Medically Necessary care and services covered by the Medicaid State Plan regardless of whether those Medically Necessary Services are covered by the MCO.
    - 4.1.9.1.1 The MCO shall deem NEMT Medically Necessary for coverage of a Member's NEMT covered service to a medical appointment originating from and returning to a nursing facility.
  - 4.1.9.2 The MCO shall provide the most cost-effective and least expensive mode of transportation to secure Covered Services for its Members. However, the MCO shall ensure that a Member's lack of personal transportation is not a barrier of accessing care. The MCO and/or any Subcontractors shall be required to comply with all of the NEMT Medicaid State Plan requirements.
  - 4.1.9.3 The MCO shall ensure that each vehicle providing NEMT Covered Services meets the following requirements:
    - 4.1.9.3.1 Has a valid vehicle registration;

Page 87 of 397 Date
---------------------

4.1.9.4

4.1.9.5

4.1.9.6

4.1.9.7



# **Exhibit B - DRAFT**

EXII	IDIT B - DRAFT
4.1.9.3.2	Has undergone a satisfactory safety inspection in accordance with the laws of the state of New Hampshire; and
4.1.9.3.3	Has no apparent need for maintenance that affects safety including, but not limited to, visible holes in the body of the vehicle, defective brakes, worn or underinflated tires, leaking fluids, or illuminated check engine light.
Mileage Rein family memb Necessary se enroll in the I	all ensure that its Members utilize a Family and Friends nbursement Program if they have a car, or a friend or er with a car, who can drive them to their Medically ervice. A Member with a car who does not want to family and Friends Program shall meet one (1) of the eria to qualify for transportation services:
4.1.9.4.1	Does not have a valid driver's license;
4.1.9.4.2	Does not have a working vehicle available in the household;
4.1.9.4.3	Is unable to travel or wait for services alone; or
4.1.9.4.4	Has a physical, cognitive, mental or developmental limitation.
cents per m encourage th	and Friends mileage reimbursement rate shall be 62.5 nile. The MCO shall create incentive programs to be utilization of the Family and Friends Program with a percent (50%) utilization.
If no car is transportation	owned or available, the Member shall use public n if:
4.1.9.6.1	The Member lives less than one half mile from a bus route;
4.1.9.6.2	The Provider is less than one half mile from the bus route; and
4.1.9.6.3	The Member is an adult under the age of sixty-five (65).
Exceptions th	ne above public transportation requirement are:
4.1.9.7.1	The Member has two (2) or more children under age six (6) who shall travel with the parent;
4.1.9.7.2	The Member has one (1) or more children over age six (6) who has limited mobility and shall accompany the parent to the appointment; or

Date \_\_\_\_\_ Page 88 of 397



# **Exhibit B - DRAFT**

	4.1.9.7.3	The Member conditions:	er has at least one (1) of the following
		4.1.9.7.3.1.	Pregnant or up to six (6) weeks post-partum;
		4.1.9.7.3.2.	Moderate to severe respiratory condition with or without an oxygen dependency;
		4.1.9.7.3.3.	Limited mobility (walker, cane, wheelchair, amputee, etc.);
		4.1.9.7.3.4.	Visually impaired;
		4.1.9.7.3.5.	Developmentally delayed;
		4.1.9.7.3.6.	Significant and incapacitating degree of mental illness; or
		4.1.9.7.3.7.	Other exception by Provider approval only.
4.1.9.8	•	ber is provide	not an option, the MCO shall ensure that d transportation from a transportation
	4.1.9.8.1		driver services, excluding public transit MCO shall ensure:
		4.1.9.8.1.1.	Background checks are performed for all NEMT drivers;
		4.1.9.8.1.2.	Each Provider and individual driver is not excluded from participation in any federal health care program (as defined in section 1128B(f) of the Act) and is not listed on the exclusion list of the Inspector General of the Department of Health and Human Services;
		4.1.9.8.1.3.	Each such individual driver has a valid driver's license;
		4.1.9.8.1.4.	Each such provider has in place a process to address any violation of a State drug law;
		4.1.9.8.1.5.	Each such provider has in place a process to disclose to the State Medicaid program the driving history, including any traffic violations, of each such individual driver employed by such provider. [Consolidated Appropriations

Date \_\_\_\_\_ Page 89 of 397

Medicaid Care Management Services Contract
New Hampshire Department of Health and Human Services
Medicaid Care Management Services



## **Exhibit B - DRAFT**

		Act, 2021 (Public Law 116-260), Division CC, Title II, Section 209];	
	4.1.9.8.1.6.	Each such individual driver consistently utilizes a Global Positioning System device (GPS) to document the date, time and location for each pick up and drop off to track on-time performance and ensure that trips take place as scheduled;	
	4.1.9.8.1.7.	All vehicles utilized in the delivery of NEMT services shall be compliant with all federal and state safety requirements during the provision of the NEMT ride; and	
	4.1.9.8.1.8.	Once a ride has been confirmed for a Member, the ride shall be provided unless cancelled by the Member.	
4.1.9.8.2	independen	ment may require the procurement of an tevaluator to measure and report on how ces are being provided.	
4.1.9.8.3	The Department reserves the right to reject, suspend, or terminate any Transportation Provider and/or individual driver from participation in the NEMT Program.		
4.1.9.8.4	The MCO shall submit a weekly issue log for NEMT services as specified in Exhibit O: Quality and Oversight Reporting Requirements, and guidance issued by the Department.		
	4.1.9.8.4.1.	NEMT Encounter Data and submission shall conform to all requirements described in Section 5.1.3 (Encounter Data) of this Agreement.	
4.1.9.8.5	are no disru issues whic	portation Coordinator shall ensure there ptions to Covered Services due to NEMT h shall be subject to liquidated damages are with Exhibit N: Liquidated Damages	
	4.1.9.8.5.1.	The MCO, through their sole responsibility to provide transportation for their Members, shall assure that ninety-five percent (95%) of all Member scheduled rides for Covered Services	

Date \_\_\_\_\_ Page 90 of 397

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



### **Exhibit B - DRAFT**

are delivered within fifteen (15) minutes of the scheduled pick-up time or shall otherwise be subject to liquidated damages in accordance with Exhibit N: Liquidated Damages Matrix.

- 4.1.9.9 The Department reserves the right to require the use of a single transportation Subcontractor.
  - 4.1.9.9.1 The MCO shall subcontract with and provide remuneration to the single transportation Subcontractor designated by the Department for NEMT services. The Department has the sole discretion to establish the subcontract terms.
  - 4.1.9.9.2 The MCO shall not make amendments to the single transportation contract without prior written approval from the Department.
- 4.1.9.10 Failure of the MCO to meet any of these requirements shall subject the MCO to liquidated damages as specified in Exhibit N: Liquidated Damages Matrix.
- 4.1.9.11 The MCO shall provide reports to the Department related to NEMT requests, authorizations, trip results, service use, late rides, and cancellations, in accordance with Exhibit O: Quality and Oversight Reporting Requirements.

#### 4.2 Pharmacy Management

#### 4.2.1 General

- 4.2.1.1 The Department reserves the right to require the use of a single Pharmacy Benefits Manager (PBM) starting in Year 3 or Year 4 of this Agreement.
  - 4.2.1.1.1 The MCO shall subcontract with and provide remuneration to the Single PBM designated by the Department for pharmacy claims payment and administrative services. The Department has the sole discretion to establish the subcontract terms.
  - 4.2.1.1.2 The MCO shall not make amendments to the Single PBM subcontract without prior written approval from the Department.

## 4.2.2 MCO and DHHS Covered Prescription Drugs

4.2.2.1 The MCO shall cover all outpatient drugs where the manufacturer has entered into the federal rebate agreement and for which the Department provides coverage as defined in Section 1927(k)(2) of the Social Security Act [42 CFR 438.3(s)(1)]. The MCO shall not

Page 91 of 397	Date

New Hampshire Department of Health and Human Services Medicaid Care Management Services



#### **Exhibit B - DRAFT**

- include drugs by manufacturers not participating in the Omnibus Budget Reconciliation Act of 1990 (OBRA 90) Medicaid rebate program on the MCO formulary without the Department's consent.
- 4.2.2.2 The Department shall include a High-Cost Pharmacy Risk Pool (HCPRP) for purposes of risk mitigation as described in Section 6.3.5.1.1 of this Agreement.
- 4.2.2.3 The MCO shall pay for all prescription drugs, including specialty and office administered drugs consistent with the MCO's formulary, pharmacy edits and Prior Authorization criteria reviewed and approved by the Department, and are consistent with the Department's Preferred Drug List (PDL) as described in Section 4.2.3 (MCO Formulary) below.
- 4.2.2.4 Current Food and Drug Administration (FDA)-approved specialty, bio-similar and orphan drugs, and those approved by the FDA in the future, shall be covered in their entirety by the MCO.
- 4.2.2.5 The MCO shall pay for, when Medically Necessary, orphan drugs that are not yet approved by the FDA for use in the United States but that may be legally prescribed on a "compassionate-use basis" and imported from a foreign country.
- 4.2.2.6 The MCO shall ensure Members diagnosed with opioid use disorder, Substance Use Disorder, and behavioral health conditions treated at Community Mental Health Programs, FQHCs, FQHC look-alikes, and Doorway network facilities with integrated on-site pharmacies have immediate access to covered specialty drugs to treat related conditions.

## 4.2.3 **MCO Formulary**

- 4.2.3.1 The Department shall establish the PDL and shall be the sole party responsible for negotiating rebates for drugs on the PDL.
- 4.2.3.2 The MCO shall use the Department's PDL and shall not negotiate any drug rebates with pharmaceutical manufacturers for prescribed drugs on the PDL.
- 4.2.3.3 The Department shall be responsible for invoicing any pharmaceutical manufacturers for federal rebates mandated under federal law and for PDL supplemental rebates negotiated by the Department.
- 4.2.3.4 The MCO shall develop a formulary that adheres to the Department's PDL for drug classes included in the PDL and is consistent with Section 4.2.2 (MCO and DHHS Covered Prescription Drugs). In the event that the Department makes changes to the PDL, the Department shall notify the MCO of the change and provide the MCO with 30 calendar days to implement the change.

Page 92 of 397	Date	

New Hampshire Department of Health and Human Services Medicaid Care Management Services



## **Exhibit B - DRAFT**

- 4.2.3.5 Negative changes shall apply to new starts within thirty (30) calendar days of notice from the Department. The MCO shall have ninety (90) calendar days to notify Members and prescribers currently utilizing medications that are to be removed from the PDL if current utilization is to be transitioned to a preferred alternative.
- 4.2.3.6 For any drug classes not included in the Department's PDL, the MCO shall determine the placement on its formulary of products within that drug class, provided the MCO covers all products for which a federal manufacturer rebate is in place and the MCO is in compliance with all Department requirements in this Agreement.
- 4.2.3.7 The Department shall maintain a uniform review and approval process through which the MCO may submit additional information and/or requests for the inclusion of additional drug or drug classes on the Department's PDL. The Department shall invite the MCO's Pharmacy Manager to attend meetings of the NH Medicaid DUR Board.
- 4.2.3.8 The MCO shall make an up-to-date version of its formulary available to all Participating Providers and Members through the MCO's website and electronic prescribing tools. The formulary shall be available to Members and Participating Providers electronically, in a machine-readable file and format, and shall, at minimum, contain information related to:
  - 4.2.3.8.1 Which medications are covered, including whether it is the generic and/or the brand drug; and
  - 4.2.3.8.2 What tier each medication is on. [42 CFR 438.10(i)(1-3)]
- 4.2.3.9 The MCO shall adhere to all relevant State and federal law, including without limitation, with respect to the criteria regarding coverage of non-preferred formulary drugs pursuant to Chapter 188, laws of 2004, Senate Bill 383-FN, Section IVa. A Member shall continue to be treated or, if newly diagnosed, may be treated with a non-preferred drug based on any one (1) of the following criteria:
  - 4.2.3.9.1 Allergy to all medications within the same class on the PDL;
  - 4.2.3.9.2 Contraindication to or drug-to-drug interaction with all medications within the same class on the PDL;
  - 4.2.3.9.3 History of unacceptable or toxic side effects to all medications within the same class on the PDL;
  - 4.2.3.9.4 Therapeutic failure of all medications within the same class on the PDL;

Page 93 of 397 Date \_\_\_\_\_

New Hampshire Department of Health and Human Services **Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

4.2.3.9.5	An indication that is unique to a non-preferred drug and is supported by peer-reviewed literature or a unique federal FDA-approved indication;
4.2.3.9.6	An age-specific indication;
4.2.3.9.7	Medical co-morbidity or other medical complication that precludes the use of a preferred drug; or;
4.2.3.9.8	Clinically unacceptable risk with a change in therapy to a preferred drug. Selection by the physician of the criteria under this subparagraph shall require an automatic approval by the pharmacy benefit program.

#### 4.2.4 **Pharmacy Clinical Policies and Prior Authorizations**

- The MCO, including any pharmacy Subcontractors, shall establish 4.2.4.1 a pharmacy Prior Authorization program that includes Prior Authorization criteria and other POS edits (such as prospective DUR edits and dosage limits), and complies with Section 1927(d)(5) of the Social Security Act [42 CFR 438.3(s)(6)] and any other applicable State and federal laws, including House Bill 517, as further described in Section 4.8.1.6 (Prior Authorization).
- 4.2.4.2 The MCO's pharmacy Prior Authorization criteria, including any pharmacy policies and programs, shall be submitted to the Department prior to the implementation of this Agreement, shall be subject to the Department's approval, and shall be submitted to the Department prior to the MCO's implementation of a modification to the criteria, policies, and/or programs.
- The MCO's pharmacy Prior Authorization criteria shall be no more 4.2.4.3 restrictive than the Prior Authorization criteria of the Fee for Service (FFS) program.
- The MCO's pharmacy Prior Authorization criteria shall meet the 4.2.4.4 requirements related to Substance Use Disorder, as outlined in (Limitations on Prior 4.13.33.3 Authorization Requirements) of this Agreement. Under no circumstances shall the MCO's Prior Authorization criteria and other POS edits or policies depart from these requirements.
  - 4.2.4.4.1 Additionally, specific to Substance Use Disorder, the MCO shall offer a pharmacy mail order opt-out program that is designed to support Members in individual instances where mail order requirements create an unanticipated and unique hardship.
  - 4.2.4.4.2 The MCO shall conduct both prospective and retrospective DUR for all Members receiving MAT for Substance Use Disorder to ensure that Members are

Page 94 of 397 Date \_\_\_\_

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

not receiving opioids and/or benzodiazepines from other health care Providers while receiving MAT.

- 4.2.4.4.3 The retrospective DUR shall include a review of medical claims to identify Members that are receiving MAT through physician administered drugs (such as methadone, Vivitrol®, etc.).
- 4.2.4.5 The MCO shall make available on its website information regarding any modifications to the MCO's pharmacy Prior Authorization criteria, pharmacy policies, and pharmacy programs no less than thirty (30) calendar days prior to the Department -approved modification effective date.
- 4.2.4.6 Further, the MCO shall notify all Members and Participating Providers impacted by any modifications to the MCO's pharmacy Prior Authorization criteria, pharmacy policies, and pharmacy programs no less than thirty (30) calendar days prior to the Department -approved modification effective date.
- 4.2.4.7 The MCO shall implement and operate a DUR program that shall be in compliance with Section 1927(g) of the Social Security Act, address Section 1004 provisions of the SUPPORT for Patient and Communities Act, and include:
  - 4.2.4.7.1 Prospective DUR;
  - 4.2.4.7.2 Retrospective DUR:
  - 4.2.4.7.3 An educational program for Participating Providers, including prescribers and dispensers; and
  - 4.2.4.7.4 DUR program features in accordance with Section 1004 provisions of the SUPPORT for Patient and Communities Act, including:
    - 4.2.4.7.4.1. Safety edit on days' supply, early refills, duplicate fills, and quantity limitations on opioids and a claims review automated process that indicates fills of opioids in excess of limitations identified by the State:
    - 4.2.4.7.4.2. Safety edits on the maximum daily morphine equivalent for treatment of pain and a claims review automated process that indicates when an individual is prescribed the morphine milligram equivalent for such treatment in excess of any limitation that may be identified by the State;

Page 95 of 397 Date \_\_\_\_\_

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

- 4.2.4.7.4.3. A claims review automated process that monitors when an individual is concurrently prescribed opioids and benzodiazepines or opioids and antipsychotics;
- 4.2.4.7.4.4. A program to monitor and manage the appropriate use of antipsychotic medications by all children including foster children enrolled under the State Plan;
- 4.2.4.7.4.5. Fraud and abuse identification processes that identifies potential Fraud or abuse of controlled substances by beneficiaries, health care providers, and pharmacies; and
- 4.2.4.7.4.6. Operate like the State's Fee-for-Service DUR program. [42 CFR 456, subpart K; 42 CFR 438.3(s)(4)].
- 4.2.4.8 The MCO shall submit to the Department a detailed description of its DUR program prior to the implementation of this Agreement and, if the MCO's DUR program changes, annually thereafter.
  - 4.2.4.8.1 In accordance with Section 1927 (d)(5)(A) of the Social Security Act, the MCO shall respond by telephone or other telecommunication device within twenty-four (24) hours of a request for Prior Authorization one hundred percent (100%) of the time and reimburse for the dispensing of at least a seventy two (72) hour supply of a covered outpatient prescription drug in an emergency situation when Prior Authorization cannot be obtained. [42 CFR 438.210(d)(3)]
- 4.2.4.9 The MCO shall develop and/or participate in other State of New Hampshire pharmacy-related quality improvement initiatives, as required by the Department and in alignment with the MCO's QAPI, further described in Section 4.14.3 (Quality Assessment and Performance Improvement Program).
  - 4.2.4.9.1 For the HEDIS Measure "Use of Opioids from Multiple Providers", the MCO shall achieve performance that is less than or equal to the average rate of New England HMO Medicaid health plans as reported by NCQA Quality Compass for the previous calendar year.

Page 96 of 397	Date
rage 90 01 397	Date

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



### **Exhibit B - DRAFT**

- 4.2.4.10 The MCO shall institute a Pharmacy Lock-In Program for Members, which has been reviewed by the Department, and complies with requirements included in Section 4.13.33.3 (Limitations on Prior Authorization Requirements). If the MCO determines that a Member meets the Pharmacy Lock-In criteria, the MCO shall be responsible for all communications to Members regarding the Pharmacy Lock-In determination. The MCO may, provided the MCO receives prior approval from the Department, implement Lock-In Programs for other medical services.
- 4.2.4.11 Members shall not be required to change covered prescription drugs more than once per calendar year, with the following exceptions:
  - 4.2.4.11.1 When a Member is new to Medicaid, or switches from one Medicaid MCO to another Medicaid MCO:
  - 4.2.4.11.2 When a covered prescription drug change is initiated by the Member's provider;
  - 4.2.4.11.3 When a biosimilar becomes available to the market;
  - 4.2.4.11.4 When FDA boxed warnings or new clinical guidelines are recognized by CMS;
  - 4.2.4.11.5 When a covered prescription drug is withdrawn from the market because it has been found to be unsafe or removed for another reason; and
  - 4.2.4.11.6 When a covered prescription is not available due to a supply shortage.

#### 4.2.5 Pharmacy Systems, Data, and Reporting Requirements

- 4.2.5.1 Systems Requirements
  - 4.2.5.1.1 The MCO shall adjudicate pharmacy claims for its Members using a POS system where appropriate. System modifications include, but are not limited to:
    - 4.2.5.1.1.1. Systems maintenance,
    - 4.2.5.1.1.2. Software upgrades, and
    - 4.2.5.1.1.3. National Drug Code sets, or migrations to new versions of National Council for Prescription Drug Programs (NCPDP).
  - 4.2.5.1.2 Transactions shall be updated and maintained to current industry standards. The MCO shall provide an automated determination during the POS transaction; in accordance with NCPDP mandated response times within an average of less than or equal to three (3) seconds.

Page 97 of 397	Date
----------------	------



## **Exhibit B - DRAFT**

EXHIBIT B - DRAFT			
4.2.5.2	Pharmacy Data and Reporting Requirements		
	4.2.5.2.1	PDL, the M	ate its compliance with the Department CO shall submit to the Department egarding its PDL compliance rate.
	4.2.5.2.2	processes in shall be resprebates, inc	ce with changes to rebate collection the Affordable Care Act, the Department consible for collecting OBRA 90 CMS lusive of supplemental, from drug s on MCO pharmacy claims.
	4.2.5.2.3	Encounter Dabilling proce Encounter D days of clair submission	hall provide all necessary pharmacy ata to the State to support the rebate ss and the MCO shall submit the ata file within fourteen (14) calendar in payment. The Encounter Data and shall conform to all requirements Section 5.1.3 (Encounter Data) of this
	4.2.5.2.4		tilization information reported to the shall, at a minimum, include information
	4		The total number of units of each dosage form,
	4	.2.5.2.4.2.	Strength, and
	4		Package size by National Drug Code of each covered outpatient drug dispensed, per Department encounter specifications. [42 CFR 438.3(s)(2); Section 1927(b) of the Social Security Act]
	4.2.5.2.5	The MCO shall establish procedures to exclude utilization data for covered outpatient drugs that are subject to discounts under the 340B Drug Pricing Program from drug utilization reports provided to the Department. [42 CFR 438.3(s)(3)]	
	4.2.5.2.6		all implement a mechanism to prevent scounts in the 340B Drug Pricing
	4.2.5.2.7	ensure that a and supplem benefit admin	all work cooperatively with the State to II data needed for the collection of CMS ental rebates by the State's pharmacy istrator is delivered in a comprehensive anner, inclusive of any payments made

Page 98 of 397 Date \_\_\_\_\_



### **Exhibit B - DRAFT**

for Members for medications covered by othe payers.
4.2.5.2.8 The MCO shall adhere to federal regulations with respect to providing pharmacy data required for the Department to complete and submit to CMS the Annual Medicaid DUR Report. [42 CFF 438.3(s)(4),(5)]
4.2.5.2.9 The MCO shall provide the Department reporting regarding pharmacy utilization, polypharmacy authorizations and the Pharmacy Lock-In Program medication management, and safety monitoring of psychotropics in accordance with Exhibit O: Quality and Oversight Reporting Requirements.
4.2.5.2.10 The MCO shall provide to the Department a detailed plan describing the exchange of Member pharmacy and medical record information between the PCP behavioral health Provider, and other appropriate parties for the purpose of medication management. This information shall be provided in a manner prescribed by the Department as permitted by State and federal law.
4.2.5.2.10.1. All Member medical records and othe medication management information exchanged between parties shall be shared with the Member's PCP in an easily identifiable format.
4.2.5.2.10.2. The MCO shall retain oversight and accountability of the medication management program, including data exchanges between parties.
4.2.5.2.10.3. The MCO shall submit its medication management plan for the Department's review and authorization at time or readiness, and prior to implementation when changes to the MCO's medication management program are proposed.

#### **Medication Management** 4.2.6

- 4.2.6.1 Medication Management for All Members
  - 4.2.6.1.1 Polypharmacy criteria for Members are defined as follows:
    - Child Members dispensed four (4) or 4.2.6.1.1.1. more maintenance drugs based on GPI

Page 99 of 397 Date \_\_\_\_\_

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

10 or an equivalent product identification code (such as HICL) over a rolling sixty (60) day period, each drug filled for at least ninety (90) days duration, allowing each drug up to one fifteen (15) day gap between fills;

- 4.2.6.1.1.2. Adult Members dispensed five (5) or more maintenance drugs based on Generic Product Identifier (GPI) 10 or an equivalent product identification code (such as HICL) over a rolling sixty (60) day period; and
- 4.2.6.1.1.3. Brand and equivalent generics (or similar relationship such as reference product and biosimilar) within same GPI or equivalent product identification code shall not be counted as separate drugs within the five (5) maintenance drugs.
- 4.2.6.2 The MCO shall support medication management for Members meeting Polypharmacy criteria, and for other Members requesting medication review to ensure the PCP, pharmacist, or other qualified health care individual pharmacist has the information necessary to conduct Polypharmacy and medication management for child/adolescent and adult Members.
- 4.2.6.3 Comprehensive Medication Review (CMR) is defined as a systematic process of collecting patient-specific information, assessing medication therapies to identify medication-related problems, developing a prioritized list of medication-related problems, and creating a plan to resolve them with the patient, caregiver and/or prescriber. This systematic process shall be used for each CMR.
  - 4.2.6.3.1 The MCO is responsible to ensure that a Member receives at least one Comprehensive Medication Review (CMR) within six (6) months from the date/quarter in which the Member was identified as meeting Polypharmacy criteria.
  - 4.2.6.3.2 The PCP, pharmacist, or other qualified individual shall participate in Polypharmacy and medication management.
  - 4.2.6.3.3 The PCP, pharmacist or other qualified individual shall provide counseling with any Member or authorized representative upon request, as described

age 100 of 397	Date	

New Hampshire Department of Health and Human Services Medicaid Care Management Services



## **Exhibit B - DRAFT**

in this section, and in Exhibit O: Quality Oversight Reporting Requirements.

- 4.2.6.3.4 The MCO shall report to the Department on a quarterly basis the total number of CMRs completed, including total number of counselling interactions with any Member, the Member names, and Provider (PCP, pharmacist, or other qualified health care provider) who performed the CMR and/or counselling interaction with the Member or authorized representative.
- 4.2.6.3.5 The related CMR counseling is an interactive personto-person, telephonic, or telehealth consultation conducted in real-time between the Member, authorized representative, and the PCP, pharmacist and/or other qualified individual with the intent to improve a Member's knowledge of their prescriptions, over-the-counter medications, herbal therapies, and dietary supplements; identify, and address problems or concerns the patient may have; and empower them to self-manage their medications and health conditions. These items shall be addressed for each Member during each CMR counselling interaction.
- 4.2.6.3.6 In the event a Member identified for Polypharmacy does not participate in such review offered by a PCP, pharmacist, or other qualified individual at least once annually, the MCO shall offer CMR and counseling at least monthly until the Member actively accepts or denies receipt of CMR services.
  - 4.2.6.3.6.1. When the Member does not engage with the PCP, pharmacist, or other qualified individual for the purpose of satisfying medication management requirements of this Agreement, the MCO may subcontract with an appropriately credentialed and licensed professional or entity to support such engagement with prior approval from the Department.
- 4.2.6.4 The MCO shall routinely monitor and address the appropriate use of behavioral health medications in children by encouraging the use of, and reimbursing for consultations with, child psychiatrists.
- 4.2.6.5 The MCO shall provide to the qualified individual conducting CMR contact information for at least five (5) in-network child/adolescent psychiatrists for the purpose of peer-to-peer consulting whenever a child/adolescent Member is identified for Polypharmacy and is

Page 101 of 397	Date	

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

prescribed behavioral health prescriptions. The qualified individual shall seek counseling for at least two (2) of the four (4) prescriptions identified for Polypharmacy.

4.2.6.6 The MCO shall monitor Members who meet criteria for Polypharmacy three (3), six (6), and twelve (12) months after the CMR is completed to see if the member continues to meet criteria for Polypharmacy, or if it has been resolved. The MCO shall report the number of members who continue to meet criteria for Polypharmacy, and the number of members who no longer meet criteria on a quarterly basis.

#### 4.2.7 Medication Management for Children with Special Health Care Needs

- 4.2.7.1 The MCO shall be responsible for active and comprehensive medication management for Children with Special Health Care Needs. The MCO shall offer to Members, their parents, and/or caregivers, comprehensive medication management services for Children with Special Health Care Needs. If comprehensive medication management services are accepted, the MCO shall develop active and comprehensive medication management protocols for Children with Special Health Care Needs that shall include, but not be limited to, the following:
  - 4.2.7.1.1 Performing or obtaining necessary health assessments;
  - 4.2.7.1.2 Formulating a medication treatment plan according to therapeutic goals agreed upon by the prescriber and the Member, parent and/or caregiver;
  - 4.2.7.1.3 Selecting, initiating, modifying, recommending changes to, or administering medication therapy:
  - 4.2.7.1.4 Monitoring, which could include lab assessments and evaluating the Member's response to therapy:
  - 4.2.7.1.5 Consulting with social service agencies on medication management services;
  - 4.2.7.1.6 Initial and on-going CMR to prevent medicationrelated problems and address drug reconciliation, including adverse drug events, followed by targeted medication reviews;
  - 4.2.7.1.7 Documenting and communicating information about care delivered to other appropriate health care Providers;
  - 4.2.7.1.8 Member education to enhance understanding and appropriate use of medications; and

Page 102 of 397	Date
-----------------	------

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

4.2.7.1.9	Coordination and integration of medication therapy
	management services with broader health Care
	Management services to ensure access to Medically
	Necessary medications wherever Member is placed,
	including access to out of network pharmacies.

- 4.2.7.2 Review of medication use shall be based on the following:
  - 4.2.7.2.1 Pharmacy claims:
  - 4.2.7.2.2 Provider progress reports;
  - 4.2.7.2.3 Comprehensive Assessments and Care Plans;
  - 4.2.7.2.4 Contact with the Member's Providers;
  - 4.2.7.2.5 Current diagnoses;
  - 4.2.7.2.6 Current behavioral health functioning;
  - 4.2.7.2.7 Information from the family, Provider, the Department, and residential or other treatment entities or Providers; and
  - 4.2.7.2.8 Information shared with DCYF around monitoring and managing the use of psychotropic medications for children in State custody/guardianship, to the extent permissible by State and federal law.

#### 4.3 Member Enrollment and Disenrollment

- 4.3.1 Eligibility
  - 4.3.1.1 The Department has sole authority to determine whether an individual meets the eligibility criteria for Medicaid as well as whether the individual shall be enrolled in the MCM program. The MCO shall comply with eligibility decisions made by the Department.
  - 4.3.1.2 The MCO and its Subcontractors shall ensure that ninety-nine percent (99%) of transfers of eligibility files are incorporated and updated within one (1) business day after successful receipt of data. The MCO shall make the Department aware, within one (1) business day, of unsuccessful uploads that go beyond twenty-four (24) hours.
  - 4.3.1.3 The Accredited Standards Committee (ASC) X12 834 enrollment file shall limit enrollment history to eligibility spans reflective of any assignment of the Member with the MCO.
  - 4.3.1.4 To ensure appropriate Continuity of Care, the Department shall provide up to six (6) months (as available) of all FFS paid claims history including: medical, pharmacy, behavioral health and LTSS claims history data for all FFS Medicaid Members assigned to the

Page 103 of 397	Date	

New Hampshire Department of Health and Human Services Medicaid Care Management Services



#### **Exhibit B - DRAFT**

- MCO. For Members transitioning from another MCO, the Department shall also provide such claims Confidential Data as well as available encounter information regarding the Member supplied by other Medicaid MCOs, as applicable.
- 4.3.1.5 The MCO shall notify the Department within five (5) business days when it identifies information in a Member's circumstances that may affect the Member's eligibility, including changes in the Member's residence, such as out-of-state claims, or the death of the Member. [42 CFR 438.608(a)(3)]
- 4.3.1.6 In accordance with separate guidance, the MCO shall outreach to Members forty-five (45) calendar days prior to each Member's Medicaid eligibility expiration date to assist the Member with completion and submission of required paperwork. The MCO shall submit their outbound call protocols for the Department's review during the Readiness Review process.
  - The MCO shall not conduct outreach to address the 4.3.1.6.1 backlog of pending Medicaid eligibility cases to Members in a manner that would constitute a violation of federal law, including, but not limited to. the Americans with Disabilities Act of 1990 (ADA), Title VI of the Civil Rights Act of 1964, Section 504 of the Rehabilitation Act of 1973 (Section 504), the Age Discrimination Act of 1975, and Section 1557 of the Affordable Care Act (Section 1557). Further, compliance with these laws includes providing reasonable accommodations to individuals with disabilities under the ADA, Section 504, and Section 1557. with eligibility and documentation requirements, understanding program rules and notices, to ensure they understand program rules and notices, as well as meeting other program requirements necessary to obtain and maintain benefits. [CMS State Health Official Letter].

#### 4.3.2 Enrollment

- 4.3.2.1 Pursuant to 42 CFR 438.54, Members who do not select an MCO as part of the Medicaid application process shall be auto-assigned to an MCO. All newly eligible Medicaid Members shall be given ninety (90) calendar days to either remain in the assigned MCO or select another MCO, if they choose. Members may not change from one (1) MCO to another outside the ninety (90) day plan selection period unless they meet the "cause" criteria as described in Section 4.3.5 (Disenrollment) of this Agreement.
- 4.3.2.2 The MCO shall accept all Members who are assigned to the MCO by the Department. The MCO shall accept for automatic re-

Page 104 of 397	Date	

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

- enrollment Members who were disenrolled due to a loss of Medicaid eligibility for a period of two (2) months or less. [42 CFR 438.56(g)]
- 4.3.2.3 The MCO shall permit each Member to choose a PCP to the extent possible and appropriate. [42 CFR 438.3(I)] In instances in which the Member does not select a PCP at the time of enrollment, the MCO shall assign a PCP to the Member.
- 4.3.2.4 When assigning a PCP, the MCO shall include the following methodology in selecting a PCP for the Member, if information is available: Member claims history; family member's Provider assignment and/or claims history; geographic proximity; special medical needs; and language/cultural preference.

#### 4.3.3 Non-Discrimination

- 4.3.3.1 The MCO shall accept new enrollment from individuals in the order in which they apply, without restriction, unless authorized by CMS. [42 CFR 438.3(d)(1)]
- 4.3.3.2 The MCO shall not discriminate against eligible persons or Members on the basis of their health or mental health history, health or mental health status, their need for health care services, amount payable to the MCO on the basis of the eligible person's actuarial class, or pre-existing medical/health conditions. [42 CFR 438.3(d)(3)]
- 4.3.3.3 The MCO shall not discriminate in enrollment, disenrollment, and re-enrollment against individuals on the basis of health status or need for health care services. [42 CFR 438.3(q)(4)]
- 4.3.3.4 The MCO shall not discriminate against individuals eligible to enroll on the basis of race, color, national origin, sex, sexual orientation, gender identity, or disability and shall not use any policy or practice that has a discriminatory effect. [42 CFR 438.3(d)(4)] [RSA 354-A]

## 4.3.4 Auto-Assignment

- 4.3.4.1 In its sole discretion, the Department shall use the following factors for auto-assignment in an order to be determined by the Department:
  - 4.3.4.1.1 Preference to an MCO with which there is already a family affiliation;
  - 4.3.4.1.2 Previous MCO enrollment, when applicable;
  - 4.3.4.1.3 Provider-Member relationship, to the extent obtainable and pursuant to 42 CFR 438 54(d)(7);
  - 4.3.4.1.4 Any members earned through the Performance-Based Auto-Assignment Program; and

Page 105 of 397	Date
age 100 01 007	Date

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



### **Exhibit B - DRAFT**

- 4.3.4.1.5 Equitable distribution among the MCOs as determined appropriate solely by the Department.
- 4.3.4.2 The Department's Performance-Based Auto-Assignment Program rewards one or more MCOs that demonstrate exceptional performance on one or more key dimensions of performance determined at the Department's sole discretion.
  - 4.3.4.2.1 High-performing MCO(s) may be rewarded with preferential auto-assigned membership in accordance with separate guidance. Such an award would potentially precede any equitable distribution of Members who do not self-select an MCO across.

#### 4.3.5 Disenrollment

- 4.3.5.1 Member Disenrollment Request
  - 4.3.5.1.1 A Member may request disenrollment "with cause" to the Department at any time during the coverage year when:
    - 4.3.5.1.1.1. The Member moves out of state;
    - 4.3.5.1.1.2. The Member needs related services to be performed at the same time; not all related services are available within the network; and receiving the services separately would subject the Member to unnecessary risk;
    - 4.3.5.1.1.3. Other reasons, including but not limited to poor quality of care, lack of access to services covered under the Agreement, violation of rights, or lack of access to Providers experienced in dealing with the Member's health care needs. [42 CFR 438.56(d)(2)]; or
    - 4.3.5.1.1.4. When the MCO does not cover the service the Member seeks because of moral or religious objections. [42 CFR 438.56(d)(2)(i-ii)].
  - 4.3.5.1.2 For Member disenrollment requests "with cause" as described in in this section of the Agreement, the Member shall first seek redress through the MCO's grievance system.
  - 4.3.5.1.3 A Member may request disenrollment "without cause" at the following times:

age 106 of 397	Date



# **Exhibit B - DRAFT**

4.	3.5.1.3.1.	During the ninety (90) calendar days following the date of the Member's initial enrollment into the MCO or the date of the Department Member notice of the initial auto-assignment/enrollment, whichever is later;
4.	3.5.1.3.2.	When Members have an established relationship with a PCP that is only in the network of a non-assigned MCO, the Member can request disenrollment during the first twelve (12) months of enrollment at any time and enroll in the non-assigned MCO;
4.	3.5.1.3.3.	Once every twelve (12) months;
4.	3.5.1.3.4.	During enrollment related to renegotiation and re-procurement;
4.	3.5.1.3.5.	For sixty (60) calendar days following an automatic re-enrollment if the temporary loss of Medicaid eligibility has caused the Member to miss the annual enrollment/disenrollment opportunity (this provision applies to redeterminations only and does not apply when a Member is completing a new application for Medicaid eligibility); and
4.	3.5.1.3.6.	When the Department imposes a sanction on the MCO. [42 CFR $438.3(q)(5)$ ; 42 CFR $438.56(c)(1)$ ; 42 CFR $438.56(c)(2)(i-iii)$ ]
4.3.5.1.4	rights at leas of each re-e an explanat	shall provide Members and their ves with written notice of disenrollment at sixty (60) calendar days before the start nrollment period. The notice shall include ion of all of the Member's disenrollment specified in this Agreement. [42 CFR
4.3.5.1.5	(or their aut	is requesting disenrollment, the Member thorized representative) shall submit an en request to the Department. [42 CFR )]
4.3.5.1.6	The MCO sl Department	hall furnish all relevant information to the for its determination regarding

Date \_\_\_\_ Page 107 of 397

4.3.5.2



## **Exhibit B - DRAFT**

		nt, within three (3) business days after e Department's request for information.
4.3.5.1.7	effective da no later th	of the reason for disenrollment, the te of an approved disenrollment shall be an the first day of the second month e month in which the Member files the
4.3.5.1.8	determination disenrollme	artment fails to make a disenrollment on within this specified timeframe, the nt is considered approved. [42 CFR 42 CFR 438.56(d)(3); 42 CFR 438.3(q); 42 CFR 4
MCO Dise	nrollment Requ	est
4.3.5.2.1	requests	shall submit involuntary disenrollment to the Department with proper ion for the following reasons:
	4.3.5.2.1.1.	Member has established out of state residence;
	4.3.5.2.1.2.	Member death;
	4.3.5.2.1.3.	Determination that the Member is ineligible for enrollment due to being deemed part of an excluded population;
	4.3.5.2.1.4.	Fraudulent use of the Member identification card; or
	4.3.5.2.1.5.	In the event of a Member's threatening or abusive behavior that jeopardizes the health or safety of Members, staff, or Providers. [42 CFR 438.56(b)(1); 42 CFR 438.56(b)(3)]
4.3.5.2.2	The MCO sl	nall not request disenrollment because of:
	4.3.5.2.2.1.	An adverse change in the Member's health status;
	4.3.5.2.2.2.	The Member's utilization of medical services;
	4.3.5.2.2.3.	The Member's diminished mental capacity;
	4.3.5.2.2.4.	The Member's uncooperative or disruptive behavior resulting from their special needs (except when their continued enrollment in the MCO

Date \_\_\_\_\_ Page 108 of 397

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

seriously impairs the entity's ability to furnish services to either the particular Member or other Members); or

- 4.3.5.2.2.5. The Member's misuse of substances, prescribed or illicit, and any legal consequences resulting from substance misuse. [Section 1903(m)(2)(A)(v) of the Social Security Act; 42 CFR 438.56(b)(2)]
- 4.3.5.2.3 If an MCO is requesting disenrollment of a Member, the MCO shall:
  - 4.3.5.2.3.1. Specify the reasons for the requested disenrollment of the Member; and
  - 4.3.5.2.3.2. Submit a request for involuntary disenrollment to the Department along with documentation and justification, for review.
    - 4.3.5.2.3.2.1 Regardless of the reason for disenrollment, the effective date of an approved disenrollment shall be no later than the first day of the second month following the month in which the MCO files the request.
    - 4.3.5.2.3.2.2 If the Department fails to make a disenrollment determination within this specified timeframe, the disenrollment is considered approved. [42 CFR 438.56(e)]

#### 4.4 Member Services

- 4.4.1 Member Information
  - 4.4.1.1 The MCO shall perform the Member Services responsibilities contained in this Agreement for all Members.
  - 4.4.1.2 PCP Information
    - 4.4.1.2.1 The MCO shall send a letter to a Member upon initial enrollment, and anytime the Member requests a new PCP, confirming the Member's PCP and providing the PCP's name, address, and telephone number.
  - 4.4.1.3 Member Identification Card

Page 109 of 397	Date	

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

- 4.4.1.3.1 The MCO shall issue a hardcopy identification card to all New Members within ten (10) calendar days following the MCO's receipt of a valid enrollment file from the Department, but no later than seven (7) calendar days after the effective date of enrollment.
- 4.4.1.3.2 The identification card shall include, but is not limited to, the following information and any additional information shall be approved by the Department prior to use on the identification card:
  - 4.4.1.3.2.1. The Member's name;
  - 4.4.1.3.2.2. The Member's Medicaid identification number assigned by the Department at the time of eligibility determination;
  - 4.4.1.3.2.3. The name of the MCO:
  - 4.4.1.3.2.4. The twenty-four (24) hours a day, seven (7) days a week toll-free Member Services telephone/hotline number operated by the MCO;
  - 4.4.1.3.2.5. The toll-free telephone number for transportation; and
  - 4.4.1.3.2.6. How to file an appeal or grievance.
- 4.4.1.3.3 The MCO shall reissue a Member identification card if:
  - 4.4.1.3.3.1. A Member reports a lost card;
  - 4.4.1.3.3.2. A Member has a name change; or
  - 4.4.1.3.3.3. Any other reason that results in a change to the information disclosed on the identification card.

#### 4.4.1.4 Member Handbook

- 4.4.1.4.1 The MCO shall publish and provide Member information in the form of a Member Handbook at the time of Member enrollment in the plan and, at a minimum, on an annual basis thereafter. The Member Handbook shall be based upon the model Member Handbook developed by the Department. [42 CFR 438.10(g)(1), 45 CFR 147.200(a); 42 CFR 438.10(c)(4)(ii)]
- 4.4.1.4.2 The MCO shall inform all Members by mail of their right to receive free of charge a written copy of the Member Handbook. The MCO shall provide program

Page 110 of 397	Date	
-----------------	------	--

New Hampshire Department of Health and Human Services Medicaid Care Management Services



## **Exhibit B - DRAFT**

content that is coordinated and collaborative with other Department initiatives. The MCO shall submit the Member Handbook to the Department for review at the time it is developed as part of Readiness Review and after any substantive revisions at least thirty (30) calendar days prior to the effective date of such change.

4.4.1.4.3 The Member Handbook shall be in easily understood language, and include, but not be limited to, the following information:

4.4.1.4.3.1. General information

4.4.1.4.3.2. A table of contents;

4.4.1.4.3.2.1 How to access Auxiliary Aids and services, including additional information in alternative formats or languages [42 CFR 438.10(g)(2)(xiii-xvi), 42 CFR 438.10(d)(5)(i-iii)];

4.4.1.4.3.2.2 The Department developed definitions, including but not limited to: appeal, Copayment, DME. Emergency Medical Condition, emergency medical transportation, emergency room care. Emergency Services. excluded services, grievance, habilitation services and devices, health insurance, home health care, hospice services, hospitalization, hospital outpatient Medically care. Non-Necessary, network, Participating Provider. Participating Provider, PCP. physician plan, services. preauthorization, premium, prescription drug coverage, prescription drugs, primary care physician, Provider, rehabilitation services and devices, skilled nursing care. specialist: and [42 urgent **CFR** care 438.10(c)(4)(i)];



**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



### **Exhibit B - DRAFT**

- 4.4.1.4.3.2.3 The medical necessity definitions used in determining whether services will be covered:
- 4.4.1.4.3.2.4 A reminder to report to the Department any change of address, as Members may be liable for premium payments paid during period of ineligibility;
- 4.4.1.4.3.2.5 Information and guidance as to how the Member can effectively use the managed care program [42 CFR 438.10(q)(2)];
- 4.4.1.4.3.3. Appointment procedures;
  - 4.4.1.4.3.3.1 How to contact Service Link
    Aging and Disability Resource
    Center and the Department's
    Medicaid Service Center that can
    provide all Members and
    potential Members choice
    counseling and information on
    managed care;
  - 4.4.1.4.3.3.2 Notice of all appropriate mailing addresses and telephone numbers to be utilized Members seeking information or authorization. including MCO's toll-free telephone line and website. the toll-free telephone number for Member Services, the toll-free telephone number for Medical Management, and the toll-free telephone number for any other unit providing services directly to Members **CFR** [42 438.10(g)(2)(xiii-xvi);
- 4.4.1.4.4 How to access the NH DHHS Office of the Ombudsman and the NH Office of the Long Term Care Ombudsman;
- 4.4.1.4.5 The policies and procedures for disenrollment;
  - 4.4.1.4.5.1. A description of the transition of care policies for potential Members and Members [42 CFR 438.62(b)(3)];

Page 112 of 397 Date \_\_\_\_\_



# **Exhibit B - DRAFT**

	4.4.1.4.5.2.	Cost-sharing requirements [42 CFR 438.10(g)(2)(viii)];
	4.4.1.4.5.3.	A description of utilization review policies and procedures used by the MCO;
	4.4.1.4.5.4.	A statement that additional information, including information on the structure and operation of the MCO plan and Physician Incentive Plans, shall be made available upon request [42 CFR 438.10(f)(3), 42 CFR 438.3(i)];
	4.4.1.4.5.5.	Information on how to report suspected Fraud or abuse [42 CFR 438.10(g)(2)(xiii-xvi)];
	4.4.1.4.5.6.	Information about the role of the PCP and information about choosing and changing a PCP [42 CFR $438.10(g)(2)(x)$ ];
	4.4.1.4.5.7.	Non-Participating Providers and cost- sharing on any benefits carved out and provided by the Department [42 CFR 438.10(g)(2)(i-ii)];
	4.4.1.4.5.8.	How to exercise Advance Directives [42 CFR 438.10(g)(2)(xii), 42 CFR 438.3(j)];
	4.4.1.4.5.9.	Advance Directive policies which include a description of current State law. [42 CFR 438.3(j)(3)];
	4.4.1.4.5.10.	Information on the parity compliance process, including the appropriate contact information, as required by Section 4.13.18. (Parity);
	4.4.1.4.5.11.	Any restrictions on the Member's freedom of choice among Participating Providers. [42 CFR 438.10(g)(2)(vi-vii)]
4.4.1.4.6	Benefits:	
	4.4.1.4.6.1.	How and where to access any benefits provided, including Maternity services, Family Planning Services and NEMT services [42 CFR 438.10(g)(2)(i-ii), (vi-

Date \_\_\_\_\_ Page 113 of 397

vii)];



# **Exhibit B - DRAFT**

4.4.1.4.6.2.	Detailed information regarding the amount, duration, and scope of all available benefits so that Members understand the benefits to which they are entitled [42 CFR 438.10(g)(2)(iii-iv)];
4.4.1.4.6.3.	How to access EPSDT services and component services if Members under age twenty-one (21) entitled to the EPSDT benefit are enrolled in the MCO;
4.4.1.4.6.4.	How and where to access EPSDT benefits delivered outside the MCO, if any [42 CFR 438.10(g)(2)(i-ii)];
4.4.1.4.6.5.	How transportation is provided for any benefits carved out of this Agreement and provided by the Department [42 CFR 438.10(g)(2)(i-ii)];
4.4.1.4.6.6.	Information explaining that, in the case of a counseling or referral service that the MCO does not cover because of moral or religious objections, the MCO shall inform Members that the service is not covered and how Members can obtain information from the Department about how to access those services [42 CFR 438.10(g)(2)(ii)(A-B), 42 CFR 438.102(b)(2)];
4.4.1.4.6.7.	A description of pharmacy policies and pharmacy programs; and
4.4.1.4.6.8.	How emergency care is provided, including:
	4.4.1.4.6.8.1 The extent to which, and how, after hours and emergency coverage are provided;
	4.4.1.4.6.8.2 What constitutes an Emergency Service and an Emergency Medical Condition; The extent to which, and how, after hours and emergency coverage are provided;
	4.4.1.4.6.8.3 The fact that Prior

Date \_\_\_\_\_ Page 114 of 397

Authorization is not required for

Emergency Services; and

New Hampshire Department of Health and Human Services Medicaid Care Management Services



### **Exhibit B - DRAFT**

4.4.1.4.6.8.4 The Member's right to use a hospital or any other setting for emergency care. [42 CFR 438.10(q)(2)(v)]

#### 4.4.1.4.6.9. Service Limitations:

- 4.4.1.4.6.9.1 An explanation of any service limitations or exclusions from coverage;
- 4.4.1.4.6.9.2 An explanation that the MCO cannot require a Member to receive prior approval prior to choosing a family planning Provider [42 CFR 438.10(g)(2)(vii)];
- 4.4.1.4.6.9.3 A description of all precertification, Prior Authorization criteria, or other requirements for treatments and services:
- 4.4.1.4.6.9.4 Information regarding Prior Authorization in the event the Member chooses to transfer to another MCO and the Member's right to continue to utilize a Provider specified in a Prior Authorization for a period of time regardless of whether the Provider is participating in the MCO network;
- 4.4.1.4.6.9.5 The policy on referrals for specialty care and for other Covered Services not furnished by the Member's PCP [42 CFR 438.10(g)(2)(iii-iv)];
- 4.4.1.4.6.9.6 Information on how to obtain services when the Member is out-of-state and for after-hours coverage [42 CFR 438.10(g)(2)(v)]; and
- 4.4.1.4.6.9.7 A notice stating that the MCO shall be liable only for those services authorized by or required of the MCO.

Page 115 of 397 Date \_\_\_\_\_

New Hampshire Department of Health and Human Services Medicaid Care Management Services



### **Exhibit B - DRAFT**

- 4.4.1.4.6.10. Rights and Responsibilities:
  - 4.4.1.4.6.10.1Member rights and protections, outlined in Section 4.4.3 (Member Rights), including the Member's right to obtain available and accessible health care services covered under the MCO. [42 CFR 438.100(b)(2)(i-vi), 42 CFR 438.10(g)(2)(ix), 42 CFR 438.10(g)(2)(ix), 42 CFR 438.100(b)(3)]
- 4.4.1.4.6.11. Grievances, Appeals, and Fair Hearings Procedures and Timeframes:
  - 4.4.1.4.6.11.1The right to file grievances and appeals;
  - 4.4.1.4.6.11.2The requirements and timeframes for filing grievances or appeals;
  - 4.4.1.4.6.11.3The availability of assistance in the filing process for grievances and appeals;
  - 4.4.1.4.6.11.4The right to request a State fair hearing after the MCO has made a determination on a Member's appeal which is adverse to the Member; and
  - 4.4.1.4.6.11.5 The right to have benefits continue pending the appeal or request for State fair hearing if the decision involves the reduction or termination of benefits, however, if the Member receives an adverse decision then the Member may be required to pay for the cost of service(s) furnished while the appeal or State fair hearing is pending. [42 CFR 438.10(q)(2)(xi)(A-E)]
- 4.4.1.4.7 Member Handbook Dissemination
  - 4.4.1.4.7.1. The MCO shall post on its website and advise the Member within ten (10) calendar days following the MCO's

Page 116 of 397 Date \_\_\_\_\_

New Hampshire Department of Health and Human Services Medicaid Care Management Services



## **Exhibit B - DRAFT**

receipt of a valid enrollment file from the Department, but no later than seven (7) calendar days after the effective date of enrollment in paper or electronic form that the Member Handbook is available on the Internet and includes the applicable Internet address, provided that enrollees with disabilities who cannot access this information online are provided auxiliary aids and services upon request at no cost. [42 CFR 438.10(q)(3)(i-iv)]

4.4.1.4.7.2.

The MCO may provide the information by any other method that can reasonably be expected to result in the Member receiving that information. The MCO shall provide the Member Handbook information by email after obtaining the Member's agreement to receive the information electronically. [42 CFR 438.10(g)(3)(i-iv)]

4.4.1.4.7.3.

The MCO shall notify all Members, at least once a year, of their right to obtain a Member Handbook and shall maintain consistent and up-to-date information on the MCO's website. [42 CFR 438.10(g)(3)(i) - (iv)] The Member information appearing on the website (also available in paper form) shall include the following, at a minimum:

- 4.4.1.4.7.3.1 Information contained in the Member Handbook;
- 4.4.1.4.7.3.2 Information on how to file grievances and appeals;
- 4.4.1.4.7.3.3 Information on the MCO's Provider network for all Provider types covered under this PCPs. Agreement (e.g., specialists, family planning Providers, pharmacies, FQHCs, RHCs. hospitals, and mental health and Substance Use Disorder Providers):

Page 117 of 397 Date \_\_\_\_\_

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



### **Exhibit B - DRAFT**

- 4.4.1.4.7.3.3.1 Names and any group affiliations;
- 4.4.1.4.7.3.3.2 Street addresses;
- 4.4.1.4.7.3.3.3 Office hours;
- 4.4.1.4.7.3.3.4 Telephone numbers;
- 4.4.1.4.7.3.3.5 Website (whenever web presence exists);
- 4.4.1.4.7.3.3.6 Specialty (if any),
- 4.4.1.4.7.3.3.7 Description of accommodations offered for people with disabilities;
- 4.4.1.4.7.3.3.8 The cultural and linguistic capabilities of Participating Providers, including languages (including American Sign Language (ASL)) offered by the Provider or a skilled medical interpreter at the Provider's office:
- 4.4.1.4.7.3.3.9 Gender of the Provider;
- 4.4.1.4.7.3.3.10 Identification of Providers that are not accepting new Members; and
- 4.4.1.4.7.3.3.11 Any restrictions on the Member's freedom of choice among Participating Providers. [42 CFR 438.10(g)(2)(vi-vii)]
- 4.4.1.4.7.4. The MCO shall produce a revised Member Handbook, or an insert, informing Members of changes to Covered Services, upon Department's notification of any change in Covered Services, and at least thirty (30) calendar days prior to the effective date of such change. This includes notification of any policy to discontinue coverage of a counseling or referral service based on moral or religious objections and how the Member can access those services. [42 CFR 438.102(b)(1)(i)(B); 42 **CFR** 438.10(g)(4)]
- 4.4.1.4.7.5. The MCO shall use Member notices, as applicable, in accordance with the model notices developed by the Department. [42 CFR 438.10(c)(4)(ii)] For any change that affects Member rights, filing

Page 118 of 397 Date \_\_\_\_\_

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



### **Exhibit B - DRAFT**

requirements, time frames for grievances, appeals, and State fair hearings, availability of assistance in submitting grievances and appeals, and toll-free numbers of the MCO grievance system resources, the MCO shall give each Member written notice of the change at least thirty (30) calendar days before the intended effective date of the change. The MCO shall utilize notices that describe transition of care policies for Members and potential Members, [42] CFR 438.62(b)(3)]

### 4.4.1.5 Provider Directory

- 4.4.1.5.1 The MCO shall publish a Provider Directory that shall be reviewed by the Department prior to initial publication and distribution. The MCO shall submit the draft Provider Directory and all substantive changes to the Department for review.
- 4.4.1.5.2 The following information shall be in the MCO's Provider Directory for all Participating Provider types covered under this Agreement (e.g., PCPs, specialists, family planning Providers, pharmacies, FQHCs, RHCs, hospitals, and mental health and Substance Use Disorder Providers, FQHCs, RHCs):
  - 4.4.1.5.2.1. Names and any group affiliations;
  - 4.4.1.5.2.2. Street addresses;
  - 4.4.1.5.2.3. Office hours;
  - 4.4.1.5.2.4. Telephone numbers;
  - 4.4.1.5.2.5. Website (whenever web presence exists);
  - 4.4.1.5.2.6. Specialty (if any),
  - 4.4.1.5.2.7. Gender;
  - 4.4.1.5.2.8. Description of accommodations offered for people with disabilities;
  - 4.4.1.5.2.9. The cultural and linguistic capabilities of Participating Providers, including languages (including ASL) offered by the Participating Provider or a skilled medical interpreter at the Provider's office:

Page 119 of 397 Date \_\_\_\_\_

4.4.1.5.8

Medicaid Care Management Services Contract
New Hampshire Department of Health and Human Services
Medicaid Care Management Services



# **Exhibit B - DRAFT**

	EXTINUIT B - BIVA	<u></u>
	4.4.1.5.2.10.	Hospital affiliations (if applicable);
	4.4.1.5.2.11.	Board certification (if applicable);
	4.4.1.5.2.12.	Identification of Participating Providers that are not accepting new patients; and
	4.4.1.5.2.13.	Any restrictions on the Member's freedom of choice among Participating Providers. [42 CFR 438.10(h)(1)(i-viii); 42 CFR 438.10(h)(2)]
4.4.1.5.3	ten (10) cald a valid enro later than so date of er Provider D informing th	chall send a letter to New Members within endar days following the MCO's receipt of collment file from the Department, but no even (7) calendar days after the effective prollment directing the Member to the Directory on the MCO's website and the Member of the right to a printed version rider Directory upon request and free of
4.4.1.5.4	Members a described in	shall disseminate Practice Guidelines to and potential Members upon request as a Section 4.8.2 (Practice Guidelines and [42 CFR 438.236(c)]
4.4.1.5.5	year, of the Provider Dil up-to-date	shall notify all Members, at least once a eir right to obtain a paper copy of the rectory and shall maintain consistent and information on the MCO's website in a eadable file and format as specified by
4.4.1.5.6	Directory at a mobile-er the MCO h directory; at later than t	hall update the paper copy of the Provider least monthly if the MCO does not have habled electronic directory, or quarterly, if as a mobile-enabled, electronic provider and shall update an electronic directory no hirty (30) calendar days after the MCO pdated provider information. [42 CFR e-4)]
4.4.1.5.7	of all Partic shall be se	shall post on its website a searchable list ipating Providers. At a minimum, this list earchable by Provider name, specialty, and whether the Provider is accepting new

Page 120 of 397 Date \_\_\_\_\_

The MCO shall update the Provider Directory on its website within seven (7) calendar days of any

New Hampshire Department of Health and Human Services Medicaid Care Management Services



## **Exhibit B - DRAFT**

changes. The MCO shall maintain an updated list of Participating Providers on its website in a Provider Directory.

- 4.4.1.5.9 Thirty (30) calendar days after the effective date of this Agreement or ninety (90) calendar days prior to the Program Start Date, whichever is later, the MCO shall develop and submit the draft website Provider Directory template to the Department for review; thirty (30) calendar days prior to Program Start Date the MCO shall submit the final website Provider Directory.
- 4.4.1.5.10 Upon the termination of a Participating Provider, the MCO shall make good faith efforts within fifteen (15) calendar days of the notice of termination to notify Members who received their primary care from, or was seen on a regular basis by, the terminated Provider. [42 CFR 438.10(f)(1)]

## 4.4.2 Language and Format of Member Information

- 4.4.2.1 The MCO shall have in place mechanisms to help potential Members and Members understand the requirements and benefits of the MCO. [42 CFR 438.10(c)(7)]
- 4.4.2.2 The MCO shall use the Department developed definitions consistently in any form of Member communication. The MCO shall develop Member materials utilizing readability principles appropriate for the population served.
- 4.4.2.3 The MCO shall provide all enrollment notices, information materials, and instructional materials relating to Members and potential Members in a manner and format that may be easily understood and readily accessible in a font size no smaller than twelve (12) point. [42 CFR 438.10(c)(1), 42 CFR 438.10(d)(6)(i-iii)]
- 4.4.2.4 The MCO's written materials shall be developed in compliance with all applicable communication access requirements at the request of the Member or prospective Member at no cost.
- 4.4.2.5 Information shall be communicated in an easily understood language and format, including alternative formats and in an appropriate manner that takes into consideration the special needs of Members or potential Members with disabilities or LEP.
- 4.4.2.6 The MCO shall inform Members that information is available in alternative formats and how to access those formats. [42 CFR 438.10(d)(3), 42 CFR 438.10(d)(6)(i-iii)]

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



### **Exhibit B - DRAFT**

- 4.4.2.7 The MCO shall make all written Member information available in English, Spanish, and any other state-defined prevalent non-English languages of MCM Members. [42 CFR 438.10(d)(1)]
- 4.4.2.8 All written Member information critical to obtaining services for potential Members shall include at the bottom, taglines printed in a conspicuously visible font size, and in the non-English languages prevalent among Members, to explain the availability of written translation or oral interpretation, and include the toll-free and teletypewriter (TTY/TDD) telephone number of the MCO's Member Services Center. [42 CFR 438.10(d)(3)]
- 4.4.2.9 The large print tagline must be printed in a conspicuously visible font size, and shall include information on how to request Auxiliary Aids and services, including materials in alternative formats. Upon request, the MCO shall provide all written Member and potential enrollee critical to obtaining services information in large print with a font size no smaller than eighteen (18) point. [42 CFR 438.10(d)(2-3), 42 CFR 438.10(d)(6)(i-iii)]
- 4.4.2.10 Written Member information shall include at a minimum:
  - 4.4.2.10.1 Provider Directories;
  - 4.4.2.10.2 Member Handbooks;
  - 4.4.2.10.3 Appeal and grievance notices; and
  - 4.4.2.10.4 Denial and termination notices.
- 4.4.2.11 The MCO shall also make oral interpretation services available free of charge to Members and potential Members for MCO Covered Services. This applies to all non-English languages, not just those that the Department identifies as languages of other major population groups. Members shall not to be charged for interpretation services. [42 CFR 438.10(d)(4)]
- 4.4.2.12 The MCO shall notify Members that oral interpretation is available for any language and written information is available in languages prevalent among MCM Members; the MCO shall notify Members of how to access those services. [42 CFR 438.10(d)(4), 42 CFR 438.10(d)(5)(i-iii)]
- 4.4.2.13 The MCO shall provide Auxiliary Aids such as TTY/TDD and ASL interpreters free of charge to Members or potential Members who require these services. [42 CFR 438.10(d)(4)]
  - 4.4.2.13.1 The MCO shall take into consideration the special needs of Members or potential Members with disabilities or LEP. [42 CFR 438.10(d)(5)(i)-(iii)]

#### 4.4.3 Member Rights

	Page 122 of 397	Date
--	-----------------	------

New Hampshire Department of Health and Human Services Medicaid Care Management Services



## **Exhibit B - DRAFT**

4.4.3.1	Member Ha	nall have written policies which shall be included in the ndbook and posted on the MCO website regarding its, such that each Member is guaranteed the right to:
	4.4.3.1.1	Receive information on the MCM program and the MCO to which the Member is enrolled;
	4.4.3.1.2	Be treated with respect and with due consideration for their dignity and privacy and the confidentiality of their PHI and PI as safeguarded by State rules and State and federal laws;
	4.4.3.1.3	Receive information on available treatment options and alternatives, presented in a manner appropriate to the Member's condition and ability to understand;
	4.4.3.1.4	Participate in decisions regarding his/her health care, including the right to refuse treatment;
	4.4.3.1.5	Be free from any form of restraint or seclusion used as a means of coercion, discipline, convenience, or retaliation;
	4.4.3.1.6	Request and receive a copy of his/her medical records free of charge, and to request that they be amended or corrected;
	4.4.3.1.7	Request and receive any MCO's written Physician Incentive Plans;
4.4.3.2		fits, including Family Planning Services and supplies, articipating Providers;
	4.4.3.2.1	Request and receive a Second Opinion; and
	4.4.3.2.2	Exercise these rights without the MCO or its Participating Providers treating the Member adversely. [42 CFR 438.100(a)(1); 42 CFR 438.100(b)(2)(i)-(vi]); 42 CFR 438.100(c); 42 CFR 438.10(f)(3); 42 CFR 438.10(g)(2)(vi)-(vii); 42 CFR 438.10(g)(2)(ix); 42 CFR 438.3(i)]
Member (	Communicati	on Supports

# 4.4.4 Member Communication Supports

- 4.4.4.1 During the Readiness Review period, the MCO shall provide a blueprint of its website, including Member portal, for review by the Department.
- 4.4.4.2 Member Call Center
  - 4.4.4.2.1 The MCO shall operate a toll-free Member Call Center Monday through Friday, and be operational on all days the Department Customer Service Center is open.

Page 123 of 397 Date \_\_\_\_\_

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



### **Exhibit B - DRAFT**

- 4.4.4.2.2 The MCO shall ensure that the Member Call Center integrates support for physical and Behavioral Health Services including meeting the requirement that the MCO have a call line that is in compliance with requirements set forth in Section 4.4 (Member Services), works efficiently to resolve issues, and is adequately staffed with qualified personnel who are trained to accurately respond to Members. At a minimum, the Member Call Center shall be operational:
  - 4.4.4.2.2.1. Two (2) days per week: eight (8:00) am Eastern Standard Time (EST) to five (5:00) pm EST;
  - 4.4.4.2.2.2. Three (3) days per week: eight (8:00) am EST to eight (8:00) pm EST; and
  - 4.4.4.2.2.3. During major program transitions, additional hours and capacity shall be accommodated by the MCO.
- 4.4.4.2.3 The Member Call Center shall meet the following minimum standards, which the Department reserves the right to modify at any time:
  - 4.4.4.2.3.1. Call Abandonment Rate: Fewer than five percent (5%) of calls shall be abandoned:
  - 4.4.4.2.3.2. Average Speed of Answer: Eighty-five percent (85%) of calls shall be answered with live voice within thirty (30) seconds; and
  - 4.4.4.2.3.3. Voicemail or answering service messages shall be responded to no later than the next business day.
- 4.4.4.2.4 The MCO shall coordinate its Member Call Center with the Department Customer Service Center, and community-based and statewide crisis lines, and at a minimum, include the development of a warm transfer protocol for Members.

#### 4.4.4.3 Welcome Call

4.4.4.3.1 The MCO shall make a welcome call or an interactive voice recognition (IVR) call to each new Member within ninety (90) calendar days of the Member's enrollment in the MCO, and include a means for the

Page 124 of 397 Date \_\_\_\_\_



# **Exhibit B - DRAFT**

		to request immediate live MCO ive support during the welcome call.
4.4.4.3.2	communica	nce with applicable law, the MCO may te with Members by text, email, phone or or electronic communications.
4.4.4.3.3	The welcom	ne call shall, at a minimum:
	4.4.4.3.3.1.	Assist the Member in selecting a PCP or confirm selection of a PCP;
	4.4.4.3.3.2.	Arrange for a Wellness Visit with the Member's PCP (either previously identified or selected by the Member from a list of available PCPs), which shall include:
	4.4	4.4.3.3.2.1 Assessments of both physical and behavioral health, including identification of urgent health care needs;
	4.4	1.4.3.3.2.2 Screening for depression, mood, suicidality, and Substance Use Disorder;
	4.4.4.3.3.3.	Support development of a Member's plan of care with the PCP;
	4.4.4.3.3.4.	Arrange for the completion of a HRA Screening in accordance with the terms of this Agreement and Section 4.10.2 (Health Risk Assessment (HRA) Screening).
	4.4.4.3.3.5.	Screening for adverse health needs, special needs, physical and behavioral health, and services of the Member. The MCO shall share the results of screening findings with the Member's PCP to support the Member's plan of care with the Provider;
	4.4.4.3.3.6.	Answer any other Member questions about the MCO;
	4.4.4.3.3.7.	Ensure Members can access information in their preferred language; and
	4.4.4.3.3.8.	Remind Members to report to the

Date \_\_\_\_\_ Page 125 of 397

Department any change of address, as

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

Members shall be liable for premium payments paid during period of ineligibility.

4.4.4.3.4 Regardless of the completion of the welcome call by the MCO, the PCP shall complete HRA Screenings as stipulated in Section 4.10.2 (Health Risk Assessment (HRA) Screening), and documented by a claim encounter.

#### 4.4.4.4 Member Hotline

- 4.4.4.4.1 The MCO shall establish a toll-free Member Service automated hotline that operates outside of the Member Call Center standard hours, Monday through Friday, and at all hours on weekends and holidays.
- 4.4.4.2 The automated system shall provide callers with operating instructions on what to do and who to call in case of an emergency, and shall also include, at a minimum, a voice mailbox for Members to leave messages.
- 4.4.4.3 The MCO shall ensure that the voice mailbox has adequate capacity to receive all messages. Return voicemail calls shall be made no later than the next business day.
- The MCO may substitute a live answering service in place of an automated system.

#### 4.4.5. Program Website

- 4.4.5.1 The MCO shall develop a website, in compliance with Section 7.7 (Website and Social Media) in this Agreement, to provide general information about the MCO's program, its Participating Provider network, its formulary, Prior Authorization requirements, the Member Handbook, its services for Members, and its Grievance Process and Member Appeal Process.
- 4.4.5.2 The solicitation or disclosure of any PHI, PI or other Confidential Information shall be subject to the requirements in Exhibit N (Liquidated Damages Matrix).
- 4.4.5.3 If the MCO chooses to provide required information electronically to Members, it shall:
  - 4.4.5.3.1 Be in a format and location that is prominent and readily accessible;
  - 4.4.5.3.2 Be provided in an electronic form which can be electronically retained and printed;

Dags 126 of 207	
Page 126 of 397 Date	

4.4.6.



# **Exhibit B - DRAFT**

	4.4.5.3.3	Be requ	consistent irements;	with	content	and	language	;
	4.4.5.3.4		y the Membe er form withou					ì
	4.4.5.3.5		ide, upon re n five (5) bus					
4.4.5.4	The MCO pro	ogram	content inclu	ıded or	the webs	ite shal	be:	
	4.4.5.4.1	Writt	en in English	and Sp	oanish;			
	4.4.5.4.2 Culturally appropriate;							
	4.4.5.4.3 Appropriate to the reading literacy of the population served; and					ì		
	4.4.5.4.4 Geared to the health needs of the enrolled MCO program population.					)		
4.4.5.5	The MCO's website shall be compliant with the federal DOJ "Accessibility of State and Local Government Websites to People with Disabilities."							
Marketing								
4.4.6.1	The MCO shall not, directly or indirectly, conduct door-to-door, telephonic, or other Cold Call Marketing to potential Members. The MCO shall submit all MCO Marketing material to the Department for approval before distribution.							
4.4.6.2	The Department shall identify any required changes to the Marketing Materials within thirty (30) calendar days. If the Department has not responded to a request for review by the thirtieth calendar day, the MCO may proceed to use the submitted materials. [42 CFR 438.104(b)(1)(i-ii), 42 CFR 438.104(b)(1)(iv-v)]							
4.4.6.3	The MCO shall comply with federal requirements for provision of information that ensures the potential Member is provided with accurate oral and written information sufficient to make an informed decision on whether or not to enroll.				1			
4.4.6.4	The MCO Marketing Materials shall not contain false or materially misleading information. The MCO shall not offer other insurance products as inducement to enroll.							
4.4.6.5	materials, is	accur	ensure that rate and does ne Departmen	s not m	nislead, co	nfuse,	or defraud	ł

Date \_\_\_\_\_ Page 127 of 397

shall not contain any written or oral assertions or statements that:

New Hampshire Department of Health and Human Services Medicaid Care Management Services



## **Exhibit B - DRAFT**

- 4.4.6.5.1 The recipient shall enroll in the MCO in order to obtain benefits or in order not to lose benefits; or
- 4.4.6.5.2 The MCO is endorsed by CMS, the State or federal government, or a similar entity. [42 CFR 438.104(b)(2)(i-ii)]
- 4.4.6.6 The MCO shall distribute Marketing Materials to the entire State. The MCO's Marketing Materials shall not seek to influence enrollment in conjunction with the sale or offering of any private insurance. The MCO shall not release and make public statements or press releases concerning the program without the prior consent of the Department. [42 CFR 438.104(b)(1)(i)-(ii), 42 CFR 438.104(b)(1)(iv-v)]
- 4.4.7. Member Engagement Strategy
  - 4.4.7.1 The MCO shall develop and facilitate an active Member Advisory Board that is composed of Members who represent its Member population.
  - 4.4.7.2 Member Advisory Board
    - 4.4.7.2.1 Representation on the Member Advisory Board shall draw from and be reflective of the MCO membership to ensure accurate and timely feedback on the MCM program.
    - 4.4.7.2.2 The Member Advisory Board shall meet at least four (4) times per year.
    - 4.4.7.2.3 The Member Advisory Board shall meet in-person or through interactive technology, including but not limited to a conference call or webinar and provide Member perspective(s) to influence the MCO's QAPI program changes (as further described in Section 4.14.3 (Quality Assessment and Performance Improvement Program).
    - 4.4.7.2.4 All costs related to the Member Advisory Board shall be the responsibility of the MCO.

## 4.4.7.3 Regional Member Meetings

- 4.4.7.3.1 The MCO shall hold in-person regional Member meetings for two-way communication where Members can provide input and ask questions, and the MCO can ask questions and obtain feedback from Members.
- 4.4.7.3.2 Regional meetings shall be held at least twice each Agreement year in demographically different locations in New Hampshire. The MCO shall make

Page 128 of 397	Date
-----------------	------

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



### **Exhibit B - DRAFT**

efforts to provide video conferencing opportunities for Members to attend the regional meetings. If video conferencing is unavailable, the MCO shall use alternate technologies as available for all meetings.

## 4.4.8. Cultural and Accessibility Considerations

- 4.4.8.1 The MCO shall participate in the Department's efforts to promote the delivery of services in a culturally and linguistically competent manner to all Members, including those with LEP and diverse cultural and ethnic backgrounds, disabilities, and regardless of gender, sexual orientation or gender identity. [42 CFR 438.206(c)(2)]
- 4.4.8.2 The MCO shall ensure that Participating Providers provide physical access, reasonable accommodations, and accessible equipment for Members with physical or behavioral disabilities. [42 CFR 438.206(c)(3)]

### 4.4.8.3 Cultural Competency Plan

- 4.4.8.3.1 In accordance with 42 CFR 438.206, the MCO shall have a comprehensive written Cultural Competency Plan describing how it will ensure that services are provided in a culturally and linguistically competent manner to all Members, including those with LEP, using qualified staff, interpreters, and translators in accordance with Exhibit O: Quality and Oversight Reporting Requirements.
- 4.4.8.3.2 The Cultural Competency Plan shall describe how the Participating Providers, and systems within the MCO will effectively provide services to people of all cultures, races, ethnic backgrounds, and religions in a manner that recognizes values, affirms and respects the worth of the each Member and protects and preserves a Member's dignity.
- 4.4.8.3.3 The MCO shall work with the Department Office of Health Equity to address cultural and linguistic considerations.

## 4.4.8.4 Communication Access

4.4.8.4.1 The MCO shall develop effective methods of communicating and working with its Members who do not speak English as a first language, who have physical conditions that impair their ability to speak clearly in order to be easily understood, as well as Members who have low-vision or hearing loss, and accommodating Members with physical and cognitive

Page 129 of 397 Date \_\_\_\_\_



# **Exhibit B - DRAFT**

	disabilities styles, and	and different literacy levels, learning capabilities.
4.4.8.4.2	methods for systems, communicating including encounters,	shall develop effective and appropriate or identifying, flagging in electronic and tracking Members' needs for tion assistance for health encounters preferred spoken language for all need for interpreter, and preferred r written information.
4.4.8.4.3	delivering la using only Qualified In	hall adhere to certain quality standards in anguage assistance services, including Qualified Bilingual/Multilingual Staff, terpreters for a Member with a Disability, terpreters for a Member with LEP, and anslators.
4.4.8.4.4	providing la use of un	nall ensure the competence of employees nguage assistance, recognizing that the trained individuals and/or minors as should be avoided. The MCO shall not:
	4.4.8.4.4.1.	Require a Member with LEP to provide their own interpreter;
	4.4.8.4.4.2.	Rely on an adult accompanying a Member with LEP to interpret or facilitate communication, except:
	4.4.8.4.4.3.	In an emergency involving an imminent threat to the safety or welfare of the Member or the public where there is no Qualified Interpreter for the Member with LEP immediately available, or
	4.4.8.4.4.4.	Where the Member with LEP specifically requests that the accompanying adult interpret or facilitate communication, the accompanying adult agrees to provide such assistance, and reliance on that adult for such assistance is appropriate under the circumstances;
	4.4.8.4.4.5.	Rely on a minor to interpret or facilitate communication, except in an emergency involving an imminent threat to the safety or welfare of a Member or the public where there is no Qualified Interpreter for the Member with LEP immediately available; or

Page 130 of 397 Date \_\_\_\_\_



# **Exhibit B - DRAFT**

	4.4.8.4.4.6.	Rely on staff other than Qualified Bilingual/Multilingual Staff to communicate directly with Members with LEP. [45 CFR 92.101(b)(2)]
4.4.8.4.5	available to regardless language wi	shall ensure interpreter services are or any Member who requests them, of the prevalence of the Member's ithin the overall program for all health planervices, exclusive of inpatient services.
4.4.8.4.6	service (such be appropring for all Members to interpretation shall be lessed to the service of	shall recognize that no one interpreter ch as over-the-phone interpretation) will ate (i.e. will provide meaningful access) pers in all situations. The most appropriate or use (in-person versus remote on) will vary from situation to situation and based upon the unique needs and sees of each Member.
4.4.8.4.7	appropriate circumstancinterpreter	the MCO shall provide the most interpretation service possible under the ces. In all cases, the MCO shall provide services when deemed clinically by the Provider of the encounter service.
4.4.8.4.8	interpreting Interpreters	shall not use low-quality video remote services. In instances where the Qualified are being provided through video remote services, the MCO's health programs and all provide:
	4.4.8.4.8.1.	Real-time, full-motion video and audio over a dedicated high-speed, wide-bandwidth video connection or wireless connection that delivers high-quality video images that do not produce lags, choppy, blurry, or grainy images, or irregular pauses in communication;
	4.4.8.4.8.2.	A sharply delineated image that is large enough to display the interpreter's face and the participating Member's face regardless of the Member's body position;
	4.4.8.4.8.3.	A clear, audible transmission of voices; and
	4.4.8.4.8.4.	Adequate training to users of the

Page 131 of 397 Date \_\_\_\_

and

other

involved

technology

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

individuals so that they may quickly and efficiently set up and operate the video remote interpreting. [45 CFR 92.101(b)(3)]

- 4.4.8.4.9 The MCO shall bear the cost of interpretive services and communication access, including ASL interpreters and translation into Braille materials as needed for Members with hearing loss and who are low-vision or visually impaired.
- 4.4.8.4.10 The MCO shall communicate in ways that can be understood by Members who are not literate in English or their native language. Accommodations may include the use of audio-visual presentations or other formats that can effectively convey information and its importance to the Member's health and health care.
- 4.4.8.4.11 If the Member declines free interpretation services offered by the MCO, the MCO shall have a process in place for informing the Member of the potential consequences of declination with the assistance of a competent interpreter to assure the Member's understanding, as well as a process to document the Member's declination.
- 4.4.8.4.12 Interpreter services shall be offered by the MCO at every new contact. Every declination requires new documentation by the MCO of the offer and decline.
- 4.4.8.4.13 The MCO shall comply with applicable provisions of federal laws and policies prohibiting discrimination, including but not limited to Title VI of the Civil Rights Act of 1964, as amended, which prohibits the MCO from discriminating on the basis of race, color, or national origin.
- 4.4.8.4.14 As clarified by Executive Order 13166, Improving Access to Services for Persons with LEP, and resulting agency guidance, national origin discrimination includes discrimination on the basis of LEP. To ensure compliance with Title VI of the Civil Rights Act of 1964, the MCO shall take reasonable steps to ensure that LEP Members have meaningful access to the MCO's programs.
- 4.4.8.4.15 Meaningful access may entail providing language assistance services, including oral and written translation, where necessary. The MCO is

Page 132 of 397 Date \_\_\_\_\_

New Hampshire Department of Health and Human Services Medicaid Care Management Services



## **Exhibit B - DRAFT**

encouraged to consider the need for language services for LEP persons served or encountered both in developing their budgets and in conducting their programs and activities. Additionally, the MCO is encouraged to develop and implement a written language access plan to ensure it is prepared to take reasonable steps to provide meaningful access to each Member with LEP who may require assistance.

4.4.8.4.16 Digital, video, and phone interpretation services must comply with Exhibit K: Information Security Requirements and Exhibit Q: IT Requirements Workbook.

### 4.5. Member Grievances and Appeals

## 4.5.1. General Requirements

- 4.5.1.1 The MCO shall develop, implement and maintain a Grievance System under which Members may challenge the denial of coverage of, or payment for, medical assistance and which includes a Grievance Process, an Appeal Process, and access to the State's fair hearing system. [42 CFR 438.402(a); 42 CFR 438.228(a)] The MCO shall ensure that the Grievance System is in compliance with this Agreement, 42 CFR 438 Subpart F, State law as applicable, and NH Code of Administrative Rules, Chapter He-C 200 Rules of Practice and Procedure.
- 4.5.1.2 The MCO shall provide to the Department a complete description, in writing and including all of its policies, procedures, notices and forms, of its proposed Grievance System for the Department's review and approval during the Readiness Review period. Any proposed changes to the Grievance System shall be reviewed by the Department thirty (30) calendar days prior to implementation.
- 4.5.1.3 The Grievance System shall be responsive to any grievance or appeal of Dual-Eligible Members. To the extent such grievance or appeal is related to a Medicaid service, the MCO shall handle the grievance or appeal in accordance with this Agreement.
- 4.5.1.4 In the event the MCO, after review, determines that the Dual-Eligible Member's grievance or appeal is solely related to a Medicare service, the MCO shall refer the Member to the State's Health Insurance Assistance Program (SHIP), which is currently administered by Service Link Aging and Disability Resource Center.
- 4.5.1.5 The MCO shall be responsible for ensuring that the Grievance System (Grievance Process, Appeal Process, and access to the State's fair hearing system) complies with the following general requirements. The MCO shall:

Page 133 of 397	Date
-----------------	------

4.5.1.6



# **Exhibit B - DRAFT**

4.5.1.5.1	completing includes, bu services an interpreter of providing we have a service of the control of	mbers with all reasonable assistance in forms and other procedural steps. This it is not limited to, providing interpreter d toll-free numbers with TTY/TDD and capability and assisting the Member in vritten consent for appeals [42 CFR 438.228(a)];
4.5.1.5.2	(including of authorized	e receipt of each grievance and appeal oral appeals), unless the Member or Provider requests expedited resolution 3.406(b)(1); 42 CFR 438.228(a)];
4.5.1.5.3	appeals and previous lev	t decision makers on grievances and I their subordinates were not involved in els of review or decision making [42 CFR 2)(i); 42 CFR 438.228(a)];
4.5.1.5.4	comments, information representati information adverse	decision makers take into account all documents, records, and other submitted by the Member or their ve without regard to whether such was submitted or considered in the initial benefit determination [42 CFR 2)(iii); 42 CFR 438.228(a)];
4.5.1.5.5	decision ma	i, if deciding any of the following, the akers are health care professionals with ertise in treating the Member's condition
	4.5.1.5.5.1.	An appeal of a denial based on lack of medical necessity;
	4.5.1.5.5.2.	A grievance regarding denial of expedited resolutions of an appeal; or
	4.5.1.5.5.3.	A grievance or appeal that involves clinical issues. [42 CFR 438.406(b)(2)(ii)(A-C); 42 CFR 438.228(a)]
4.5.1.5.6	and State fa	Members are permitted to file appeals air hearings after receiving notice that an ion is upheld. [42 CFR 438.402(c)(1); 42 8]

Date \_\_\_\_\_ Page 134 of 397

The MCO shall send written notice to Members and Participating Providers of any changes to the Grievance System at least thirty

(30) calendar days prior to implementation.

New Hampshire Department of Health and Human Services Medicaid Care Management Services



## **Exhibit B - DRAFT**

4.5.1.7	The MCO shall provide information as specified in 42 CFR
	438.10(g) about the Grievance System to Providers and
	Subcontractors at the time they enter into a contact or Subcontract.
	The information shall include, but is not limited to:

- 4.5.1.7.1 The Member's right to file grievances and appeals and requirements and timeframes for filing;
- 4.5.1.7.2 The Member's right to a State fair hearing, how to obtain a hearing, and the rules that govern representation at a hearing;
- 4.5.1.7.3 The availability of assistance with filing;
- 4.5.1.7.4 The toll-free numbers to file oral grievances and appeals;
- 4.5.1.7.5 The Member's right to request continuation of benefits during an appeal or State fair hearing filing and, if the MCO's action is upheld in a hearing, that the Member may be liable for the cost of any continued benefits; and
- 4.5.1.7.6 The Provider's right to appeal the failure of the MCO to pay for or cover a service.
- 4.5.1.8 The MCO shall make available training to Providers in supporting and assisting Members in the Grievance System.
- 4.5.1.9 The MCO shall maintain records of grievances and appeals, including all matters handled by delegated entities, for a period not less than ten (10) years. [42 CFR 438.416(a)]
- 4.5.1.10 At a minimum, such records shall include a general description of the reason for the grievance or appeal, the name of the Member, the dates received, the dates of each review, the dates of the grievance or appeal, the resolution and the date of resolution. [42 CFR 438.416(b)(1-6)]
- 4.5.1.11 In accordance with Exhibit O: Quality and Oversight Reporting Requirements, the MCO shall provide reports on all actions related to Member grievances and appeals, including all matters handled by delegated entities, including timely processing, results, and frequency of grievance and appeals.
- 4.5.1.12 The MCO shall review Grievance System information as part of the State quality strategy and in accordance with this Agreement and 42 CFR 438.402. The MCO shall regularly review appeals Confidential Data for process improvement which should include but not be limited to reviewing:

age 135 of 397	Date
age 133 01 381	Date

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

- 4.5.1.12.1 Reversed appeals for issues that could be addressed through improvements in the Prior Authorization process; and
- 4.5.1.12.2 Overall appeals to determine further Member and Provider education in the Prior Authorization process.
- 4.5.1.13 The MCO shall make such information accessible to the State and available upon request to CMS. [42 CFR 438.416(c)]

## 4.5.2. Member Grievance Process

- 4.5.2.1 The MCO shall develop, implement, and maintain a Grievance Process that establishes the procedure for addressing Member grievances and which is compliant with RSA 420-J:5, 42 CFR 438 Subpart F and this Agreement.
- 4.5.2.2 The MCO shall permit a Member, or the Member's authorized representative with the Member's written consent, to file a grievance with the MCO either orally or in writing at any time. [42 CFR 438.402(c)(1)(i-ii); 42 CFR 438.408; 42 CFR 438.402(c)(2)(i); 42 CFR 438.402(c)(3)(i)]
- 4.5.2.3 The Grievance Process shall address Member's expression of dissatisfaction with any aspect of their care other than an adverse benefit determination. Subjects for grievances include, but are not limited to:
  - 4.5.2.3.1 The quality of care or services provided;
  - 4.5.2.3.2 Aspects of interpersonal relationships such as rudeness of a Provider or employee;
  - 4.5.2.3.3 Failure to respect the Member's rights;
  - 4.5.2.3.4 Dispute of an extension of time proposed by the MCO to make an authorization decision:
  - 4.5.2.3.5 Members who believe that their rights established by RSA 135-C:56-57 or He-M 309 have been violated; and
  - 4.5.2.3.6 Members who believe the MCO is not providing mental health or Substance Use Disorder benefits in accordance with 42 CFR 438, subpart K.
- 4.5.2.4 The MCO shall complete the resolution of a grievance and provide notice to the affected parties as expeditiously as the Member's health condition requires, but not later than forty-five (45) calendar days from the day the MCO receives the grievance or within fifty-nine (59) calendar days of receipt of the grievance for grievances extended for up to fourteen (14) calendar days even if the MCO does not have all the information necessary to make the decision,

Page 136 of 397 Date \_\_\_\_\_

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

for ninety-eight percent (98%) of Members filing a grievance.	[42
CFR 438.408(a); 42 CFR 438.408(b)(1)]	_

- 4.5.2.5 The MCO may extend the timeframe for processing a grievance by up to fourteen (14) calendar days:
  - 4.5.2.5.1 If the Member requests the extension; or
  - 4.5.2.5.2 If the MCO shows that there is need for additional information and that the delay is in the Member's interest (upon State request). [42 CFR 438.408(c)(1)(i-ii); 438.408(b)(1)]
- 4.5.2.6 If the MCO extends the timeline for a grievance not at the request of the Member, the MCO shall:
  - 4.5.2.6.1 Make reasonable efforts to give the Member prompt oral notice of the delay; and
  - 4.5.2.6.2 Give the Member written notice, within two (2) calendar days, of the reason for the decision to extend the timeframe and inform the Member of the right to file a grievance if he or she disagrees with that decision. [42 CFR 438.408(c)(2)(i-ii); 42 CFR 438.408(b)(1)]
- 4.5.2.7 If the Member requests disenrollment, then the MCO shall resolve the grievance in time to permit the disenrollment (if approved) to be effective no later than the first day of the second month in which the Member requests disenrollment. [42 CFR 438.56(d)(5)(ii); 42 CFR 438.56(e)(1); 42 CFR 438.228(a)]
- 4.5.2.8 The MCO shall notify Members of the resolution of grievances. The notification may be orally or in writing for grievances not involving clinical issues. Notices of resolution for clinical issues shall be in writing. [42 CFR 438.408(d)(1); 42 CFR 438.10]
- 4.5.2.9 Members shall not have the right to a State fair hearing in regard to the resolution of a grievance.

# 4.5.3. Member Appeal Process

- 4.5.3.1 The MCO shall develop, implement, and maintain an Member Appeal Process that establishes the procedure for addressing Member requests for review of any action taken by the MCO and which is in compliance with 42 CFR 438 Subpart F and this Agreement. The MCO shall have only one (1) level of appeal for Members. [42 CFR 438.402(b); 42 CFR 438.228(a)]
- 4.5.3.2 The MCO shall permit a Member, or the Member's authorized representative, or a Provider acting on behalf of the Member and with the Member's written consent, to request an appeal orally or in

Page 137 of 397	Date

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



### **Exhibit B - DRAFT**

- writing of any MCO action. [42 CFR 438.402(c)(3)(ii); 42 CFR 438.402(c)(1)(ii)]
- 4.5.3.3 The MCO shall include as parties to the appeal, the Member and the Member's authorized representative, or the legal representative of the deceased Member's estate. [42 CFR 438.406(b)(6)]
- 4.5.3.4 The MCO shall permit a Member to file an appeal, either orally or in writing, within sixty (60) calendar days of the date on the MCO's notice of action. [42 CFR 438.402(c)(2)(ii)]
- 4.5.3.5 The MCO shall ensure that oral inquiries seeking to appeal an action are treated as appeals and confirm those inquires in writing, unless the Member or the authorized Provider requests expedited resolution. [42 CFR 438.406(b)(3)]
- 4.5.3.6 If the Department receives a request to appeal an action of the MCO, the Department shall forward relevant information to the MCO and the MCO shall contact the Member and acknowledge receipt of the appeal. [42 CFR 438.406(b)(1); 42 CFR 438.228(a)]
- 4.5.3.7 The MCO shall ensure that any decision to deny a service authorization request or to authorize a service in an amount, duration, or scope that is less than requested, shall be made by a health care professional who has appropriate clinical expertise in treating the Member's condition or disease.
- 4.5.3.8 The MCO shall permit the Member a reasonable opportunity to present evidence, and allegations of fact or law, in person as well as in writing [42 CFR 438.406(b)(4)]. The MCO shall inform the Member of the limited time available for this in the case of expedited resolution.
- 4.5.3.9 The MCO shall provide the Member and/or the Member's representative an opportunity to receive the Member's case file, free of charge prior to and sufficiently in advance of the resolution timeframe for standard and expedited appeal resolutions. [42 CFR 438.406(b)(5); 438.408(b-c)]
- 4.5.3.10 The MCO may offer peer-to-peer review support with a like clinician, upon request from a Member's Provider prior to the appeal decision. Any such peer-to-peer review should occur in a timely manner.
- 4.5.3.11 The MCO shall resolve ninety-eight percent (98%) of standard Member appeals within thirty (30) calendar days from the date the appeal was filed with the MCO. [42 CFR 438.408(a); 42 CFR 438.408(b)(2)]
- 4.5.3.12 The date of filing shall be considered either the date of receipt of an oral request for appeal or a written request for appeal from either the Member or Provider, whichever date is the earliest.

Page 138 of 397	Date	

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

- 4.5.3.13 Members who believe the MCO is not providing mental health or Substance Use Disorder benefits, in violation of 42 CFR 42 CFR 438, subpart K, may file an appeal.
- 4.5.3.14 If the MCO fails to adhere to notice and timing requirements, established in 42 CFR 438.408, then the Member is deemed to have exhausted the MCO's appeals process, and the Member may initiate a State fair hearing. [42 CFR 438.408; 42 CFR 438.402(c)(1)(i)(A)]

## 4.5.4. Member Adverse Actions

- 4.5.4.1 The MCO shall permit the appeal of any action taken by the MCO. Actions shall include, but are not limited to the following:
  - 4.5.4.1.1 Denial or limited authorization of a requested service, including the type or level of service;
  - 4.5.4.1.2 Reduction, suspension, or termination of a previously authorized service;
  - 4.5.4.1.3 Denial, in whole or in part, of payment for a service;
  - 4.5.4.1.4 Failure to provide services in a timely manner, as defined by this Agreement;
  - 4.5.4.1.5 Untimely service authorizations;
  - 4.5.4.1.6 Failure of the MCO to act within the timeframes set forth in this Agreement or as required under 42 CFR 438 Subpart F and this Agreement; and
  - 4.5.4.1.7 At such times, if any, that the Department has an Agreement with fewer than two (2) MCOs, for a rural area resident with only one (1) MCO, the denial of a Member's request to obtain services outside the network, in accordance with 42 CFR 438.52(b)(2)(ii).

## 4.5.5. Expedited Member Appeal

- 4.5.5.1 The MCO shall develop, implement, and maintain an expedited appeal review process for appeals when the MCO determines, as the result of a request from the Member, or a Provider request on the Member's behalf or supporting the Member's request, that taking the time for a standard resolution could seriously jeopardize the Member's life or health or ability to attain, maintain, or regain maximum function. [42 CFR 438.410(a)]
- 4.5.5.2 The MCO shall inform Members of the limited time available to present evidence and testimony, in person and in writing, and make legal and factual arguments sufficiently in advance of the resolution timeframe for expedited appeals. [42 CFR 438.406(b)(4); 42 CFR 438.408(b); 42 CFR 438.408(c)]

Page 139 of 397	Date	

New Hampshire Department of Health and Human Services Medicaid Care Management Services



## **Exhibit B - DRAFT**

4.5.5.3	The MCO shall make a decision on the Member's request for expedited appeal and provide notice, as expeditiously as the Member's health condition requires, but no later than seventy-two (72) hours after the MCO receives the appeal. [42 CFR 438.408(a); 42 CFR 438.408(b)(3)]

- 4.5.5.4 The MCO may extend the seventy-two (72) hour time period by up to fourteen (14) calendar days if the Member requests an extension, or if the MCO justifies a need for additional information and how the extension is in the Member's interest. The MCO shall also make reasonable efforts to provide oral notice. [42 CFR 438.408(c)(1); 42 CFR 438.408(b)(2)]
- 4.5.5.5 The date of filing of an expedited appeal shall be considered either an oral request for appeal or a written request from either the Member or Provider, whichever date is the earliest.
- 4.5.5.6 If the MCO extends the timeframes not at the request of the Member, it shall:
  - 4.5.5.6.1 Make reasonable efforts to give the Member prompt oral notice of the delay by providing a minimum of three (3) oral attempts to contact the Member at various times of the day, on different days within two (2) calendar days of the MCO's decision to extend the timeframe as detailed in He-W 506.08(i);
  - 4.5.5.6.2 Within two (2) calendar days give the Member written notice of the reason for the decision to extend the timeframe and inform the Member of the right to file a grievance if he or she disagrees with that decision;
  - 4.5.5.6.3 Resolve the appeal as expeditiously as the Member's health condition requires and no later than the date the extension expires. [42 CFR 438.408(c)(2)(i-iii); 42 CFR 438.408(b)(2-3)]
- 4.5.5.7 The MCO shall meet the timeframes above for ninety-eight percent (98%) of requests for expedited appeals.
- 4.5.5.8 The MCO shall ensure that punitive action is not taken against a Provider who requests an expedited resolution or supports a Member's appeal.
- 4.5.5.9 If the MCO denies a request for expedited resolution of an appeal, it shall transfer the appeal to the timeframe for standard resolution and make reasonable efforts to give the Member prompt oral notice of the denial, and follow up within two (2) calendar days with a written notice. [42 CFR 438.410(c); 42 CFR 438.408(b)(2); 42 CFR 438.408(c)(2)]

Page 140 of 397	Date	
-----------------	------	--

New Hampshire Department of Health and Human Services Medicaid Care Management Services



## **Exhibit B - DRAFT**

4.5.5.10 The Member has a right to file a grievance regarding the MCOs denial of a request for expedited resolution. The MCO shall inform the Member of his/her right and the procedures to file a grievance in the notice of denial.

### 4.5.6. Content of Member Appeal Notices

- 4.5.6.1 The MCO shall notify the requesting Provider, and give the Member written notice of any decision to deny a service authorization request, or to authorize a service in an amount, duration, or scope that is less than requested. [42 CFR 438.210(c); 42 CFR 438.404] Such notice shall meet the requirements of 42 CFR 438.404, except that the notice to the Provider need not be in writing.
- 4.5.6.2 The MCO shall utilize NCQA compliant Department model notices for all adverse actions and appeals. MCO adverse action and appeal notices shall be submitted for the Department review during the Readiness Review process. Each notice of adverse action shall contain and explain:
  - 4.5.6.2.1 The action the MCO or its Subcontractor has taken or intends to take [42 CFR 438.404(b)(1)];
  - 4.5.6.2.2 The reasons for the action, including the right of the Member to be provided, upon request and free of charge, reasonable access to and copies of all documents, records, and other information relevant to the adverse action [42 CFR 438.404(b)(2)];
  - 4.5.6.2.3 The Member's or the Provider's right to file an appeal, including information on exhausting the MCO's one (1) level of appeal and the right to request a State fair hearing if the adverse action is upheld [42 CFR 438.404(b)(3); 42 CFR 438.402(b-c)];
  - 4.5.6.2.4 Procedures for exercising Member's rights to file a grievance or appeal [42 CFR 438.404(b)(4)];
  - 4.5.6.2.5 Circumstances under which expedited resolution is available and how to request it [42 CFR 438.404(b)(5)]; and
  - 4.5.6.2.6 The Member's rights to have benefits continue pending the resolution of the appeal, how to request that benefits be continued, and the circumstances under which the Member may be required to pay the costs of these continued benefits. [42 CFR 438.404(b)(6)]
- 4.5.6.3 The MCO shall ensure that all notices of adverse action be in writing and shall meet the following language and format requirements:

Page 141 of 397	Date
-----------------	------



# **Exhibit B - DRAFT**

	4.5.6.3.1	Written notice shall be translated for the Members who speak one (1) of the commonly encountered languages spoken by MCM Members (as defined by the State per 42 CFR 438.10(d));
	4.5.6.3.2	Notice shall include language clarifying that oral interpretation is available for all languages and how to access it; and
	4.5.6.3.3	Notices shall use easily understood language and format, and shall be available in alternative formats, and in an appropriate manner that takes into consideration those with special needs. All Members shall be informed that information is available in alternative formats and how to access those formats.
4.5.6.4		all mail the notice of adverse action by the date of the any of the following occur:
	4.5.6.4.1	The Member has died;
	4.5.6.4.2	The Member submits a signed written statement requesting service termination;
	4.5.6.4.3	The Member submits a signed written statement including information that requires service termination or reduction and indicates that he understands that the service termination or reduction shall result;
	4.5.6.4.4	The Member has been admitted to an institution where he or she is ineligible under the Medicaid State Plan for further services;
	4.5.6.4.5	The Member's address is determined unknown based on returned mail with no forwarding address;
	4.5.6.4.6	The Member is accepted for Medicaid services by another state, territory, or commonwealth;
	4.5.6.4.7	A change in the level of medical care is prescribed by the Member's physician;
	4.5.6.4.8	The notice involves an adverse determination with regard to preadmission screening requirements of section 1919(e)(7) of the Social Security Act; or
	4.5.6.4.9	The transfer or discharge from a facility shall occur in an expedited fashion. [42 CFR 438.404(c)(1); 42 CFR 431.213; 42 CFR 431.231(d); section 1919(e)(7) of the Social Security Act; 42 CFR 483.12(a)(5)(i); 42 CFR 483.12(a)(5)(ii)]

Date \_\_\_\_\_ Page 142 of 397

New Hampshire Department of Health and Human Services Medicaid Care Management Services



## **Exhibit B - DRAFT**

# 4.5.7. Timing of Member Notices

- 4.5.7.1 For termination, suspension or reduction of previously authorized Medicaid Covered Services, the MCO shall provide Members written notice at least ten (10) calendar days before the date of action, except the period of advance notice shall be no more than five (5) calendar days in cases where the MCO has verified facts that the action should be taken because of probable Fraud by the Member. [42 CFR 438.404(c)(1); 42 CFR 431.211; 42 CFR 431.214]
- 4.5.7.2 In accordance with 42 CFR 438.404(c)(2), the MCO shall mail written notice to Members on the date of action when the adverse action is a denial of payment or reimbursement.
- 4.5.7.3 For standard service authorization denials or partial denials, the MCO shall provide Members with written notice as expeditiously as the Member's health condition requires but may not exceed fourteen (14) calendar days following a request for initial and continuing authorizations of services. [42 CFR 438.210(d)(1); 42 CFR 438.404(c)(3)] An extension of up to an additional fourteen (14) calendar days is permissible, if:
  - 4.5.7.3.1 The Member, or the Provider, requests the extension; or
  - 4.5.7.3.2 The MCO justifies a need for additional information and how the extension is in the Member's interest. [42 CFR 438.210(d)(1)(i)-(ii); 42 CFR 438.210(d)(2)(ii); 42 CFR 438.404(c)(4); 42 CFR 438.404(c)(6)]
- 4.5.7.4 When the MCO extends the timeframe, the MCO shall give the Member written notice of the reason for the decision to extend the timeframe and inform the Member of the right to file a grievance if he or she disagrees with that decision. [42 CFR 438.210(d)(1)(ii); 42 CFR 438.404(c)(4)(i)] Under such circumstance, the MCO shall issue and carry out its determination as expeditiously as the Member's health condition requires and no later than the date the extension expires. [42 CFR 438.210(d)(1)(ii); 42 CFR 438.404(c)(4)(ii)]
- 4.5.7.5 For cases in which a Provider indicates, or the MCO determines, that following the standard timeframe could seriously jeopardize the Member's life or health or ability to attain, maintain, or regain maximum function, the MCO shall make an expedited authorization decision and provide notice as expeditiously as the Member's health condition requires and no later than seventy-two (72) hours after receipt of the request for service. [42 CFR 438.210(d)(2)(i); 42 CFR 438.404(c)(6)]

Page 143 of 397 Date \_\_\_\_\_

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

- 4.5.7.6 The MCO may extend the seventy-two (72) hour time period by up to fourteen (14) calendar days if the Member requests an extension, or if the MCO justifies a need for additional information and how the extension is in the Member's interest.
- 4.5.7.7 The MCO shall provide notice on the date that the timeframes expire when service authorization decisions are not reached within the timeframes for either standard or expedited service authorizations. [42 CFR 438.404(c)(5)]

#### 4.5.8. Continuation of Member Benefits

- 4.5.8.1 The MCO shall continue the Member's benefits if:
  - 4.5.8.1.1 The appeal is filed timely, meaning on or before the later of the following:
    - 4.5.8.1.1.1. Within ten (10) calendar days of the MCO mailing the notice of action, or
    - 4.5.8.1.1.2. The intended effective date of the MCO's proposed action;
    - 4.5.8.1.1.3. The appeal involves the termination, suspension, or reduction of a previously authorized course of treatment;
    - 4.5.8.1.1.4. The services was ordered by an authorized Provider;
    - 4.5.8.1.1.5. The authorization period has not expired;
    - 4.5.8.1.1.6. The Member files the request for an appeal within sixty (60) calendar days following the date on the adverse benefit determination notice; and
    - 4.5.8.1.1.7. The Member requests extension of benefits, orally or in writing. [42 CFR 438.420(a); 42 CFR 438.420(b)(1-5); 42 CFR 438.402(c)(2)(ii)]
- 4.5.8.2 If the MCO continues or reinstates the Member's benefits while the appeal is pending, the benefits shall be continued until one (1) of the following occurs:
  - 4.5.8.2.1 The Member withdraws the appeal, in writing;
  - 4.5.8.2.2 The Member does not request a State fair hearing within ten (10) calendar days from when the MCO mails an adverse MCO decision regarding the Member's MCO appeal;

	Page 144 of 397	Date
--	-----------------	------

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

4.5.8.2.3	A State fair hearing decision adverse to the Member
	is made; or

- 4.5.8.2.4 The authorization expires or authorization service limits are met. [42 CFR 438.420(c)(1-3); 42 CFR 438.408(d)(2)]
- 4.5.8.3 If the final resolution of the appeal upholds the MCO's action, the MCO may recover from the Member the amount paid for the services provided to the Member while the appeal was pending, to the extent that they were provided solely because of the requirement for continuation of services. [42 CFR 438.420(d); 42 CFR 431.230(b)]
- 4.5.8.4 A Provider acting as an authorized representative shall not request a Member's continuation of benefits pending appeal even with the Member's written consent.

#### 4.5.9. **Resolution of Member Appeals**

- 4.5.9.1 The MCO shall resolve each appeal and provide notice, as expeditiously as the Member's health condition requires, within the following timeframes:
  - 4.5.9.1.1 For standard resolution of appeals and for appeals for termination, suspension, or reduction of previously authorized services, a decision shall be made within thirty (30) calendar days after receipt of the appeal even if the MCO does not have all the information necessary to make the decision, unless the MCO notifies the Member that an extension is necessary to complete the appeal.
  - 4.5.9.1.2 The MCO may extend the timeframes up to fourteen (14) calendar days if:
    - 4.5.9.1.2.1. The Member requests an extension, orally or in writing, or
    - 4.5.9.1.2.2. The MCO shows that there is a need for additional information and the MCO shows that the extension is in the Member's best interest; [42 CFR 438.408(c)(1)(i-ii); 438.408(b)(1)]
  - 4.5.9.1.3 If the MCO extends the timeframes not at the request of the Member then it shall:
    - 4.5.9.1.3.1. Make reasonable efforts to give the Member prompt oral notice of the delay,
    - 4.5.9.1.3.2. Within two (2) calendar days give the Member written notice of the reason for

Page 145 of 397	Date

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

the decision to extend the timeframe and inform the Member of the right to file a grievance if he or she disagrees with that decision; and resolve the appeal as expeditiously as the Member's health condition requires and no later than the date the extension expires. [42 CFR 438.408(c)(2)(i-ii); 42 CFR 438.408(b)(1); 42 CFR 438.408(b)(3)]

- 4.5.9.2 Under no circumstances may the MCO extend the appeal determination beyond forty-five (45) calendar days from the day the MCO receives the appeal request even if the MCO does not have all the information necessary to make the decision.
- 4.5.9.3 The MCO shall provide written notice of the resolution of the appeal, which shall include the date completed and reasons for the determination in easily, understood language.
- 4.5.9.4 The MCO shall include a written statement, in simple language, of the clinical rationale for the decision, including how the requesting Provider or Member may obtain the Utilization Management clinical review or decision-making criteria. [42 CFR 438.408(d)(2)(i); 42 CFR 438.10; 42 CFR 438.408(e)(1-2)]
- 4.5.9.5 For notice of an expedited resolution, the MCO shall provide written notice, and make reasonable efforts to provide oral notice. [42 CFR 438.408(d)(2)(ii)]
- 4.5.9.6 For appeals not resolved wholly in favor of the Member, the notice shall:
  - 4.5.9.6.1 Include information on the Member's right to request a State fair hearing:
  - 4.5.9.6.2 How to request a State fair hearing;
  - 4.5.9.6.3 Include information on the Member's right to receive services while the hearing is pending and how to make the request; and
  - 4.5.9.6.4 Inform the Member that the Member may be held liable for the amount the MCO pays for services received while the hearing is pending, if the hearing decision upholds the MCO's action. [42 CFR 438.408(d)(2)(i); 42 CFR 438.10; 42 CFR 438.408(e)(1-2)]

#### 4.5.10. State Fair Hearing for Member Appeals

4.5.10.1 The MCO shall inform Members regarding the State fair hearing process, including but not limited to Members' right to a State fair

Page 146 of 397	Date
_	

New Hampshire Department of Health and Human Services Medicaid Care Management Services



#### **Exhibit B - DRAFT**

hearing and how to obtain a State fair hearing in accordance with its informing requirements under this Agreement and as required under 42 CFR 438 Subpart F.

- 4.5.10.2 The parties to the State fair hearing include the MCO as well as the Member and their representative or the representative of a deceased Member's estate.
- 4.5.10.3 The MCO shall ensure that Members are informed, at a minimum, of the following:
  - 4.5.10.3.1 That Members shall exhaust all levels of resolution and appeal within the MCO's Grievance System prior to filing a request for a State fair hearing with the Department; and
  - 4.5.10.3.2 That if a Member does not agree with the MCO's resolution of the appeal, the Member may file a request for a State fair hearing within one hundred and twenty (120) calendar days of the date of the MCO's notice of the resolution of the appeal. [42 CFR.408(f)(2)]
- 4.5.10.4 If the Member requests a State fair hearing, the MCO shall provide to the Department and the Member, upon request, within three (3) business days, all MCO-held documentation related to the appeal, including but not limited to any transcript(s), records, or written decision(s) from Participating Providers or delegated entities.
- 4.5.10.5 A Member may request an expedited resolution of a State fair hearing if the Administrative Appeals Unit (AAU) determines that the time otherwise permitted for a State fair hearing could seriously jeopardize the Member's life, physical or mental health, or ability to attain, maintain, or regain maximum function, and:
  - 4.5.10.5.1 The MCO adversely resolved the Member's appeal wholly or partially; or
  - 4.5.10.5.2 The MCO failed to resolve the Member's expedited appeal within seventy-two (72) hours and failed to extend the seventy-two (72)-hour deadline in accordance with 42 CFR 408(c) and He-W 506.08(i).
- 4.5.10.6 If the Member requests an expedited State fair hearing, the MCO shall provide to the Department and the Member, upon request within twenty-four (24) hours, all MCO-held documentation related to the appeal, including but not limited to any transcript(s), records, or written decision(s) from Participating Providers or delegated entities.
- 4.5.10.7 If the AAU grants the Member's request for an expedited State fair hearing, then the AAU shall resolve the appeal within three (3)

Page 147 of 397	Date	

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

- business days after the Unit receives from the MCO the case file and any other necessary information. [He-W 506.09(g)]
- 4.5.10.8 The MCO shall appear and defend its decision before the Department AAU. The MCO shall consult with the Department regarding the State fair hearing process. In defense of its decisions in State fair hearing proceedings, the MCO shall provide supporting documentation, affidavits, and providing the Medical Director or other staff as appropriate, at no additional cost. In the event the State fair hearing decision is appealed by the Member, the MCO shall provide all necessary support to the Department for the duration of the appeal at no additional cost.
- 4.5.10.9 The Department AAU shall notify the MCO of State fair hearing determinations. The MCO shall be bound by the fair hearing determination, whether or not the State fair hearing determination upholds the MCO's decision. The MCO shall not object to the State intervening in any such appeal.

#### 4.5.11. Effect of Adverse Decisions of Member Appeals and Hearings

- 4.5.11.1 If the MCO or the Department reverses a decision to deny, limit, or delay services that were not provided while the appeal or State fair hearing were pending, the MCO shall authorize or provide the disputed services promptly, and as expeditiously as the Member's health condition requires but no later than 72 hours from the date it receives notice reversing the determination. [42 CFR 438.424(a)]
- 4.5.11.2 If the MCO or the Department reverses a decision to deny authorization of services, and the Member received the disputed services while the appeal or State fair hearing were pending, the MCO shall pay for those services. [42 CFR 438.424(b)]

#### 4.5.12. Survival of Member Appeals and Grievances

4.5.12.1 The obligations of the MCO to fully resolve all grievances and appeals, including but not limited to providing the Department with all necessary support and providing a Medical Director or similarly qualified staff to provide evidence and testify at proceedings until final resolution of any grievance or appeal shall survive the termination of this Agreement.

#### 4.6. Provider Appeals

#### 4.6.1. **General**

4.6.1.1 The MCO shall develop, implement, and maintain a Provider Appeals Process under which Providers may challenge any Provider adverse action by the MCO, and access the State's fair hearing system in accordance with RSA 126-A:5, VIII.

Page 148 of 397	Date
-----------------	------

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

- 4.6.1.2 The MCO shall provide to the Department a complete description of its Provider Appeals Process, in writing, including all policies and procedures, notices and forms, of its proposed Provider Appeals Process for the Department's review and approval during the Readiness Review period.
- 4.6.1.3 Any proposed changes to the Provider Appeals Process shall be approved by the Department at least thirty (30) calendar days in advance of implementation.
- 4.6.1.4 The MCO shall clearly articulate its Provider Appeals Process in the MCO's Provider manual, and reference it in the Provider agreement.
- 4.6.1.5 The MCO shall ensure its Provider Appeals Process complies with the following general requirements:
  - 4.6.1.5.1 Gives reasonable assistance to Providers requesting an appeal of a Provider adverse action;
  - 4.6.1.5.2 Ensures that the decision makers involved in the Provider Appeals Process and their subordinates were not involved in previous levels of review or decision making of the Provider's adverse action;
  - 4.6.1.5.3 Ensures that decision makers take into account all comments, documents, records, and other information submitted by the Provider to the extent such materials are relevant to the appeal; and
  - 4.6.1.5.4 Advises Providers of any changes to the Provider Appeals Process at least thirty (30) calendar days prior to implementation.

#### 4.6.2. Provider Adverse Actions

- 4.6.2.1 The Provider shall have the right to file an appeal with the MCO and utilize the Provider Appeals Process for any adverse action, in accordance with RSA 126-A:5, VIII, except for Member appeals or grievances described in Section 4.5 (Member Grievances and Appeals). The Provider shall have the right to file an appeal within thirty (30) calendar days of the date of the MCO's notice of adverse action to the Provider. Reasons may include, but are not limited to:
  - 4.6.2.1.1 Action against the Provider for reasons related to program integrity;
  - 4.6.2.1.2 Termination of the Provider's agreement before the agreement period has ended for reasons other than when the Department, MFCU or other government agency has required the MCO to terminate such agreement;

Page 149 of 397 Date \_\_\_\_\_

New Hampshire Department of Health and Human Services Medicaid Care Management Services



## **Exhibit B - DRAFT**

- 4.6.2.1.3 Denial of claims for services rendered that have not been filed as a Member appeal; and
- 4.6.2.1.4 Violation of the agreement between the MCO and the Provider.
- 4.6.2.2 The MCO shall not be precluded from taking an immediate adverse action even if the Provider requests an appeal; provided that, if the adverse action is overturned during the MCO's Provider Appeals Process or State fair hearing, the MCO shall immediately take all steps to reverse the adverse action within ten (10) calendar days.

#### 4.6.3. **Provider Appeal Process**

- 4.6.3.1 The MCO shall provide written notice, and electronic notice if available, to the Provider of any adverse action, and include in its notice a description of the basis of the adverse action, and the right to appeal the adverse action.
- 4.6.3.2 Providers shall submit a written request for an appeal to the MCO, together with any evidence or supportive documentation it wishes the MCO to consider, within sixty (60) calendar days of:
  - 4.6.3.2.1 The date of the MCO's written notice advising the Provider of the adverse action to be taken; or
  - 4.6.3.2.2 The date on which the MCO should have taken a required action and failed to take such action.
- 4.6.3.3 The MCO shall be permitted to extend the decision deadline to issue the Resolution Notice by an additional sixty (60) calendar days to allow the Provider to submit additional evidence or supportive documentation, and for other good cause determined by the MCO.
- 4.6.3.4 The MCO shall ensure that all Provider Appeal Process decisions are determined by an administrative or clinical professional with expertise in the subject matter of the Provider appeal.
- 4.6.3.5 The MCO may offer peer-to-peer review support with a like clinician, upon request, for Providers who receive an adverse decision from the MCO. Any such peer-to-peer review should occur in a timely manner and before the Provider seeks recourse through the Provider Appeal Process or State fair hearing process.
- 4.6.3.6 The MCO shall maintain a log and records of all Provider Appeals, including for all matters handled by delegated entities, for a period not less than ten (10) years. At a minimum, log records shall include:
  - 4.6.3.6.1 General description of each appeal;
  - 4.6.3.6.2 Name of the Provider;

Page 150 of 397	Date

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

4.6.3.6.3	Date(s) of rece	ipt of the	appea	I and su	pporting
	documentation,	decision,	and	effectuat	ion, as
	applicable; and				

- 4.6.3.6.4 Name(s), title(s), and credentials of the reviewer(s) determining the appeal decision.
- 4.6.3.7 If the MCO fails to adhere to notice and timing requirements established in this Agreement, then the Provider is deemed to have exhausted the MCO's Provider Appeal Process and may initiate a State fair hearing.

#### 4.6.4. MCO Resolution of Provider Appeals

- 4.6.4.1 The MCO shall provide timely written notice of Provider appeal resolution (Resolution Notice) at a rate of ninety-five percent (95%) within thirty (30) calendar days from either the date the MCO receives the appeal request, or if an extension is granted to the Provider to submit additional evidence, the date on which the Provider's evidence is received by the MCO.
- 4.6.4.2 The Resolution Notice shall include, without limitation:
  - 4.6.4.2.1 The MCO's decision;
  - 4.6.4.2.2 The reasons for the MCO's decision;
  - 4.6.4.2.3 The Provider's right to request a State fair hearing in accordance with RSA 126-A:5, VIII; and
  - 4.6.4.2.4 For overturned appeals, the MCO shall take all steps to reverse the adverse action within ten (10) calendar days.

## 4.6.5. State Fair Hearing for Provider Appeals

- 4.6.5.1 The MCO shall inform its Participating Providers regarding the State fair hearing process consistent with RSA 126-A:5, VIII, including but not limited to how to obtain a State fair hearing in accordance with its informing requirements under this Agreement.
- 4.6.5.2 The parties to the State fair hearing include the MCO as well as the Provider.
- 4.6.5.3 The Participating Provider shall exhaust the MCO's Provider Appeals Process before pursuing a State fair hearing.
- 4.6.5.4 If a Participating Provider requests a State fair hearing, the MCO shall provide to the Department and the Participating Provider, upon request, within three (3) business days, all MCO-held documentation related to the Provider Appeal, including but not limited to, any transcript(s), records, or written decision(s).

Page 151 of 397	Date

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

- 4.6.5.5 The MCO shall consult with the Department regarding the State fair hearing process. In defense of its decisions in State fair hearing proceedings, the MCO shall provide supporting documentation, affidavits, and availability of the Medical Director and/or other staff as appropriate, at no additional cost.
- 4.6.5.6 The MCO shall appear and defend its decision before the Department AAU. Nothing in this Agreement shall preclude the MCO from representation by legal counsel.
- 4.6.5.7 The Department AAU shall notify the MCO of State fair hearing determinations within sixty (60) calendar days of the date of the MCO's Notice of Resolution.
- 4.6.5.8 The MCO shall:
  - 4.6.5.8.1 Not object to the State intervening in any such appeal;
  - 4.6.5.8.2 Be bound by the State fair hearing determination, whether or not the State fair hearing determination upholds the MCO's Final Determination; and
  - 4.6.5.8.3 Take all steps to reverse any overturned adverse action within ten (10) calendar days.
- 4.6.5.9 Reporting
  - 4.6.5.9.1 The MCO shall provide to the Department, as detailed in Exhibit O: Quality and Oversight Reporting Requirements, Provider complaint and appeal logs. [42 CFR 438.66(c)(3)]

#### 4.7. Access

## 4.7.1. Participating Provider Network

- 4.7.1.1 The MCO shall implement written policies and procedures for selection and retention of Participating Providers. [42 CFR 438.12(a)(2); 42 CFR 438.214(a)]
- 4.7.1.2 The MCO shall develop and maintain a statewide Participating Provider network that adequately meets all covered medical, mental health, Serious Mental Illness, Serious Emotional Disturbance, Substance Use Disorder and psychosocial needs of the covered population in a manner that provides for coordination and collaboration among multiple Providers and disciplines and Equal Access to services. In developing its Participating Provider network, the MCO shall consider and address the following factors to ensure network adequacy for each Member:
  - 4.7.1.2.1 Current and anticipated NH Medicaid enrollment;

age 152 of 397	Date	

New Hampshire Department of Health and Human Services Medicaid Care Management Services

4.7.1.3

4.7.1.4

4.7.1.5



# **Exhibit B - DRAFT**

4.7.1.2.2	The expected utilization of services, taking into consideration the characteristics and health care needs of the covered NH Medicaid population;	
4.7.1.2.3	The number and type (in terms of training and experience and specialization) of Providers required to furnish the contracted services;	
4.7.1.2.4	The number of network Participating Providers limiting NH Medicaid patients' access to the Participating Provider or not accepting new or any NH Medicaid patients;	
4.7.1.2.5	The geographic location of Providers and Members, considering distance, travel time, and the means of transportation ordinarily used by NH Members;	
4.7.1.2.6	The linguistic capability of Providers to communicate with Members in non-English languages, including oral and American Sign Language;	
4.7.1.2.7	The availability of screening systems, as well as the use of telemedicine, e-visits, and/or other evolving and innovative technological solutions, in compliance with Exhibit K: Information Security Requirements and Exhibit Q: IT Requirements Workbook;	
4.7.1.2.8	Adequacy of the primary care Participating Provider network to offer each Member a choice of at least two (2) appropriate PCPs that are accepting new Medicaid patients;	
4.7.1.2.9	Access standards identified in this Agreement; and	
4.7.1.2.10	Required access standards set forth by the NHID, including RSA. 420-J; and N.H. Code of Administrative Rules Ins 2700	
The MCO shall meet the Participating Provider network adequacy standards included in this Agreement in all geographic areas in which the MCO operates for all Provider types covered under this Agreement.		
in terms of ti	all ensure that services are as accessible to Members meliness, amount, duration and scope as those that to Members covered by the Department under FFS	

Medicaid within the same service area.

Page 153 of 397 Date \_\_\_\_\_

The MCO shall ensure Participating Providers comply with the

accessibility standards of the ADA. Participating Providers shall demonstrate physical access, reasonable accommodations, and

New Hampshire Department of Health and Human Services Medicaid Care Management Services



#### **Exhibit B - DRAFT**

- accessible equipment for all Members including those with physical or cognitive disabilities. [42 CFR 438.206(c)(3)]
- 4.7.1.6 The MCO shall demonstrate that there are sufficient Participating Indian Health Care Providers (IHCPs) in the Participating Provider network to ensure timely access to services for American Indians who are eligible to receive services. If Members are permitted by the MCO to access out-of-state IHCPs, or if this circumstance is deemed to be good cause for disenrollment, the MCO shall be considered to have met this requirement. [42 CFR 438.14(b)(1); 42 CFR 438.14(b)(5)]
- 4.7.1.7 The MCO shall maintain an updated list of Participating Providers on its website in a Provider Directory, as specified in Section 4.4.1.5 (Provider Directory) of this Agreement.

## 4.7.2. Assurances of Adequate Capacity and Services

- 4.7.2.1 The MCO's Participating Provider network shall have Participating Providers in sufficient numbers, and with sufficient capacity and expertise for all Covered Services to meet the geographic standards in Section 4.7.3 (Time and Distance Standards), the timely provision of services requirements in Section 4.7.5 (Timely Access to Service Delivery), Equal Access, and reasonable choice by Members to meet their needs [42 CFR 438.207(a)].
- 4.7.2.2 The MCO shall submit documentation to the Department, in the format and frequency specified by the Department in Exhibit O: Quality and Oversight Reporting Requirements, that fulfills the following requirements:
  - 4.7.2.2.1 The MCO shall give assurances and provide supporting documentation to the Department that demonstrates that it has the capacity to serve the expected enrollment in its service area in accordance with the Department's standards for access and timeliness of care. [42 CFR 438.207(a); 42 CFR 438.68; 42 CFR 438.206(c)(1)]
  - 4.7.2.2.2 The MCO offers an appropriate range of preventive, primary care, and specialty services that is adequate for the anticipated number of Members for the service area. [42 CFR 438.207(b)(1)];
  - 4.7.2.2.3 The MCO's Participating Provider network includes sufficient family planning Providers to ensure timely access to Covered Services. [42 CFR 438.206(b)(7)];
  - 4.7.2.2.4 The MCO is complying with the Department's requirements for availability, accessibility of services, and adequacy of the Participating Provider network

Page 154 of 397	Date
age 134 01 397	Date

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

including	pediatric	subspecialists	as	described	in
Section 4.	.7.5.11 (A	ccess Standards	s for	Children w	ith
Special H	ealth Care	e Needs);			

- 4.7.2.2.5 The MCO is complying with the Department's requirements for Behavioral Health Services, as specified in Section 4.13, including but not limited to Substance Use Disorder treatment services and recovery, Mental Health services, Community Mental Health services, and
- 4.7.2.2.6 The MCO demonstrates Equal Access to services for all populations in the MCM program, as described in Section 4.7.5 (Timely Access to Service Delivery).
- 4.7.2.3 To permit the Department to determine if access to private duty nursing services is increasing, as indicated by the Department in Exhibit O: Quality and Oversight Reporting Requirements, the MCO shall provide to the Department the following information:
  - 4.7.2.3.1 The number of pediatric private duty nursing hours authorized by day/weekend/night, and intensive (ventilator dependent) modifiers; and
  - 4.7.2.3.2 The number of pediatric private duty nursing hours delivered by day/weekend/night, and intensive (ventilator dependent) modifiers.
- 4.7.2.4 The MCO shall submit documentation to the Department to demonstrate that it maintains an adequate network of Participating Providers that is sufficient in number, mix, and geographic distribution to meet the needs of the anticipated number of Members in the service area, in accordance with Exhibit O: Quality and Oversight Reporting Requirements:
  - 4.7.2.4.1 During the Readiness Review period, prior to the Program Start Date;
  - 4.7.2.4.2 Annually; and
  - 4.7.2.4.3 At any time there has been a significant change (as defined by the Department) in the entity's operations that would affect adequate capacity and services, including but not limited to changes in services, benefits, geographic service area, or payments; and/or enrollment of a new population in the MCO. [42 CFR 438.207(b-c)]
- 4.7.2.5 For purposes of providing assurances of adequate capacity and services, the MCO shall base the anticipated number of Members on the "NH MCM Fifty Percent (50%) Population Estimate by Zip Code" report provided by the Department.

Page 1	155 of 397	Date	

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

#### 4.7.3. Time and Distance Standards

4.7.3.1 At a minimum, the MCO shall meet the geographic access standards described in the Table below for all Members, in addition to maintaining in its network a sufficient number of Participating Providers to provide all services and Equal Access to its Members, subject to alternative CMS requirements. [42 CFR 438.68(b)(1)(i-viii); 42 CFR 438.68(b)(3)]

Geographic Access Standards			
Provider/Service	Requirement		
PCPs (Adult and Pediatric)	Two (2) within forty (40) driving minutes or fifteen (15) driving miles		
Adult Specialists	One (1) within sixty (60) driving minutes or forty-five (45) driving miles		
Pediatric Specialists	One (1) within one hundred twenty (120) driving minutes or eighty driving (80) miles		
OB/GYN Providers	One (1) within sixty (60) driving minutes or forty-five (45) driving miles		
Hospitals	One (1) within sixty (60) driving minutes or forty-five (45) driving miles		
Mental Health Providers (Adult and Pediatric)	One (1) within forty-five (45) driving minutes or twenty-five (25) driving miles		
Community Mental Health Programs	One (1) within forty-five (45) driving minutes or twenty-five (25) driving miles		
Mobile Crisis Service Providers <sup>12</sup>	One (1) within sixty (60) driving minutes or forty-five (45) driving miles		
Pharmacies	One (1) within forty-five (45) driving minutes or fifteen (15) driving miles		
Tertiary or Specialized Services (e.g., Trauma, Neonatal)	One (1) within one hundred twenty (120) driving minutes or eighty driving (80) miles		

<sup>&</sup>lt;sup>12</sup> Mobile crisis services are provided by CMH Programs but subject to a different Geographic Access Standard requirement pursuant to the Department's selected Mobile Crisis System model.

Date \_\_\_\_\_

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

Geographic Access Standards		
Provider/Service	Requirement	
Individual/Group MLADCs	One (1) within forty-five (45) minutes or fifteen (15) miles	
Substance Use Disorder Programs	One (1) within sixty (60) minutes or forty-five (45) miles.	
Adult Medical Day Care	One (1) within sixty (60) driving minutes or forty-five (45) driving miles	
Hospice	One (1) within sixty (60) driving minutes or forty-five (45) driving miles	
Office-based Physical Therapy/Occupation-al Therapy/Speech Therapy	One (1) within sixty (60) driving minutes or forty-five (45) driving miles	

- 4.7.3.2 The MCO shall report annually how specific provider types meet the time and distance standards for Members in each county within NH in accordance with Exhibit O: Quality and Oversight Reporting Requirements.
- 4.7.3.3 The Department shall continue to assess where additional access requirements, whether time and distance or otherwise, shall be incorporated. The Department may provide additional guidance to the MCO regarding its Participating Provider network adequacy requirements in accordance with Members' ongoing access to care needs.
- 4.7.3.4 The MCO shall contract with qualified Substance Use Disorder Providers who request to join its Participating Provider network pending the Substance Use Disorder Provider's agreement to the terms of the MCO's contract.
- 4.7.3.5 Additional Behavioral Health Provider Standards

Provider/Service	Requirement
MLADCs	The MCO's Participating Provider network shall include seventy percent (70%) of all such Providers licensed and practicing in NH and no less than two (2) Providers in any public health region unless there are less than two (2) such Providers in the region
Opioid Treatment Programs (OTPs)	The MCO's Participating Provider network shall include seventy-five percent (75%) of all such Providers licensed and practicing in NH and no less than two (2) Providers in any public health region unless there are less than two (2) such Providers in the region

Page 157 of 397	Date

New Hampshire Department of Health and Human Services Medicaid Care Management Services



## **Exhibit B - DRAFT**

Provider/Service	Requirement
Buprenorphine Prescribers	The Participating Provider network shall include seventy-five percent (75%) of all such Providers actively prescribing Buprenorphine in their practice and licensed and practicing in NH and no less than two (2) Providers in any public health region unless there are less than two (2) such Providers in the region
Residential Substance Use Disorder Treatment Programs	The MCO's Participating Provider network shall include fifty percent (50%) of all such Providers licensed and practicing in NH and no less than two (2) in any public health region unless there are less than two (2) such Providers in the region
Peer Recovery Programs	The MCO's Participating Provider network shall include one hundred percent (100%) of all such willing Programs in NH
Residential Programs for Serious Mental Illness	The MCO's Participating Provider network shall include 100% of all such Providers, located in NH, if they are operated by or under contract with Community Mental Health Programs, and 100% of all such Providers if they are otherwise under contract with the Department and are appropriately licensed or certified by the Department under He-P 800 or He-M 1000.
Psychiatric Residential Treatment Facilities	The MCO's Participating Provider network shall include 100% of all such Providers, located in NH, if they are owned or operated by, under contract with, or are otherwise determined or designated by the Department to provide this service, and are appropriately licensed or certified by the Department or a Department approved alternative certification entity.

#### 4.7.4. Standards for Geographic Accessibility

- 4.7.4.1 The MCO may request reasonable exceptions from the Agreement's Participating Provider network standards after demonstrating its efforts to contract a sufficient network of Participating Providers. The Department reserves the right to approve or disapprove these requests, at its discretion.
- 4.7.4.2 Should the MCO be unable to contract a sufficient number of Participating Providers to meet the geographic and timely access to service delivery standards, and should the MCO be unable, with the assistance of the Department and after good faith negotiations, continue to be unable to meet geographic and timely access to service delivery standards, then for a period of up to sixty (60) calendar days of the Program Start Date or at any time during the contract term, Liquidated Damages described in Section 5.5.2 (Liquidated Damages) and Exhibit N: Liquidated Damages Matrix shall apply.
- 4.7.4.3 Except within a period of sixty (60) calendar days after the start date where Liquidated Damages shall not apply, should the MCO, after

Page 158 of 397	Date	

New Hampshire Department of Health and Human Services **Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

good faith negotiations, be unable to create a sufficient number of Participating Providers to meet the geographic and timely access to service delivery standards, and should the MCO be unable, after good faith negotiations with assistance of the Department, continue to be unable to meet geographic and timely access to service delivery standards the Department may, at its discretion, provide temporary exemption to the MCO from Liquidated Damages.

- 4.7.4.4 At any time the provisions of this section may apply, the MCO shall ensure Members have reasonable access to Covered Services.
- The MCO shall ensure that an adequate number of participating 4.7.4.5 physicians have admitting privileges at participating acute care hospitals in the Participating Provider network to ensure that necessary admissions can be made.
- 4.7.4.6 Exceptions
  - 4.7.4.6.1 The MCO may request exceptions, via a Request for Exception, from the Participating Provider network adequacy standards after demonstrating its efforts to create a sufficient network of Participating Providers to meet these standards. [42 CFR 438.68(d)(1)]
  - 4.7.4.6.2 The Department may grant the MCO an exception in the event that:
    - 4.7.4.6.2.1. The MCO demonstrates that an insufficient number qualified of Providers or facilities that are willing to contract with the MCO are available to meet the Participating Provider network adequacy standards in this Agreement and as otherwise defined by the NHID and the Department;
    - 4.7.4.6.2.2. The MCO demonstrates, to satisfaction of the Department, that the MCO's failure to develop a Participating Provider network that meets the requirements is due to the refusal of a Provider to accept a reasonable rate, fee, term, or condition and that the MCO has taken steps to effectively mitigate the detrimental impact on covered
    - 4.7.4.6.2.3. The MCO demonstrates that the required specialist services can be obtained through the use of telemedicine or telehealth from a

persons; or

Page 159 of 397

New Hampshire Department of Health and Human Services Medicaid Care Management Services



#### **Exhibit B - DRAFT**

Participating Provider that is a physician, physician assistant, nurse practitioner, clinic nurse specialist, nurse-midwife, clinical psychologist, clinical social worker, registered dietitian or nutrition professional, certified registered nurse anesthetist, or other behavioral health specialists licensed by the NH Board of Medicine. [RSA 167:4-d]

4.7.4.6.2.4. The MCO is permitted to use telemedicine as a tool for ensuring access to needed services in accordance with telemedicine coverage policies reviewed and approved by the Department, but the MCO shall not use telemedicine to meet the Participating Provider network adequacy standards unless the Department has specifically approved a Request for Exception.

4.7.4.6.3 The MCO shall report on Participating Provider network adequacy and exception requests in accordance with Exhibit O: Quality and Oversight Reporting Requirements.

#### 4.7.5. Timely Access to Service Delivery

- 4.7.5.1 The MCO shall meet the following timely access standards for all Members, in addition to maintaining in its network a sufficient number of Participating Providers to provide all services and Equal Access to its Members.
- 4.7.5.2 The MCO shall make Covered Services available for Members twenty-four (24) hours a day, seven (7) days a week, when Medically Necessary. [42 CFR 438.206(c)(1)(iii)]
- 4.7.5.3 The MCO shall require that all Participating Providers offer hours of operation that provide Equal Access and are no less than the hours of operation offered to commercial Members or are comparable to Medicaid FFS patients, if the Provider serves only Medicaid Members. [42 CFR 438.206(c)(1)(ii)]
- 4.7.5.4 The MCO shall encourage Participating Providers to offer afterhours office care in the evenings and on weekends.
- 4.7.5.5 The MCO's Participating Provider network shall meet minimum timely access to care and services standards as required per 42 CFR 438.206(c)(1)(i). Health care services shall be made accessible on a timely basis in accordance with medically

Page 160 of 397	Date

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

appropriate	guidelines	consistent	with	generally	accepted
standards of care.					

- 4.7.5.6 The MCO shall have in its Participating Provider network the capacity to ensure that waiting times for appointments do not exceed the following:
  - 4.7.5.6.1 Non-Symptomatic Office Visits (i.e., diagnostic, preventive care) shall be available from the Member's PCP or another Provider within forty-five (45) calendar days.
  - 4.7.5.6.2 A Non-Symptomatic Office Visit may include, but is not limited to, well/preventive care such as physical examinations, annual gynecological examinations, or child and adult immunizations.
  - 4.7.5.6.3 Non-Urgent, Symptomatic Office Visits (i.e., routine care) shall be available from the Member's PCP or another Provider within ten (10) calendar days of a request for the visit. Non-Urgent, Symptomatic Office Visits are associated with the presentation of medical signs or symptoms not requiring immediate attention.
  - 4.7.5.6.4 Urgent, Symptomatic Office Visits shall be available from the Member's PCP or another Provider within forty-eight (48) hours. An Urgent, Symptomatic Office Visit is associated with the presentation of medical signs or symptoms that require immediate attention, but are not life threatening and do not meet the definition of Emergency Medical Condition.
  - 4.7.5.6.5 Transitional Health Care shall be available from a primary care or specialty Provider for clinical assessment and care planning within two (2) business days of discharge from inpatient or institutional care for physical or behavioral health disorders or discharge from a Substance Use Disorder treatment program.
    - Transitional Home Care shall be available with a home care nurse, licensed counselor, and/or therapist (physical therapist or occupational therapist) within two (2) calendar days of discharge from inpatient or institutional care for physical or mental health disorders, if ordered by the Member's PCP or Specialty Care Provider or as part of the discharge plan.

4.7.5.6.6

Page 161 of 397

Date		

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

- 4.7.5.6.7 Obstetrics and gynecological care shall be available within fifteen (15) calendar days from the date of the Member's appointment request.
- 4.7.5.7 The MCO shall establish mechanisms to ensure that Participating Providers comply with the timely access standards.
- 4.7.5.8 The MCO shall regularly monitor its Participating Provider network to determine compliance with timely access and shall provide an annual report to the Department documenting its compliance with 42 CFR 438.206(c)(1)(iv) and (v), in accordance with Exhibit O: Quality and Oversight Reporting Requirements.
- 4.7.5.9 The MCO shall monitor waiting times for obtaining appointments with approved CMH Programs and report case details on a semi-annual basis.
- 4.7.5.10 The MCO shall develop and implement a CAP if it or its Participating Providers fail to comply with timely access provisions in this Agreement in compliance with 42 CFR 438.206(c)(1)(vi).
- 4.7.5.11 Access Standards for Children with Special Health Care Needs
  - 4.7.5.11.1 The MCO shall contract with specialists that have pediatric expertise where the need for pediatric specialty care significantly differs from adult specialty care.
  - 4.7.5.11.2 In addition to the "specialty care" Participating Provider network adequacy requirements, the MCO shall contract with Providers who offer the following specialty services:
    - 4.7.5.11.2.1. Pediatric Critical Care;
    - 4.7.5.11.2.2. Pediatric Child Development;
    - 4.7.5.11.2.3. Pediatric Genetics:
    - 4.7.5.11.2.4. Pediatric Physical Medicine and Rehabilitation;
    - 4.7.5.11.2.5. Pediatric Ambulatory Tertiary Care;
    - 4.7.5.11.2.6. Neonatal-Perinatal Medicine:
    - 4.7.5.11.2.7. Pediatrics-Adolescent Medicine: and
    - 4.7.5.11.2.8. Pediatric Psychiatry.
  - 4.7.5.11.3 The MCO shall have adequate Participating Provider networks of pediatric Providers, sub-specialists, children's hospitals, pediatric regional centers and ancillary Providers to provide care to Children with Special Health Care Needs.

Page 162 of 397 Date \_\_\_\_\_

New Hampshire Department of Health and Human Services Medicaid Care Management Services



## **Exhibit B - DRAFT**

4.7.5.11.4	The MCO shall specify, in their listing of mental health and Substance Use Disorder Provider directories, which Providers specialize in children's services.
4.7.5.11.5	The MCO shall ensure that Members have access to specialty centers in and out of NH for diagnosis and treatment of rare disorders.
4.7.5.11.6	The MCO shall permit a Member who meets the definition of Children with Special Health Care Needs following plan enrollment and who requires specialty services to request approval to see a Non-Participating Provider to provide those services if the MCO does not have a Participating specialty Provider with the same level of expertise available.
4.7.5.11.7	The MCO shall develop and maintain a program for Children with Special Health Care Needs, which includes, but is not limited to methods for ensuring and monitoring timely access to pediatric specialists,

4.7.5.11.8 The MCO shall ensure PCPs and specialty care Providers are available to provide consultation to DCYF regarding medical and psychiatric matters for Members who are children in State custody/guardianship.

subspecialists, ancillary therapists and specialized equipment and supplies; these methods may include standing referrals or other methods determined by

4.7.5.12 Access Standards for Behavioral Health

the MCO.

- 4.7.5.12.1 The MCO shall have in its Participating Provider network the capacity to ensure that Transitional Health Care by a Provider shall be available from a primary or specialty Provider for clinical assessment and care planning within two (2) business days of discharge from inpatient or institutional care for physical or mental health disorders or discharge from a Substance Use Disorder treatment program.
- 4.7.5.12.2 Emergency medical and behavioral health care shall be available twenty-four (24) hours a day, seven (7) days a week. Behavioral health care shall be available, and the MCO shall have in its Participating Provider network the capacity to ensure that waiting times for appointments and/or service availability do not exceed the following:

Page 163 of 397 Date \_\_\_\_\_

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

- 4.7.5.12.2.1. Within six (6) hours for a non-life threatening emergency;
- 4.7.5.12.2.2. Within forty-eight (48) hours for urgent care: and
- 4.7.5.12.2.3. Within ten (10) business days for a routine office visit appointment.
- 4.7.5.12.3 American Society of Addiction Medicine (ASAM) Level of Care
  - 4.7.5.12.3.1. The MCO shall ensure Members timely access to care through a network of Participating Providers in each ASAM Level of Care. During the Readiness Review process and in accordance with Exhibit O: Quality and Oversight Reporting Requirements:
    - 4.7.5.12.3.1.1The MCO shall submit a plan describing on-going efforts to continually work to recruit and maintain sufficient networks of Substance Use Disorder service Providers so that services are accessible without unreasonable delays; and
    - 4.7.5.12.3.1.2The MCO shall have a specified number of Providers able to provide services at each level of care required; if supply precludes compliance, the MCO shall notify the Department and, within thirty (30) calendar days, submit an updated plan that identifies the specific steps that shall be taken to increase capacity, including milestones by which to evaluate progress.
- 4.7.5.12.4 The MCO shall ensure that Providers under contract to provide Substance Use Disorder services shall respond to inquiries for Substance Use Disorder services from Members or referring agencies as soon as possible and no later than two (2) business days following the day the call was first received. The Substance Use Disorder Provider is required to conduct an initial eligibility screening for services as

Page 164 of 397 Date \_\_\_\_\_

New Hampshire Department of Health and Human Services **Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

soon as possible, ideally at the time of first contact (face-to-face communication by meeting in person or electronically or by telephone conversation) with the Member or referring agency, but not later than two (2) business days following the date of first contact.

The MCO shall ensure that Members who have 4.7.5.12.5 screened positive for Substance Use Disorder services shall receive an ASAM Level of Care Assessment within two (2) business days of the initial eligibility screening and a clinical evaluation as soon as possible following the ASAM Level of Care Assessment and no later than (3) business days after admission.

4.7.5.12.6 The MCO shall ensure that Members identified for withdrawal management, outpatient or intensive outpatient services shall start receiving services within seven (7) business days from the date ASAM Level of Care Assessment was completed until such a time that the Member is accepted and starts receiving services by the receiving agency. Members identified for partial hospitalization or rehabilitative residential services shall start receiving interim services (services at a lower level of care than that identified by the ASAM Level of Care Assessment) or the identified service type within seven (7) business days from the date the ASAM Level of Care Assessment was completed and start receiving the identified level of care no later than fourteen (14) business days from the date the ASAM Level of Care Assessment was completed.

> If the type of service identified in the ASAM Level of Care Assessment is not available from the Provider that conducted the initial assessment within fortyeight (48) hours, the MCO shall ensure that the Provider provides interim Substance Use Disorder services until such a time that the Member starts receiving the identified level of care. If the type of service is not provided by the ordering Provider, and the ordering Provider does not make a referral for the Covered Service within three (3) business days from initial contact, then the MCO is responsible, in collaboration with the Member's care team, for making a closed loop referral for that type of service (for the identified level of care), and to the applicable Doorway Program location within three (3) business

4.7.5.12.7

Page 165 of 397

Dale	Date
------	------

New Hampshire Department of Health and Human Services Medicaid Care Management Services



#### **Exhibit B - DRAFT**

days thereafter. The MCO is responsible for ensuring that the Member has access to interim Substance Use Disorder services until such a time that the Member is accepted and starts receiving services by the receiving agency.

- 4.7.5.12.8 When the level of care identified by the initial assessment becomes available by the receiving agency or the agency of the Member's choice, Members being provided interim services shall be reassessed for ASAM level of care.
- 4.7.5.12.9 The MCO shall ensure that pregnant women are admitted to the identified level of care within twenty-four (24) hours of the ASAM Level of Care Assessment. If the MCO is unable to admit a pregnant woman for the needed level of care within twenty-four (24) hours, the MCO shall:
  - 4.7.5.12.9.1. Assist the pregnant woman with identifying alternative Providers and with accessing services with these Providers. This assistance shall include actively reaching out to identify Providers on the behalf of the Member;
- 4.7.5.12.10 Provide interim services until the appropriate level of care becomes available at either the agency or an alternative Provider. Interim services shall include: at least one (1) sixty (60) minute individual or group outpatient session per week; Recovery support services as needed by the Member; and daily calls to the Member to assess and respond to any emergent needs.
- 4.7.5.12.11 Pregnant women seeking treatment shall be provided access to childcare and transportation to aid in treatment participation.

#### 4.7.6. Women's Health

- 4.7.6.1 The MCO shall provide Members with direct access to a women's health specialist within the network for Covered Services provide necessary to provide women's routine and preventive health care services. This is in addition to the Member's designated source of primary care if that source is not a women's health specialist. [42 CFR 438.206(b)(2)]
- 4.7.6.2 The MCO shall provide access to Family Planning Services to Members without the need for a referral or prior-authorization.

Page 166 of 397	Date
1 age 100 01 001	

New Hampshire Department of Health and Human Services Medicaid Care Management Services



#### **Exhibit B - DRAFT**

- Additionally, Members shall be able to access these services by Providers whether they are in or out of the MCO's network.
- 4.7.6.3 Enrollment in the MCO shall not restrict the choice of the Provider from whom the Member may receive Family Planning Services and supplies. [Section 1902(a)(23) of the Social Security Act; 42 CFR 431.51(b)(2)]
- 4.7.6.4 The MCO shall only provide for abortions in the following situations:
  - 4.7.6.4.1 If the pregnancy is the result of an act of rape or incest; or
  - 4.7.6.4.2 In the case where a woman suffers from a physical disorder, physical injury, or physical illness, including a life-endangering physical condition, caused by, or arising from, the pregnancy itself, that would, as certified by a physician, place the woman in danger of death unless an abortion is performed. [42 CFR 441.202; Consolidated Appropriations Act of 2008]
- 4.7.6.5 The MCO shall not provide abortions as a benefit, regardless of funding, for any reasons other than those identified in this Agreement.

## 4.7.7. Access to Special Services

- 4.7.7.1 The MCO shall ensure Members have access to DHHS-designated Level I and Level II Trauma Centers within the State, or hospitals meeting the equivalent level of trauma care in the MCO's service area or in close proximity to such service area. The MCO shall have written, out-of-network reimbursement arrangements with the DHHS-designated Level I and Level II Trauma Centers or hospitals meeting equivalent levels of trauma care if the MCO does not include such a Trauma Center in its network.
- 4.7.7.2 The MCO shall ensure accessibility to other specialty hospital services, including major burn care, organ transplantation, specialty pediatric care, specialty out-patient centers for HIV/AIDS, sickle cell disease, hemophilia, cranio-facial and congenital anomalies, home health agencies, and hospice programs. To the extent that the above specialty services are available within the State, the plan shall not exclude NH Providers from its network if the negotiated rates are commercially reasonable.
- 4.7.7.3 The MCO shall only pay for organ transplants when the Medicaid State Plan provides, and the MCO follows written standards that provide for similarly situated Members to be treated alike and for any restriction on facilities or practitioners to be consistent with the accessibility of high-quality care to Members. [Section 1903(i) of the

Page 167 of 397 Date \_\_\_\_\_

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

- Social Security Act, final sentence; section 1903(i)(1) of the Social Security Act]
- 4.7.7.4 The MCO may offer such tertiary or specialized services at socalled "centers of excellence". The tertiary or specialized services shall be offered within the New England region, if available. The MCO shall not exclude NH Providers of tertiary or specialized services from its network provided that the negotiated rates are commercially reasonable.

## 4.7.8. Non-Participating Providers

- 4.7.8.1 If the MCO's network is unable to provide necessary medical, behavioral health or other services covered under the Agreement to a particular Member, the MCO shall adequately and in a timely manner cover these services for the Member through Non-Participating Providers, for as long as the MCO's Participating Provider network is unable to provide them. [42 CFR 438.206(b)(4)].
- 4.7.8.2 The MCO shall inform the Non-Participating Provider that the Member cannot be balance billed.
- 4.7.8.3 The MCO shall coordinate with Non-Participating Providers regarding payment utilizing a single case agreement. For payment to Non-Participating Providers, the following requirements apply:
  - 4.7.8.3.1 If the MCO offers the service through a Participating Provider(s), and the Member chooses to access non-emergent services from a Non-Participating Provider, the MCO is not responsible for payment.
  - 4.7.8.3.2 If the service is not available from a Participating Provider and the Member requires the service and is referred for treatment to a Non-Participating Provider, the payment amount is a matter between the MCO and the Non-Participating Provider.
  - 4.7.8.3.3 The MCO shall ensure that cost to the Member is no greater than it would be if the service were furnished within the network. [42 CFR 438.206(b)(5)]

#### 4.7.9. Access to Providers During Transitions of Care

- 4.7.9.1 The MCO shall use a standard definition of "Ongoing Special Condition" which shall be defined as follows:
  - 4.7.9.1.1 In the case of an acute illness, a condition that is serious enough to require medical care or treatment to avoid a reasonable possibility of death or permanent harm.

age 168 of 397 Date	
---------------------	--

Medicaid Care Management Services Contract New Hampshire Department of Health and Human Services Medicaid Care Management Services

4.7.9.2

4.7.9.3



## **Exhibit B - DRAFT**

4.7.9.1.2	In the case of a chronic illness or condition, a disease or condition that is life threatening, degenerative, or disabling, and requires medical care or treatment over a prolonged period of time.
4.7.9.1.3	In the case of pregnancy, pregnancy from the start of the second trimester.
4.7.9.1.4	In the case of a terminal illness, a Member has a medical prognosis that the Member's life expectancy is six (6) months or less.
4.7.9.1.5	In the case of a child with Special Health Care Needs as defined in Section 4.11.2 (MCO-Delivered Care Management for Required Priority Populations).
transitions in (including o	hall permit that, in the instances when a Member nto the MCO from FFS Medicaid, another MCO ne that has terminated its agreement with the or another type of health insurance coverage and:
4.7.9.2.1	The Member is in ongoing course of treatment, has an Ongoing Special Condition (not including pregnancy or terminal illness), or is a Child with Special Health Care Needs, the Member is permitted to continue seeing their Provider(s), regardless of whether the Provider is a Participating or Non-Participating Provider, for up to ninety (90) calendar days from the Member's enrollment date or until the completion of a medical necessity review, whichever occurs first;
4.7.9.2.2	The Member is pregnant and in the second or third trimester, the Member may continue seeing her Provider(s), whether the Provider is a Participating or Non-Participating Provider, through her pregnancy and up to sixty (60) calendar days after delivery;
4.7.9.2.3	The Member is determined to be terminally ill at the time of the transition, the Member may continue seeing his or her Provider, whether the Provider is a Participating or Non-Participating Provider, for the remainder of the Member's life with respect to care directly related to the treatment of the terminal illness or its medical manifestations.

Page 169 of 397 Date \_\_\_\_\_

The MCO shall permit that, in instances when a Member with an Ongoing Special Condition transitions into the MCO from FFS Medicaid or another MCO and at the time has a currently prescribed medication, the MCO shall cover such medications for ninety (90)

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

calendar days from the Member's enrollment date or until the completion of a medical necessity review, whichever occurs first.

- 4.7.9.4 The MCO shall permit that, in instances in which a Provider in good standing leaves an MCO's network and:
  - 4.7.9.4.1 The Member is in ongoing course of treatment, has a special condition (not including pregnancy or terminal illness), or is a Child with Special Health Care Needs, the Member is permitted to continue seeing their Provider(s), whether the Provider is a Participating or Non-Participating Provider, for up to ninety (90) calendar days;
  - 4.7.9.4.2 The Member is pregnant and in the second or third trimester, the Member may continue seeing her Provider(s), whether the Provider is a Participating or Non-Participating Provider, through her pregnancy and up to sixty (60) calendar days after delivery;
  - 4.7.9.4.3 The Member is determined to be terminally ill at the time of the transition, the Member may continue seeing his or her Provider, whether the Provider is a Participating or Non-Participating Provider, for the remainder of the Member's life with respect to care directly related to the treatment of the terminal illness or its medical manifestations.
- 4.7.9.5 The MCO shall maintain a transition plan providing for Continuity of Care in the event of Agreement termination, or modification limiting service to Members, between the MCO and any of its contracted Providers, or in the event of site closing(s) involving a PCP with more than one (1) location of service. The transition plan shall describe how Members shall be identified by the MCO and how Continuity of Care shall be provided.
- 4.7.9.6 The MCO shall provide written notice of termination of a Participating Provider to all affected Members, defined as those who:
  - 4.7.9.6.1 Have received services from the terminated Provider within the sixty (60)-day period immediately preceding the date of the termination; or
  - 4.7.9.6.2 Are assigned to receive primary care services from the terminated Provider.
- 4.7.9.7 The MCO shall make a good faith effort to give written notice of termination of a contracted Provider, as follows:
  - 4.7.9.7.1 Written notice to the Department, the earlier of: (1) fifteen (15) calendar days after the receipt or

Page 170 of 397 Date \_\_\_\_\_

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

issuance of the termination notice, or (2) fifteen (15) calendar days prior to the effective date of the termination; and

- 4.7.9.7.2 Written notice to each Member who received their care from, or was seen on a regular basis by, the terminated Provider, the later of:
  - 4.7.9.7.2.1. Thirty (30) calendar days prior to the effective date of the termination; or
  - 4.7.9.7.2.2. Fifteen (15) calendar days after receipt or issuance of the termination notice by the terminated Provider.
- 4.7.9.8 The MCO shall have a transition plan in place for affected Members described in this section within three (3) calendar days prior to the effective date of the termination.
- 4.7.9.9 In addition to notification of the Department of Provider terminations, the MCO shall provide reporting in accordance with Exhibit O: Quality and Oversight Reporting Requirements.
- 4.7.9.10 If a Member is in a prior authorized ongoing course of treatment with a Participating Provider who becomes unavailable to continue to provide services, the MCO shall notify the Member in writing within seven (7) calendar days from the date the MCO becomes aware of such unavailability and develop a transition plan for the affected Member.
- 4.7.9.11 If the terminated Provider is a PCP to whom the MCO Members are assigned, the MCO shall:
  - 4.7.9.11.1 Describe in the notice to Members the procedures for selecting an alternative PCP;
  - 4.7.9.11.2 Explain that the Member shall be assigned to an alternative PCP if they do not actively select one; and
  - 4.7.9.11.3 Ensure the Member selects or is assigned to a new PCP within thirty (30) calendar days of the date of notice to the Member.
- 4.7.9.12 If the MCO is receiving a new Member it shall facilitate the transition of the Member's care to a new Participating Provider and plan a safe and medically appropriate transition if the Non-Participating Provider refuses to contract with the MCO.
- 4.7.9.13 The MCO shall actively assist Members in transitioning to a Participating Provider when there are changes in Participating Providers, such as when a Provider terminates its contract with the MCO. The Member's Care Management team shall provide this

Page 171 of 397	Date	

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

assistance to Members who have chronic or acute medical or behavioral health conditions, and Members who are pregnant.

- 4.7.9.14 To minimize disruptions in care, the MCO shall:
  - 4.7.9.14.1 With the exception of Members in their second or third trimester of pregnancy, provide continuation of the terminating Provider's services for up to ninety (90) calendar days or until the Member may be reasonably transferred to a Participating Provider without disruption of care, whichever is less; and
  - 4.7.9.14.2 For Members in their second or third trimester of pregnancy, permit continued access to the Member's prenatal care Provider and any Provider currently treating the Member's chronic or acute medical or behavioral health condition or currently providing LTSS, through the postpartum period.

## 4.7.10. **Second Opinion**

4.7.10.1 The MCO shall provide for a Second Opinion from a qualified health care professional within the Participating Provider network, or arrange for the Member to obtain one (1) outside the network, at no cost to the Member. The MCO shall clearly state its procedure for obtaining a Second Opinion in its Member Handbook. [42 CFR 438.206(b)(3)]

#### 4.7.11. Provider Choice

4.7.11.1 The MCO shall permit each Member to choose their Provider to the extent possible and appropriate. [42 CFR 438.3(I)]

#### 4.8. Utilization Management

#### 4.8.1. Policies and Procedures

- 4.8.1.1 The MCO's policies and procedures related to the authorization of services shall be in compliance with all applicable laws and regulations including but not limited to 42 CFR 438.210 and RSA Chapter 420-E.
- 4.8.1.2 The MCO shall ensure that the Utilization Management program assigns responsibility to appropriately licensed clinicians, including but not limited to physicians, nurses, therapists, and behavioral health Providers (including Substance Use Disorder professionals).

#### 4.8.1.3 Amount, Duration, and Scope

4.8.1.3.1 The MCO shall ensure that each service provided to adults is furnished in an amount, duration and scope that is no less than the amount, duration and scope

Page 172 of 397	Date	
-----------------	------	--

New Hampshire Department of Health and Human Services Medicaid Care Management Services



#### **Exhibit B - DRAFT**

for the same services provided under FFS Medicaid. [42 CFR 438.210(a)(2)]

- 4.8.1.3.2 The MCO shall also provide services for Members under the age of twenty-one (21) to the same extent that services are furnished to individuals under the age of twenty-one (21) under FFS Medicaid. [42 CFR 438.210(a)(2)] Services shall be sufficient in amount, duration, or scope to reasonably achieve the purpose for which the services are furnished. [42 CFR 438.210(a)(3)(i)]
- 4.8.1.3.3 Authorization duration for certain Covered Services shall be as follows:
  - 4.8.1.3.3.1. Private duty nursing authorizations shall be issued for no less than six (6) months unless the Member is new to the private duty nursing benefit. Initial authorizations for Members new to the private duty nursing benefit shall be no less than two (2) weeks;
  - 4.8.1.3.3.2. Personal Care Attendant (PCA) authorizations shall be issued for no less that one (1) year unless the Member is new to the PCA benefit. Initial authorizations for Members new to the PCA benefit shall be no less than three (3) months.
  - 4.8.1.3.3. Occupational therapy, physical therapy, and speech therapy authorizations that exceed the service limit of twenty (20) visits for each type of therapy shall be issued for no less than three (3) months initially. Subsequent authorizations for continuation of therapy services shall be issued for no less than six (6) months if the therapy is for habilitative purposes directed at functional impairments.
- 4.8.1.4 Written Utilization Management Policies
  - 4.8.1.4.1 The MCO shall develop, operate, and maintain a Utilization Management program that is documented through a program description and defined structures, policies, and procedures that are reviewed and approved by the Department. The MCO shall

Page 173 of 397 Date \_\_\_\_\_

New Hampshire Department of Health and Human Services **Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

ensure that the Utilization Management Program has
criteria and policies that:

- 4.8.1.4.1.1. Are practicable, objective and based on evidence-based criteria, to the extent possible:
- 4.8.1.4.1.2. Are based on current, nationally accepted standards of medical practice and are developed with input from appropriate actively practicing practitioners in the MCO's service area, and are consistent with the Practice Guidelines described in Section 4.8.2 (Practice Guidelines and Standards);
- Are reviewed annually and updated as 4.8.1.4.1.3. appropriate. including as new applications, treatments. and technologies emerge (the Department shall approve any changes to the clinical criteria before the criteria are utilized);
- 4.8.1.4.1.4. Are applied based on individual needs and circumstances (including healthrelated social needs);
- 4.8.1.4.1.5. Are applied based on an assessment of the local delivery system;
- 4.8.1.4.1.6. Involve appropriate practitioners in developing, adopting and reviewing the criteria: and
- 4.8.1.4.1.7. Conform to the standards of NCQA Health Plan Accreditation as required by Section 4.14.2 (Health Plan Accreditation).
- 4.8.1.4.2 The MCO's written Utilization Management policies, procedures, and criteria shall describe the categories of health care personnel that perform utilization review activities and where they are licensed. Such policies, procedures and criteria shall address, at a

  - certification;

minimum: 4.8.1.4.2.1. Second Opinion programs; 4.8.1.4.2.2. Pre-hospital admission certification; 4.8.1.4.2.3. Pre-inpatient service eligibility Page 174 of 397 Date \_\_\_\_

Medicaid Care Management Services Contract New Hampshire Department of Health and Human Services Medicaid Care Management Services



## **Exhibit B - DRAFT**

		4.8.1.4.2.4.	Concurrent hospital review to determine appropriate length of stay;
		4.8.1.4.2.5.	The process used by the MCO to preserve confidentiality of medical information.
	4.8.1.4.3	communica	least thirty (30) calendar days in advance
	4.8.1.4.4	shall be sub	tion Management Program descriptions omitted by the MCO to the Department for approval prior to the Program Start Date.
	4.8.1.4.5	Manageme accordance	the MCO shall report on the Utilization on the Program as part of annual reporting in with Exhibit O: Quality and Oversight Requirements.
	4.8.1.4.6	Utilization M	shall communicate any changes to Management processes at least thirty (30) bys prior to implementation.
	4.8.1.4.7	procedures	written Utilization Management policies, and criteria shall be made available upon the Department, Participating Providers, ers.
	4.8.1.4.8	Committee committee Managemer with Exhibi	shall provide the Medical Management (or the MCO's otherwise named responsible for medical Utilization (nt) reports and minutes in accordance to C. Quality and Oversight Reporting (nts. [42 CFR 438.66 (c)(7)]
4.8.1.5	Service Li	mits	
	4.8.1.5.1	on the basis CFR 438.2 provided th	may place appropriate limits on a service s of criteria such as medical necessity [42 210(a)(4)(i)]; or for utilization control, e services furnished can reasonably be to achieve their purpose. [42 CFR

438.210(a)(4)(ii)(A)]

4.8.1.5.2.1.

for utilization control, provided:

4.8.1.5.2

Page 175 of 397 Date \_\_\_\_\_

The MCO may place appropriate limits on a service

The services supporting Members with

ongoing or Chronic Conditions are authorized in a manner that reflects the

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

Member's ongoing need for such services and supports [42 CFR 438.210(a)(4)(ii)(B)]. This includes allowance for up to six (6) skilled nursing visits per benefit period without a Prior Authorization; and

4.8.1.5.2.2. Family Planning Services are provided in a manner that protects and enables the Member's freedom to choose the method of Family Planning to be used. [42 CFR 438.210(a)(4)(ii)(C)]

#### 4.8.1.6 Prior Authorization

- 4.8.1.6.1 The MCO and, if applicable, its Subcontractors shall have in place and follow written policies and procedures as described in the Utilization Management policies for processing requests for initial and continuing authorizations of services and including conditions under which retroactive requests shall be considered. Any Prior Authorization for Substance Use Disorder shall comply with RSA 420-J:17 and RSA 420-J:18 as described in Section 4.13.33.3 (Limitations on Prior Authorization Requirements). [42 CFR 438.210(b)(1)]
- 4.8.1.6.2 Authorizations shall be based on a comprehensive and individualized needs assessment that addresses all needs including health-related social needs and a subsequent person-centered planning process.
- 4.8.1.6.3 The MCO's Prior Authorization requirements shall comply with parity in mental health and Substance Use Disorder, as described in Section 4.13.18.4 (Restrictions on Treatment Limitations). [42 CFR 438.910(d)]
- 4.8.1.6.4 The MCO shall use the NH MCM standard Prior Authorization form, as applicable. The MCO shall also work in good faith with the Department, as initiated by the Department, to adopt Prior Authorization form practices with consistent information and documentation requirements from Providers wherever feasible. Providers shall be able to submit the Prior Authorizations forms electronically, by mail, or fax.
- 4.8.1.6.5 The MCO shall have in effect mechanisms to ensure consistent application of review criteria for

Page 176 of 397 Date \_\_\_\_\_

Medicaid Care Management Services Contract
New Hampshire Department of Health and Human Services
Medicaid Care Management Services



# **Exhibit B - DRAFT**

	authorization decisions, including but not limited to interrater reliability monitoring, and consult with the requesting Provider when appropriate and at the request of the Provider submitting the authorization [42 CFR 438.210(b)(2)(i)-(ii)].
4.8.1.6.6	The MCO shall ensure that any decision to deny a service authorization request or to authorize a service in an amount, duration, or scope that is less than requested, be made by a health care professional who has appropriate clinical expertise in treating the Member's condition or disease. [42 CFR 438.210(b)(3)]
4.8.1.6.7	The MCO shall not arbitrarily deny or reduce the amount, duration, or scope of a required service solely because of the diagnosis, type of illness, or condition of the Member.
4.8.1.6.8	The MCO shall comply with all relevant federal regulations regarding inappropriate denials or reductions in care. [42 CFR 438.210(a)(3)(ii)]
4.8.1.6.9	The MCO shall issue written denial notices within timeframes specified by federal regulations and this Agreement.
4.8.1.6.10	The MCO shall permit Members to appeal service determinations based on the Grievance and Appeal Process required by federal law and regulations and this Agreement.
4.8.1.6.11	Compensation to individuals or entities that conduct Utilization Management activities shall not be structured so as to provide incentives for the individual or entity to deny, limit, or discontinue Medically Necessary services to any Member. [42 CFR 438.210(e)]
4.8.1.6.12	Medicaid State Plan services and/or pharmaceutical Prior Authorizations, including those for specialty drugs, in place at the time a Member transitions to an MCO shall be honored for ninety (90) calendar days or until completion of a medical necessity review, whichever comes first.
4.8.1.6.13	The MCO shall, in the Member Handbook, provide information to Members regarding Prior Authorization in the event the Member chooses to transfer to another MCO.

Date \_\_\_\_\_ Page 177 of 397

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

- 4.8.1.6.14 Upon receipt of Prior Authorization information from the Department, the new MCO shall honor Prior Authorizations in place by the former MCO as described in Section 4.7.9. (Access to Providers During Transitions of Care). The new MCO shall review the service authorization in accordance with the urgent determination requirements of Section 4.8.4.2 (Urgent Determinations and Covered/Extended Services).
- 4.8.1.6.15 In the event that the Prior Authorization specifies a specific Provider, that MCO shall continue to utilize that Provider, regardless of whether the Provider is a Participating Provider, until such time as services are available in the MCO's network.
- 4.8.1.6.16 The MCO shall ensure that the Member's needs are met continuously and shall continue to cover services under the previously issued Prior Authorization until the MCO issues new authorizations that address the Member's needs.
- 4.8.1.6.17 The MCO shall ensure that Subcontractors or any other party performing utilization review are licensed in NH in accordance with Section 3.10.2 (Contracts with Subcontractors).
- 4.8.1.6.18 The MCO shall ensure that Subcontractors or any other party performing utilization reviews applicable to inpatient psychiatric treatment at New Hampshire Hospital and other State determined IMDs for mental illness, conduct authorization for services as follows:
  - 4.8.1.6.18.1. For a Member's initial admission, an automatic five (5) business days (excluding holidays) shall be authorized for the Member's initial involuntary emergency psychiatric admission to an IMD facility.
  - 4.8.1.6.18.2. Reauthorization of the Member's continuous admission, shall be rendered promptly within 24 hours of the request for reauthorization of the initial involuntary emergency psychiatric admission.

#### 4.8.2. Practice Guidelines and Standards

4.8.2.1 The MCO shall adopt evidence-based clinical Practice Guidelines in compliance with 42 CFR 438.236 and with NCQA's requirements

Page 178 of 397	Date

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

for health plan accreditation. T	ne Practice Guidelines adopted by
the MCO shall:	

- 4.8.2.1.1 Be based on valid and reasonable clinical evidence or a consensus of Providers in the particular field,
- 4.8.2.1.2 Consider the needs of the MCO's Members,
- 4.8.2.1.3 Be adopted in consultation with Participating Providers, and
- 4.8.2.1.4 Be reviewed and updated periodically as appropriate. [42 CFR 438.236(b)(1-3); 42 CFR 438.236(b)(4)]
- 4.8.2.2 The MCO shall develop Practice Guidelines based on the health needs and opportunities for improvement identified as part of the QAPI Program.
- 4.8.2.3 The MCO shall adopt Practice Guidelines consistent with the standards of care and evidence-based practices of specific professional specialty groups, as identified by the Department. These include, but are not limited to:
  - 4.8.2.3.1 ASAM, as further described in Section 4.13.26 (Substance Use Disorder Clinical Evaluations and Treatment Plans);
  - 4.8.2.3.2 The recommendations of the U.S. Preventive Services Task Force for the provision of primary and secondary care to adult, adolescent, and pediatric populations, rated A or B; as well as State specified requirements which include, but are not limited to, pediatric lead testing rates of fifty-five percent (55%) for 12-month olds and forty-four percent (44%) for 24 month olds in the first year of the Agreement, increasing by five percent 5% each year thereafter until the final year of the Agreement when the goals will be seventy-five percent (75%) for 12-month olds and sixty-four percent (64%) for 24-month olds.
  - 4.8.2.3.3 The preventive services recommended by the AAP Bright Futures program; and
  - 4.8.2.3.4 The Zero Suicide Consensus Guide for Emergency Departments.
- 4.8.2.4 The MCO may substitute generally recognized, accepted guidelines to replace the U.S. Preventive Services Task Force and AAP Bright Futures program requirements, provided that the MCO meets all other Practice Guidelines requirements indicated within this Section 4.8.2 (Practice Guidelines and Standards) of the

Page 179 of 397 Date \_\_\_\_\_

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

- Agreement and that such substitution is reviewed by the Department prior to implementation.
- 4.8.2.5 The MCO shall disseminate Practice Guidelines to the Department and all affected Providers and make Practice Guidelines available, including but not limited to the MCO's website, and, upon request, to Members and potential Members. [42 CFR 438.236(c)]
- 4.8.2.6 The MCO's decisions regarding Utilization Management, Member education, and coverage of services shall be consistent with the MCO's clinical Practice Guidelines. [42 CFR 438.236(d)]

#### 4.8.3. Medical Necessity Determination

- 4.8.3.1 The MCO shall specify what constitutes "Medically Necessary" services in a manner that:
  - 4.8.3.1.1 Is no more restrictive than the NH DHHS FFS Medicaid program including quantitative and non-quantitative treatment limits, as indicated in State laws and regulations, the Medicaid State Plan, and other State policies and procedures [42 CFR 438.210(a)(5)(i)]; and
  - 4.8.3.1.2 Addresses the extent to which the MCO is responsible for covering services that address [42 CFR 438.210(a)(5)(ii)(A)-(C)]:
    - 4.8.3.1.2.1. The prevention, stabilization, diagnosis, and treatment of a Member's diseases, condition, and/or disorder that results in health impairments and/or disability;
    - 4.8.3.1.2.2. The ability for a Member to achieve ageappropriate growth and development; and
    - 4.8.3.1.2.3. The ability for a Member to attain, maintain, or regain functional capacity.
- 4.8.3.2 For Members twenty-one (21) years of age and older, "Medically Necessary" shall be as defined in Section 2.1 (Definitions).
- 4.8.3.3 For Members under twenty-one (21) years of age, per EPSDT, "Medically Necessary" shall be as defined in Section 2.1 (Definitions).

#### 4.8.4. **Notices of Coverage Determinations**

4.8.4.1 The MCO shall provide the requesting Provider and the Member with written notice of any decision by the MCO to deny a service authorization request, or to authorize a service in an amount,

age 180 of 397 Date	age 180 of 397 Date
---------------------	---------------------

New Hampshire Department of Health and Human Services **Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

duration, or scope that is less than requested. The notice shall meet the requirements of 42 CFR 438.210(c) and 438.404.

- 4.8.4.2 Urgent Determinations and Continued/Extended Services
  - The MCO shall make Utilization Management 4.8.4.2.1 decisions in a timely manner. The following minimum standards shall apply:
    - 4.8.4.2.1.1. Urgent Determinations: Determination of an authorization involving urgent care shall be made as soon as possible, taking into account the medical exigencies, but in no event later than seventy-two (72) hours after receipt of the request for service for ninety-eight percent (98%) of requests, unless the Member or Member's representative fails to provide sufficient information to determine whether, or to what extent, benefits are covered or payable. [42] 438.210(d)(2)(i); CFR 42 438.404(c)(6)]
    - 4.8.4.2.1.2. In the case of such failure, the MCO shall notify the Member or Member's representative within twenty-four (24) hours of receipt of the request and shall advise the Member or Member's representative of the specific information necessary to make a coverage determination.
    - 4.8.4.2.1.3. The Member Member's or representative shall be afforded a reasonable amount of time, taking into account the circumstances, but not less than forty-eight (48) hours, to provide the specified information.
    - Thereafter, notification of the benefit 4.8.4.2.1.4. determination shall be made as soon as possible, but in no case later than fortyeight (48) hours after the earlier of the MCO's receipt of the specified additional information; or the end of the period afforded the Member or Member's representative to provide the specified additional information.

Page 181 of 397

Date					

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

4.8.4.2.1.5. Continued/Extended Services: The determination of an authorization involving urgent care and relating to the extension of an ongoing course of treatment and involving a question of medical necessity shall be made within twenty-four (24) hours of receipt of the request for ninety-eight percent (98%) of requests, provided that the request is made at least twenty-four (24) hours prior to the expiration of the prescribed period of time or course of treatment.

#### 4.8.4.3 All Other Determinations

- 4.8.4.3.1 The determination of all other authorizations for preservice benefits shall be made within a reasonable time period appropriate to the medical circumstances, but shall not exceed fourteen (14) calendar days for ninety-five percent (95%) of requests after the receipt of a request.
- 4.8.4.3.2 An extension of up to fourteen (14) calendar days is permissible for non-diagnostic radiology determinations if the Member or the Provider requests the extension, or the MCO justifies a need for additional information.
- 4.8.4.3.3 If an extension is necessary due to a failure of the Member or Member's representative to provide sufficient information to determine whether, or to what extent, benefits are covered as payable, the notice of extension shall specifically describe the required additional information needed, and the Member or Member's representative shall be given at least forty-five (45) calendar days from receipt of the notice within which to provide the specified information.
- 4.8.4.3.4 Notification of the benefit determination following a request for additional information shall be made as soon as possible, but in no case later than fourteen (14) calendar days after the earlier of:
  - 4.8.4.3.4.1. The MCO's receipt of the specified additional information; or
  - 4.8.4.3.4.2. The end of the period afforded the Member or Member's representative to provide the specified additional information.

Page 182 of 397	Date

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

- 4.8.4.3.4.3. When the MCO extends the timeframe, the MCO shall give the Member written notice of the reason for the decision to extend the timeframe and inform the Member of the right to file a grievance if he or she disagrees with that decision. Under such circumstance, the MCO issue and shall carry out determination as expeditiously as the Member's health condition requires and no later than the date the extension expires.
- 4.8.4.3.5 Ninety-five percent (95%) of post service authorization determinations shall be made within thirty (30) calendar days of the date of filing. In the event the Member fails to provide sufficient information to determine the request, the MCO shall notify the Member within fifteen (15) calendar days of the date of filing, as to what additional information is required to process the request and the Member shall be given at least forty-five (45) calendar days to provide the required information.
- 4.8.4.3.6 The thirty (30) calendar day period for determination shall be tolled until such time as the Member submits the required information.
- 4.8.4.3.7 Whenever there is an adverse determination, the MCO shall notify the ordering Provider and the Member. For an adverse standard authorization decision, the MCO shall provide written notification within three (3) calendar days of the decision.
- 4.8.4.3.8 The MCO shall provide Utilization Management Confidential Data to include but not be limited to timely processing, results, and frequency of service authorizations in accordance with Exhibit O: Quality and Oversight Reporting Requirements.

#### 4.8.5. Advance Directives

- 4.8.5.1 The MCO shall adhere to all State and federal laws pertaining to Advance Directives including, but not limited to, RSA 137-J:20.
- 4.8.5.2 The MCO shall maintain written policies and procedures that meet requirements for Advance Directives in Subpart I of 42 CFR 489.
- 4.8.5.3 The MCO shall adhere to the definition of Advance Directives as defined in 42 CFR 489.100.

Page 183 of 397	Date

New Hampshire Department of Health and Human Services Medicaid Care Management Services



## **Exhibit B - DRAFT**

- 4.8.5.4 The MCO shall maintain written policies and procedures concerning Advance Directives with respect to all adult Members. [42 CFR 438.3(j)(1)-(2); 42 CFR 422.128(a); 42 CFR 422.128(b); 42 CFR 489.102(a)]
- 4.8.5.5 The MCO shall educate staff concerning policies and procedures on Advance Directives. [42 CFR 438.3(j)(1)-(2); 42 CFR 422.128(b)(1)(ii)(H); 42 CFR 489.102(a)(5)]
- 4.8.5.6 The MCO shall not condition the provision of care or otherwise discriminate against a Member or potential Member based on whether or not the Member has executed an Advance Directive. [42 CFR 438.3(j)(1)-(2); 42 CFR 422.128(b)(1)(ii)(F); 42 CFR 489.102(a)(3)]
- 4.8.5.7 The MCO shall provide information in the Member Handbook with respect to how to exercise an Advance Directive, as described in Section 4.4.1.4 (Member Handbook). [42 CFR 438.10(g)(2)(xii); 42 CFR 438.3(j)]
- 4.8.5.8 The MCO shall reflect changes in State law in its written Advance Directives information as soon as possible, but no later than ninety (90) calendar days after the effective date of the change. [42 CFR 438.3(j)(4)]

#### 4.9. Member Education and Incentives

#### 4.9.1. **General Provisions**

- 4.9.1.1 The MCO shall develop and implement evidenced-based wellness and prevention programs for its Members. The MCO shall seek to promote and provide wellness and prevention programming aligned with similar programs and services promoted by the Department, including the National Diabetes Prevention Program. The MCO shall also participate in other public health initiatives at the direction of the Department.
- 4.9.1.2 The MCO shall provide Members with general health information and provide services to help Members make informed decisions about their health care needs. The MCO shall encourage Members to take an active role in shared decision-making.
- 4.9.1.3 The MCO shall promote personal responsibility through the use of incentives and care management. The MCO shall reward Members for activities and behaviors that promote good health, health literacy and Continuity of Care. The Department shall review and approve all reward activities proposed by the MCO prior to their implementation.

## 4.9.2. **Member Health Education**

Page 184 of 397	Date	
-----------------	------	--

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

- 4.9.2.1 The MCO shall develop and initiate a Member health education program that supports the overall wellness, prevention, and Care Management programs, with the goal of empowering patients to actively participate in their health care.
- 4.9.2.2 The MCO shall actively engage Members in both wellness program development and in program participation and shall provide additional or alternative outreach to Members who are difficult to engage or who utilize EDs inappropriately.

## 4.9.3. **Member Cost Transparency**

- 4.9.3.1 The MCO shall publish on its website and incorporate in its Care Coordination programs cost transparency information related to the relative cost of Participating Providers for MCO-selected services and procedures, with clear indication of which setting and/or Participating Provider is most cost-effective, referred to as "Preferred Providers."
- 4.9.3.2 The cost transparency information published by the MCO shall be related to select, non-emergent services, designed to permit Members to select between Participating Providers of equal quality, including the appropriate setting of care as assessed by the MCO. The services for which cost transparency data is provided may include, for example, services conducted in an outpatient hospital and/or ambulatory surgery center. The MCO should also include information regarding the appropriate use of EDs relative to low-acuity, non-emergent visits.
- 4.9.3.3 The information included on the MCO's website shall be accessible to all Members and also be designed for use specifically by Members that participate in the MCO's Reference-Based Pricing Incentive Program, as described in Section 4.9.4 (Member Incentive Programs) of this Agreement.

## 4.9.4. **Member Incentive Programs**

- 4.9.4.1 The MCO shall develop at least one (1) Member Healthy Behavior Incentive Program and at least one (1) Reference-Based Pricing Incentive Program, as further described within this section of the Agreement. The MCO shall ensure that all incentives deployed are cost-effective and have a linkage to the APM initiatives described in Section 4.15 (Alternative Payment Models) of this Agreement as appropriate.
- 4.9.4.2 For all Member Incentive Programs developed, the MCO shall provide to participating Members that meet the criteria of the MCO-designed program cash or other incentives that:
  - 4.9.4.2.1 May include incentives such as gift cards for specific retailers, vouchers for a farmers' market,

age 185 of 397	Date

Medicaid Care Management Services Contract
New Hampshire Department of Health and Human Services
Medicaid Care Management Services



# **Exhibit B - DRAFT**

		contribution used for membership			ints that machases,	ay be gym
	4.9.4.2.2		given fiscal year otal monetary va (\$250.00).			
	4.9.4.2.3	and approv	hall submit to that al all Member rior to implemen	Incentive		
4.9.4.3		•	al, the MCO by the Departm		clude adeo	quate
	4.9.4.3.1	The progra	m meets the rec t; and	quiremen	ts of the S	Social
	4.9.4.3.2	Department Behavior In	m meets the cr as described in centive Prograr Based Pricing In	Section on section of the section of	4.9.4.5 (He Section 4.	althy 9.4.6
4.9.4.4	results of ar twelve (12) indicated by	ny Member I months, inc the Departme	the Departmer ncentive Progra luding the follow ent, in accordand Requirements:	ms in eff wing met	fect in the trics and	prior hose
	4.9.4.4.1		er of Members as determined b	•	•	arget
	4.9.4.4.2	payments,	r of Members that and the num amount as a res	ber that	received	the
	4.9.4.4.3	The total va	lue of the incent	ive paym	ents;	
	4.9.4.4.4	An analysis program; ar	of the statistica nd	lly releva	nt results o	of the
	4.9.4.4.5	Identificatio informed by	n of goals and obter the data.	ojectives t	for the next	year
4.9.4.5	Healthy Beh	avior Incentiv	e Programs			
	4.9.4.5.1		hall develop and r Healthy Beha :			
	4	.9.4.5.1.1.	Incorporate incomplete a		or Members Screening	

Page 186 of 397 Date \_\_\_\_\_

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

	compliance with Section 4.10.2 of this Agreement;
4.9.4.5.1.2.	Increase the timeliness of prenatal care, particularly for Members at risk of having a child with NAS;
4.9.4.5.1.3.	Address obesity;
4.9.4.5.1.4.	Prevent diabetes;
4.9.4.5.1.5.	Support smoking cessation;
4.9.4.5.1.6.	Increase lead screening rates in one- and two-year old Members; and/or
4.9.4.5.1.7.	Other similar types of healthy behavior incentive programs in consultation with the collaboration with the Department's Division of Public Health New Hampshire Tobacco Cessation Program, Quitline.

# 4.9.4.6 Reference-Based Pricing Incentive Programs

4.9.4.6.1

The MCO shall develop at least one (1) Reference-Based Pricing Member Incentive Program that encourages Members to use, when reasonable, Preferred Providers as assessed and indicated by the MCO and on its website in compliance with the Cost Transparency requirements included in Section 4.9.3 (Member Cost Transparency). The Reference-Based Pricing Member Incentive Program shall also include means for encouraging members' appropriate use of EDs and opportunities to direct Members to other settings for low acuity, non-emergent visits.

4.9.4.6.2 The MCO's Reference-Based Pricing Member Incentive Program shall be designed such that the Member may gain and lose incentives (e.g., through the development of a points system that is monitored throughout the year) based on the Member's adherence to the terms of the program throughout the course of the year.

#### 4.9.5. Collaboration with New Hampshire Tobacco Cessation Programs

- 4.9.5.1 The MCO shall promote and utilize the Department-approved tobacco treatment quitline, 1-800-QUITNOW (1-800-784-8669) to provide:
  - 4.9.5.1.1 Intensive tobacco cessation treatment through a DHHS-approved tobacco cessation quitline;

Page 187 of 397 Date
----------------------

New Hampshire Department of Health and Human Services Medicaid Care Management Services

4.9.5.2



## **Exhibit B - DRAFT**

4.9.5.1.2	Individual tobacco cessation coaching/counseling in conjunction with tobacco cessation medication;
4.9.5.1.3	The following FDA-approved over-the-counter agents: nicotine patch; nicotine gum; nicotine lozenge; and any future FDA-approved therapies, as indicated by the Department; and
4.9.5.1.4	Combination therapy, when available through quitline, meaning the use of a combination of medicines, including but not limited to: long-term nicotine patch and other nicotine replacement therapy (gum or nasal spray); nicotine patch and inhaler; or nicotine patch and bupropion sustained-release.
The MCO sha minimum:	nall provide tobacco cessation treatment to include, at
4.9.5.2.1	Tobacco cessation coaching/counseling in addition to the quitline;
4.9.5.2.2	In addition to the quitline, the following FDA-approved over-the-counter agents: nicotine patch; nicotine gum; nicotine lozenge; and any future FDA-approved therapies, as indicated by the Department;
4.9.5.2.3	In addition to the quitline, Combination therapy, meaning the use of a combination of medicines, including but not limited to: long-term nicotine patch and other nicotine replacement therapy (gum or nasal spray); nicotine patch and inhaler; or nicotine patch and bupropion sustained-release; and
4.9.5.2.4	Covered FDA-approved tobacco cessation prescription medications that qualify for rebates under the Medicaid Prescription Drug Rebate Program, including:
4	.9.5.2.4.1. Non-nicotine prescription medications; and
4.9.5.2.5	Inhalers and nasal sprays.
	all report on tobacco cessation activities in accordance O: Quality and Oversight Reporting Requirements.

## 4.10. Primary Care and Prevention Focused Care Model

## 4.10.1 General Requirements

4.9.5.3

4.10.1.1 Under the Primary Care and Prevention Focused Model of Care, Primary Care services shall be furnished by or through a general

Page 188 of 397 Date \_\_\_\_\_

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

practitioner, family physician, internal medicine physician, obstetrician/gynecologist, pediatrician, physician assistant, or nurse practitioner, including alternative provider types as designated by the Department.

- 4.10.1.2 The MCO shall ensure that the Primary Care and Preventive Care Focused Model of Care shall be administered in accordance with this Agreement, including:
  - 4.10.1.2.1 Assurance of comprehensive PCP participation in the Model of Care wholly supported by the MCO;
  - 4.10.1.2.2 Guaranteed access to related services for all Members;
  - 4.10.1.2.3 Delivery of services in a manner that are both clinically and developmentally appropriate, patient-focused, and that consider the Member's, parent's, caregiver's and other networks of support the Member may rely upon, in accordance with this Agreement and all applicable State and federal laws and regulations:
  - 4.10.1.2.4 PCP (and other Providers who share responsibility for primary care of the Member) responsibility for Provider-Delivered Care Coordination as described at Section 4.1.7 (Provider-Delivered Care Coordination and Integration with Social Services and Community Care) of this Agreement consistent with Practice Guidelines and Standards required and stipulated in Section 4.8.2, including a plan for integration of these programs;
  - 4.10.1.2.5 Member's selection or assignment of a PCP within fifteen (15) calendar days of enrollment with the MCO:
  - 4.10.1.2.6 Completion of the Member welcome call as stipulated at Section 4.4..4.3 (Welcome Call);
  - 4.10.1.2.7 Member receipt of a Wellness Visit with their PCP, as defined in Section 4.10.3 (Wellness Visits), at least once annually. If the Member is assigned a new PCP, the MCO shall ensure the Member receives a Wellness Visit with the new PCP regardless of when the last Wellness Visit occurred with another Provider;
  - 4.10.1.2.8 Initial and regular reporting to PCPs the names of Members attributed to the PCP's panel within thirty (30) calendar days of PCP assignment or selection,

Page 189 of 397	Date	
-		

New Hampshire Department of Health and Human Services **Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

including the date of the attributed Member's last Wellness Visit and HRA Screening, as available, and/or the absence of such visit and screenings if there have been none:

- 4.10.1.2.9 Demonstration of the authentic engagement between the Member and PCP. At a minimum, as demonstrated through claim encounters initially within ninety (90) days of PCP selection/assignment, and routinely thereafter.
- 4.10.1.2.10 Provider reimbursement for provision of the following Member services:
  - 4.10.1.2.10.1. Wellness Visits in accordance with 4.10.3 (Wellness Section Visits), including assurance there are no barriers to professional claim billing and payment outside of a Wellness Visit for U.S. Preventive Services Task Force (USPSTF) recommended services that utilize a standardized tool in the screening for obesity. anxietv. depression and suicide risk, unhealthy alcohol use, unhealthy drug use, and falls prevention;
  - 4.10.1.2.10.2. HRA Screening as stipulated in Section 4.10.2 (Health Risk Assessment (HRA) Screening) can occur during a visit for any separate acute service and is not solely restricted to a Wellness Visit;
  - 4.10.1.2.10.3. Preventive screenings in accordance with the Practice Guidelines and Standards (Section 4.8.2), including but not limited to the recommendations of the U.S. USPSTF for the provision of primary and secondary care for adult, adolescent, and pediatric populations, rated Level A or B and other preventive screening and services as required by
  - 4.10.1.2.10.4. Medically Necessary diagnostic and treatment Covered Services based on the findings or risk factors identified in the annual Wellness Visit. completion of a HRA Screening, or during routine,

the Department; and urgent, or emergent health care visits.

Page 190 of 397 Date \_\_\_\_\_

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

- 4.10.1.2.11 Provider and Member incentives for completion of the following:
  - 4.10.1.2.11.1. A Wellness Visit;
  - 4.10.1.2.11.2. A HRA Screening; and
  - 4.10.1.2.11.3. Preventive screenings.
- 4.10.1.3 Support the PCP to engage Members to complete the HRA Screening in accordance with Section 4.10.2.
- 4.10.1.4 Comprehensive Medication Reviews for children and adults meeting Polypharmacy criteria as stipulated in Section 4.2.6.
- 4.10.1.5 Provider-Delivered Care Coordination utilization of closed-loop referral processes in accordance with Section 4.11.7 (Provider-Delivered Care Coordination and Integration with Social Services and Community Care), including:
  - 4.10.1.5.1 PCP initiation and coordination of closed-loop referrals for clinical and non-clinical services the Member needs, including but not limited to Behavioral Health Services and health-related social needs, with the Provider remaining engaged with clinical and non-clinical Provider(s) throughout the course of treatment for the referred service(s);
  - 4.10.1.5.2 PCP adoption and utilization of closed-loop referral processes, and the Department's closed-loop referral system, as it becomes available, to promote efficiency and optimal communication among and between Providers; and
  - 4.10.1.5.3 PCP education and training to ensure that the PCP knows when and how to utilize a closed-loop referral system.
- 4.10.1.6 The MCO shall ensure the Primary Care and Prevention Focused Model satisfies care and coordination of services as follows: [42.CFR 438.208]
  - 4.10.1.6.1 The MCO shall ensure that each Member has a designated PCP who shall serve as an ongoing source of case appropriate to his or her needs and the Member shall be provided information on how to contact their designated PCP;
  - 4.10.1.6.2 The MCO shall provide Care Management services for times at which a Member does not have an established and designated PCP (e.g., corrections populations, DCYF children and youth);

Page 191 of 397 Date \_\_\_\_\_

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

4.10.1.6.3	The MCO shall also cover Transitions of Care
	Management (TCM) codes for Participating Providers
	to perform care transition assistance including
	coordinating appropriate services between settings of
	care;

- 4.10.1.6.4 The MCO shall make best effort to connect each Member to a PCP and to pay network PCPs to conduct an initial screening of each Member's needs within ninety (90) calendar days of the effective date of enrollment for new Members:
- 4.10.1.6.5 The MCO shall request documentation from the Participating Provider regarding HRA Screenings and initial screenings of Member needs should the MCO determine that additional entities such as the State, PIHPs and PAHPs serving the enrollee to prevent duplication of those activities;
- 4.10.1.6.6 The MCO shall ensure Participating Providers that furnish services to Members maintains and shares, as appropriate, Member health records in accordance with professional standards; and
- 4.10.1.6.7 The MCO shall ensure that in the process of coordinating care, each Member's privacy is protected in accordance with state and federal privacy requirements in to the extent that they are applicable. [45 CFR parts 160 and 164 subparts A and E]

#### 4.10.2 Health Risk Assessment (HRA) Screening

- 4.10.2.1 The HRA Screening process shall identify the need for the Member's Care Coordination and Care Management services and the need for clinical and non-clinical services, including closed-loop referrals to specialists, not limited to Behavioral Health services Providers, and community resources.
- 4.10.2.2 The MCO shall implement a process to allow professional services billing and payment for Participating Providers who complete and review a Member's HRA Screening, and shall create incentive programs to facilitate the Participating Provider's completion and review of the HRA Screening.
- 4.10.2.3 The MCO shall ensure a HRA Screening of all existing and newly enrolled Members within ninety (90) calendar days of the effective date of MCO enrollment and annually thereafter to identify Members who may have unmet health care needs. [42 CFR 438.208(c)(1)]

Page 192 of 397 Date	Page 192 of 397	Date
----------------------	-----------------	------

New Hampshire Department of Health and Human Services Medicaid Care Management Services



- 4.10.2.4 The HRA Screening tool shall be the same for each MCO. The HRA Screening tool shall be developed by the Department and made available for Provider use.
- 4.10.2.5 The MCO shall empower and support the PCP to engage Members to complete the HRA screening in an agency office/clinic setting, during a scheduled home visit or medical appointment. The HRA Screening may be conducted in-person or through a HIPAA compliant electronic means, telephonic means, or through completion of the written form by the Member. The MCO shall verify the PCP has made at least three (3) reasonable attempts to contact a Member at the phone number and address most recently reported by the Member. [42 CFR 438.208(b)(3)]
  - 4.10.2.5.1 For Members determined eligible for Community Mental Health services pursuant to He-M 401, the MCO shall encourage the Member's PCP to coordinate completion of the HRA Screening (Section 4.10.2) with the Member's applicable Community Mental Health program (a Community Mental Health Center) or other Community Mental Health Provider, if the Member consents, to enable the Community Mental Health Provider to provide support for effective completion of the Health Risk Assessment Screening by the PCP and the Member.
- 4.10.2.6 The MCO shall report the number of Members who received a HRA Screening, using claims data, in accordance with Exhibit O: Quality and Oversight Reporting Requirements.
- 4.10.2.7 The MCO shall share with the Department the results of any identification and assessment of Member's needs to prevent duplication of activities as described in separate guidance. [42 CFR 438.208(b)(4)]
  - 4.10.2.7.1 The PCP shall review HRA Screening data and make appropriate referrals to social service agencies or other entities whether the data is collected in-person, digitally or electronically, telephonically, in-person, digitally or electronically, telephonically, or through completion of the written form by the Member.
  - 4.10.2.7.2 The Provider conducting the HRA Screening shall share Member HRA results with the MCO upon request.
- 4.10.2.8 The MCO shall ensure, through incentives or professional provider reimbursement, that the Participating Provider reviews the HRA Screening results and make appropriate referrals to social service agencies or other entities whether the HRA is collected inperson,

Page 193 of 397	Date
-----------------	------

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



- electronically, telephonically, or through completion of the written form by the Member.
- 4.10.2.9 The Participating Provider conducting the HRA Screening shall share Member HRA Screening results with the MCO upon request.
- 4.10.2.10 The MCO shall ensure Participating Providers complete and review Member HRA Screenings at least annually as follows:
  - 4.10.2.10.1 By the end of Year 1 (SFY 2025 (June 30, 2025)), the HRA minimum completion rate requirement is twenty percent (20%) of the plan's total membership;
  - 4.10.2.10.2 By the end of Year 2 (SFY 2026 (June 30, 2026)), the HRA minimum completion rate requirement is forty percent (40%) of the plan's total membership;
  - 4.10.2.10.3 By the end of Year 3 (SFY 2027 (June 30, 2027)), the HRA minimum completion rate completion rate requirement is sixty percent (60%) of the plan's total membership; and
  - 4.10.2.10.4 By the end of Year 4 (SFY 2028 (June 30, 2028) and through the end of the contract term, the HRA minimum completion rate requirement is seventy-five percent (75%) of the plan's total membership.
- 4.10.2.11 The evidence-based HRA Screening tool shall identify, at minimum, the following information about Members:
  - 4.10.2.11.1 Demographics;
  - 4.10.2.11.2 Chronic and/or acute conditions;
  - 4.10.2.11.3 Chronic pain;
  - 4.10.2.11.4 The unique needs of children with developmental delays, Special Health Care Needs or involved with the juvenile justice system and child protection agencies (i.e., DCYF);
  - 4.10.2.11.5 Behavioral Health needs, including depression or other Substance Use Disorders as described in sections, including but not limited to Section 4.13.9 (Comprehensive Assessment and Care Plans for Behavioral Health Needs), Section 4.11.5.4 (Comprehensive Assessment and Care Plans), and Section 4.13.25 (Provision of Substance Use Disorder Services);
  - 4.10.2.11.6 The need for assistance with personal care such as dressing or bathing or home chores and grocery shopping;

Page 194 of 397	1	Oate

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

- 4.10.2.11.7 Tobacco Cessation needs;
- 4.10.2.11.8 Health-related social needs, including housing, childcare, food insecurity, transportation and/or other interpersonal risk factors such as safety concerns/caregiver stress; and
- 4.10.2.11.9 Other factors or conditions about which the MCO shall need to be aware to arrange available interventions for the Member.

#### 4.10.3 Wellness Visits

- 4.10.3.1 For all Members the MCO shall support the Member to arrange a Wellness Visit with his or her PCP, either previously identified or selected by the Member from a list of available PCPs. If the Member changes their PCP, the MCO shall authorize a new Wellness Visit with the new PCP, even if within a calendar year of the last Wellness Visit with the previous PCP.
- 4.10.3.2 The Wellness Visit conducted by the PCP or other qualified Provider shall include health risk and social determinant of health screening assessments for the purpose of determining a Member's health wellness and development of a plan of care, including evaluations of:
  - 4.10.3.2.1 Both physical and behavioral health, including screening for depression;
  - 4.10.3.2.2 Mood, suicidality; and
  - 4.10.3.2.3 Substance Use Disorder.

#### 4.10.4 Prior Authorization for Primary Care and Preventive Services

4.10.4.1 Notwithstanding other provisions of Section 4.8.1.6, Prior Authorizations for any preventive services, as defined in Section 4.8.2.3.2 of this Agreement, and as stipulated to in Practice Guidelines and Standards at Section 4.8.2 shall be prohibited. This prohibition shall include medically appropriate follow-up testing related to the initial test results, as well as any claims or encounters associated with the PCP's coordination and collaboration with Behavioral Health Services to support the Member's participation in preventive services activities.

#### 4.10.5 Primary Care and Prevention Focused Model Implementation Plan

4.10.5.1 The MCO shall submit a plan for implementing the Primary Care and Prevention Focused Model of Care in accordance with Exhibit O: Quality and Oversight Reporting.

age 195 of 397 Date
---------------------

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

## 4.11. Care Coordination and Care Management

#### 4.11.1. General Requirements

- 4.11.1.1 The MCO shall be responsible for ensuring effective management, coordination, and Continuity of Care for all Members, including oversight of Provider-Delivered Care Coordination for the PCPs' attributed Members, and shall develop and maintain policies and procedures to address these responsibilities.
- 4.11.1.2 The MCO shall submit a plan at time of Readiness Review and implement procedures to facilitate integrated Provider-Delivered Care Coordination and MCO-Delivered Care Management to ensure each Member has an ongoing source of care appropriate to their needs, and includes procedures for confidentiality, consent, or informed consent. [42 CFR 438.208(b)]
- 4.11.1.3 The MCO shall ensure the services described in this section are provided for all Members who need Care Coordination regardless of their acuity level.
- 4.11.1.4 The MCO shall implement and monitor Provider-Delivered Care Coordination and MCO-Delivered Care Management, as appropriate, in order to achieve the following goals:
  - 4.11.1.4.1 Improve care of Members;
  - 4.11.1.4.2 Improve health outcomes:
  - 4.11.1.4.3 Increase collaboration among the Member's Providers, including but not limited to Behavioral Health Services Providers;
  - 4.11.1.4.4 Reduce inpatient hospitalizations including readmissions;
  - 4.11.1.4.5 Improve Continuity of Care;
  - 4.11.1.4.6 Improve transition planning;
  - 4.11.1.4.7 Improve medication management;
  - 4.11.1.4.8 Improve U.S. Preventive Services Task Force (USPSTF) recommended Level A and B preventive screenings; as well as State specified screenings
  - 4.11.1.4.9 Reduce utilization of unnecessary Emergency Services;
  - 4.11.1.4.10 Reduce unmet resource needs related to health-related social needs;
  - 4.11.1.4.11 Decrease total costs of care; and

age 196 of 397 Date	
---------------------	--

New Hampshire Department of Health and Human Services Medicaid Care Management Services



#### **Exhibit B - DRAFT**

- 4.11.1.4.12 Increase Member satisfaction with their health care experience.
- 4.11.1.5 The MCO shall implement and oversee a process that ensures its Participating Providers coordinate care among and between Providers serving a Member, including PCPs, specialists, Behavioral Health Service Providers, and social service resources, and include related documentation in the Member Care Plan.
  - 4.11.1.5.1 The MCO and its Participating Providers shall utilize, leverage and partner with the Department's closed-loop referral system, if available, or 2-1-1 NH if it is not, which is a New Hampshire statewide information and referral service, using closed-loop referral processes to ensure warm transfers are completed and outcomes are reported for all closed-loop referrals.
- 4.11.1.6 The MCO shall implement procedures to coordinate services the MCO furnishes to the Member with the services the Member receives from another MCO. [42 CFR 438.208(b)(2)(ii)]
- 4.11.1.7 The MCO shall also implement procedures to coordinate services the MCO furnishes to the Required Priority Population Member with the services the Member receives in FFS Medicaid, including Medicaid dental services, as applicable. For other Members not included in the Required Priority Population, the PCP shall coordinate these services. [42 CFR 438.208(b)(2)(iii)].
- 4.11.1.8 The MCO shall provide Care Management support for Required Priority Population Members who utilize services not covered by this Agreement (e.g., Medicaid, commercial, or government health insurance programs). In such cases, the MCO's responsibility shall include coordination and referrals in compliance with 42 CFR 438.208(b)(2)(iii-iv). The MCO shall use the Department's closed-loop referral solution, if available, to initiate and support the Required Priority Population Member's access to other services to which the MCO, or its applicable PCP or other Participating Provider is referring the Member.

#### 4.11.2. MCO-Delivered Care Management for Required Priority Populations

- 4.11.2.1 Required Priority Populations are most likely to have Care Management needs that shall be met with the MCO-Delivered Care Management processes described in this Agreement.
- 4.11.2.2 The following high-risk groups are identified as Required Priority Populations in need of Care Management focus by the MCO:

Page 197 of 397 Date \_\_\_\_\_

New Hampshire Department of Health and Human Services **Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

4.11.2.2.1	Individuals who have required an inpatient admission for a behavioral health diagnosis within the previous twelve (12) months;
4.11.2.2.2	Infants, children and youth who are involved in the

- State's protective services and juvenile justice system, Division for Children Youth and Families (DCYF), including those in foster care, and/or those who have elected voluntary supportive services;
- 4.11.2.2.3 Infants diagnosed with low birth weight and/or neonatal abstinence syndrome (NAS);
- Individuals with behavioral health needs (e.g., 4.11.2.2.4 substance use disorder, mental health) who are incarcerated in the State's prisons and eligible for participation in the Department's Community Reentry demonstration waiver pending CMS approval; 13 and
- 4.11.2.2.5 Other Required Priority Populations identified by the Department with advance notification to the MCO with an effective date for Care Management services within ninety (90) calendar days of written notice from the Department.
- 4.11.2.3 The MCO may identify other Members who may benefit from the plan's Care Management services at the plan's option; however, MCO-Delivered Care Management requirements specified in this Agreement apply only to the Required Priority Populations identified by the Department, which may be expanded from time to time with advance notification to the MCO.

# 4.11.3. Comprehensive Assessment for Required Priority Population.

- 4.11,3.1 The MCO shall implement mechanisms to conduct a Comprehensive Assessment for each Member identified as meeting Required Priority Population criteria in order to identify whether they have Special Health Care Needs and any ongoing special conditions that require a course of treatment or regular care monitoring. [42 CFR 438.208(c)(2)]
- 4.11.3.2 The Comprehensive Assessment shall identify a Member's health condition that requires a course of treatment that is either episodic. which is limited in duration or significance to a particular medical episode, or requires ongoing Provider-Delivered Care Coordination or MCO-Delivered Care Management monitoring to ensure the Member is managing his or her medical and/or behavioral health

Date \_\_\_\_

<sup>&</sup>lt;sup>13</sup> Substance Use Disorder Serious Mental Illness Serious Emotional Disturbance Treatment and Recovery Access Section 1115(a) Research and Demonstration Waiver available on July 18, 2023 at https://www.dhhs.nh.gov/sites/q/files/ehbemt476/files/documents2/sed-extention-request.pdf.

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



- care needs (including screening for depression, mood, suicidality, and Substance Use Disorder).
- 4.11.3.3 The Comprehensive Assessment shall be a person-centered assessment of a Member's medical and behavioral care needs, functional status, accessibility needs, strengths and supports, health care goals and other characteristics that shall inform whether the Member should receive Care Management and shall inform the Member's ongoing Care Plan and treatment. The MCO shall incorporate into the Comprehensive Assessment information obtained as a result of Provider referral, or the Wellness Visit.
- 4.11.3.4 The MCO shall make best efforts to complete the Comprehensive Assessment within thirty (30) calendar days of identifying a Member as being part of one or more Required Priority Population as identified through Medicaid enrollment records, HRA Screening, risk scoring and stratification or other means at the MCO's discretion.
- 4.11.3.5 The MCO shall not withhold any Medically Necessary Covered Services including EPSDT services per Section 4.1.8 (Early and Periodic Screening, Diagnostic, and Treatment) for Members while awaiting the completion of the Comprehensive Assessment but may conduct utilization review for any services requiring Prior Authorization.
- 4.11.3.6 The MCO shall conduct the Comprehensive Assessment in a location of the Member's, parent's or guardian's choosing, as applicable, and shall endeavor to conduct the Comprehensive Assessment in-person for populations where the quality of information may be compromised if provided telephonically (e.g., for Members whose physical or behavioral health needs may impede the ability to provide comprehensive information by telephone), including others in the person's life in the assessment process such as family members, paid and natural supports as agreed upon and appropriate to the Member/Member's parent, if a minor, or guardian to the maximum extent practicable.
- 4.11.3.7 Additionally, participation in the Comprehensive Assessment shall be extended to the Member's Care Team or Case Management staff, including but not limited to Area Agencies, CFI waiver, CMH Programs, Special Medical Services, and 1915(i) case managers as practicable, with Member consent to the extent required by State and federal law.
- 4.11.3.8 The MCO shall develop and implement a Comprehensive Assessment tailored to Members that include, at a minimum, the following domains/content:
  - 4.11.3.8.1 Members' immediate care needs;

Page 199 of 397	Date	

Medicaid Care Management Services Contract New Hampshire Department of Health and Human Services Medicaid Care Management Services



## **Exhibit B - DRAFT**

	4.11.3.8.2	Demographics;
	4.11.3.8.3	Education;
	4.11.3.8.4	Housing;
	4.11.3.8.5	Employment and entitlements;
	4.11.3.8.6	Legal involvement;
	4.11.3.8.7	Risk assessment, including suicide risk;
	4.11.3.8.8	Other State or local community and family support services currently used;
	4.11.3.8.9	Medical and other health conditions;
	4.11.3.8.10	Physical, I/DDs;
	4.11.3.8.11	Functional status (activities of daily living (ADL)/instrumental activities of daily living (IADL)) including cognitive functioning;
	4.11.3.8.12	Medications;
	4.11.3.8.13	Available informal, caregiver, or social supports, including peer supports;
	4.11.3.8.14	Current and past mental health and substance use status and/or disorders;
	4.11.3.8.15	Health-related social needs; and
	4.11.3.8.16	Exposure to adverse childhood experiences or other trauma (e.g., parents with mental health or Substance Use Disorders that affect their ability to protect the safety of the child, child abuse or neglect).
4.11.3.9	Comprehens related to co	shall provide to the Department a copy of the ive Assessment tool and all policies and procedures onducting the Comprehensive Assessment for the review as part of Readiness Review and annually
4.11.3.10		nall conduct a re-assessment of the Comprehensive for a Member receiving ongoing Care Management:
	4.11.3.10.1	At least annually;
	4.11.3.10.2	When the Member's circumstances or needs change significantly;
	4.11.3.10.3	At the Member's request; and/or
	4.11.3.10.4	Upon the Department's request.
4.11.3.11		shall share the results of the Comprehensive in writing with the Member's Care Team within 14

Page 200 of 397 Date \_\_\_\_\_

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

calendar days of completion of the assessment to inform care and treatment planning, with Member consent to the extent required by State and federal law.

- 4.11.3.12 The MCO shall report to the Department the following in accordance with Exhibit O: Quality and Oversight Reporting Requirements:
  - 4.11.3.12.1 Assessments conducted as a percentage (%) of total Members and by Required Priority Population category;
  - 4.11.3.12.2 Assessments completed by a Subcontractor entity, such as but not limited to CMH Programs, Special Medical Services, HCBS case managers, and Area Agencies;
  - 4.11.3.12.3 Timeliness of assessments:
  - 4.11.3.12.4 Timeliness of dissemination of assessment results to PCPs, specialists, behavioral health Providers and other members of the local community based care team; and
  - 4.11.3.12.5 Quarterly report of unmet resource needs, aggregated by county, based on the care screening and Comprehensive Assessment tool to include number of Members reporting in accordance with Exhibit O: Quality and Oversight Reporting Requirements.

#### 4.11.4. Member Care Management Engagement

- 4.11.4.1 The MCO shall assign a designated Care Manager for every Required Priority Population Member.
- 4.11.4.2 For any Member identified as part of a Required Priority Population relative to behavioral health, as described in this Agreement, and subsequently identified by the MCO as not needing Care Management, the MCO shall provide documentation to the Member's PCP and behavioral health provider(s), if applicable, of this decision, and to the Department. If, based on Member utilization data or consultation with the behavioral health provider, the Department notifies the MCO that the Member's utilization history is of continuing concern to the Department, such that Care Management is still warranted, the Department will notify the MCO and the MCO shall provide Care Management and designate a Care Manager for the Member.
- 4.11.4.3 Members selected for MCO-Delivered Care Management shall be informed of:

Page 201 of 397	Date
-----------------	------

Medicaid Care Management Services Contract New Hampshire Department of Health and Human Services Medicaid Care Management Services



# **Exhibit B - DRAFT**

	4.11.4.3.1	The nature of the Care Management engagement relationship;
	4.11.4.3.2	Circumstances under which information shall be disclosed to third parties, consistent with State and federal law;
	4.11.4.3.3	The availability of a grievance and appeals process;
	4.11.4.3.4	The rationale for implementing Care Management services; and
	4.11.4.3.5	The processes for opting out of and terminating Care Management services.
4.11.4.4	The MCO's minimum:	Care Management responsibilities shall include, at a
	4.11.4.4.1	Coordination of physical, mental health, Substance Use Disorder and social services using Provider engagement approaches not inconsistent with those described in this Agreement for certain Department identified Required Priority Populations and Behavioral Providers, including but not limited to Community Mental Health Programs and Certified Community Behavioral Health Centers;
	4.11.4.4.2	Quarterly medication reconciliation;
	4.11.4.4.3	Monthly telephonic contact with the Member;
	4.11.4.4.4	Monthly communication with the care team either in writing or telephonically for coordination and updating of the Care Plan for dissemination to care team participants;
	4.11.4.4.5	Referral follow-up monthly;
	4.11.4.4.6	Peer support;
	4.11.4.4.7	Support for unmet resource needs;
	4.11.4.4.8	Training on disease self-management, as relevant; and
	4.11.4.4.9	Transitional Care Management as defined in Section 4.11.6 (Transitional Care Management).
4.11.4.5	Priority Pop Management Member, dep	nall convene an initial Care Team for each Required pulation Member receiving MCO-Delivered Care to where necessary to improve health outcomes for the pendent on a Member's needs including, including but the Marshar constalor(s) and guarding(s).

Page 202 of 397 Date \_\_\_\_\_

not limited to, the Member, caretaker(s) and guardian(s), PCP, behavioral health Provider(s), specialist(s), targeted case

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

managers, children's behavioral health system coordinators, Critical Time Intervention coaches, Supportive Housing casing managers, transitional case managers, school personnel, nutritionist(s), and/or pharmacist(s) based on applicable need to participate to effectively support achievement of improved health outcomes for the Member.

- 4.11.4.6 The ongoing Care Team shall be chosen or approved by the Member, or their parent(s) or guardian(s) if a minor, or their guardian(s) if an adult and applicable, whose composition best meets the unique care needs to be addressed and with whom the Member has already established relationships.
- 4.11.4.7 The MCO shall identify the information necessary to support improved health outcomes for the Member to be shared among all Care Team participants concerned with a Member's care to achieve safer, more effective health care delivery and improved health outcomes for the Member, including how the Provider-Delivered Care Coordination and MCO-Delivered Care Management programs interface with the Member's PCP, behavioral health providers for mental illness, SMI, SPMI, SED, I/DD, and Substance Use Disorder, and other applicable specialist Providers and existing community resources and supports. The MCO shall communicate this information, with the Member or their parent(s)' or guardian(s)' consent in compliance with state and federal laws and regulations.
- 4.11.4.8 The MCO shall work with the Member's Care Team to identify responsibilities for the Member's Care Plan which is optimally maintained by the PCP, in collaboration with the Care Team participants within thirty (30) calendar days of the completed Comprehensive Assessment, for each Priority Population Member identified through a Comprehensive Assessment or other means as in need of a course of treatment or regular Care Management monitoring. [42 CFR 438.208(c)(3)]
  - 4.11.4.8.1 The MCO shall ensure that each Provider furnishing services to Members maintains and shares Member health records in accordance with professional standards. [42 CFR 438.208(b)(5)]
  - 4.11.4.8.2 The MCO shall use and disclose individually identifiable health information, such as medical records and any other health or enrollment information that identifies a particular Member in accordance with confidentiality requirements in 45 CFR 160 and 164, this Agreement, and all other applicable laws and regulations. [42 CFR 438.208(b)(6); 42 CFR 438.224; 45 CFR 160; 45 CFR 164]

Page 203 of 397 Date \_\_\_\_\_

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

- 4.11.4.8.3 The MCO shall develop and implement a strategy to address how the Interoperability Standards Advisory standards, from the Office of the National Coordinator for Health Information Technology, informs the MCO system development and interoperability.
- 4.11.4.8.4 The MCO shall contribute to the Member's Care Plan as follows:
  - 4.11.4.8.4.1. At least quarterly;
  - 4.11.4.8.4.2. When a Member's circumstances or needs change significantly;
  - 4.11.4.8.4.3. At the Member's request;
  - 4.11.4.8.4.4. When a re-assessment occurs; and
  - 4.11.4.8.4.5. Upon the Department's request.
- 4.11.4.8.5 The MCO shall submit coordinating Care Plan processes to the Department for review as part of the Readiness Review process and annually thereafter.
- 4.11.4.9 The MCO shall track the Member's progress through routine Care Team conferences, the frequency to be determined by the MCO based on the Member's level of need.
- 4.11.4.10 The MCO shall develop policies and procedures that describe when Members should be discharged from the Care Management program, should the Care Team determine that the Member no longer requires a course of treatment which was episodic or no longer needs ongoing care monitoring.
  - 4.11.4.10.1 Policies and procedures for discharge shall include a Member notification process.
- 4.11.4.11 For Members who have been determined, through a Comprehensive Assessment, to need a course of treatment or regular care monitoring, the MCO shall ensure there is a mechanism in place to permit such Members to directly access a specialist as appropriate for the Member's condition and identified needs. [42 CFR 438.208(c)(4)]

## 4.11.5. MCO Care Managers

- 4.11.5.1 The MCO shall formally designate a Care Manager that is primarily responsible for MCO-Delivered Care Management for each Required Priority Population Member, including regular contact with the Member's PCP who is responsible for Provider-Delivered Care Coordination as defined in this Agreement.
- 4.11.5.2 The MCO shall provide to Members information on how to contact their designated Care Manager. [42 CFR 438.208(b)(1)]

Page 204 of 397	Date	

New Hampshire Department of Health and Human Services Medicaid Care Management Services



## **Exhibit B - DRAFT**

4.11.5.3	Care Manag following are	ers shall have qualifications and competency in the as:
	4.11.5.3.1	All aspects of person-centered needs assessments and Care Planning;
	4.11.5.3.2	Motivational interviewing and self-management;
	4.11.5.3.3	Trauma-informed care;
	4.11.5.3.4	Cultural and linguistic competency;
	4.11.5.3.5	Understanding and addressing unmet resource needs including expertise in identifying, accessing and utilizing available social support and resources in the Member's community; and
	4.11.5.3.6	Adverse childhood experiences and trauma.
4.11.5.4	Care Managers shall be trained in the following:	
	4.11.5.4.1	Disease self-management;
	4.11.5.4.2	Person-centered needs assessment and Care Planning including coordination of care needs;
	4.11.5.4.3	Integrated and coordinated physical and behavioral health, including as they intersect with and are served within the State's Community Mental Health system, Substance Use Disorder system, and Children's Behavioral Health system;
	4.11.5.4.4	The State's Behavioral Health Crisis Response system and available resources (for Care Managers with assigned Members with behavioral health needs);
	4.11.5.4.5	Cultural and linguistic competency;
	4.11.5.4.6	Family support; and
	4.11.5.4.7	Understanding and addressing unmet resource needs, including expertise in identifying and utilizing available social supports and resources in the Member's community.
4.11.5.5	•	ers shall remain conflict-free which shall be defined as lated by blood or marriage to a Member, financially

Page 205 of 397 Date \_\_\_\_\_

responsible for a Member, or with any legal power to make financial

Required Priority Populations and consistent follow up with Providers and Members to assure that Members are making

4.11.5.6 The MCO shall provide real-time, high-touch, Care Management for

or health related decisions for a Member.

progress with their Care Plans.

New Hampshire Department of Health and Human Services Medicaid Care Management Services



#### **Exhibit B - DRAFT**

- 4.11.5.7 The MCO shall design an effective Care Management structure for the Required Priority Population Members.
  - 4.11.5.7.1 At a minimum by the measurement period ending June 30, 2026 (SFY 2026), the MCO shall have no less than fifty percent (50%) of each Required Priority Population in MCO-Delivered Care Management.
- 4.11.5.8 The MCO shall, as described in Section 4.11.6 (Transitional Care Management), demonstrate that it has active access to an Admission, Discharge, Transfer (ADT) data source(s) that correctly identifies when empaneled Members are admitted, discharged, or transferred to/from an ED or hospital or DRF in real-time or near real-time.
  - 4.11.5.8.1 The MCO shall ensure that ADT data from applicable hospitals be made available to the Member's PCP, behavioral health Providers, Care Team, and applicable community-based agencies within twelve (12) hours of the admission, discharge, or transfer.

#### 4.11.6. Transitional Care Management

- 4.11.6.1 For all Members, the MCO shall be responsible, in collaboration with the Member's Care Team, as applicable, which may include the Member's PCP, behavioral health provider(s), specialist(s), targeted case managers, children's behavioral health system service coordinators, Critical Time Intervention coaches, Supportive Housing case managers, and transitional case managers, school personnel as needed, pharmacists, and others as appropriate, for managing transitions of care for all Members moving from one (1) clinical setting to another, including step-up or step-down treatment programs for Members in need of continued mental health and Substance Use Disorder services, to prevent unplanned or unnecessary readmissions, ED visits, or adverse health outcomes.
- 4.11.6.2 The MCO shall maintain and operate a formalized hospital and/or institutional discharge planning program that includes effective post-discharge Transitional Care Management for all Members, including appropriate discharge planning for short-term and long-term hospital and institutional stays. [42 CFR 438.208(b)(2)(i)]
- 4.11.6.3 The MCO shall develop policies and procedures for the Department's review, as part of Readiness Review and annually thereafter, which describe how transitions of care between settings shall be effectively managed including data systems that trigger notification that a Member is in transition.

age 206 of 397 Date
---------------------

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



- 4.11.6.4 The MCO's transition of care policies shall be consistent with federal requirements that meet the State's transition of care requirements. [42 CFR 438.62(b)(12)]
- 4.11.6.5 The MCO shall have a documented process to, at a minimum:
  - 4.11.6.5.1 Coordinate appropriate follow-up services from any inpatient or facility stay;
  - 4.11.6.5.2 Support continuity of care for Members when they move from home to foster care placement; foster care to independent living; return from foster care placement to community; change in legal status from foster care to adoption; or when the Member moves from one level of care to another within the State's behavioral health system for Community Mental Health, Substance Use Disorders, or children's behavioral health;
  - 4.11.6.5.3 Schedule a face-to-face visit to complete a Comprehensive Assessment and update a Member's Care Plan when a Member is hospitalized;
  - 4.11.6.5.4 Evaluate Members for continued mental health and Substance Use Disorder services upon discharge from an inpatient psychiatric facility or residential treatment center as described in Section 4.13.20 (Agreement for New Hampshire State-Owned Hospital Agreement(s) and Other State Determined IMDs for Mental Illness), and upon discharge from an ED due to a mental illness or substance use disorder; and
  - 4.11.6.5.5 Coordinate with inpatient discharge planners for Members referred for subacute treatment in a nursing facility.
- 4.11.6.6 The MCO shall have an established process, inclusive of but not limited to use of the Department's event notification system and closed-loop referral solution, if available, to work with Providers (including hospitals regarding notice of admission and discharge) to ensure appropriate communication among Providers and between Providers and the MCO to ensure that Members receive appropriate follow-up care and are in the most integrated and cost-effective delivery setting appropriate for their needs.
- 4.11.6.7 The MCO shall implement a protocol to identify Members who use ED services inappropriately, analyze reasons why each Member did so and provide additional services to assist the Member to access appropriate levels of care including assistance with scheduling and attending follow-up care with PCPs and/or

Page 207 of 397	Date
-----------------	------

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



- appropriate specialists to improve Continuity of Care, resolve Provider access issues, and establish a medical home.
- 4.11.6.8 The MCO shall demonstrate, at a minimum, it has active access to ADT data source(s) that correctly identifies when empaneled Members are admitted, discharged, or transferred to/from an ED or hospital in real-time or near real-time.
- 4.11.6.9 The MCO shall ensure that ADT data from applicable hospitals be made available to PCPs, and behavioral health Providers within twelve (12) hours of the admission, discharge, or transfer.
- 4.11.6.10 The MCO shall ensure that Transitional Care Management includes, at a minimum:
  - 4.11.6.10.1 Care Management or other services to ensure the Member's Care Plan continues:
  - 4.11.6.10.2 Facilitating clinical hand-offs;
  - 4.11.6.10.3 Obtaining a copy of the discharge plan/summary prior to the day of discharge, if available, otherwise, as soon as it is available, and documenting that a follow-up outpatient visit is scheduled, ideally before discharge;
  - 4.11.6.10.4 Communicating with the Member's PCP about discharge plans and any changes to the Care Plan;
  - 4.11.6.10.5 Conducting medication reconciliation within forty-eight (48) business hours of discharge;
  - 4.11.6.10.6 Ensuring that a Care Manager is assigned to manage the transition, and that the Care Manager collaborates with the Member's applicable Community Mental Health system, Substance Use Disorder system, or Children's Behavioral Health system providers to support the Member's effective transition and continuous access to needed services throughout the transitional period;
  - 4.11.6.10.7 Follow-up by the assigned Care Manager, or otherwise designated member of the Member's care management team, within forty-eight (48) business hours of discharge of the Member;
  - 4.11.6.10.8 Determining when a follow up visit should be conducted in a Member's home;
  - 4.11.6.10.9 Supporting Members to keep outpatient appointments; and

Page 208 of 397	Date	

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



- 4.11.6.10.10 A process to assist with supporting continuity of care for the transition and enrollment of children being placed in foster care, including children who are currently enrolled in the plan and children in foster care who become enrolled in the plan, including prospective enrollment so that any care required prior to effective data of enrollment is covered.
- 4.11.6.11 The MCO shall assist with coordination between the children and adolescent service delivery system as these Members transition into the adult mental health service delivery system, through activities such as communicating treatment plans and exchange of information.
- 4.11.6.12 The MCO shall coordinate inpatient and community services, including the following requirements related to hospital admission and discharge:
  - 4.11.6.12.1 The outpatient Provider shall be involved in the admissions process when possible; if the outpatient Provider is not involved, the outpatient Provider shall be notified promptly of the Member's hospital admission:
  - 4.11.6.12.2 Psychiatric hospital and residential treatment facility discharges shall not occur without a discharge plan (i.e. an outpatient visit shall be scheduled before discharge to ensure access to proper Provider/medication follow-up; and an appropriate placement or housing site shall be secured prior to discharge);
  - 4.11.6.12.3 The hospital's evaluation shall be performed prior to discharge to determine what, if any, mental health or Substance Use Disorder services are Medically Necessary. Once deemed Medically Necessary, the outpatient Provider shall be involved in the discharge planning, the evaluation shall include an assessment for any social services needs such as housing and other necessary supports the young adults need to assist in their stability in their community; and
  - 4.11.6.12.4 A procedure to ensure Continuity of Care regarding medication shall be developed and implemented.
- 4.11.7. Provider-Delivered Care Coordination and Integration with Social Services and Community Care
  - 4.11.7.1 The MCO shall implement and provide administrative support of a Provider-Delivered Care Coordination Program that includes reimbursement and other incentives to enable Participating

Page 209 of 397	Date
-----------------	------

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

Providers to coordinate health-related and community support services for Members.

- 4.11.7.2 The MCO shall provide program administrative support that includes, at a minimum:
  - 4.11.7.2.1 Secure transmission of data and other information to Providers about their attributed Members' service utilization and care coordination needs;
  - 4.11.7.2.2 Provider assistance with securing:
    - 4.11.7.2.2.1. Health-related services and community support services, including but not limited to housing, that can improve health and family well-being, including assistance filling out and submitting applications; and
    - 4.11.7.2.2.2. Access to medical-legal partnership for legal issues adversely affecting health, subject to the availability and capacity of a medical-legal assistance Provider.
  - 4.11.7.2.3 Provider education and training, including:
    - 4.11.7.2.3.1. How to access information about community support services, and housing for Members; and
    - 4.11.7.2.3.2. How to facilitate Member closed-loop referrals utilizing the Department's event notification system and closed-loop referral solution, if available, or another closed-loop referral solution.
  - 4.11.7.2.4 Incentivizing the Provider's use of closed-loop referrals for effective care coordination in accordance with Exhibit O: Quality and Oversight Reporting Requirements.
- 4.11.7.3 The MCO shall assist Providers to actively link Members with other State, local, and community programs that may provide or assist Members with health and social services including, but not limited to: [42 CFR 438.208(b)(2)(iv)]
  - 4.11.7.3.1 Juvenile Justice and Adult Community Corrections;
  - 4.11.7.3.2 Locally administered social services programs including, but not limited to, Women, Infants, and Children, Head Start Programs, Community Action Programs, local income and nutrition assistance programs, housing, etc.;

Page 210 of 397 Date \_\_\_\_\_

New Hampshire Department of Health and Human Services Medicaid Care Management Services



## **Exhibit B - DRAFT**

4.11.7.3.3	Family Organizations, Youth Organizations, Consumer Organizations, and Faith Based Organizations;
4.11.7.3.4	Public Health Agencies;
4.11.7.3.5	Schools;
4.11.7.3.6	The court system;
4.11.7.3.7	ServiceLink Resource Network;
4.11.7.3.8	2-1-1 NH;
4.11.7.3.9	Housing; and
4.11.7.3.10	VA Hospital and other programs and agencies serving service Members, veterans and their families.

4.11.7.4 The MCO shall report on the number of referrals for social services and community care provided to Required Priority Population Members by Member type, consistent with the format and content requirements in accordance with Exhibit O: Quality and Oversight Reporting Requirements.

#### 4.12. Behavioral Health

## 4.12.1 General Coordination Requirements

- 4.12.1.1 This section describes the delivery and coordination of Behavioral Health Services and supports for mental health, Serious Mental Illness, Substance Use Disorders, and Serious Emotional Disturbances, delivered to children, youth and transition-aged youth/young adults, and adults.
- 4.12.1.2 The MCO shall ensure Behavioral Health Services are delivered in a manner that is both clinically and developmentally appropriate, and that considers the Member, parents, caregivers, and other networks of support the Member may rely upon.
- 4.12.1.3 The delivery of service shall be Member-centered and align with the principles of system of care, recovery, and resiliency.
- 4.12.1.4 The MCO shall provide Behavioral Health Services in accordance with this Agreement and all applicable State and federal laws and regulations.
- 4.12.1.5 The MCO shall be responsible for providing a full continuum of physical health and Behavioral Health Services, as authorized under the State's Medicaid State Plans and in accordance with the applicable NH Administrative Rules identified in this Agreement specific to Behavioral Health Services; ensuring continuity and coordination of care between covered physical health and

Page 211 of 397	Date	

New Hampshire Department of Health and Human Services **Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

Behavioral Health Services Providers; and requiring collaboration between physical health and Behavioral Health Providers.

- The continuum of Behavioral Health Services shall include the 4.12.1.6 following categories of Providers approved by the Department for providing one or more types of services under the State Medicaid Plan, certain Administrative Rules, and under contracts with the Department when necessary to ensure Member access to higher levels of care for Serious Mental Illness, Substance Use Disorder, Serious Emotional Disturbance, and I/DD:
  - Mental Health Services, including but not limited to 4.12.1.6.1 psychotherapy, psychological evaluation and testing, authorized in the Medicaid State Plan under Attachment 3.1-A for Medical, Remedial Care and Services. These services shall be provided by appropriately licensed and certified Providers who are not providing the service on behalf of or under agreement with a Community Mental Health Program (also known as Community Mental Health Center) or a Community Mental Health Provider. The MCO shall not authorize payment of these services under Attachment 3.1-A for Other Diagnostic, Screening, Preventative and Rehabilitative Services, which represents services at a higher level of care for Members who are currently eligible for that level of care under He-M 401 and which are only a covered service if provided by Community Mental Health Programs or Community Mental Health Providers.
  - 4.12.1.6.2 Community Mental Health Services (CMH Services), authorized in the Medicaid State Plan under Attachment 3.1-A for Other Diagnostic, Screening, Preventative and Rehabilitative Services, which represents services at a higher level of care for Members with current He-M 401 eligibility and which are provided by:
    - 4.12.1.6.2.1. **Community Mental Health Programs** (CMH Programs), also known as Community Mental Health Centers (CMHC) that are currently approved by the Department pursuant to He-M 403; there are ten such programs in NH; or by
    - **Community Mental Health Providers** 4.12.1.6.2.2. (CMH Providers) that have been approved previously by

Commissioner of the Department of

Page 212 of 397 Date \_\_\_\_

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

Health and Human Services to provide Community Mental Health Services identified in He-M 426.07-He-M 426.17 for which they have received approval to provide pursuant to He-M 426.04 and remain in compliance with the requirements specified in He-M 426.04.

4.12.1.6.3 **Substance Use Disorder Services** authorized in accordance with the Medicaid State Plan, He-W 513, and where applicable, He-W 300 for Opioid Treatment Programs (OTP).

#### 4.12.2 Behavioral Health Subcontracts

- 4.12.2.1 If the MCO enters into a Subcontractor relationship with a behavioral health (Mental Health, Community Mental Health or Substance Use Disorder Provider) Subcontractor to provide or manage Behavioral Health Services, the MCO shall provide a copy of the agreement between the MCO and the Subcontractor to the Department for review and approval, including but not limited to any agreements with CMH Providers as required in Section 4.13.19 Community Mental Health Services).
- 4.12.2.2 Such subcontracts shall address the coordination of services provided to Members by the Subcontractor, as well as the approach to Prior Authorization, claims payment, claims resolution, contract disputes, performance metrics, quality health outcomes, performance incentives, and reporting.
- 4.12.2.3 The MCO remains responsible for ensuring that all requirements of this Agreement are met, including requirements to ensure continuity and coordination between physical health and Behavioral Health Services, and that any Subcontractor adheres to all requirements and guidelines, as outlined in Section 3.10 (Subcontractors).

#### 4.12.3 **Promotion of Integrated Care**

- 4.12.3.1 The MCO shall ensure physical and behavioral health Providers provide co-located or Integrated Care as defined in the Substance Abuse and Mental Health Services Administration's (SAMHSA's) Six Levels of Collaboration/Integration or the Collaborative Care Model to the maximum extent feasible.
- 4.12.3.2 In accordance with Exhibit O: Quality and Oversight Reporting Requirements, the MCO shall include in its Behavioral Health Strategy Plan and Report efforts towards continued progression of the SAMHSA Integration Framework at all contracted primary and behavioral health Providers.

## 4.12.4 Approach to Behavioral Health Services

Page 213 of 307	Date
Page 213 of 397	Date

New Hampshire Department of Health and Human Services Medicaid Care Management Services



#### **Exhibit B - DRAFT**

- 4.12.4.1 The MCO shall ensure that its clinical standard and operating procedures are consistent with trauma-informed models of care, as defined by SAMHSA<sup>14</sup> and reflect a focus on Recovery and resiliency.<sup>15</sup>
- 4.12.4.2 The MCO shall offer training inclusive of mental health first aid training, to MCO staff who manage the behavioral health contract and Participating Providers, including Care Managers, physical health Providers, and Providers on Recovery and resiliency, Trauma-Informed Care, and Community Mental Health Services and resources available within the applicable region(s).
- 4.12.4.3 The MCO shall track training rates and monitor usage of Recovery and resiliency and Trauma-Informed Care practices.
- 4.12.4.4 In accordance with Section 4.8.2 (Practice Guidelines and Standards), the MCO shall ensure that Providers, including those who do not serve behavioral health Members, are trained in Trauma-Informed models of Care.

## 4.12.5 Behavioral Health Strategy Plan and Report

- 4.12.5.1 The MCO shall submit to the Department an initial plan describing its program, policies and procedures regarding the continuity and coordination of covered physical and Behavioral Health Services and integration between physical health and behavioral health Providers. In accordance with Exhibit O: Quality and Oversight Reporting Requirements, the initial Plan shall address but not be limited to how the MCO shall:
  - 4.12.5.1.1 Assure Participating Providers meet SAMHSA Standard Framework for Levels of Integrated Healthcare:
  - 4.12.5.1.2 Assure the appropriateness of the diagnosis, treatment, and referral of behavioral health disorders commonly seen by PCPs;
  - 4.12.5.1.3 Assure the promotion of Integrated Care;
  - 4.12.5.1.4 Reduce Psychiatric Boarding described in Section 4.13.19.17 (Psychiatric Boarding);
  - 4.12.5.1.5 Reduce Behavioral Health Readmissions described in Section 4.13.10 (Reduction in Behavioral Health Readmissions and Emergency Department Utilization);

Page 214 of 397 Date \_\_\_\_\_

<sup>&</sup>lt;sup>14</sup> Substance Abuse and Mental Health Services Administration, "Trauma-Informed Approach and Trauma-Specific Interventions," available at <a href="https://www.samhsa.gov/nctic/trauma-interventions">https://www.samhsa.gov/nctic/trauma-interventions</a>.

<sup>&</sup>lt;sup>15</sup> Substance Abuse and Mental Health Services Administration, "Recovery and Recovery Support," available at <a href="https://www.samhsa.gov/recovery">https://www.samhsa.gov/recovery</a>.

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

Care: and

4.12.5.1.6	Reduce Behavioral Health related emergency department utilization as described in Section 4.13.10 (Reduction in Behavioral Health Readmissions and Emergency Department Utilization);
4.12.5.1.7	Support the NH 10-Year Mental Health Plan <sup>16</sup> ;
4.12.5.1.8	Assure the appropriateness of psychopharmacological medication;
4.12.5.1.9	Assure access to appropriate services;
4.12.5.1.10	Implement a training plan that includes, but is not limited to, Trauma-Informed Care and Integrated

- 4.12.5.1.11 Other information in accordance with Exhibit O: Quality and Oversight Reporting Requirements.
- 4.12.5.2 On an annual basis and in accordance with Exhibit O: Quality and Oversight Reporting Requirements, the MCO shall provide an updated Behavioral Health Strategy Plan and Report which shall include an effectiveness analysis of the initial Plan's program, policies and procedures.
  - 4.12.5.2.1 The analysis shall include MCO interventions which require improvement, including improvements in SAMHSA Standard Framework for Levels of Integrated Healthcare, continuity, coordination (i.e., enhanced Care Coordination and Care Management to minimize inpatient readmissions, emergency department utilization, and psychiatric boarding), and collaboration for physical health and Behavioral Health Services.

#### 4.12.6 Collaboration with the Department

- 4.12.6.1 At the discretion of the Department, the MCO shall provide mental health and Substance Use Disorder updates as requested by the Department during regular behavioral health meetings between the MCO and the Department.
- 4.12.6.2 To improve health outcomes for Members and ensure that delivery of services are provided at the appropriate intensity and duration, the MCO shall meet with behavioral health programs and the Department at least four (4) times per year to discuss quality assurance activities conducted by the MCO, such as PIPs and

. Year%20Mental%20Health%20Plan%20is%20the%20result,health%20needs%20of%20people%20across%20their%20life%20span

Date \_\_\_\_\_

<sup>&</sup>lt;sup>16</sup> New Hampshire Department of Health and Human Services, New Hampshire 10-Year Mental Health Plan (January 2019), available on July 20, 2023 at https://www.dhhs.nh.gov/programs-services/health-care/behavioral-health/10-year-mental-health-plan#:~:text=The%2010-

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



- APMs, and to review quality improvement plans and outstanding needs.
- 4.12.6.3 Quarterly meetings shall also include a review of progress against deliverables, improvement measures, and select data reports as detailed in Exhibit O: Quality and Oversight Reporting Requirements. Progress and data reports shall be produced and exchanged between the MCO and the Department two (2) weeks prior to each quarterly meeting.
  - 4.12.6.3.1 At each meeting, the MCO shall update the Department on the following topics:
    - 4.12.6.3.1.1.1 Updates related to the MCO's

      Behavioral Health Strategy

      Report and interventions to improve outcomes;
    - 4.12.6.3.1.1.2Utilization of ACT services and any waitlists for ACT services;
    - 4.12.6.3.1.1.3 Current EBSE rates;
    - 4.12.6.3.1.1.4 Current compliance with New Hampshire Hospital discharge performance standards;
    - 4.12.6.3.1.1.5 Current compliance with ED discharge performance standards for overdoses and Substance Use Disorder:
    - 4.12.6.3.1.2. Updates regarding services identified in Section 4.13 (Behavioral Health);
    - 4.12.6.3.1.3. Updates on Mental Health and Substance Use Disorder PIPs; and
    - 4.12.6.3.1.4. Other topics requested by the Department.
- 4.12.6.4 For all Members, the MCO shall work in collaboration with the Department and the NH Suicide Prevention Council to promote suicide prevention awareness programs, including the Zero Suicide program.
- 4.12.6.5 The MCO shall submit to the Department, as specified by the Department in Exhibit O: Quality and Oversight Reporting Requirements, its implementation plan for incorporating the "Zero Suicide" program into its operations; the plan shall include, in addition to any other requirements specified in Exhibit O: Quality

Page 216 of 397	Date	

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

and Oversight Reporting Requirements related to the plan, how the	е
MCO shall:	

- 4.12.6.5.1 Incorporate efforts to implement standardized provider screenings and other preventative measures; and
- 4.12.6.5.2 Incorporate the Zero Suicide Consensus Guide for Emergency Departments, as described in Section 4.8.2 (Practice Guidelines and Standards).

## 4.12.7 Primary Care Provider Screening for Behavioral Health Needs

- 4.12.7.1 The MCO shall ensure that the need for Behavioral Health Services is systematically identified by and addressed by the Member's PCP at the earliest possible time following initial enrollment of the Member and ongoing thereafter or after the onset of a condition requiring mental health and/or Substance Use Disorder treatment.
- 4.12.7.2 At a minimum, this requires timely access to a PCP for mental health and/or Substance Use Disorder screening, coordination and a closed loop referral to behavioral health Providers if clinically necessary.
- 4.12.7.3 The MCO shall encourage PCPs and other Providers to use a screening tool approved by the Department, as well as other mechanisms to facilitate early identification of behavioral health needs.
- 4.12.7.4 The MCO shall require all PCPs and behavioral health Providers to incorporate the following domains into their screening and assessment process:
  - 4.12.7.4.1 Demographic,
  - 4.12.7.4.2 Medical.
  - 4.12.7.4.3 Substance Use Disorder.
  - 4.12.7.4.4 Housing.
  - 4.12.7.4.5 Family & support services,
  - 4.12.7.4.6 Education,
  - 4.12.7.4.7 Employment and entitlement,
  - 4.12.7.4.8 Legal, and
  - 4.12.7.4.9 Risk assessment including suicide risk and functional status (ADL, IADL, cognitive functioning).
- 4.12.7.5 The MCO shall require that pediatric Providers ensure that all children receive standardized, validated developmental screening, such as the Ages and Stages Questionnaire and/or Ages and

Page 217 of 397	Date	

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

Stages Questionnaires: Social Emotional at nine (9), eighteen (18) and twenty-four (24)/thirty (30) month pediatric visits; and use Bright Futures or other AAP recognized developmental and behavioral screening system. The assessment shall include universal screening via full adoption and integration of, at minimum, two (2) specific evidenced-based screening practices:

- 4.12.7.5.1 Depression screening (e.g., PHQ 2 & 9); and
- 4.12.7.5.2 Screening, Brief Intervention, and Referral to Treatment (SBIRT) in primary care.

#### 4.12.8 **Referrals**

- 4.12.8.1 The MCO shall ensure through its HRA Screening (Section 4.10.2) and risk scoring and stratification or other means at the MCO's discretion that Members with a potential need for Behavioral Health Services, particularly Required Priority Population Members as described in Section 4.11.2 (4.11.2. MCO-Delivered Care Management for Required Priority Populations) are appropriately and timely referred to behavioral health Providers if co-located care is not available.
- 4.12.8.2 This shall include education about Behavioral Health Services, including the Recovery process, Trauma-Informed Care, resiliency, CMH Programs/CMH Providers and Substance Use Disorder treatment Providers in the applicable region(s).
- 4.12.8.3 The MCO shall develop a referral process to be used by Participating Providers, including what information shall be exchanged and when to share this information, as well as notification to the Member's Care Manager.
- 4.12.8.4 The MCO shall develop and provide Provider education and training materials to ensure that physical health providers know when and how to refer Members who need specialty Behavioral Health Services.
- 4.12.8.5 The MCO shall ensure that Members with both physical health and behavioral health needs are appropriately and timely referred to their PCPs for treatment of their physical health needs when Integrated Care is not available.
- 4.12.8.6 The MCO shall develop a referral process to be used by its Providers. The referral process shall include providing a copy of the physical health consultation and results to the behavioral health Provider.
- 4.12.8.7 The MCO shall develop and provide Provider education and training materials to ensure that behavioral health Providers know when and how to refer Members who need physical health services.

Page 218 of 397	Date
-----------------	------

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

#### 4.12.9 Prior Authorization for Behavioral Health Services

- 4.12.9.1 As of September 2017, the MCO shall comply with the Prior Authorization requirements of House Bill 517 for behavioral health drugs, including use of the universal online Prior Authorization form provided by the Department for drugs used to treat mental illness.
- 4.12.9.2 The MCO shall ensure that any Subcontractor, including any CMH Program/CMH Provider, complies with all requirements included in the bill.

## 4.12.10 Comprehensive Assessment and Care Plans for Behavioral Health Needs

- 4.12.10.1 The MCO's policies and procedures shall identify the role of physical health and behavioral health Providers in assessing a Member's behavioral health needs as part of the Comprehensive Assessment and developing a Care Plan.
- 4.12.10.2 For Members with chronic physical conditions that require ongoing treatment who also have behavioral health needs and who are not already treated by an integrated Provider team, the MCO shall ensure participation of the Member's physical health Provider (PCP or specialist), behavioral health Provider, and, if applicable, Care Manager, in the Comprehensive Assessment and Care Plan development process as well as the ongoing provision of services.

# 4.12.11 Reduction in Behavioral Health Readmissions and Emergency Department Utilization

- 4.12.11.1 Within the MCO's annual Behavioral Health Strategy Plan and Report in accordance with Exhibit O: Quality and Oversight Reporting Requirements, subject to approval by the Department, the MCO shall develop and detail its plan to reduce readmissions and emergency department utilization attributed to a Member's behavioral health. The plan shall include but is not limited to:
  - 4.12.11.1.1 The MCO's approach to monitoring the thirty (30)-day, ninety (90)-day, and one hundred and eighty (180)-day readmission rates to New Hampshire Hospital, other State determined IMDs for mental illness, designated receiving facilities and other equivalent facilities to review Member specific data with each of the CMH Programs, and other CMH Providers and Mental Health providers, as applicable, and implement measurable strategies within ninety (90) calendar days of the execution of this Agreement to reduce thirty (30)-day, ninety (90)-day and one hundred and eighty (180)-day readmission.
  - 4.12.11.1.2 The MCO's approach to monitoring the thirty (30)-day, ninety (90)-day, and one hundred and eighty

Page 219 of 397 Date \_\_\_\_\_

New Hampshire Department of Health and Human Services Medicaid Care Management Services



#### **Exhibit B - DRAFT**

(180)-day readmission rates to acute care hospitals attributed to substance misuse and Substance Use Disorder, to review Member specific data with the Member's community-based care team, which may include the Member's PCP and other Mental Health or Substance Use Disorder Treatment Programs, as applicable, and implement measurable strategies within ninety (90) calendar days of the execution of this Agreement to reduce these rates.

- 4.12.11.1.3 The MCO's approach to monitoring the thirty (30)-day, ninety (90)-day, and one hundred and eighty (180)-day repeated ED utilization rates attributed to mental illness, to review Member specific data with each of the CMH Programs, and other CMH Providers and Mental Health providers, as applicable, and implement measurable strategies within ninety (90) calendar days of the execution of this Agreement to reduce these rates.
- 4.12.11.1.4 The MCO's approach to monitoring Members' repeated ED utilization rates within thirty (30)-days and ninety (90)-days attributed to substance misuse and Substance Use Disorder, to review Member specific data with the Member's community-based care team, which may include the Member's PCP and other Mental Health or Substance Use Disorder Treatment Programs, as applicable, and implement measurable strategies within ninety (90) calendar days of the execution of this Agreement to reduce these rates.
- 4.12.11.1.5 The MCO's approach to ensuring Members experiencing readmissions or repeated ED utilization have access to a full array of Medically Necessary outpatient medication and Behavioral Health Services after discharge from inpatient or ED care due to a Behavioral Health reason, with sufficient frequency and amounts, to support the Member's progress on achieving their Behavioral Health goals.
- 4.12.11.1.6 For Members with readmissions to any inpatient psychiatric setting within thirty (30) days and one hundred and eighty (180) days, the MCO shall report on the CMH and related service utilization that directly proceeded readmission in accordance with Exhibit O: Quality and Oversight Reporting Requirements. This data shall be shared with the Member's CMH Program/CMH Provider, if

Page 220 of 397 Date \_\_\_\_\_

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

applicable, and the Department in order to evaluate if appropriate levels of care were provided to decrease the likelihood of re-hospitalization.

#### 4.12.12 Written Consent for Release of Behavioral Health Information

- 4.12.12.1 Per 42 CFR Part 2 and NH Code of Administrative Rules, Chapter He-M 309, the MCO shall ensure that both the PCP and behavioral health Provider request written consent from Members to release information to coordinate care regarding mental health services or Substance Use Disorder services, or both, and primary care.
- 4.12.12.2 The MCO shall conduct a review of a sample of case files where written consent was required to determine if a release of information was included in the file.
- 4.12.12.3 The MCO shall report instances in which consent was not given, and, if possible, the reason why, and submit this report in accordance with Exhibit O: Quality and Oversight Reporting Requirements.

## 4.12.13 Coordination Among Behavioral Health Providers

- 4.12.13.1 The MCO shall support communication and coordination between mental health and Substance Use Disorder service Providers and PCPs by providing access to data and information when the Member consent has been documented in accordance with State and federal law, including:
  - 4.12.13.1.1 Assignment of a responsible party to ensure communication and coordination occur and that Providers understand their role to effectively coordinate and improve health outcomes;
  - 4.12.13.1.2 Determination of the method of mental health screening to be completed by Substance Use Disorder service Providers;
  - 4.12.13.1.3 Determination of the method of Substance Use Disorder screening to be completed by mental health service Providers;
  - 4.12.13.1.4 Description of how treatment plans shall be coordinated among Behavioral Health Service Providers; and
  - 4.12.13.1.5 Assessment of cross training of behavioral health Providers (i.e. mental health Providers being trained on Substance Use Disorder issues and Substance Use Disorder Providers being trained on mental health issues).

#### 4.12.14 Member Service Line

Page 221 of 397 Date
----------------------

New Hampshire Department of Health and Human Services Medicaid Care Management Services



## **Exhibit B - DRAFT**

- 4.12.14.1 As further outlined in Section 4.4.4.2 (Member Call Center), the MCO shall operate a Member Services toll-free phone line that is used by all Members, regardless of whether they are calling about physical health or Behavioral Health Services.
- 4.12.14.2 The MCO shall not have a separate number for Members to call regarding Behavioral Health Services, but may either route the call to another entity or conduct a transfer to another entity after identifying and speaking with another individual at the receiving entity to accept the call (i.e., a "warm transfer").
- 4.12.14.3 If the MCO's nurse triage/nurse advice line is separate from its Member Services line, the nurse triage/nurse advice line shall be the same for all Members, regardless of whether they are calling about physical health and/or behavioral health term services.

## 4.12.15 Provision of Services Required by Courts

4.12.15.1 The MCO shall pay for all NH Medicaid State Plan services that are within the Managed Care Program including, but not limited to, assessment and diagnostic evaluations, for its Members as ordered by any court within the State. Court ordered treatment services shall be delivered at an appropriate level of care only when consistent with Medical Necessity for the service.

#### 4.12.16 Behavioral Health Member Experience of Care Survey

- 4.12.16.1 The MCO shall contract with a third party to conduct a Member behavioral health experience of care survey on an annual basis.
- 4.12.16.2 The survey shall be designed by the Department and the MCO's results shall be reported in accordance with Exhibit O: Quality and Oversight Reporting Requirements. The survey shall comply with necessary NCQA Health Plan Accreditation standards.

## 4.12.17 Behavioral Health Emergency Services

- 4.12.17.1 The MCO shall ensure that all types of behavioral health crisis response services are included, such as mobile crisis and office-based crisis services.
  - 4.12.17.1.1 For the period September 1, 2024 through June 30, 2025, certain mobile crisis services may be carved-out of the at-risk services under the MCM benefit package by the Department, as described in separate guidance or otherwise incorporated under MCM through contracts with CMH Programs within the MCO/CMH Subcontracts entered into with the Department's approval.
- 4.12.17.2 Emergency Services shall be accessible to Members anywhere in the region served by the CMH Program.

Page 222 of 397	Date	

New Hampshire Department of Health and Human Services Medicaid Care Management Services



#### **Exhibit B - DRAFT**

- 4.12.17.2.1 Mobile crisis services may be provided by CMH Programs outside of their designated CMH Region to ensure accessibility to Members in crisis 24 hours a day / 7 days a week and within the Geographic Access Standard requirement. Mobile crisis services provided outside of the applicable CMH region are also included.
- 4.12.17.2.2 CMH Program-delivered emergency services that are not delivered by mobile crisis teams, such as for use in determining whether involuntary emergency admission is required, and applying an existing client's crisis safety plan in an office setting, are also included in the meaning of emergency services, and shall be provided within the CMH Program's applicable CMH region only.

#### 4.12.18 Behavioral Health Training Plan

- 4.12.18.1 In accordance with Exhibit O: Quality and Oversight Reporting Requirements, the MCO shall develop a behavioral health training plan each year outlining how it will strengthen behavioral health service and accessibility capacity for Members within the state and to support the efforts of its Behavioral Health provider network to hire, retain and train qualified staff including, but not limited to, CMH Programs, other Community Mental Health Providers of services covered under He-M 426, Substance Use Disorder harm reduction, treatment and recovery providers, and other providers of behavioral health services in the MCO's network that provide services under the Medicaid State Plan
- 4.12.18.2 The MCO shall coordinate its behavioral health training plan's training offerings with the Department to reduce duplication of training efforts, and shall submit the behavioral health training plan to the Department prior to program start, and annually thereafter, inclusive of the training schedule and target Provider audiences.
- 4.12.18.3 As part of the behavioral health training plan, the MCO shall also incorporate strategies to engage Providers in accessing the training opportunities, including explaining the benefits of participating in the training, how it may increase or improve provider competence, and how the knowledge gained will lead to improved quality of care. The MCO's approach shall include opportunities for skill-enhancement through its training opportunities and consultation, through either the MCO or other consultants with expertise in the subject of the training.
- 4.12.18.4 The MCO training plan shall include at least twenty-four (24) hours of training designed to sustain and expand the use of the:

Page 223 of 397	Date

Medicaid Care Management Services Contract
New Hampshire Department of Health and Human Services
Medicaid Care Management Services



## **Exhibit B - DRAFT**

4.12.18.4.1	Trauma Focused Cognitive Behavioral Therapy;
4.12.18.4.2	Trauma Informed Care;
4.12.18.4.3	Motivational Interviewing;
4.12.18.4.4	Interventions for Nicotine Education and Treatment;
4.12.18.4.5	Dialectical Behavioral Therapy (DBT);
4.12.18.4.6	Cognitive Behavioral Therapy;
4.12.18.4.7	Client Centered Treatment Planning;
4.12.18.4.8	Family Psychoeducation;
4.12.18.4.9	Crisis Intervention;
4.12.18.4.10	SBIRT for PCPs;
4.12.18.4.11	Depression Screening for PCPs;
4.12.18.4.12	Managing Cardiovascular and Metabolic Risk for People with SMI; and
4.12.18.4.13	MAT (including education on securing a SAMHSA waiver to provide MAT and, for Providers that already have such waivers, the steps required to increase the number of waiver slots).
administer the	Plan shall also outline the MCO's plan to develop and e following behavioral health trainings for all Providers that are involved in the delivery of Behavioral Health lembers:
4.12.18.5.1	Training for primary care clinics on best practices for behavioral health screening and Integrated Care for common depression, anxiety and Substance Use Disorders;
4.12.18.5.2	Training to physical health Providers on how and when to refer Members for Behavioral Health Services;
4.12.18.5.3	Training to behavioral health Providers on how and when to refer Members for physical health services;
4.12.18.5.4	Cross training to ensure that Mental Health Providers receive Substance Use Disorder training and Substance Use Disorder Providers receive Mental Health training;
4.12.18.5.5	
	New models for behavioral health interventions that can be implemented in primary care settings;

Page 224 of 397 Date \_\_\_\_\_

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

- 4.12.18.5.7 Community-based resources to address health-related social needs.
- 4.12.18.6 The MCO shall offer a minimum of two (2) hours of training each Agreement year to all contracted CMH Program staff on suicide risk assessment, suicide prevention and post intervention strategies in keeping with the Department's objective of reducing the number of suicides in NH.
- 4.12.18.7 The MCO shall provide, on at least an annual basis, training on appropriate billing practices to Participating Providers. The Department reserves the discretion to change training plan areas of focus in accordance with programmatic changes and objectives.
- 4.12.18.8 In accordance with Exhibit O: Quality and Oversight Reporting Requirements, the MCO shall summarize in the annual Behavioral Health Strategy Plan and Report the training that was provided, a copy of the agenda for each training, a participant registration list, and a summary, for each training provided, of the evaluations done by program participants, and the proposed training for the next fiscal year.

## 4.12.19 **Parity**

- 4.12.19.1 The MCO and its Subcontractors shall comply with the Mental Health Parity and Addiction Equity Act of 2008, 42 CFR 438, subpart K, which prohibits discrimination in the delivery of mental health and Substance Use Disorder services and in the treatment of Members with, at risk for, or recovering from a mental health or Substance Use Disorder.
- 4.12.19.2 Semi-Annual Report on Parity
  - 4.12.19.2.1 The MCO shall complete the Department's Parity Compliance Report which shall include, at a minimum:
    - 4.12.19.2.1.1. All Non-Quantitative and Quantitative Treatment Limits identified by the MCO pursuant to the Department's criteria;
    - 4.12.19.2.1.2. All Member grievances and appeals regarding a parity violation and resolutions;
    - 4.12.19.2.1.3. The processes, strategies, evidentiary standards. other factors or determining access to Non-Participating Providers for health mental Substance Use Disorder benefits that are comparable to, and applied no more stringently processes, than, the

Page 225 of 397	Date	

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

strategies, evidentiary standards, or other factors in determining access to Non-Participating Providers for medical/surgical benefits in the same classification:

- 4.12.19.2.1.4. A comparison of payment for services that ensure comparable access for people with mental health diagnoses; and
- 4.12.19.2.1.5. Any other requirements identified in Exhibit O: Quality and Oversight Reporting Requirements. [61 Fed. Reg. 18413, 18414 and 18417 (March 30, 2016)]
- 4.12.19.2.2 The MCO shall review its administrative and other practices, including those of any contracted behavioral health organizations or third party administrators, for the prior calendar year for compliance with the relevant provisions of the federal Mental Health Parity Law, regulations and guidance issued by State and federal entities.
- 4.12.19.2.3 The MCO shall annually submit a certification signed by the CEO and chief medical officer (CMO) stating that the MCO has completed a comprehensive review of the administrative, clinical, and utilization practices of the MCO for the prior calendar year for compliance with the necessary provisions of State Mental Health Parity Laws and federal Mental Health Parity Law and any guidance issued by State and federal entities.
- 4.12.19.2.4 If the MCO determines that any administrative, clinical, or utilization practices were not in compliance with relevant requirements of the federal Mental Health Parity Law or guidance issued by State and federal entities during the calendar year, the certification shall state that not all practices were in compliance with federal Mental Health Parity Law or any guidance issued by state or federal entities and shall include a list of the practices not in compliance and the steps the MCO has taken to bring these practices into compliance.
- 4.12.19.2.5 A Member enrolled in any MCO may file a complaint with the Department at nhparity@dhhs.nh.gov if services are provided in a way that is not consistent

Page 226 of 397 Date \_\_\_\_\_

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

with applicable federal Mental Health Parity laws, regulations or federal guidance.

4.12.19.2.6 As described in Section 4.4 (Member Services), the MCO shall describe the parity compliant process, including the appropriate contact information, in the Member Handbook.

#### 4.12.19.3 Prohibition on Lifetime or Annual Dollar Limits

4.12.19.3.1 The MCO shall not impose aggregate lifetime or annual dollar limits on mental health or Substance Use Disorder benefits. [42 CFR 438.905(b)]

#### 4.12.19.4 Restrictions on Treatment Limitations

- 4.12.19.4.1 The MCO shall not apply any financial requirement or treatment limitation applicable to mental health or Substance Use Disorder benefits that are more restrictive than the predominant treatment limitations applied to substantially all medical and surgical benefits covered by the plan (or coverage), and the MCO shall not impose any separate treatment limitations that are applicable only with respect to mental health or Substance Use Disorder benefits. [42 CFR 438.910(b)(1)]
- 4.12.19.4.2 The MCO shall not apply any cumulative financial requirements for mental health or Substance Use Disorder benefits in a classification that accumulates separately from any established for medical/surgical benefits in the same classification. [42 CFR 438.910(c)(3)]
- 4.12.19.4.3 If an MCO Member is provided mental health or Substance Use Disorder benefits in any classification of benefit, the MCO shall provide mental health or Substance Use Disorder benefits to Members in every classification in which medical/surgical benefits are provided. [42 CFR 438.910(b)(2)]
- 4.12.19.4.4 The MCO shall not impose Non-Quantitative Treatment Limits for Community Mental Health or Substance Use Disorder benefits in any classification unless, under the policies and procedures of the MCO as written and in operation, any processes, strategies, evidentiary standards, or other factors used in applying the Non-Quantitative Treatment Limits to mental health or Substance Use Disorder benefits in the classification are comparable to, and are applied no more stringently than, the processes.

Page 227 of 397 Date \_\_\_\_\_

New Hampshire Department of Health and Human Services Medicaid Care Management Services



## **Exhibit B - DRAFT**

strategies, evidentiary standards, or other factors used in applying the limitation for medical/surgical benefits in the classification. [42 CFR 438.910(d)]

## 4.12.20 Community Mental Health Services

4.12.20.1 General Requirements

- 4.12.20.1.1 The MCO shall be required to enter into a Department approved capitation model of contracting with every CMH Program that is currently approved by the Department pursuant to NH Code of Administrative Rules, Chapter He-M 403, which is essential to supporting Member access to the full continuum of Community Mental Health Services under NH Code of Administrative Rules, Chapter He-M 426 in the MCM program. The MCOs shall utilize a Department provided standard contract for this purpose to ensure continuity of services and care across the Community Mental Health Services systems for Members.
- 4.12.20.1.2 The MCO shall reach agreements and enter into contracts with all CMH Programs that meet the terms specified by the Department no later than ninety (90) calendar days after the MCM program's Agreement execution.
- 4.12.20.1.3 For the purposes of this paragraph, Agreement execution means that the Agreement has been signed by the MCO and the State, and approved by all required State authorities and is generally expected to occur in September 2024.
- 4.12.20.1.4 The MCO shall be subject to payment requirements described in Section 4.17 (Provider Payments: Community Mental Health Programs).
- 4.12.20.1.5 The MCO shall comply with key administrative functions and processes for CMH Services delivered by CMH Programs (CMHCs), which may include, but are not limited to:
  - 4.12.20.1.5.1. Timely processing of CMH Services Member eligibility lists, which shall be provided to the MCO by the CMH Programs and shall indicate the Member's eligibility for CMH Services pursuant to the eligibility categories under NH Code of Administrative Rules, Chapter He-M 401. The MCO shall validate the eligibility lists through a

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

process developed in collaboration with the CMH Programs and approved by the Department;

- 4.12.20.1.5.2. Determining whether Members are eligible for the DHHS-required CMH Services Capitation Payments to CMH Programs, or whether the CMH Program should be paid on a FFS basis for the service the Member received:
- 4.12.20.1.5.3. Providing detailed MCO data submissions to DHHS and the CMH Program for purposes of reconciling payments and performance pursuant to the MCO-CMH Program Contract, and for CMH Services provided by a CMH Provider not under subcontract with a CMH Program for the applicable service for purposes of reconciling payments and performance (e.g., 835 file);
- 4.12.20.1.5.4. Establishing a coordinated effort for Substance Use Disorder treatment in collaboration with CMH Programs by CMH Region, as defined in NH Code of Administrative Rules, Chapter He-M 425, and with CMH Providers not under subcontract with a CMH Program, to ensure Members have access to Substance Use Disorder treatment services they may need from other providers, if not provided by the CMH Program or the CMH Provider under NH Code of Administrative Rules, Chapter He-M 426; and
- 4.12.20.1.5.5. Monitoring of CMH Program performance through quality metrics and oversight procedures
- 4.12.20.1.5.6. Ensuring compliance with this Agreement, where applicable, and all applicable State and federal laws, rules and regulations.
- 4.12.20.1.5.7. Overseeing, enforcing, and remedying contract disputes between the MCO and CMH Program.

Page 229 of 397 Date \_\_\_\_\_

New Hampshire Department of Health and Human Services Medicaid Care Management Services



#### **Exhibit B - DRAFT**

- 4.12.20.1.5.8. All additional capabilities set forth by DHHS during the Readiness Review process.
- 4.12.20.1.6 In the event a CMH Program is designated by the Department as a Certified Community Behavioral Health Clinic, the MCO shall enter into a different contractual relationship and payment model for the payment and delivery of the full continuum of Community Mental Health Services delivered by the agency, Mental Health Services available at lower levels of care, and applicable Substance Use Disorder services.
- 4.12.20.2 MCO Agreements and Payment for Community Mental Health Services CMH Providers
  - 4.12.20.2.1 Consistent with 4.14, Network Requirements, the MCO shall maintain and monitor a network of CMH Providers for the provision of Community Mental Health Services described in NH Code of Administrative Rules, Chapter He-M 426 on behalf of Medicaid Members who are eligible for such services in accordance with He-M 401.
  - 4.12.20.2.2 The MCO shall provide for monitoring of CMH Provider performance through quality metrics and oversight procedures detailed in the MCO's provider or network agreement with each CMH Provider.
  - 4.12.20.2.3 The MCO shall ensure that its agreements with CMH Providers meet the following requirements:
    - 4.12.20.2.3.1. Comply with the requirements of this Agreement and all applicable State and federal laws, rules and regulations;
    - 4.12.20.2.3.2. Define the role of the MCO versus the CMH Provider;
    - 4.12.20.2.3.3. Include procedures for communication and coordination between the MCO and the CMH Provider, other Providers serving the same Member, CMH Programs as may be required by He-M 426 for CMH Provider provided services and the need to collaborate with the applicable CMH Program, and the Department;

Page 230 of 397 Date \_\_\_\_\_

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

- 4.12.20.2.3.4. Include provisions for data sharing on Members;
- 4.12.20.2.3.5. Include data reporting between the CMH Provider and the MCO and the Department; and
- 4.12.20.2.3.6. Include provisions for oversight, enforcement, and remedies for contract disputes.
- 4.12.20.2.4 The MCO shall ensure that Community Mental Health Services provided by CMH Providers are provided in accordance with the Medicaid State Plan and He-M 401.02, He-M 403.02 and He-M 426.
- 4.12.20.2.5 This includes, but is not limited to, ensuring that Community Mental Health Services for which the CMH Provider is currently approved by the Department to provide, are appropriately provided to eligible Members.
- 4.12.20.2.6 For all Community Mental Health Services provided by a CMH Provider, the CMH Provider shall comply with He-M 426.04, including but not limited to, ensuring that all Members receiving CMH Services from the CMH Provider have been identified as currently eligible Members to receive CMH Services by a CMH Program, pursuant to He-M 401, and that the CMH Provider has a method for collaborative service planning and service delivery with the regional CMH Program, including joint development and approval of an Individual Service Plan for each Member.

## 4.12.20.3 Community Mental Health Services Continuum

- 4.12.20.3.1 Eligible Members shall be offered the provisions of supports for illness self-management and recovery;
- 4.12.20.3.2 Eligible Members shall be provided with coordinated care when entering and leaving a designated receiving facility.
- 4.12.20.3.3 The MCO shall ensure that all Providers providing Community Mental Health Services comply with the requirements of He-M 426.
- 4.12.20.3.4 As described in He-M 400, only Members who are currently eligible for Community Mental Health Services are eligible to receive Community Mental

age 231 of 397	Date

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

Health Services. Eligibility shall be determined by a CMH Program pursuant to He-M 401, due to a:

- 4.12.20.3.4.1. Severe or persistent mental illness (SPMI) for an adult;
- 4.12.20.3.4.2. SMI for an adult;
- 4.12.20.3.4.3. SPMI or SMI with low service utilization for an adult;
- 4.12.20.3.4.4. SED for a child; or
- 4.12.20.3.4.5. SED and interagency involvement for a child.
- 4.12.20.3.5 Any MCO quality monitoring or audits of the performance of the CMH Programs or of CMH Providers shall be available to the Department upon request.
- 4.12.20.3.6 To improve health outcomes for Members and ensure that the delivery of services is provided at the appropriate intensity and duration, the MCO shall meet with CMH Programs, CMH Providers, and the Department at least quarterly to coordinate data collection and ensure data sharing.
- 4.12.20.3.7 At a minimum, this shall include sharing of quality assurance activities conducted by the MCO and the Department and a review of quality improvement plans, data reports, Care Coordination activities, and outstanding needs. Reports shall be provided in advance of quarterly meetings.
- 4.12.20.3.8 The MCO shall work in collaboration with the Department, CMH Programs and CMH Providers to support and sustain evidenced-based practices that have a profound impact on Providers and Member outcomes.

## 4.12.20.4 Comprehensive Assessment and Care Plans

4.12.20.4.1 The MCO shall ensure, through its regular quality improvement activities, on-site reviews for children and youth, and reviews of the Department administered quality service reviews for adults, that Community Mental Health Services are delivered in the least restrictive community based environment possible and based on a person-centered approach where the Member and his or her family's personal

Page 232 of 397 Date \_\_\_\_\_

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

- goals and needs are considered central in the development of the individualized service plans.
- 4.12.20.4.2 The MCO shall ensure that initial and updated Care Plans are based on a Comprehensive Assessment conducted by a CMH Program using an evidenced-based assessment tool, such as the NH version of the Child and Adolescent Needs and Strengths Assessment (CANS) and the Adult Needs and Strengths Assessment (ANSA).
- 4.12.20.4.3 If the MCO, or a CMH Program acting on behalf of the MCO, elects to permit clinicians to use an evidenced-based assessment tool other than CANS or ANSA, the MCO shall notify and receive approval of the specific tool from the Department.
- 4.12.20.4.4 The MCO shall ensure that clinicians conducting or contributing to a Comprehensive Assessment are certified in the use of NH's CANS and ANSA, or an alternative evidenced based assessment tool approved by the Department within one hundred and twenty (120) calendar days of implementation by the Department of a web-based training and certification system.
- 4.12.20.4.5 The MCO shall require that CMH Program's certified clinicians use the CANS, ANSA, or an alternative evidenced-based assessment tool approved by the Department for any newly evaluated Member and for an existing Member no later than at the Member's first eligibility renewal determination for CMH Services, following certification.
- 4.12.20.5 Assertive Community Treatment (ACT)
  - 4.12.20.5.1 The MCO shall work in collaboration with DHHS, CMH Programs, and CMH Providers to ensure that Members identified as needing ACT services are provided ACT services pursuant to He-M 426.16, and in sufficient quantity to ensure applicable Members have appropriate access to these service.
  - 4.12.20.5.2 In collaboration with the Department, the MCO shall support CMH Programs and CMH Providers, if applicable, to achieve program improvement goals outlined in the ACT Quality Improvement Plan on file with the Department to achieve full implementation of ACT.

age 233 of 397	Date

New Hampshire Department of Health and Human Services Medicaid Care Management Services



#### **Exhibit B - DRAFT**

- 4.12.20.5.3 In accordance with Exhibit O: Quality and Oversight Reporting Requirements, the MCO shall report quarterly on the rate at which the MCO's Medicaid Members eligible for Community Mental Health Services are receiving ACT services.
- 4.12.20.6 Mental Health Performance Improvement Project
  - 4.12.20.6.1 As outlined in Section 4.14.3.8 (Performance Improvement Projects), the MCO shall focus on the Department's objectives outlined in the NH MCM Quality Strategy.
- 4.12.20.7 Services for the Homeless
  - 4.12.20.7.1 The MCO shall provide care to Members who are homeless or at risk of homelessness by conducting outreach to Members with a history of homelessness and establishing partnerships with community-based organizations to connect such Members to housing services.
  - 4.12.20.7.2 In its contract with CMH Programs, the MCO shall describe how it shall provide appropriate oversight of CMH Program responsibilities, including:
    - 4.12.20.7.2.1. Identifying housing options for Members at risk of experiencing homelessness;
    - 4.12.20.7.2.2. Assisting Members in filing applications for housing and gathering necessary documentation;
    - 4.12.20.7.2.3. Coordinating the provision of supportive housing; and
    - 4.12.20.7.2.4. Coordinating housing-related services amongst CMH Programs, the MCO and NH's Housing Bridge Subsidy Program.
  - 4.12.20.7.3 The contract with CMH Programs shall require quarterly assessments and documentation of housing status and homelessness for all Members.
  - 4.12.20.7.4 The MCO shall ensure that any Member discharged into homelessness is connected to Care Management as described in Section 4.11.7 (Provider-Delivered Care Coordination and Integration with Social Services and Community Care) within twenty-four (24) hours upon release.

4.12.20.8 Supported Employment

Page 234 of 397	Date	

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

- 4.12.20.8.1 In coordination with CMH Programs and CMH Providers, if applicable, the MCO shall actively promote an Evidence Based Supported Employment (EBSE) or an Individual Placement and Support Model of Supported Employment (IPS-SE) to eligible Members, whichever is the Department approved model pursuant to He-M 426.
- 4.12.20.8.2 The MCO shall obtain fidelity review reports from the Department to inform EBSE team's adherence to fidelity with the expectation of at least good fidelity implementation for each CMH Program and CMH Provider, if providing supported employment services.
- 4.12.20.8.3 In collaboration with DHHS, the MCO shall support the CMH Programs and CMH Providers to achieve program improvement goals outlined in the applicable model's Quality Improvement Plan on file with DHHS to achieve full implementation of the model.
- 4.12.20.8.4 Based on data provided by the Department, the MCO shall support DHHS's goals to ensure that at least nineteen percent (19%) of adult CMH eligible Members are engaged in a Department approved supported employment model of supported employment services and that employment status is updated by the CMH Program and CMH Provider, if applicable on a quarterly basis.
- 4.12.20.8.5 The MCO shall report the Supported Employment participation rate to the Department in accordance with Exhibit O: Quality and Oversight Reporting Requirements and provide updates as requested by DHHS during regular behavioral health meetings between the MCO and the Department.
- 4.12.20.9 Illness Management and Recovery (IMR)
  - 4.12.20.9.1 In coordination with CMH Programs and CMH Providers, if applicable, the MCO shall actively promote the delivery of, and increased penetration rates of, Illness Management and Recovery to Members with SMI and SPMI.
  - 4.12.20.9.2 The MCO shall provide updates as requested by DHHS during regular behavioral health meetings between the MCO and the Department.
- 4.12.20.10 Dialectical Behavioral Therapy (DBT)

Page 235 of 397	Date	

New Hampshire Department of Health and Human Services Medicaid Care Management Services



#### **Exhibit B - DRAFT**

- 4.12.20.10.1 In coordination with CMH Programs, the MCO shall actively promote the delivery of DBT to Members with diagnoses, including but not limited to SMI, SPMI, and Borderline Personality Disorder.
- 4.12.20.10.2 The MCO shall provide updates, such as the rate at which eligible Members receive meaningful levels of DBT services, as requested by the Department during regular behavioral health meetings between the MCO and DHHS.
- 4.12.20.11 Peer Support Services (PSS)
  - 4.12.20.11.1 In coordination with CMH Programs, the MCO shall actively promote the delivery of PSS provided by Peer Support Specialists who are employees of CMH Programs.
  - 4.12.20.11.2 The MCOs, in coordination with CMH Programs, the Department and Peer Support Agencies authorized by the Department under He-M 402, shall actively promote in a variety of settings, such as New Hampshire Hospital, primary care clinics, EDs, CMH Programs, and CMH Provider sites, the delivery of peer support services provided by Peer Support Agencies under He-M 402.
  - 4.12.20.11.3 The MCO shall provide updates as requested by the Department during regular behavioral health meetings between the MCO and DHHS on its efforts to promote Peer Support Services delivered in CMH Program and those provided by Peer Support Agencies under He-M 402.
- 4.12.20.12 Modular Approach to Therapy for Children with Anxiety, Depression, Trauma, or Conduct Problems.
  - 4.12.20.12.1 In coordination with CMH Programs, the MCO shall actively promote the delivery of Modular Approach to Therapy for Children with Anxiety, Depression, Trauma, or Conduct Problems<sup>17</sup> for children and youth Members experiencing anxiety, depression, trauma and conduct issues.
  - 4.12.20.12.2 The MCO shall provide updates as requested by the Department during regular behavioral health meetings between the MCO and the Department.

4.12.20.13 First Episode Psychosis

Page 236 of 397 Date \_\_\_\_\_

<sup>&</sup>lt;sup>17</sup> Available at: <a href="http://www.practicewise.com/portals/0/match\_public/index.html">http://www.practicewise.com/portals/0/match\_public/index.html</a>.

New Hampshire Department of Health and Human Services Medicaid Care Management Services



#### **Exhibit B - DRAFT**

- 4.12.20.13.1 In coordination with CMH Programs, the MCO shall actively promote the delivery of programming to address early symptoms of psychosis.
- 4.12.20.13.2 The MCO shall provide updates as requested by the Department during regular behavioral health meetings between the MCO and the Department.
- 4.12.20.14 Child Parent Psychotherapy
  - 4.12.20.14.1 In coordination with CMH Programs, the MCO shall actively promote delivery of Child Parent Psychotherapy for young children.
  - 4.12.20.14.2 The MCO shall provide updates as requested by the Department during regular behavioral health meetings between the MCO and the Department.
- 4.12.20.15 Changes in Healthy Behavior
  - 4.12.20.15.1 The MCO shall promote Community Mental Health Service recipients' whole health goals to address health disparities.
  - 4.12.20.15.2 Efforts can encompass interventions (e.g., tobacco cessation, "InShape") or other efforts designed to improve health.
  - 4.12.20.15.3 The MCO shall gather smoking status data on all Members and report to the Department in accordance with Exhibit O: Quality and Oversight Reporting Requirements.
  - 4.12.20.15.4 The MCO shall support CMH Programs to establish incentive programs for Members to increase their engagement in healthy behavior change initiatives.

#### 4.12.20.16 Psychiatric Boarding

4.12.20.16.1 The MCO shall provide assistance and support to Members, directly or through the Member's care team, to reduce the frequency and duration of the Member's wait for psychiatric services needed on an acute or crisis basis, regardless of the facility type best-suited to meet the Member's immediate care and treatment needs. The MCO's assistance shall include a beneficiary-specific plan for discharge, treatment, admittance or transfer to New Hampshire Hospital, or other State determined facility or IMDs for mental illness or Substance Use Disorder services.

Page 237 of 397 Date \_\_\_\_\_

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

- 4.12.20.16.2 At the request of the Department, the MCO shall participate in meetings with hospitals to address Psychiatric Boarding.
- 4.12.20.16.3 The MCO shall pay no less than the rate paid by NH Medicaid FFS program for all inpatient and outpatient service categories for billable services related to psychiatric boarding.
- 4.12.20.16.4 The MCO's capitation rates related to psychiatric services shall reflect utilization levels consistent with best practices for clinical path protocols, ED Psychiatric Boarding services, and discharge/readmission management at or from New Hampshire Hospital or other State determined IMDs for mental illness or Substance Use Disorder services.
- 4.12.20.16.5 The MCO shall describe its plan for reducing Psychiatric Boarding in its Annual Behavioral Health Strategy Plan and Report, in accordance with Exhibit O: Quality and Oversight Reporting Requirements.
- 4.12.20.16.6 At minimum, the Plan shall address how:
  - 4.12.20.16.6.1. The MCO identifies when its Members are in the ED awaiting psychiatric placement or in a hospital setting awaiting an inpatient psychiatric bed;
  - 4.12.20.16.6.2. Policies for ensuring a prompt crisis team consultation and face-to-face evaluation:
  - 4.12.20.16.6.3. Strategies for identifying placement options or alternatives to hospitalization; and
  - 4.12.20.16.6.4. Coordination with the CMH Programs and CMH Providers, as applicable, serving Members.
- 4.12.20.16.7 In accordance with Exhibit O: Quality and Oversight Reporting Requirements, the MCO shall provide a monthly report on the number of its Members awaiting placement in the ED or in a hospital setting for twenty-four (24) hours or more; the disposition of those awaiting placement; and the average length of stay in the ED and medical ward for both children and adult Members, and the rate of recidivism for Psychiatric Boarding.

Page 238 of 397 Date \_\_\_\_\_

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

## 4.12.21 Agreements for New Hampshire State-Owned Hospital Agreement(s) and Other State Determined IMDs for Mental Illness

- 4.12.21.1 The MCO shall maintain a written collaborative agreement with New Hampshire state-owned hospitals (New Hampshire Hospital and Hampstead Hospital), hereinafter State-Owned Hospitals, and other State determined IMDs for mental illness.
- 4.12.21.2 This collaborative agreement shall be subject to the approval of DHHS and shall address the ADA requirement that Members be served in the most integrated setting appropriate to their needs, include the responsibilities of the CMH Program and CMH Provider, as applicable, to ensure a seamless transition of care upon admission and discharge to the community, and detail information sharing and collaboration between the MCO and State-Owned Hospitals and other State determined IMDs for mental illness.
- 4.12.21.3 The collaborative agreement shall also include mutually developed admission and utilization review criteria bases for determining the appropriateness of admissions to or continued stays both within and external to State-Owned Hospitals and other State determined IMDs for mental illness.
- 4.12.21.4 Prior to admission to State-Owned Hospitals or other State determined IMDs for mental illness, the MCO shall ensure that a crisis team consultation has been completed for all Members evaluated by a licensed physician or psychologist.
- 4.12.21.5 The MCO shall ensure that a face-to-face evaluation by a mandatory pre-screening agent is conducted to assess eligibility for emergency involuntary admission to State-Owned Hospitals and determine whether all available less restrictive alternative services and supports are unsuitable.

## 4.12.22 Discharge Planning

- 4.12.22.1 The MCO shall ensure that upon discharge from a State-Owned Hospital, inpatient psychiatric facility, or other State determined IMDs for mental illness, the Member has immediate access to an appropriate living situation rather than a homeless shelter.
- 4.12.22.2 The MCO shall track any Member discharges that the MCO, through its Provider network, was unable to place into the community and Members who instead were discharged to a shelter or into homelessness.
- 4.12.22.3 Also included in Section 3.11.2 (Other MCO Required Staff), the MCO shall designate an off-site liaison with privileges to continue the Member's Care Management, and assist in facilitating a coordinated discharge planning process for Members admitted to

age 239 of 397	Date

New Hampshire Department of Health and Human Services Medicaid Care Management Services



#### **Exhibit B - DRAFT**

State-Owned Hospitals or other State determined IMDs for mental illness.

- 4.12.22.4 In the event the Member is attributed to a CMH Program upon their admission or discharge, the MCO's liaison shall assist and collaborate with the applicable CMH Program to expedite discharge and engagement in ongoing CMH Services provided by the CMH Program or CMH Provider, as may be applicable, which may include the Member's participation in Critical Time Intervention, Supportive Housing Services, or other Department approved evidence based practices covered as an In Lieu of Service, a 1915(i) service, or under a Department approved contract for Transitional Housing Services.
- 4.12.22.5 In the event the Member is not attributed to a CMH Program upon admission or discharge, the MCO's shall actively participate in State-Owned Hospital and other State determined IMDs for mental illness treatment team meetings and discharge planning meetings to ensure that Members receive treatment in the least restrictive environment complying with the ADA and other applicable State and federal regulations.
- 4.12.22.6 The MCO shall actively participate, and assist State-Owned Hospitals and other State determined IMDs for mental illness staff in the development of a written discharge plan within twenty-four (24) hours of admission.
- 4.12.22.7 The MCO shall ensure that the final State-Owned Hospitals or other State determined IMDs for mental illness discharge instruction sheet shall be provided to the Member and the Member's authorized representative prior to discharge, or the next business day, for at least ninety-eight percent (98%) of Members discharged.
- 4.12.22.8 The MCO shall ensure that the discharge progress note shall be provided to the aftercare Provider within seven (7) calendar days of Member discharge for at least ninety-eight percent (98%) of Members discharged.
- 4.12.22.9 For ACT team service recipients, the MCO shall ensure that the discharge progress note is provided to the CMH Program or CMH Provider, if applicable, within twenty-four (24) hours of Member discharge.
- 4.12.22.10 If a Member lacks a reasonable means of communicating with a plan prior to discharge, the MCO shall identify an alternative viable means for communicating with the Member in the discharge plan.
- 4.12.22.11 The MCO shall make at least three (3) attempts to contact Members within three (3) business days of discharge from State-Owned Hospitals and other State determined IMDs for mental illness in order to review the discharge plan, support the Member in

age 240 of 397	Date

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

- attending any scheduled follow-up appointments, support the continued taking of any medications prescribed, and answer any questions the Member may have.
- 4.12.22.12 The performance metric shall be that one hundred percent (100%) of Members discharged shall have been attempted to be contacted within three (3) business days.
  - 4.12.22.12.1 For any Member the MCO does not make contact with within three (3) business days, the MCO shall contact the aftercare Provider and request that the aftercare Provider make contact with the Member within twenty-four (24) hours.
  - 4.12.22.12.2 The MCO shall ensure an appointment with a CMH Program or CMH Provider or other appropriate mental health clinician is scheduled and that transportation has been arranged for the appointment prior to discharging a Member.
- 4.12.22.13 Such appointment shall occur within seven (7) calendar days after discharge.
- 4.12.22.14 Members receiving ACT team services shall be seen within twenty-four (24) hours of discharge by the applicable CMH Program or CMH Provider.
- 4.12.22.15 For Members discharged from psychiatric hospitalization who are not currently attributable to a CMH Program, the Member shall have an intake appointment that is scheduled to occur with the CMH Program assigned to the CMH Region in which the Member resides within seven (7) calendar days after discharge.
- 4.12.22.16 The MCO shall work with DHHS and the applicable CMH Program and CMH Provider to review cases of Members that New Hampshire Hospital and other State determined IMDs for mental illness have indicated a difficulty returning back to the community, identify barriers to discharge, and develop an appropriate transition plan back to the community.

## 4.12.23 Administrative Days and Post Stabilization Care Services

- 4.12.23.1 The MCO shall perform Member in-reach activities within State-Owned Hospitals and other State determined IMDs for mental illness and other State determined IMDs for mental illness designed to accomplish transitions to the community in collaboration with the CMH Program applicable to the CMH Region to which the Member's town of residence is attributed. These activities shall include, but not be limited to:
  - 4.12.23.1.1 The MCO's use of the Department's event notification system and closed-loop referral solution, if available,

Page 241 of 397 Date \_\_\_\_\_

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

to facilitate sharing of clinical, care, transition to other levels of care, discharge planning, CMH eligibility assessment, and final discharge information;

- 4.12.23.1.2 The MCO's and CMH Program's meaningful and effective collaboration with applicable members of the IMD's care team assigned to the Member to ensure that the MCO and CMH Program are appropriately informed of the Member's ongoing care needs post-discharge.
- In the event the Member declines to consent to the 4.12.23.1.3 CMH Program's involvement in discharge planning and the CMH Program becoming their post-discharge ongoing provider of CMH Services, the MCO shall follow this same approach to in-reach activities utilizing the Member's CMH Provider, if applicable, or other Mental Health Services provider of covered services at levels lower than CMH Services. In such cases, the MCO shall directly, or through the other CMH Provider or Mental Health services, connect, in sufficient frequency and effective duration, with the Member post-discharge to ensure the Member's access to the post-discharge services is sufficient to support the Member's continued progress toward achieving the behavioral health related goals.

#### 4.12.24 Substance Use Disorder

- 4.12.24.1 The MCO's policies and procedures related to Substance Use Disorder shall be in compliance with State and federal law, including but not limited to, Chapter 420-J, Section J:15 through Section J:19 and shall comply with all State and federal laws related to confidentiality of Member behavioral health information.
- 4.12.24.2 In addition to services covered under the Medicaid State Plan, the MCO shall cover the services necessary for compliance with the requirements for parity in mental health and Substance Use Disorder benefits. [42 CFR 438, subpart K; 42 CFR 438.3(e)(1)(ii)]
- 4.12.24.3 The MCO shall ensure that the full continuum of care required for Members with Substance Use Disorders is available and provided to Members in accordance with NH Code of Administrative Rules, Chapter He-W 500, Part He-W 513.

#### 4.12.25 Contracting for Substance Use Disorder

age 242 of 397	Date

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

- 4.12.25.1 The MCO shall contract with Substance Use Disorder service programs and Providers to deliver Substance Use Disorder services for eligible Members, as defined in He-W 513.<sup>18</sup>
- 4.12.25.2 The contract between the MCO and the Substance Use Disorder programs and Participating Providers shall be submitted to DHHS for review and approval prior to implementation in accordance with Section 3.10.2 (Contracts with Subcontractors).
- 4.12.25.3 The contract shall, at minimum, address the following:
  - 4.12.25.3.1 The scope of services to be covered;
  - 4.12.25.3.2 Compliance with the requirements of this Agreement and applicable State and federal law;
  - 4.12.25.3.3 The role of the MCO versus the Substance Use Disorder program and/or Provider;
  - 4.12.25.3.4 Procedures for communication and coordination between the MCO and the Substance Use Disorder program and/or Provider;
  - 4.12.25.3.5 Other Providers serving the same Member, and DHHS as applicable;
  - 4.12.25.3.6 The approach to payment, including payment for MAT services;
  - 4.12.25.3.7 Data sharing on Members;
  - 4.12.25.3.8 Data reporting between the Substance Use Disorder programs and/or Providers and the MCO, and DHHS as applicable; and
  - 4.12.25.3.9 Oversight, enforcement, and remedies for contract disputes.
- 4.12.25.4 The contract shall provide for monitoring of Substance Use Disorder service performance through quality metrics and oversight procedures specified in the contract.
- 4.12.25.5 When contracting with Peer Recovery Programs, the MCO shall contract with all Willing Providers in the State through the PRSS Facilitating Organization or other accrediting body approved by DHHS, unless the Provider requests a direct contract.
- 4.12.25.6 The MCO shall reimburse Peer Recovery Programs in accordance with rates that are no less than the equivalent DHHS FFS rates.
- 4.12.25.7 When contracting with methadone clinics, the MCO shall contract with and have in its network all Willing Providers in the state.

Page 243 of 397 Date \_\_\_\_\_

<sup>&</sup>lt;sup>18</sup> Available at <a href="http://www.gencourt.state.nh.us/rules/state\_agencies/he-w.html">http://www.gencourt.state.nh.us/rules/state\_agencies/he-w.html</a>.

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

#### 4.12.26 Provision of Substance Use Disorder Services

- 4.12.26.1 The MCO shall ensure that Substance Use Disorder services are provided in accordance with the Medicaid State Plan and He-W 513, this includes but is not limited to all of the MCO's Substance Use Disorder service providers' compliance with the Covered Services provisions in He-W 513.0 applicable to their provider type, to Opioid Treatment Programs, other Substance Use Disorder Treatment, and Recovery Services providers. This includes, but is not limited to:
  - 4.12.26.1.1 Ensuring that the full continuum of care is appropriately provided to eligible Members including, but not limited to the provision of treatment and services that meet the Member's assessed ASAM level of care needs, and subject to the following additional conditions associated with certain providers of Substance Use Disorder services:
    - 4.12.26.1.1.1. For those providers for whom the MCO is contracted with under a Departmentapproved directed payment model, such as Community Mental Health Programs, or a prospective payment system model. Certified such as Community Behavioral Health Clinics, the MCO's obligation to ensure the provision of the continuum of care shall be achieved through the MCO's review of services provided to Members, audits of clinical records no less than annually, and through its collaboration between those providers and the balance of the Member's care team, as appropriate;
    - 4.12.26.1.1.2. Ensuring that eligible Members are provided with recovery support services; and
    - 4.12.26.1.1.3. Ensuring that eligible Members are provided with coordinated care by the current treatment program provider and the provider(s) to whom the Member is being referred for ongoing treatment and services when entering or leaving a treatment program.
  - 4.12.26.1.2 For those providers for whom the MCO is contracted with under a Department-approved directed payment model, such as Community Mental Health Programs,

Page 244 of 397 Date \_\_\_\_\_

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

or a prospective payment system model, such as Certified Community Behavioral Health Clinics, the MCO's obligation to ensure the provision of coordinated care shall be achieved through the MCO's direct involvement that facilitates connection between the providers, or at minimum confirms that the connection has been made between the providers.

- 4.12.26.1.3 In the event the MCO cannot ensure or otherwise confirm that the Member has been connected to subsequent treatment or other services identified as necessary for the Member, within a time period that is sufficient to support effective continuity of care, including authorization of pharmacotherapy, the MCO shall contact the Member directly to facilitate connection to such services.
- 4.12.26.2 The MCO shall ensure that all Providers providing Substance Use Disorder services comply with the requirements of He-W 513, through mechanisms including but not limited to claims utilization review, record audits, reauthorizations when applicable, and provider enrollment qualifications and certification audits.
  - 4.12.26.2.1 The MCO shall conduct reviews and audits of clinical records and claims for Members receiving Substance Use Disorder treatment services provided by Substance Use Disorder Programs and Medication Assisted Treatment Services provided by Opioid Treatment Programs (OTP), as described in separate quidance.
  - 4.12.26.2.2 For Providers of Substance Use Disorder services that are delivered through CMH Programs under a Department approved APM, and Certified Community Behavioral Health Clinic under a Department approved PPS, this shall be limited to analysis of utilization patterns, provider and Department released quality reviews, and MCO conducted audits as required by the Department in this Agreement.
- 4.12.26.3 The MCO shall work in collaboration with DHHS and Substance Use Disorder programs and/or Providers to support and sustain evidenced-based practices that have a profound impact on Provider and Member outcomes, including, but is not limited to, enhanced rate or incentive payments for evidenced-based practices.
- 4.12.26.4 The MCO shall ensure that the full continuum of care required for Members with Substance Use Disorders is available and provided

Page 245 of 397	Date	
-----------------	------	--

New Hampshire Department of Health and Human Services Medicaid Care Management Services



## **Exhibit B - DRAFT**

to Members in accordance with NH Code of Administrative Rules, Chapter He-W 500, Part He-W 513.

- 4.12.26.5 This includes, but is not limited to:
  - 4.12.26.5.1 Ensuring that Members at-risk of experiencing Substance Use Disorder are assessed using a standardized evidence-based assessment tool consistent with ASAM Criteria; and
  - 4.12.26.5.2 Providing access to the full range of services available under the DHHS's Substance Use Disorder benefit, including Peer Recovery Support without regard to whether a Peer Recovery Support Service (PRSS) is an aspect of an additional service provided to the Member.
- 4.12.26.6 The MCO shall make PRSS available to Members both as a standalone service (regardless of an assessment), and as part of other treatment and Recovery services.
- 4.12.26.7 The provision of services to recipients enrolled in an MCO shall not be subject to more stringent service coverage limits than specified under this Agreement or State Medicaid policies.

## 4.12.27 Substance Use Disorder Clinical Evaluations and Treatment Plans

- 4.12.27.1 The MCO shall ensure, through its regular quality improvement activities and reviews of DHHS administered quality monitoring and improvement activities, that Substance Use Disorder treatment services are delivered in the least restrictive community based environment possible and based on a person-centered approach where the Member and their family's personal goals and needs are considered central in the development of the Individualized service plans.
- 4.12.27.2 A clinical evaluation is a biopsychosocial evaluation completed in accordance with SAMHSA Technical Assistance Publication (TAP) 21: Addiction Counseling Competencies.
- 4.12.27.3 The MCO shall ensure that all services provided include a method to obtain clinical evaluations using DSM five (5) diagnostic information and a recommendation for a level of care based on the ASAM Criteria, published in October, 2013 or as revised by ASAM.
- 4.12.27.4 The MCO shall ensure that a clinical evaluation is completed for each Member prior to admission as a part of interim services or within three (3) business days following admission.
- 4.12.27.5 For a Member being transferred from or otherwise referred by another Provider, the Provider shall use the clinical evaluation

age 246 of 397	Date

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

- completed by a licensed behavioral health professional from the referring agency, which may be amended by the receiving Provider.
- 4.12.27.6 The Provider shall complete individualized treatment plans for all Members based on clinical evaluation data within three (3) business days of the clinical evaluation (or three (3) sessions, if the Member is meeting with an outpatient treatment provider no more than once per week), that addresses problems in all ASAM 2013 domains which justify the Member's admittance to a given level of care and that include individualized treatment plan goals, objectives, and interventions written in terms that are specific, measurable, attainable, realistic, and time relevant (SMART).
- 4.12.27.7 The treatment plan shall include the Member's involvement in identifying, developing, and prioritizing goals, objectives, and interventions.
- 4.12.27.8 Treatment plans shall be updated based on any changes in any ASAM domain and at minimal intervals as described by ASAM (2013) for each level of care.
- 4.12.27.9 Treatment plan updates shall include:
  - 4.12.27.9.1 Documentation of the degree to which the Member is meeting treatment plan goals and objectives;
  - 4.12.27.9.2 Modification of existing goals or addition of new goals based on changes in the Member's functioning relative to ASAM domains and treatment goals and objectives, as appropriate;
  - 4.12.27.9.3 The Provider's assessment of whether or not the Member needs to move to a different level of care based on ASAM continuing care, transfer and discharge criteria; and
  - 4.12.27.9.4 The signature of the Member and the Provider agreeing to the updated treatment plan, or if applicable, documentation of the Member's refusal to sign the treatment plan.

## 4.12.28 Substance Use Disorder Performance Improvement Project

4.12.28.1 In compliance with the requirements outlined in Section 4.14.3 (Quality Assessment and Performance Improvement Program), the MCO shall, at a minimum, conduct at least one (1) PIP designed to improve the delivery of Substance Use Disorder services.

#### 4.12.29 **Reporting**

4.12.29.1 The MCO shall report to DHHS Substance Use Disorder-related metrics in accordance with Exhibit O: Quality and Oversight Reporting Requirements including, but not limited to, measures

Page 247 of 397	Date	

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

- related to access to services, engagement, clinically appropriate services, Member engagement in treatment, treatment retention, safety monitoring, and service utilization.
- 4.12.29.2 The MCO shall provide, in accordance with Exhibit O: Quality and Oversight Reporting Requirements, an assessment of any prescribing rate and pattern outliers and how the MCO plans to follow up with Providers identified as having high-prescribing patterns.
- 4.12.29.3 The MCO shall conduct reviews and audits of clinical records and claims for Members receiving Substance Use Disorder treatment services provided by Substance Use Disorder Programs and Medication Assisted Treatment Services provided by Opioid Treatment Programs (OTP).
- 4.12.29.4 The MCO shall utilize audit tool(s) provided by or approved by DHHS, collected via one or more mediums made available or approved by DHHS, to assess the activities of Substance Use Disorder Providers and Opioid Treatment Programs (OTPs), to ensure compliance with the He-W 513 rules, He-A 304 rules, and the MCO Contract, and this Agreement. The MCO shall provide to DHHS copies of all findings from any audit or assessment of Providers related to Substance Use Disorder conducted by the MCO or on behalf of the MCO.
  - 4.12.29.4.1 The MCO shall provide to DHHS copies of all findings from any audit or assessment of Providers related to Substance Use Disorders conducted by the MCO or on behalf of the MCO.
  - 4.12.29.4.2 The MCO shall report on SUD Provider compliance with service provisions outlined in the SUD audit tool in accordance with Exhibit O: Quality and Oversight Reporting Requirements.
- 4.12.29.5 On a monthly basis, the MCO shall provide directly to Participating Providers comparative prescribing data, including the average Morphine Equivalent Dosing (MED) levels across patients and identification of Members with MED at above average levels, as determined by the MED levels across Members.
- 4.12.29.6 The MCO shall also provide annual training to Participating Providers.

#### 4.12.30 Services for Members Who are Homeless or At-Risk of Homelessness

4.12.30.1 In coordination with Substance Use Disorder programs and/or Providers, the MCO shall provide care to Members who are homeless or at risk of homelessness as described in Section 4.13.19.8 (Services for the Homeless).

Page 248 of 397	Date
1 ago 2 10 01 001	

New Hampshire Department of Health and Human Services Medicaid Care Management Services



## **Exhibit B - DRAFT**

## 4.12.31 Peer Recovery Support Services

4.12.31.1 In coordination with Peer Recovery Programs and Peer Recovery Coaches, as defined in He-W 513, the MCO shall actively promote delivery of PRSS provided by Peer Recovery Coaches who are also certified Recovery support workers in a variety of settings such as Peer Recovery Programs, clinical Substance Use Disorder programs, EDs, and primary care clinics.

## 4.12.32 Naloxone Availability

4.12.32.1 The MCO shall work with each contracted Substance Use Disorder program and/or Provider to ensure that naloxone kits are available on-site and training on naloxone administration and emergency response procedures are provided to program and/or Provider staff at a minimum annually.

## 4.12.33 Prescription Drug Monitoring Program

- 4.12.33.1 The MCO shall include in its Provider agreements the requirement that prescribers and dispensers comply with the NH PDMP requirements, including but not limited to opioid prescribing guidelines.
- 4.12.33.2 The Provider agreements shall require Participating Providers to provide to the MCO, to the maximum extent possible, data on substance dispensing to Members prior to releasing such medications to Members.
- 4.12.33.3 The MCO shall monitor harmful prescribing rates and, at the discretion of the Department, may be required to provide ongoing updates on those Participating Providers who have been identified as overprescribing.

#### 4.12.34 Response After Overdose

- 4.12.34.1 Whenever a Member receives emergency room or inpatient hospital services as a result of a non-fatal overdose, the MCO shall work with hospitals to ensure a seamless transition of care upon admission and discharge to the community, and detail information sharing and collaboration between the MCO and the participating hospital.
- 4.12.34.2 Whenever a Member discharges themselves against medical advice, the MCO shall make a good faith effort to ensure that the Member receives a clinical evaluation, referral to appropriate treatment, Recovery support services and intense Case Management within forty-eight (48) hours of discharge or the MCO being notified, whichever is sooner.
- 4.12.34.3 Limitations on Prior Authorization Requirements

Page 249 of 397	Date

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

- 4.12.34.3.1 To the extent permitted under State and federal law, the MCO shall cover MAT.
- 4.12.34.3.2 Methadone received at a methadone clinic shall not require Prior Authorization.
- 4.12.34.3.3 Methadone used to treat pain shall require Prior Authorization.
  - 4.12.34.3.3.1. Any Prior Authorization for office based MAT shall comply with RSA 420-J:17 and RSA 420-J:18.
- 4.12.34.3.4 The MCO shall not impose any Prior Authorization requirements for MAT urine drug screenings (UDS) unless a Provider exceeds thirty (30) UDSs per month per treated Member.
- 4.12.34.3.5 In the event a Provider exceeds thirty (30) UDS per month per treated Member, the MCO shall impose Prior Authorization requirements on usage.
- 4.12.34.3.6 The MCO is precluded from imposing any Prior Authorization on screening for multiple drugs within a daily drug screen.
- 4.12.34.3.7 The MCO may require prior authorization for SUD treatments, excluding MAT services.
- 4.12.34.3.8 The MCO shall utilize ASAM Criteria when determining medical necessity for continuation of covered services.
- 4.12.34.3.9 Nothing in this section shall be construed to require coverage for services provided by a non-participating provider.
- 4.12.34.3.10 The MCO may require prior authorization for covered services only if the MCO has a medical clinician or licensed alcohol and drug counselor available on a 24-hour hotline to make the medical necessity determination and assist with placement at the appropriate level of care, and the MCO provides a prior authorization decision as soon as practicable after receipt from the treating clinician of the clinical rationale consistent with the ASAM criteria, but in no event more than 6 hours of receiving such information; provided that until such hotline determination is made, coverage for substance use disorder services shall be provided at an appropriate level of care consistent with the ASAM criteria, as defined in RSA 420-J:15, I.

Page 250 of 397 Date \_\_\_\_\_

New Hampshire Department of Health and Human Services Medicaid Care Management Services



#### **Exhibit B - DRAFT**

- 4.12.34.3.11 The Department may grant exceptions to this provision in instances where it is necessary to prevent Fraud, Waste or Abuse.
- 4.12.34.3.12 For Members who enter the Pharmacy Lock-In Program as described in Section 4.2.4 (Pharmacy Clinical Policies and Prior Authorizations), the MCO shall evaluate the need for Substance Use Disorder treatment.
- 4.12.34.4 Opioid Prescribing Requirements
  - 4.12.34.4.1 The MCO shall require Prior Authorization documenting the rationale for the prescriptions of more than one hundred (100) mg daily MED of opioids for Members.
  - 4.12.34.4.2 As required under the NH Board Administrative Rule MED 502 Opioid Prescribing, the MCO shall adhere to MED procedures for acute and chronic pain, taking actions, including but not limited to:
    - 4.12.34.4.2.1. A pain management consultation or certification from the Provider that it is due to an acute medical condition:
    - 4.12.34.4.2.2. Random and periodic UDS; and
    - 4.12.34.4.2.3. Utilizing written, informed consent.
  - 4.12.34.4.3 The MCO shall ensure that Participating Providers prescribe and dispense Naloxone for patients receiving a one hundred (100) mg MED or more per day for longer than ninety (90) calendar days.
    - 4.12.34.4.3.1. If the NH Board Administrative Rule MED 502 Opioid Prescribing is updated in the future, the MCO shall implement the revised policies in accordance with the timelines established or within sixty (60) calendar days if no such timeline is provided.

#### 4.12.34.5 Neonatal Abstinence Syndrome

4.12.34.5.1 For those Members with a diagnosis of Substance Use Disorder and all infants with a diagnosis of NAS, or that are otherwise known to have been exposed prenatally to opioids, alcohol or other drugs, the MCO shall provide Care Management services to provide for coordination of their physical and behavioral

Page 251 of 397 Date \_\_\_\_\_

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

health, according to the safeguards relating to redisclosure set out in 42 CFR Part 2.

- 4.12.34.5.2 Substance Use Disorder Care Management features shall include, but not be limited to:
  - 4.12.34.5.2.1. Conducting outreach to Members who would benefit from treatment (for example, by coordinating with emergency room staff to identify and engage with Members admitted to the ED following an overdose).
  - 4.12.34.5.2.2. Ensuring that Members are receiving the appropriate level of Substance Use Disorder treatment services.
  - 4.12.34.5.2.3. Scheduling Substance Use Disorder treatment appointments and following up to ensure appointments are attended.
  - 4.12.34.5.2.4. Coordinating care among prescribing Providers, clinician case managers, pharmacists, behavioral health Providers and social service agencies.
  - 4.12.34.5.2.5. The MCO shall make every attempt to coordinate and enhance Care Management services being provided to the Member by the treating Provider.
- 4.12.34.5.3 The MCO shall work with DCYF to provide Substance Use Disorder treatment referrals and conduct a follow-up after thirty (30) calendar days to determine the outcome of the referral and determine if additional outreach and resources are needed.
- 4.12.34.5.4 The MCO shall work with DCYF to ensure that health care Providers involved in the care of infants identified as being affected by prenatal drug or alcohol exposure, create and implement the Plan of Safe Care.
  - 4.12.34.5.4.1. The Plan of Safe Care shall be developed in collaboration with health care Providers and the family/caregivers of the infant to address the health of the infant and Substance Use Disorder treatment needs of the family or caregiver.

Page 252 of 397	Date

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



- 4.12.34.5.5 The MCO shall establish protocols for Participating Providers to implement a standardized screening and treatment protocol for infants at risk of NAS.
- 4.12.34.5.6 The MCO shall provide training to Providers serving infants with NAS on best practices, including:
  - 4.12.34.5.6.1. Opportunities for the primary care giver(s) to room-in;
  - 4.12.34.5.6.2. Transportation and childcare for the primary care giver(s);
  - 4.12.34.5.6.3. Priority given to non-pharmaceutical approaches (e.g., quiet environment, swaddling);
  - 4.12.34.5.6.4. Education for primary care giver(s) on caring for newborns;
  - 4.12.34.5.6.5. Coordination with social service agencies proving supports, including coordinated case meetings and appropriate developmental services for the infant;
  - 4.12.34.5.6.6. Information on family planning options; and
  - 4.12.34.5.6.7. Coordination with the family and Providers on the development of the Plan of Safe Care for any infant born with NAS.
- 4.12.34.5.7 The MCO shall work with the Department and Providers eligible to expand/develop services to increase capacity for specialized services for this population which address the family as a unit and are consistent with Northern New England Perinatal Quality Improvement Network's (NNEPQIN) standards.
- 4.12.34.6 Discharge Planning After Substance Use Disorder Event
  - 4.12.34.6.1 In all cases where the MCO is notified or otherwise learns that a Member has had an ED visit or is hospitalized for an overdose or Substance Use Disorder, the MCO's Care Coordination staff shall actively participate and assist hospital staff in the development of a written discharge plan.
  - 4.12.34.6.2 The MCO shall ensure that the final discharge instruction sheet shall be provided to the Member and

Page 253 of 397	Date

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

the Member's authorized representative prior to discharge, or the next business day, for at least ninety-eight (98%) of Members discharged.

- 4.12.34.6.3 The MCO shall ensure that the discharge progress note shall be provided to any treatment Provider within seven (7) calendar days of Member discharge for at least ninety-eight percent (98%) of Members discharged.
  - 4.12.34.6.3.1. If a Member lacks a reasonable means of communicating with a plan prior to discharge, the MCO shall identify an alternative viable means for communicating with the Member in the discharge plan.
- 4.12.34.6.4 The MCO shall ensure that any referrals necessary to connect the Member to post-discharge treatment Provider(s) are made as closed-loop referrals prior to the Member's discharge, including those that may be necessary for an ASAM evaluation.
- 4.12.34.6.5 The MCO shall track all Members discharged into the community who do not receive MCO contact (including outreach or a referral to a Substance Use Disorder program and/or Provider).
- 4.12.34.6.6 The MCO shall make at least three (3) attempts to contact Members within three (3) business days of discharge from the ED to review the discharge plan, support the Member in attending any scheduled follow-up appointments, support the continued taking of any medications prescribed, and answer any questions the Member may have.
- 4.12.34.6.7 At least ninety-five percent (95%) of Members discharged shall have been attempted to be contacted within three (3) business days.
- 4.12.34.6.8 For any Member the MCO does not make contact with within three (3) business days, the MCO shall contact the treatment Provider and request that the treatment Provider make contact with the Member within twenty-four (24) hours.
- 4.12.34.6.9 The MCO shall ensure an appointment for treatment other than evaluation with a Substance Use Disorder program and/or Provider for the Member is scheduled prior to discharge when possible and that transportation has been arranged for the

age 254 of 397	Date

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

- appointment. Such appointments shall occur within seven (7) calendar days after discharge.
- 4.12.34.6.10 In accordance with 42 CFR Part 2, the MCO shall work with DHHS during regularly scheduled meetings to review cases of Members that have been seen for more than three (3) overdose events within a thirty (30) calendar day period or those that have had a difficulty engaging in treatment services following referral and Care Coordination provided by the MCO.
- 4.12.34.6.11 The MCO shall also review Member cases with the applicable Substance Use Disorder program and/or Provider to promote strategies for reducing overdoses and increase engagement in treatment services.

#### 4.13 Quality Management

## 4.13.1 General Provisions

- 4.13.1.1 The MCO shall provide for the delivery of quality care with the primary goal of improving the health status of its Members and, where the Member's condition is not amenable to improvement, maintain the Member's current health status by implementing measures to prevent any further decline in condition or deterioration of health status.
- 4.13.1.2 The MCO shall work in collaboration with the Department, Members and Providers to actively improve the quality of care provided to Members, consistent with the MCO's quality improvement goals and all other requirements of the Agreement.
- 4.13.1.3 The MCO shall provide mechanisms for Member Advisory Board and the Provider Advisory Board to actively participate in the MCO's quality improvement activities.
- 4.13.1.4 The MCO shall support and comply with the most current version of the Quality Strategy for the MCM program.
- 4.13.1.5 The MCO shall approach all clinical and non-clinical aspects of QAPI based on principles of CQI/Total Quality Management and shall:
  - 4.13.1.5.1 Evaluate performance using objective quality indicators and recognize that opportunities for improvement are unlimited;
  - 4.13.1.5.2 Foster data-driven decision-making;
  - 4.13.1.5.3 Solicit Member and Provider input on the prioritization and strategies for QAPI activities;

age 255 of 397	Date

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

4.13.1.5.4	Support continuous ongoing measurement of clinical		
	and non-clinical health plan effectiveness, health		
	outcomes improvement and Member and Provider		
	satisfaction;		

- 4.13.1.5.5 Support programmatic improvements of clinical and non-clinical processes based on findings from ongoing measurements; and
- 4.13.1.5.6 Support re-measurement of effectiveness, health outcomes improvement and Member satisfaction, and continued development and implementation of improvement interventions as appropriate.

#### 4.13.2 Health Plan Accreditation

- 4.13.2.1 The MCO shall achieve health plan accreditation from the NCQA, including the NCQA Medicaid Module.
- 4.13.2.2 If the MCO participated in the MCM program prior to the Program Start Date, the MCO shall maintain its health plan accreditation status throughout the period of the Agreement, and complete the NCQA Medicaid Module within eighteen (18) months of the Program Start Date.
- 4.13.2.3 If the MCO is newly participating in the MCM program, the MCO shall achieve health plan accreditation from NCQA, including the Medicaid Module, within eighteen (18) months of the Program Start Date.
- 4.13.2.4 To demonstrate its progress toward meeting this requirement, the newly participating MCO shall complete the following milestones:
  - 4.13.2.4.1 Within sixty (60) calendar days of the Program Start Date, the MCO shall notify the Department of the initiation of the process to obtain NCQA Health Plan Accreditation; and
  - 4.13.2.4.2 Within thirty (30) calendar days of the date of the NCQA survey on-site review, the MCO shall notify the Department of the date of the scheduled on-site review.
- 4.13.2.5 The MCO shall inform the Department of whether it has been accredited by any private independent accrediting entity, in addition to NCQA Health Plan Accreditation.
- 4.13.2.6 The MCO shall authorize NCQA, and any other entity from which it has received or is attempting to receive accreditation, to provide a copy of its most recent accreditation review to the Department, including [42 CFR 438.332(a)]:

Page 256 of 397	Date	
-----------------	------	--

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

4.13.2.6.1	editation cable);	status,	survey	type,	and	level	(a	S
4 40 0 0 0								

- 4.13.2.6.2 Accreditation results, including recommended actions or improvements, CAPs, and summaries of findings; and
- 4.13.2.6.3 Expiration date of the accreditation. [42 CFR 438.332(b)(1-3)]
- 4.13.2.7 To avoid duplication of mandatory activities with accreditation reviews, DHHS may indicate in its quality strategy the accreditation review standards that are comparable to the standards established through federal EQR protocols and that the Department shall consider met on the basis of the MCO's achievement of NCQA accreditation. [42 CFR 438.360]
- 4.13.2.8 An MCO going through an NCQA renewal survey shall complete the full Accreditation review of all NCQA Accreditation Standards.
- 4.13.2.9 During the renewal survey, the MCO shall:
  - 4.13.2.9.1 Request from NCQA the full review of all NCQA Accreditation Standards and cannot participate in the NCQA renewal survey option that allows attestation for certain requirements; and
  - 4.13.2.9.2 Submit to the Department a written confirmation from NCQA stating that the renewal survey for the MCO will be for all NCQA Accreditation Standards without attestation.

#### 4.13.3 Quality Assessment and Performance Improvement Program

- 4.13.3.1 The MCO shall have an ongoing comprehensive QAPI program for the services it furnishes to Members consistent with the requirements of this Agreement and federal requirements for the QAPI program [42 CFR 438.330(a)(1); 42 CFR 438.330(a)(3)].
- 4.13.3.2 The MCO's QAPI program shall be documented in writing (in the form of the "QAPI Plan"), approved by the MCO's governing body, and submitted to the Department for its review annually.
- 4.13.3.3 In accordance with Exhibit O: Quality and Oversight Reporting Requirements, the QAPI Plan shall contain at a minimum, the following elements:
  - 4.13.3.3.1 A description of the MCO's organization-wide QAPI program structure;
  - 4.13.3.3.2 The MCO's annual goals and objectives for all quality activities, including but not limited to:
    - 4.13.3.3.2.1. Department-required PIPs;

Page 257 of 397	Date

Medicaid Care Management Services Contract New Hampshire Department of Health and Human Services Medicaid Care Management Services



# **Exhibit B - DRAFT**

	4.	13.3.3.2.2.	Department-required quality performance data;	
	4.	13.3.3.2.3.	Department-required quality reports; and	
	4.	13.3.3.2.4.	Implementation of EQRO recommendations from annual technical reports;	
	4.	13.3.3.2.5.	Mechanisms to detect both underutilization and overutilization of services; [42 CFR 438.330(b)(3)]	
	4.	13.3.3.2.6.	Mechanisms to assess the quality and appropriateness of care for Members with Special Health Care Needs (as defined by the Department in the quality strategy) [42 CFR 438.330(b)(4)] in order to identify any Ongoing Special Conditions of a Member that require a course of treatment or regular care monitoring; and	
	4.	13.3.3.2.7.	Mechanisms to assess and address disparities in the quality of, and access to, health care, based on age, race, ethnicity, sex, primary language, and disability status (defined as whether the individual qualified for Medicaid on the basis of a disability). [42 CFR 438.340(b)(6)]	
4.13.3.4	evaluation ar	nd improvem	and ongoing process for monitoring, ent of the quality and appropriateness of es provided to Members.	
4.13.3.5	includes a pl non-clinical p	nall maintain a well-defined QAPI program structure that blanned systematic approach to improving clinical and processes and outcomes. At a minimum, the MCO shall the QAPI program structure:		
	4.13.3.5.1	Is organizat within the or	ion-wide, with clear lines of accountability ganization;	
	4.13.3.5.2	for the over defined and	set of functions, roles, and responsibilities rsight of QAPI activities that are clearly d assigned to appropriate individuals, sysicians, clinicians, and non-clinicians;	
	4.13.3.5.3		nual objectives and/or goals for planned activities including clinical and non-clinical	

Page 258 of 397 Date \_\_\_\_\_

New Hampshire Department of Health and Human Services **Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

programs	or	initiatives	and	measurement	activities;
and					

- Evaluates the effectiveness of clinical and non-4.13.3.5.4 clinical initiatives.
- 4.13.3.6 If the MCO subcontracts any of the essential functions or reporting requirements contained within the QAPI program to another entity, the MCO shall maintain detailed files documenting work performed by the Subcontractor. The file shall be available for review by the Department or its designee upon request, and a summary of any functions that have been delegated to Subcontractor(s) shall be indicated within the MCO's QAPI Plan submitted to the Department annually.
- 4.13.3.7 Additional detail regarding the elements of the QAPI program and the format in which it should be submitted to the Department is provided in Exhibit O: Quality and Oversight Reporting Requirements.
- 4.13.3.8 Performance Improvement Projects
  - 4.13.3.8.1 The MCO shall conduct any and all PIPs required by CMS. [42 CFR 438.330(a)(2)]
  - Throughout the contract period, the MCO shall 4.13.3.8.2 conduct at least three (3) clinical PIPs that meet the following criteria [42 CFR 438.330 (d)(1)]:
    - At least one (1) clinical PIP shall have a 4.13.3.8.2.1. focus on the Department's objectives outlined in the NH MCM Quality Strategy;
    - 4.13.3.8.2.2. At least one (1) clinical PIP shall have a focus on Substance Use Disorder, as defined in Section 4.13.23 (Substance Use Disorder):
    - 4.13.3.8.2.3. At least (1) clinical PIP shall focus on improving quality performance in an area that the MCO performed lower than the fiftieth (50th) percentile nationally, as documented in the most recent EQRO technical report or as otherwise indicated by the Department;
    - If the MCO's individual experience is not 4.13.3.8.2.4. reflected in the most recent EQRO technical report, the MCO shall incorporate a PIP in an area that the MCOs participating in the MCM program

Page 259 of 397 Date \_\_\_\_\_

New Hampshire Department of Health and Human Services Medicaid Care Management Services



#### **Exhibit B - DRAFT**

at the time of the most recent EQRO technical report performed below the seventy-fifth (75th) percentile; and

- 4.13.3.8.2.5. Should no quality measure have a lower than seventy-fifth (75th) percentile performance, the MCO shall focus the PIP on one (1) of the areas for which its performance (or, in the event the MCO is not represented in the most recent report, the other MCOs' collective performance) was lowest.
- 4.13.3.8.3 Throughout the five-year contract term, the MCO shall conduct at least one (1) non-clinical PIP, which shall be related to one (1) of the following topic areas and approved by the Department:
  - 4.13.3.8.3.1. Addressing health-related social needs; and
  - 4.13.3.8.3.2. Integrating physical and behavioral health.
- 4.13.3.8.4 The non-clinical PIP may include clinical components, but shall have a primary focus on non-clinical outcomes.
- 4.13.3.8.5 The MCO shall ensure that each PIP is designed to achieve significant improvement, sustained over time, in health outcomes and Member satisfaction [42 CFR 438.330(d)(2)], and shall include the following elements:
  - 4.13.3.8.5.1. Measurement(s) of performance using objective quality indicators [42 CFR 438.330(d)(2)(i)];
  - 4.13.3.8.5.2. Implementation of interventions to achieve improvement in the access to and quality of care [42 CFR 438.330(d)(2)(ii)];
  - 4.13.3.8.5.3. Evaluation of the effectiveness of the interventions based on the performance measures used as objective quality indicators [42 CFR 438.330(d)(2)(iii)]; and
  - 4.13.3.8.5.4. Planning and initiation of activities for increasing or sustaining improvement [42 CFR 438.330(d)(2)(iv)].

	 38.330(d)(2)(iv)	•
Page 260 of 397		Date

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

4.13.3.8.6	Each PIP shall be approved by the Department and
	shall be completed in a reasonable time period so as
	to generally permit information on the success of
	PIPs in the aggregate to produce new information on
	quality of care every year.

- 4.13.3.8.7 In accordance with Exhibit O: Quality and Oversight Reporting Requirements, the MCO shall include in its QAPI Plan, to be submitted to the Department annually, the status and results of each PIP conducted in the preceding twelve (12) months and any changes it plans to make to PIPs or other MCO processes in the coming years based on these results or other findings [42 CFR 438.330(d)(1) and (3)].
- 4.13.3.8.8 At the sole discretion of the Department, the PIPs may be delayed in the event of a public health emergency.

## 4.13.4 Member Experience of Care Survey

- 4.13.4.1 The MCO shall be responsible for administering the Consumer Assessment of Healthcare Providers and Systems (CAHPS) survey on an annual basis, and as required by NCQA for Medicaid health plan accreditation for both adults and children, including:
  - 4.13.4.1.1 CAHPS Health Plan Survey 5.1H, Adult Version or later version as specified by the Department;
  - 4.13.4.1.2 CAHPS Health Plan Survey 5.1H, Child Version with Children with Chronic Conditions Supplement or later version as specified by the Department.
- 4.13.4.2 Each CAHPS survey administered by the MCO shall include up to twelve (12) other supplemental questions for each survey as defined by the Department and indicated in Exhibit O: Quality and Oversight Reporting Requirements. Supplemental questions, including the number, are subject to NCQA approval each October preceding the survey fielding timeframe.
- 4.13.4.3 The MCO shall obtain the Department approval of instruments prior to fielding the CAHPS surveys.

#### 4.13.5 Quality and Administrative Reporting Deliverables

4.13.5.1 Required quality and administrative reporting deliverables appear in this Agreement and/or in Exhibit O: Quality and Oversight Reporting Requirements. For ease of reference, the Department shall list quality deliverables in Exhibit O: Quality and Oversight Reporting Requirements where possible. When a reporting requirement is included in the Agreement, but not Exhibit O, or vice

Page 261 of 397	Date

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



versa, the	MCO shall	still fulfill t	the requiren	nent. These	e deliverables
include:			•		

- 4.13.5.1.1 Quality measures;
- 4.13.5.1.2 Narrative Reports;
- 4.13.5.1.3 Plans; and
- 4.13.5.1.4 Templates.
- 4.13.5.2 The MCO shall report the following quality measure sets annually according to the current industry/regulatory standard definitions, in accordance with the submission frequency established in Exhibit O: Quality and Oversight Reporting Requirements [42 CFR 438.330(b)(2); 42 CFR 438.330(c)(1) and (2); 42 CFR 438.330(a)(2)]:
  - 4.13.5.2.1 Any CMS-mandated measures [42 CFR 438.330(c)(1)(i)] to include;
    - 4.13.5.2.1.1. CMS Child Core Set of Health Care Quality deliverables for Medicaid and CHIP, as specified by the Department;
    - 4.13.5.2.1.2. Deliverables included in any future CMS Universal Foundation Measure list;
    - 4.13.5.2.1.3. CMS Adult Core Set of Health Care Quality Measures deliverables for Medicaid, as specified by the Department;
    - 4.13.5.2.1.4. Deliverables indicated by the Department as a requirement for fulfilling CMS waiver requirements; and
    - 4.13.5.2.1.5. Deliverables indicated by the Department as a requirement for the CMS Managed Care Program Annual Report [42 CFR 438.66(e)].
  - 4.13.5.2.2 NCQA Medicaid Accreditation measures, including race and ethnicity stratification, which shall be generated without NCQA Allowable Adjustments and validated by submission to NCQA.
    - 4.13.5.2.2.1. The MCO shall include supplemental Confidential Data in HEDIS measures identified in Exhibit O: Quality and Oversight Reporting Requirements for NCQA Accreditation and reporting

Page 262 of 397	Date	

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

through the Interactive Confidential Data Submission System.

- 4.13.5.2.2.2. The MCO shall report Member level Confidential Data for audited HEDIS measures as identified in Exhibit O: Quality and Oversight Reporting Requirements.
- 4.13.5.2.3 All available CAHPS measures and sections and additional supplemental questions defined by the Department;
- 4.13.5.2.4 Select measures to monitor MCO Member and Provider operational quality and Care Coordination efforts; and
- 4.13.5.2.5 Select measures specified by the Department as priority measures for use in assessing and addressing local challenges to high-quality care and access;
- 4.13.5.3 Where the Department, NCQA, CMS or other key stakeholders require the use of electronic clinical data in deliverable calculation, the MCO shall obtain this data as stipulated in measure specifications and by the measure stewards.
- 4.13.5.4 If additional measures are added to the NCQA or CMS measure sets, the MCO shall include any such new measures in its reports to the Department.
- 4.13.5.5 For measures that are no longer part of the measure sets, the Department may, at its option, continue to require those measures; any changes to MCO quality measure reporting requirements shall be communicated to MCOs and documented within a format similar to Exhibit O: Quality and Oversight Reporting Requirements.
- 4.13.5.6 The MCO shall report all quality deliverables in accordance with Exhibit O: Quality and Oversight Reporting Requirements, regardless of whether the MCO has achieved accreditation from NCQA.
- 4.13.5.7 The MCO shall submit all quality deliverables in the formats and schedule in Exhibit O: Quality and Oversight Reporting Requirements or otherwise identified by the Department.
- 4.13.5.8 The MCO shall work with the Department to ensure their understanding of Department deliverable specifications, deliverable submission processes, and deliverable review processes. This includes, as determined by the Department:

Page 263 of 397	Date	

Medicaid Care Management Services Contract
New Hampshire Department of Health and Human Services
Medicaid Care Management Services



# **Exhibit B - DRAFT**

	EXHIBIT D - DITAL I
4.13.5.8	.1 The MCO shall gain access to and utilize the NH Medicaid Quality Information System, to include participation in any Department-required training deemed necessary;
4.13.5.8	.2 The MCO shall gain access to and utilize the Department SharePoint site utilized for deliverables other than measures, to include any deliverables which contain confidential data;
4.13.5.8	.3 The MCO shall attend all meetings with relevant MCO subject matter experts to discuss specifications for deliverables indicated in Exhibit O: Quality and Oversight Reporting Requirements; and
4.13.5.8	.4 The MCO shall communicate and distribute all specifications and templates provided by the Department for deliverables in Exhibit O: Quality and Oversight Reporting Requirements, to all MCO subject matter experts involved in the production of deliverables in Exhibit O: Quality and Oversight Reporting Requirements.
day not	partment shall provide the MCO, with a ninety (90) calendar ce, any additions or modifications to the deliverables and eliverable specifications.
Confide expando Care Co and ong quality	time as the Department provides access to Medicare ntial Data sets to the MCO, the MCO shall integrate and Medicare Confidential Data sets into its QAPI Plan and ordination and Quality Programs, and include a systematic oing process for monitoring, evaluating, and improving the and appropriateness of services provided to Medicaide dual Members. The MCO shall:
4.13.5.1	O.1 Collect Confidential Data, and monitor and evaluate for improvements to physical health outcomes, behavioral health outcomes and psycho-social outcomes resulting from Care Coordination of the dual Members;
4.13.5.1	0.2 Include Medicare Confidential Data in the Department quality reporting; and
4.13.5.1	O.3 Sign Confidential Data use Agreements and submit Confidential Data management plans, as required by the Department and CMS.
4.13.5.1	0.4 For failure to submit required reports and quality Confidential Data to the Department, NCQA, the

Page 264 of 397 Date \_\_\_\_\_

EQRO, and/or other Department-identified entities,

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

the MCO shall be subject to liquidated damages as described in Section 5.5.2 (Liquidated Damages).

#### 4.13.6 **Evaluation**

- 4.13.6.1 The Department shall, at a minimum, collect the following information, and the information specified throughout the Agreement and within Exhibit O: Quality and Oversight Reporting Requirements, in order to improve the performance of the MCM program [42 CFR 438.66(c)(6)-(8)]:
  - 4.13.6.1.1 Performance on required quality measures; and
  - 4.13.6.1.2 The MCO's QAPI Plan.
- 4.13.6.2 Starting in the second year of the Term of this Agreement, the MCO shall include in its QAPI Plan a detailed report of the MCO's performance against its QAPI Plan throughout the duration of the preceding twelve (12) months, and how its development of the proposed, updated QAPI plan has taken those results into account. The report shall include detailed information related to:
  - 4.13.6.2.1 Completed and ongoing quality management activities, including all delegated functions;
  - 4.13.6.2.2 Performance trends on QAPI measures to assess performance in quality of care and quality of service (QOS) for all activities identified in the QAPI Plan;
  - 4.13.6.2.3 An analysis of whether there have been any demonstrated improvements in the quality of care or service for all activities identified in the QAPI Plan;
  - 4.13.6.2.4 An analysis of actions taken by the MCO based on MCO specific recommendations identified by the EQRO's Technical Report and other Quality Studies; and
  - 4.13.6.2.5 An evaluation of the overall effectiveness of the MCO's quality management program, including an analysis of barriers and recommendations for improvement.
- 4.13.6.3 The annual evaluation report, developed in accordance with Exhibit O: Quality and Oversight Reporting Requirements, shall be reviewed and approved by the MCO's governing body and submitted to the Department for review [42 CFR 438.330(e)(2)].
- 4.13.6.4 The MCO shall establish a mechanism for periodic reporting of QAPI activities to its governing body, practitioners, Members, and appropriate MCO staff, as well as for posting on the web.

Page 265 of 397	Date

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

4.13.6.5 In accordance with Exhibit O: Quality and Oversight Reporting Requirements, the MCO shall ensure that the findings, conclusions, recommendations, actions taken, and results of Quality Management activity are documented and reported on a semi-annual basis to the Department and reviewed by the appropriate individuals within the organization.

## 4.13.7 Accountability for Quality Improvement

- 4.13.7.1 External Quality Review
  - 4.13.7.1.1 The MCO shall collaborate and cooperate fully with the Department's EQRO in the conducting of CMS EQR activities to identify opportunities for MCO improvement [42 CFR 438.358].
  - 4.13.7.1.2 Annually, the MCO shall undergo external independent reviews of the quality, timeliness, and access to services for Members [42 CFR 438.350].
  - 4.13.7.1.3 To facilitate this process, the MCO shall supply information, including but not limited to:
    - 4.13.7.1.3.1. Claims data,
    - 4.13.7.1.3.2. Medical records,
    - 4.13.7.1.3.3. Operational process details, and
    - 4.13.7.1.3.4. Source code used to calculate performance measures to the EQRO as specified by the Department.

#### 4.13.7.2 Auto-Assignment Algorithm

- 4.13.7.2.1 As indicated in Section 4.3.4 (Auto-Assignment), the auto-assignment algorithm shall, over time, reward high-performing MCOs that offer high-quality, accessible care to its Members.
- 4.13.7.2.2 The measures used to determine auto-assignment shall not be limited to alignment with the priority measures assigned to the program MCM Withhold and Incentive Program, as determined by the Department.

## 4.13.7.3 Quality Performance Withhold

4.13.7.3.1 As described in Section 5.4 (MCM Withhold and Incentive Program), the MCM program incorporates a withhold and incentive arrangement; the MCO's performance in the program may be assessed on the basis of the MCO's quality performance, as

age 266 of 397 Date
---------------------

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

determined by the Department and indicated to the MCO in periodic guidance.

## 4.14 **Network Management**

## 4.14.1 **Network Requirements**

- 4.14.1.1 The MCO shall maintain and monitor a network of appropriate Participating Providers that is:
  - 4.14.1.1.1 Supported by written agreements; and
  - 4.14.1.1.2 Sufficient to provide adequate access to all services covered under this Agreement for all Members, including those with LEP or disabilities. [42 CFR 438.206(b)(1)]
- 4.14.1.2 In developing its Participating Provider network, the MCO's Provider selection policies and procedures shall not discriminate against Providers that serve high-risk populations or specialize in conditions that require costly treatment [42 CFR 438.214(c)].
- 4.14.1.3 The MCO shall not employ or contract with Providers excluded from participation in federal health care programs [42 CFR 438.214(d)(1); 42 CFR 455.101; Section 1932(d)(5) of the Act].
- 4.14.1.4 The MCO shall not employ or contract with Providers who fail to provide Equal Access to services.
- 4.14.1.5 The MCO shall ensure its Participating Providers and Subcontractors meet all state and federal eligibility criteria, reporting requirements, and any other applicable statutory rules and/or regulations related to this Agreement. [42 CFR 438.230]
- 4.14.1.6 All Participating Providers shall be licensed and or certified in accordance with the laws of NH and not be under sanction or exclusion from any Medicare or Medicaid program. Participating Providers shall have a NH Medicaid identification number and unique National Provider Identifier (NPI) for every Provider type in accordance with 45 CFR 162, Subpart D.
- 4.14.1.7 The MCO shall provide reasonable and adequate hours of operation, including twenty-four (24) hour availability of information, referral, and treatment for Emergency Medical Conditions. [42 CFR 438.3(q)(1)]
- 4.14.1.8 The MCO shall make arrangements with or referrals to, a sufficient number of physicians and other practitioners to ensure that the services under this Agreement can be furnished promptly and without compromising the quality of care. [42 CFR 438.3(q)(3)]

Page 267 of 397	Date
-----------------	------

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

- 4.14.1.9 The MCO shall permit Non-Participating IHCPs to refer an American Indian/Alaskan Native Member to a Participating Provider. [42 CFR 438.14(b)(6)]
- 4.14.1.10 The MCO shall implement and maintain arrangements or procedures that include provisions to verify, by sampling or other methods, whether services that have been represented to have been delivered by Participating Providers were received by Members and the application of such verification processes on a regular basis. [42 CFR 438.608(a)(5)]
- 4.14.1.11 When contracting with DME Providers, the MCO shall contract with and have in its network all qualified Willing Providers in the State.

#### 4.14.2 **Provider Enrollment**

- 4.14.2.1 The MCO shall ensure that its Participating Providers are enrolled with NH Medicaid.
- 4.14.2.2 The MCO shall prepare and submit a Participating Provider report during the Readiness Review period in a format prescribed by the Department for determination of the MCO's network adequacy.
  - 4.14.2.2.1 The report shall identify fully credentialed and contracted Providers, and prospective Participating Providers.
  - 4.14.2.2.2 Prospective Participating Providers shall have executed letters of intent to contract with the MCO.
  - 4.14.2.2.3 The MCO shall confirm its provider network with the Department and post to its website no later than thirty (30) calendar days prior to the Member enrollment period.
- 4.14.2.3 The MCO shall not discriminate relative to the participation, reimbursement, or indemnification of any Provider who is acting within the scope of their license or certification under applicable State law, solely on the basis of that license or certification.
- 4.14.2.4 If the MCO declines to include individual Provider or Provider groups in its network, the MCO shall give the affected Providers written notice of the reason for its decision. [42 CFR 438.12(a)(1); 42 CFR 438.214(c)]
- 4.14.2.5 The requirements in 42 CFR 438.12(a) shall not be construed to:
  - 4.14.2.5.1 Require the MCO to contract with Providers beyond the number necessary to meet the needs of its Members;

Page 268 of 397	Date	

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

4.14.2.5.2	Preclude	the	MCO	from	using	different
	reimburser	nent ai	mounts fo	or differe	nt specia	Ities or for
	different pr	actition	ners in th	e same	specialty	; or

- 4.14.2.5.3 Preclude the MCO from establishing measures that are designed to maintain QOS and control costs and is consistent with its responsibilities to Members. [42 CFR 438.12(a)(1); 42 CFR 438.12(b)(1-3)]
- 4.14.2.6 The MCO shall ensure that Participating Providers are enrolled with the Department as Medicaid Providers consistent with Provider disclosure, screening and enrollment requirements. [42 CFR 438.608(b); 42 CFR 455.100-107; 42 CFR 455.400-470]

## 4.14.3 Provider Screening, Credentialing and Re-Credentialing

- 4.14.3.1 The Department shall screen and enroll, and periodically revalidate all MCO Participating Providers as Medicaid Providers. [42 CFR 438.602(b)(1)].
- 4.14.3.2 The MCO shall rely on the Department's NH Medicaid providers' affirmative screening in accordance with federal requirements and the current NCQA Standards and Guidelines for the credentialing and re-credentialing of licensed independent Providers and Provider groups with whom it contracts or employs and who fall within its scope of authority and action. [42 CFR 455.410; 42 CFR 438.206)(b)(6)]
- 4.14.3.3 The MCO shall utilize a universal provider Confidential Data source, at no charge to the provider, to reduce administrative requirements and streamline Confidential Data collection during the credentialing and re-credentialing process.
- 4.14.3.4 The MCO shall demonstrate that its Participating Providers are credentialed, and shall comply with any additional Provider selection requirements established by the Department. [42 CFR 438.12(a)(2); 42 CFR 438.214(b)(1); 42 CFR 438.214(c); 42 CFR 438.214(e); 42 CFR 438.206(b)(6)]
- 4.14.3.5 The MCO's Provider selection policies and procedures shall include a documented process for credentialing and re-credentialing Providers who have signed contracts with the MCO. [42 CFR 438.214(b)]
- 4.14.3.6 The MCO shall submit for the Department review during the Readiness Review period, policies and procedures for onboarding Participating Providers, which shall include its subcontracted entity's policies and procedures.
- 4.14.3.7 For Providers not currently enrolled with NH Medicaid, the MCO shall:

Page 269 of 397	Date
ago 200 01 001	Bate

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



4.14.3.7.1	Make	reaso	nable	eff	orts	to	strean	nline	the
	creden	tialing	proces	S	in	collabo	oration	with	the
	Depart	ment;							

- 4.14.3.7.2 Conduct outreach to prospective Participating Providers within ten (10) business days after the MCO receives notice of the Providers' desire to enroll with the MCO:
- 4.14.3.7.3 Concurrently work through MCO and the Department contracting and credentialing processes with Providers in an effort to expedite the Providers' network status; and
- 4.14.3.7.4 Educate prospective Participating Providers on optional Member treatment and payment options while credentialing is underway, including:
  - 4.14.3.7.4.1. Authorization of out-of-network services;
  - 4.14.3.7.4.2. Single case agreements for an individual Member; and
  - 4.14.3.7.4.3. If agreed upon by the prospective Participating Provider, an opportunity for the Provider to accept a level of risk to receive payment after affirmative credentialing is completed in exchange for the prospective Participating Provider's compliance with network requirements and practices.
- 4.14.3.8 The MCO shall process credentialing applications from all types of Providers within prescribed timeframes as follows:
  - 4.14.3.8.1 For PCPs, within thirty (30) calendar days of receipt of clean and complete credentialing applications;
  - 4.14.3.8.2 For specialty care Providers, within forty-five (45) calendar days of receipt of clean and complete credentialing applications; and
  - 4.14.3.8.3 For any Provider submitting new or missing information for its credentialing application, the MCO shall act upon the new or updated information within ten (10) business days.
- 4.14.3.9 The start time for the approval process begins when the MCO has received a Provider's clean and complete application, and ends on the date of the Provider's written notice of network status.
- 4.14.3.10 A "clean and complete" application is an application that is signed and appropriately dated by the Provider, and includes:

Page 270 of 397	Date	
-----------------	------	--

New Hampshire Department of Health and Human Services Medicaid Care Management Services



#### **Exhibit B - DRAFT**

- 4.14.3.10.1 Evidence of the Provider's NH Medicaid ID; and
- 4.14.3.10.2 Other applicable information to support the Provider application, including Provider explanations related to quality and clinical competence satisfactory to the MCO.
- 4.14.3.11 In the event the MCO does not process a Provider's clean and complete credentialing application within the timeframes set forth in this Agreement, the MCO shall pay the Provider retroactive to thirty (30) calendar days or forty five (45) calendar days after receipt of the Provider's clean and complete application, depending on the prescribed timeframe for the Provider type as defined in this section.
- 4.14.3.12 For each day a clean and complete application is delayed beyond the prescribed timeframes in this Agreement as determined by periodic audit of the MCO's Provider enrollment records by the Department or its designee, the MCO shall be fined in accordance with Exhibit N(Liquidated Damages Matrix).
- 4.14.3.13 Nothing in this Agreement shall be construed to require the MCO to select a health care professional as a Participating Provider solely because the health care professional meets the NH Medicaid screening and credentialing verification standards, or to prevent an MCO from utilizing additional criteria in selecting the health care professionals with whom it contracts.

## 4.14.4 Provider Engagement

- 4.14.4.1 Provider Support Services
  - 4.14.4.1.1 The MCO shall develop and make available Provider support services which include, at a minimum:
    - 4.14.4.1.1.1. A website with information and a dedicated contact number to assist and support Providers who are interested in becoming Participating Providers;
    - 4.14.4.1.1.2. A dedicated contact number to MCO staff located in New Hampshire available from 8:00 a.m. to 6:00 p.m. Monday through Friday, and 9:00 a.m. to 12:00 p.m. on Saturday for the purposes of answering questions related to contracting, billing and service provision, except Department-approved holidays.

Page 271 of 397 Date \_\_\_\_\_

New Hampshire Department of Health and Human Services **Medicaid Care Management Services** 



4.14.4.1.1.3.	Ability for F	Providers to co	ntact the	MCO
	regarding	contracting,	billing,	and
	service pro	visions;		

- 4.14.4.1.1.4. Training specific to integration of physical and behavioral health, personcentered Care Management, healthrelated social needs, and quality, privacy and confidentiality of certain conditions;
- 4.14.4.1.1.5. Training curriculum, to be developed, in coordination with the Department that addresses clinical components necessary to meet the needs of Children with Special Health Care Needs. Examples of clinical topics shall include: federal requirements for EPSDT; unique needs of Children with Special Health Care Needs; family-driven, youthperson-centered treatment guided, planning and service provisions; impact of adverse childhood experiences; utilization of evidence-based practices; trauma-informed care; Recovery and resilience principles; and the value of person-centered Care Management that includes meaningful engagement of families/caregivers;
- 4.14.4.1.1.6. Training on billing and required documentation;
- guidance 4.14.4.1.1.7. Assistance and/or on identified opportunities for quality improvement:
- 4.14.4.1.1.8. Training to Providers in supporting and assisting Members in grievances and appeals, as described this Agreement; and
- Training to Providers in MCO claims 4.14.4.1.1.9. submittal through the MCO Provider portal.
- 4.14.4.1.2 The MCO shall establish and maintain a Provider services function to respond timely and adequately to Provider questions, comments, and inquiries.
- .. .. .... 4.14.4.1.3 at

As part of this f	uncti	on, the MC	CO shall	opera	ate a to	١c
free telephone	line	(Provider	service	line)	from,	а
•		`		,	•	
Page 272 of 397				Date _		

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

minimum, eight (8:00) am to five (5:00) pm EST, Monday through Friday, with the exception of Department-approved holidays. The Provider call center shall meet the following minimum standards, which may be modified by the Department as necessary:

- 4.14.4.1.3.1. Call abandonment rate: fewer than five percent (5%) of all calls shall be abandoned;
- 4.14.4.1.3.2. Average speed of answer: eighty percent (80%) of all calls shall be answered with live voice within thirty (30) seconds; and
- 4.14.4.1.3.3. Average speed of voicemail response: ninety percent (90%) of voicemail messages shall be responded to no later than the next business day (defined as Monday through Friday, with the exception of the Department-approved holidays).
- 4.14.4.1.4 The MCO shall ensure that, after regular business hours, the Provider inquiry line is answered by an automated system with the capability to provide callers with information regarding operating hours and instructions on how to verify enrollment for a Member.
- 4.14.4.1.5 The MCO shall have a process in place to handle after-hours inquiries from Providers seeking a service authorization for a Member with an urgent or emergency medical or behavioral health condition.
- 4.14.4.1.6 The MCO shall track the use of State-selected and nationally recognized clinical Practice Guidelines for Children with Special Health Care Needs.
- 4.14.4.1.7 The Department may provide additional guidelines to MCOs pertaining to evidence-based practices related to the following: Trauma-Focused Cognitive Behavioral Therapy; Trauma Informed Child-Parent Psychotherapy; Multi-systemic Therapy; Functional Family Therapy; Multi-Dimensional Treatment Foster Care; DBT; Multidimensional Family Therapy; Adolescent Community Reinforcement; and Assertive Continuing Care.

Page 273 of 397 Date \_\_\_\_\_

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

4.14.4.1.8 The MCO shall track and trend Provider inquiries, complaints and requests for information and take systemic action as necessary and appropriate pursuant to Exhibit O: Quality and Oversight Reporting Requirements.

#### 4.14.5 Provider Advisory Board

- 4.14.5.1 The MCO shall develop and facilitate an active Provider Advisory Board that is composed of a broad spectrum of Provider types. Provider representation on the Provider Advisory Board shall draw from and be reflective of Member needs and should ensure accurate and timely feedback on the MCM program, and shall include representation from at least one (1) FQHC, and at least one (1) CMH Program.
- 4.14.5.2 The Provider Advisory Board should meet face-to-face and/or via webinar or conference call a minimum of four (4) times each Agreement year. Minutes of the Provider Advisory Board meetings shall be provided to DHHS upon request.

# 4.14.6 Provider Contract Requirements

- 4.14.6.1 General Provisions
  - 4.14.6.1.1 The MCO's agreement with health care Providers shall:
    - 4.14.6.1.1.1. Be in writing;
    - 4.14.6.1.1.2. Be in compliance with applicable State and federal laws and regulations; and
    - 4.14.6.1.1.3. Include the requirements in this Agreement.
  - 4.14.6.1.2 The MCO shall submit all model Provider contracts to the Department for review before execution of the Provider contracts with NH Medicaid Providers.
  - 4.14.6.1.3 The MCO shall re-submit the model Provider contracts any time it makes substantive modifications.
  - 4.14.6.1.4 The Department retains the right to reject or require changes to any Provider contract.
  - 4.14.6.1.5 In all contracts with Participating Providers, the MCO shall comply with requirements in 42 CFR 438.214, RSA 420-F, and RSA 420-J:4 which includes selection and retention of Participating Providers, credentialing and re-credentialing requirements, and non-discrimination.

Page 274 of 397 Date \_\_\_\_\_

New Hampshire Department of Health and Human Services Medicaid Care Management Services



- 4.14.6.1.6 In all contracts with Participating Providers, the MCO shall follow a documented process for credentialing and re-credentialing of Participating Providers. [42 CFR 438.12(a)(2); 42 CFR 438.214(b)(2)]
- 4.14.6.1.7 The MCO's Participating Providers shall not discriminate against eligible Members because of race, color, creed, religion, ancestry, marital status, sexual orientation, sexual identity, national origin, age, sex, physical or mental handicap in accordance with Title VI of the Civil Rights Act of 1964, 42 U.S.C. Section 2000d, Section 504 of the Rehabilitation Act of 1973, 29 U.S.C. Section 794, the ADA of 1990, 42 U.S.C. Section 12131 and rules and regulations promulgated pursuant thereto, or as otherwise provided by law or regulation.
- 4.14.6.1.8 The MCO shall require Participating Providers and Subcontractors to not discriminate against eligible persons or Members on the basis of their health or behavioral health history, health or behavioral health status, their need for health care services, amount payable to the MCO on the basis of the eligible person's actuarial class, or pre-existing medical/health conditions.
- 4.14.6.1.9 The MCO shall keep Participating Providers informed and engaged in the QAPI program and related activities, as described in Section 4.14.3 (Quality Assessment and Performance Improvement Program).
- 4.14.6.1.10 Within 90 days upon availability or in accordance with applicable law, the MCO shall include in Provider contracts or MCO provider office reference manual a requirement securing cooperation with the QAPI program, and shall align the QAPI program to other Provider initiatives, including Advanced Payment Models (APMs), further described in Section 4.15 (Alternative Payment Models).
- 4.14.6.1.11 The MCO shall keep Participating Providers informed and engaged in the QAPI program and related activities, as described in Section 4.14.3 (Quality Assessment and Performance Improvement Program).
- 4.14.6.1.12 The MCO shall include in Provider contracts a requirement securing cooperation with the QAPI program, and shall align the QAPI program to other

Page 275 of 397	Date

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



- MCO Provider initiatives, including Advanced Payment Models (APMs), further described in Section 4.15 (Alternative Payment Models).
- 4.14.6.1.13 The MCO may execute Participating Provider agreements and begin credentialing, pending the outcome of screening and enrollment in NH Medicaid, of up to one hundred and twenty (120) calendar days duration but shall terminate a Participating Provider immediately upon notification from the Department that the Participating Provider cannot be enrolled, or the expiration of one (1) one hundred and twenty (120) day period without enrollment of the Provider, and notify affected Members. [42 CFR 438.602(b)(2)]
- 4.14.6.1.14 The MCO shall notify the Department no later than fourteen (14) calendar days in advance of the one hundred twenty (120) calendar day termination period to request the Department's assistance with NH Medicaid provider enrollment which may be available for pending enrollment applications.
- 4.14.6.1.15 The MCO shall notify impacted Members upon the MCO's Provider termination at the end of the one hundred twenty (120) day period.
- 4.14.6.1.16 The MCO shall maintain a Provider relations presence in NH, as approved by the Department.
- 4.14.6.1.17 The MCO shall provide training to all Participating Providers and their staff regarding the requirements of this Agreement, including the grievance and appeal system.
  - 4.14.6.1.17.1. The MCO's Provider training shall be completed within thirty (30) calendar days of entering into a contract with a Provider.
  - 4.14.6.1.17.2. The MCO shall provide ongoing training to new and existing Providers as required by the MCO, or as required by the Department.
  - 4.14.6.1.17.3. Provider materials shall comply with State and federal laws and the Department and NHID requirements.
  - 4.14.6.1.17.4. The MCO shall submit any Provider Manual(s) and Provider training materials to the Department for review

Page 276 of 397	Date

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



- during the Readiness Review period and sixty (60) calendar days prior to any substantive revisions.
- 4.14.6.1.17.5. Any revisions to the Provider Manual(s) and Provider training materials required by the Department shall be provided to the MCO within thirty (30) calendar days.
- 4.14.6.1.18 The MCO shall prepare and issue Provider Manual(s) upon request to all newly contracted and credentialed Providers and all Participating Providers, including any necessary specialty manuals (e.g., behavioral health).
  - 4.14.6.1.18.1. The Provider Manual shall be available and easily accessible on the web and updated no less than annually.
  - 4.14.6.1.18.2. The Provider Manual shall consist of, at a minimum:
    - 4.14.6.1.18.2.1 A description of the MCO's enrollment and credentialing process;
    - 4.14.6.1.18.2.2 How to access MCO Provider relations assistance;
    - 4.14.6.1.18.2.3 A description of the MCO's medical management and Case Management programs;
    - 4.14.6.1.18.2.4 Detail on the MCO's Prior Authorization processes;
    - 4.14.6.1.18.2.5 A description of the Covered Services and Benefits for Members, including EPSDT and pharmacy;
    - 4.14.6.1.18.2.6 A description of Emergency Services coverage;
    - 4.14.6.1.18.2.7 Member parity;
    - 4.14.6.1.18.2.8 The MCO Payment policies and processes; and
    - 4.14.6.1.18.2.9 The MCO Member and Provider Grievance System.

Page 277 of 397	Date _	

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



- 4.14.6.1.19 The MCO shall require that Providers not bill Members for Covered Services any amount greater than the Medicaid cost-sharing owed by the Member (i.e., no balance billing by Providers). [Section 1932(b)(6) of the Social Security Act; 42 CFR 438.3(k); 42 CFR 438.230(c)(1-2)]
  - 4.14.6.1.19.1. The MCO shall require the Provider to hold the Member harmless for the costs of Medically Necessary Covered Services except for applicable Cost Sharing and patient liability amounts indicated by the Department in this Agreement. [RSA 420-J:8.I.(a)]
- 4.14.6.1.20 In all contracts with Participating Providers, the MCO shall require Participating Providers to remain neutral when assisting potential Members and Members with enrollment decisions.
- 4.14.6.2 Compliance with MCO Policies and Procedures
  - 4.14.6.2.1 The MCO shall require Participating Providers to comply with all MCO policies and procedures, including without limitation:
    - 4.14.6.2.1.1. The MCO's DRA policy;
    - 4.14.6.2.1.2. The Provider Manual;
    - 4.14.6.2.1.3. The MCO's Compliance Program;
    - 4.14.6.2.1.4. The MCO's Grievance and Appeal Processes and Provider Appeal Processes:
    - 4.14.6.2.1.5. Clean Claims and Prompt Payment requirements;
    - 4.14.6.2.1.6. ADA requirements;
    - 4.14.6.2.1.7. Clinical Practice Guidelines; and
    - 4.14.6.2.1.8. Prior Authorization requirements.
  - 4.14.6.2.2 The MCO shall inform Participating Providers, at the time they enter into a contract with the MCO and periodically thereafter, about the following requirements:
    - 4.14.6.2.2.1. Member grievance and appeal processes as described in Section 4.5 (Member Grievances and Appeals), including:

Page 278 of 397	Date

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

- 4.14.6.2.2.1.1 Member grievance, appeal, and fair hearing procedures and timeframes:
- 4.14.6.2.2.1.2The Member's right to file grievances and appeals and the requirements and timeframe for filing;
- 4.14.6.2.2.1.3 The availability of assistance to the Member with filing grievances and appeals; [42 CFR 438.414; 42 CFR 438.10(g)(2)(xi)(A-C)];
- 4.14.6.2.2.1.4 The Member's right to request a State fair hearing after the MCO has made a determination on a Member's appeal which is adverse to the Member; and [42 CFR 438.414; 42 CFR 438.10(g)(2)(xi)(D)]; and
- 4.14.6.2.2.1.5 The Member's right to request continuation of benefits that the MCO seeks to reduce or terminate during an appeal of State fair hearing filing, if filed within the permissible although timeframes. the Member may be liable for the cost of any continued benefits while the appeal or State fair hearing is pending if the final decision is adverse to the Member. [42 CFR 438.414: 42 CFR 438.10(g)(2)(xi)(E)]

# 4.14.6.3 Requirement to Return Overpayment

- 4.14.6.3.1 Requirements for the Provider to comply with the Affordable Care Act and the MCO's policies and procedures that require the Provider to report and return any Overpayments identified within sixty (60) calendar days from the date the Overpayment is identified, and to notify the MCO in writing of the reason for the Overpayment. [42 CFR 438.608(d)(2)]
  - 4.14.6.3.1.1. Overpayments that are not returned within sixty (60) calendar days from the

Page 279 of 397 Date	Page 279 of 397	Date
----------------------	-----------------	------

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

date the Overpayment was identified may be a violation of State or federal law.

## 4.14.6.4 Background Screening

- 4.14.6.4.1 The Provider shall conduct background screening of its staff prior to contracting with the MCO and monthly thereafter against the Exclusion Lists.
- 4.14.6.4.2 In the event the Provider identifies that any of its staff is listed on any of the Exclusion Lists, the Provider shall notify the MCO within three (3) business days of learning that such staff Member is listed on any of the Exclusion Lists and immediately remove such person from providing services under the agreement with the MCO.

#### 4.14.6.5 Books and Records Access

- 4.14.6.5.1 The selected MCO must maintain the following records during the resulting contract term where appropriate and as prescribed by the Department:
  - 4.14.6.5.1.1. Books, records, documents and other electronic or physical Confidential Data evidencing and reflecting all costs and other expenses incurred by the selected Vendor(s) in the performance of the resulting contract(s), and all income received or collected by the selected Vendor(s).
  - All records must be maintained in 4.14.6.5.1.2. accordance with accounting procedures and practices, which sufficiently and properly reflect all such costs and expenses, and which are acceptable to the Department, and to include, without limitation, all ledgers, books, records, and original evidence of costs such as purchase requisitions and orders, vouchers, requisitions for materials, valuations inventories. of in-kind contributions, labor time cards, payrolls, and other records requested or required by the Department.
  - 4.14.6.5.1.3. Statistical, enrollment, attendance or visit records for each recipient of services, which shall include all records

Page 280 of 397	Date
-----------------	------

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

of application and eligibility (including all forms required to determine eligibility for each such recipient), records regarding the provision of services and all invoices submitted to the Department to obtain payment for such services.

- 4.14.6.5.2 Medical records on each patient/recipient of services.
- 4.14.6.5.3 During the term of the resulting contract(s) and the 10-year period for retention, the Department, the United States Department of Health and Human Services, and any of their designated representatives shall have access to all reports and records maintained pursuant to the resulting contract(s) for purposes of audit, examination, excerpts and transcripts. Upon the purchase by the Department of the maximum number of units provided for in the resulting contract(s) and upon payment of the price limitation hereunder, the selected Vendor(s) and all the obligations of the parties hereunder (except such obligations as, by the terms of the resulting contract(s) are to be performed after the end of the term of the contract(s) and/or survive the termination of the contract(s)) shall terminate, provided however, that if, upon review of the Final Expenditure Report the Department shall disallow any expenses claimed by the selected Vendor(s) as costs hereunder the Department shall retain the right, at its discretion, to deduct the amount of such expenses as are disallowed or to recover such sums from the selected Vendor(s).
- 4.14.6.5.4 The MCO shall require that all Participating Providers comply with MCO and State policies related to transition of care policies set forth in this Agreement and in the MCO's Member Handbook.
- 4.14.6.6 Continuity of Care
  - 4.14.6.6.1 The MCO shall require that all Participating Providers comply with MCO and State policies related to transition of care policies set forth by the Department and included in the Department's Model Member Handbook.
- 4.14.6.7 Anti-Gag Clause
  - 4.14.6.7.1 The MCO shall not prohibit, or otherwise restrict, a Provider acting within the lawful scope of practice,

Page 281 of 397	Date

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

from advising or advocating on behalf of a Member who is their patient:

- 4.14.6.7.1.1. For the Member's health status, medical care, or treatment options, including any alternative treatment that may be self-administered:
- 4.14.6.7.1.2. For any information the Member needs in order to decide among all relevant treatment options;
- 4.14.6.7.1.3. For the risks, benefits, and consequences of treatment or non-treatment: or
- 4.14.6.7.1.4. For the Member's right to participate in decisions regarding their health care, including the right to refuse treatment, and to express preferences about future treatment decisions. [Section 1932(b) of the Social Security Act; 42 CFR 438.102(a)(1)(i)-(iv); SMDL 2/20/98]
- 4.14.6.7.2 The MCO shall not take punitive action against a Provider who either requests an expedited resolution or supports a Member's appeal, consistent with the requirements in Section 4.5.5 (Expedited Appeal). [42 CFR 438.410(b)]

#### 4.14.6.8 Anti-Discrimination

- 4.14.6.8.1 The MCO shall not discriminate with respect to participation, reimbursement, or indemnification as to any Provider who is acting within the scope of the Provider's license or certification under applicable State law, solely on the basis of such license or certification or against any Provider that serves highrisk populations or specializes in conditions that require costly treatment.
- 4.14.6.8.2 This paragraph shall not be construed to prohibit an organization from:
  - 4.14.6.8.2.1. Including Providers only to the extent necessary to meet the needs of the organization's Members;
  - 4.14.6.8.2.2. Establishing any measure designed to maintain quality and control costs consistent with the responsibilities of the organization; or

age 282 of 397	Date
----------------	------

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

- 4.14.6.8.2.3. Using different reimbursement amounts for different specialties or for different practitioners in the same specialty.
- 4.14.6.8.3 If the MCO declines to include individual or groups of Providers in its network, it shall give the affected Providers written notice of the reason for the decision.
- 4.14.6.8.4 In all contracts with Participating Providers, the MCO's Provider selection policies and procedures shall not discriminate against particular Providers that service high-risk populations or specialize in conditions that require costly treatment. [42 CFR 438.12(a)(2); 42 CFR 438.214(c)]

### 4.14.6.9 Access and Availability

4.14.6.9.1 The MCO shall ensure that Providers comply with the time and distance and wait standards, including but not limited to those described in Section 4.7.3 (Time and Distance Standards) and Section 4.7.5 (Timely Access to Service Delivery).

#### 4.14.6.10 Payment Models

- 4.14.6.10.1 The MCO shall negotiate rates with Providers in accordance with Section 4.15 (Alternative Payment Models) and Section 4.17 (Provider Payments) of this Agreement, unless otherwise specified by the Department (e.g., minimum Medicaid fee schedule rates, directed payments).
- 4.14.6.10.2 The MCO Provider contract shall contain full and timely disclosure of the method and amount of compensation, payments, or other consideration, to be made to and received by the Provider from the MCO, including for Providers paid by an MCO Subcontractor, such as the PBM.
- 4.14.6.10.3 The MCO Provider contract shall detail how the MCO shall meet its reporting obligations to Providers as described within this Agreement.

#### 4.14.6.11 Non-Exclusivity

4.14.6.11.1 The MCO shall not require a Provider or Provider group to enter into an exclusive contracting arrangement with the MCO as a condition for network participation.

#### 4.14.6.12 Proof of Membership

age 283 of 397	Date
----------------	------

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

4.14.6.12.1 The MCO Provider contract shall require Providers in the MCO network to accept the Member's Medicaid identification card as proof of enrollment in the MCO until the Member receives his/her MCO identification card.

#### 4.14.6.13 Other Provisions

- 4.14.6.13.1 The MCO's Provider contract shall also contain:
  - 4.14.6.13.1.1. All required activities and obligations of the Provider and related reporting responsibilities;
  - 4.14.6.13.1.2. Requirements to comply with all applicable Medicaid laws, regulations, including applicable subregulatory guidance and applicable provisions of this Agreement; and
  - 4.14.6.13.1.3. A requirement to notify the MCO within one (1) business day of being cited by any State or federal regulatory authority.

#### 4.14.7 Reporting

- 4.14.7.1 The MCO shall comply with and complete all reporting in accordance with Exhibit O: Quality and Oversight Reporting Requirements, this Agreement, and as further specified by the Department.
- 4.14.7.2 The MCO shall implement and maintain arrangements or procedures for notification to the Department when it receives information about a change in a Participating Provider's circumstances that may affect the Participating Provider's eligibility to participate in the managed care program, including the termination of the Provider agreement with the MCO. [42 CFR 438.608(a)(4)]
- 4.14.7.3 The MCO shall notify the Department within seven (7) calendar days of any significant changes to the Participating Provider network.
- 4.14.7.4 As part of the notice, the MCO shall submit a Transition Plan to the Department to address continued Member access to needed service and how the MCO shall maintain compliance with its contractual obligations for Member access to needed services.
- 4.14.7.5 A significant change is defined as:
  - 4.14.7.5.1 A decrease in the total number of PCPs by more than five percent (5%);

age 284 of 397	Date

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

- 4.14.7.5.2 A loss of all Providers in a specific specialty where another Provider in that specialty is not available within time and distance standards outlined in Section 4.7.3 (Time and Distance Standards) of this Agreement; and
- 4.14.7.5.3 A loss of a hospital in an area where another contracted hospital of equal service ability is not available within time and distance standards outlined in Section 4.7.3 (Time and Distance Standards) of this Agreement; and/or
- 4.14.7.5.4 Other adverse changes to the composition of the network, which impair or deny the Members' adequate access to Participating Providers.
- 4.14.7.6 The MCO shall provide to the Department and/or the Department's Subcontractors (e.g., the EQRO) Provider participation reports on an annual basis or as otherwise determined by the Department in accordance with Exhibit O: Quality and Oversight Reporting Requirements; these may include but are not limited to Provider participation by geographic location, categories of service, Provider type categories, Providers with open panels, and any other codes necessary to determine the adequacy and extent of participation and service delivery and analyze Provider service capacity in terms of Member access to health care.

## 4.15 Alternative Payment Models

### 4.15.1 **General**

- 4.15.1.1 The Department is committed to implementing clinically and actuarially sound incentives designed to improve care quality and utilization. The Department will define a Medicaid APM Strategy that may include supporting guidance, worksheets, and templates that will build upon the parameters set forth in this Agreement.
- 4.15.1.2 The Department will implement strategies to expand use of APMs that promote the goals of the Medicaid program to provide the right care at the right time, and in the right place through the delivery of high-quality, cost-effective care for the whole person, with a focus on engaged primary and preventive care model and in a manner that is transparent to the Department, Providers, and the stakeholder community.
- 4.15.1.3 In developing and refining its APM strategy, the Department relies on the framework established by the Health Care Payment Learning and Action Network APM framework (or the "HCP-LAN APM framework") in order to:

age 285 of 397	Date

New Hampshire Department of Health and Human Services Medicaid Care Management Services



4.15.1.3.1	Clearly and effectively communicate the Department
	requirements through use of the defined categories
	established by HCP-LAN;

- 4.15.1.3.2 Encourage the MCO to align MCM APM offerings to other payers' APM initiatives to minimize Provider burden; and
- 4.15.1.3.3 Provide an established framework for monitoring MCO performance on APMs.
- 4.15.2 Prior to and/or over the course of the Term of this Agreement, the Department shall develop the Department's Medicaid APM Strategy, which shall result in additional guidance, templates, worksheets, required provider contractual provisions and other material that elucidate the requirements to which the MCO is subject under this Agreement.
- 4.15.3 The MCOs shall develop APMs consistent with guidance in the Department's Medicaid APM Strategy including, but not limited to:
  - 4.15.3.1 Incentivize primary care clinicians to engage attributed Members in Primary and Prevention Focused Model and Provider Delivered Care Coordination.
- 4.15.4 According to models that incentivize consistent quality outcomes as prescribed by the Department.
- 4.15.5 Within the guidance parameters established and issued by the Department and subject to Department approval, the MCO shall design Qualifying APMs as defined in Section 4.16.9 (Qualifying Alternative Payment Models) consistent with the Department Medicaid APM strategy and in conformance with CMS guidance.
- 4.15.6 The MCO shall support the Department in developing the Department's Medicaid APM Strategy through participation in regular stakeholder meetings and planning efforts, implementing required provider contractual provisions, providing all required and otherwise requested information related to APMs, sharing Confidential Data and analysis, and other activities as specified by the Department.
- 4.15.7 For any APMs that direct the MCO's expenditures under 42 CFR 438.6(c)(1)(i) or (ii), the MCO and the Department shall ensure that it:
  - 4.15.7.1 Makes participation in the APM available, using the same terms of performance, to a class of Providers providing services under the contract related to the reform or improvement initiative;
  - 4.15.7.2 Uses a common set of quality performance measures across all similarly situated Providers as directed in the Department's Medicaid APM Strategy;
  - 4.15.7.3 Does not set the amount or frequency of the expenditures; and

age 200 01 397 Date	age 286 of 397 Date	
---------------------	---------------------	--

New Hampshire Department of Health and Human Services Medicaid Care Management Services



- 4.15.7.4 Does not permit the Department to recoup any unspent funds allocated for these arrangements from the MCO. [42 CFR 438.6(c)]
- 4.15.8 Required Use of Alternative Payment Models
  - 4.15.8.1 The MCO shall ensure through its APM Implementation Plan as described in Section 4.17 (MCO Alternative Payment Model Implementation P, and confirmed through Exhibit O: Quality and Oversight Reporting Requirements, reporting that fifty percent (50%) of all Covered Services medical expenditures are in Qualifying APMs, as defined by the Department, subject to the following:
    - 4.15.8.1.1 If the MCO is newly participating in the MCM program as of the Program Start Date, the MCO shall have twelve (12) months to meet this requirement; and
    - 4.15.8.1.2 If the MCO determines that circumstances materially inhibit its ability to meet the APM implementation requirement, the MCO shall detail to DHHS in its proposed APM Implementation Plan an extension request: the reasons for its inability to meet the requirements of this section and any additional information required by DHHS.
  - 4.15.8.2 If approved by DHHS, the MCO may use its alternative approach, but only for the period of time requested and approved by DHHS, which is not to exceed an additional six (6) months after the initial 12-month period.
  - 4.15.8.3 The MCO shall implement the Qualifying APM models as directed by the Department in the Department's Medicaid APM Strategy including, but not limited to, directed payments with quality incentives for achieving statewide outcomes for Community Mental Health Centers and providers, total cost of care models with large providers including quality metrics incentivizing Provider-Delivered Care Coordination and Primary Care and Prevention Focused care.
  - 4.15.8.4 For failure to meet Section 4.16.8 (Required Use of Alternative Payment Models), the Department reserves the right to issue remedies as described in Section 5.5.2 (Liquidated Damages) and Exhibit N (Liquidated Damages Matrix).
  - 4.15.8.5 Consistent with RSA 126-AA, the MCO shall include, through APMs and other means, Provider alignment incentives to leverage the combined Department, MCO, and Providers to achieve the purpose of the incentives.
  - 4.15.8.6 The MCO shall be subject to incentives, at the Department's sole discretion, and/or penalties to achieve improved performance, including preferential auto-assignment of new Members, use of the

Page 287 of 397	Date	

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

MCM Withhold and Incentive Program (including the shared incentive pool), and other incentives.

## 4.15.9 Qualifying Alternative Payment Models

- 4.15.9.1 A Qualifying APM is a payment approach required and approved by the Department as consistent with the standards specified in this Section 4.16.9 (Qualifying Alternative Payment Models) and the Department's Medicaid APM Strategy.
- 4.15.9.2 At minimum, a Qualifying APM shall meet the requirements of the HCP-LAN APM framework Category 2B based on the refreshed 2017 framework released on July 11, 2017 and all subsequent revisions.
- 4.15.9.3 HCP-LAN Categories 3A, 3B, 4A, 4B and 4C shall all also be considered Qualifying APMs, and the MCO shall increasingly adopt such APMs over time in accordance with its APM Implementation Plan and the Department's Medicaid APM Strategy.
- 4.15.9.4 The Department shall determine, on the basis of the Standardized Assessment of APM Usage described in Section 4.16.12.3 (Standardized Assessment of Alternative Payment Model Usage) below and the additional information available to the Department, the HCP-LAN Category to which the MCO's APM(s) is/are aligned.
- 4.15.9.5 Under no circumstances shall the Department consider as a Qualifying APM a payment methodology that takes cost of care into account without also paying providers for achieving quality outcomes consistent with those set forth in the Department's Medicaid APM Strategy. Providers participating in Qualifying APMs shall have the opportunity to share in cost savings through a formula that is no less than 50/50 split in favor of the participating providers and shall incorporate a opportunity to share up to an additional #% of total payments paid through the APM as provider incentive payments for achieving quality outcomes as part of the Qualifying APM.
- 4.15.9.6 At the sole discretion of the Department, additional APMs specifically required by and defined as an APM by the Department shall also be deemed to meet the definition of a Qualifying APM under this Agreement.
- 4.15.9.7 Standards for Large Providers and Provider Systems
  - 4.15.9.7.1 The MCO shall predominantly adopt a total cost of care model with shared savings for large Provider systems to the maximum extent feasible, and as further defined by the Department's Medicaid APM Strategy, including incentives for the Primary Care

Page 288 of 397 Date \_\_\_\_\_

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

and Prevention Focused Model inclusive of Provider Delivered Care Coordination.

- 4.15.9.8 Treatment of Payments to Community Mental Health Programs
  - 4.15.9.8.1 The CMH Program payment model prescribed by DHHS in Section 4.13.19 (Community Mental Health Services) shall be deemed to meet the definition of a Qualifying APM under this Agreement.
- 4.15.9.9 Alternative Payment Models for Substance Use Disorder Treatment
  - 4.15.9.9.1 The MCO shall include in its APM Implementation Plan:
    - 4.15.9.9.1.1. At least one (1) APM that promotes the coordinated and cost-effective delivery of high-quality care to birthing parents and infants born affected by exposure to substance use; and
    - 4.15.9.9.1.2. At least one (1) APM that promotes greater use of Medication for treatment of substance use disorders through a bundled payment as set forth in the Department's Medicaid APM Strategy.
- 4.15.9.10 Accommodations for Other Providers
  - 4.15.9.10.1 The MCO may develop Qualifying APM models appropriate for other primary care Providers, and Federally Qualified Health Centers (FQHCs), as further defined by the DHHS Medicaid APM Strategy to incentivize engaged primary and preventive care.
  - 4.15.9.10.2 For example, the MCO may propose to the Department models that incorporate pay-for-performance bonus incentives and/or per Member per month payments related to Providers' success in meeting actuarially-relevant cost and quality targets.
- 4.15.9.11 Alignment with Existing Alternative Payment Models and Promotion of Integration with Behavioral Health
  - 4.15.9.11.1 The MCO shall incentivize Providers participating in the Qualifying APMs by paying incentives for achieving quality outcomes established by the Department in the Department's Medicaid APM Strategy.
  - 4.15.9.11.2 The MCO shall align APM offerings to current and emerging APMs in NH, both within Medicaid and across other payers (e.g., Medicare and commercial

Page 289 of 397	Date

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

shared savings arrangements) to reduce Provider burden, incentivize primary and preventive care and promote the integration of Behavioral Health.

4.15.9.11.3 The MCO may incorporate APM design elements into its Qualifying APMs that permit Participating Providers to attest to participation in an "Other Advanced APM."

## 4.15.10 MCO Alternative Payment Model Implementation Plan

- 4.15.10.1 The MCO shall submit to the Department for review and approval an APM Implementation Plan in accordance with Exhibit O: Quality and Oversight Reporting Requirements.
- 4.15.10.2 The APM Implementation Plan shall meet the requirements of this section and of any subsequent guidance issued as part of the Department Medicaid APM Strategy.
- 4.15.10.3 Additional details on the timing, format, and required contents of the MCO APM Implementation Plan shall be specified by the Department in Exhibit O: Quality and Oversight Reporting Requirements and/or through additional guidance.
- 4.15.10.4 Alternative Payment Model Transparency
- 4.15.10.5 The MCO shall describe in its APM Implementation Plan, for each APM offering and as is applicable, the actuarial and public health basis for the MCO's methodology, as well as the basis for developing and evaluating Participating Provider performance in the APM, as described in Section 4.16.10.11 (Alternative Payment Model Transparency and Reporting Requirements). The APM Implementation Plan shall also outline how the MCO intends to achieve Provider to Member engagement in Primary Care and Prevention model and Provider-Delivered Care Coordination including health and wellness assessments and integrated behavioral health care through the APM.

## 4.15.10.6 Provider Engagement and Support

- 4.15.10.6.1 The APM Implementation Plan shall describe a logical and reasonably achievable approach to implementing APMs, supported by an understanding of NH Medicaid Providers' readiness for participation in APMs, and the strategies the MCO shall use to assess and advance such readiness over time.
- 4.15.10.6.2 The APM Implementation Plan shall outline in detail what strategies the MCO plans to use, such as, meetings with Providers, as appropriate, and the frequency of such meetings, the provision of technical support, and a Confidential Data sharing strategy for

Page 290 of 397 Date \_\_\_\_\_

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

Providers reflecting the transparency, reporting and Confidential Data sharing obligations herein and in the Department Medicaid APM Strategy. The MCO shall ensure regular and consistent engagement with Providers around APMs on at least a quarterly basis through direct or virtual visits by the MCOs key staff responsible for the MCOs provider relations and APM Implementation Plan.

- 4.15.10.6.3 The MCO APM Implementation Plan shall ensure Providers, as appropriate, are supported by Confidential Data sharing and performance analytic feedback systems and tools that make actuarially sound and actionable provider level and system level clinical, cost, and performance Confidential Data available to Providers in a timely manner for purposes of developing APMs and analyzing performance and payments pursuant to APMs.
- 4.15.10.6.4 MCO shall provide the financial support for the Provider infrastructure necessary to develop and implement APM arrangements that increase in sophistication over time.

#### 4.15.10.7 Implementation Approach

- 4.15.10.7.1 The MCO shall include in the APM Implementation Plan a detailed description of the steps the MCO shall take to advance its APM Implementation Plan:
  - 4.15.10.7.1.1. In advance of the Program Start Date;
  - 4.15.10.7.1.2. During the first year of this Agreement; and
  - 4.15.10.7.1.3. Into the second year and beyond, clearly articulating its long-term vision and goals for the advancement of APMs over time.
- 4.15.10.7.2 The APM Implementation Plan shall include the MCO's plan for providing the necessary Confidential Data and information to participating APM Providers to ensure Providers' ability to successfully implement and meet the performance expectations included in the APM, including how the MCO shall ensure that the information received by Participating Providers is meaningful and actionable.
- 4.15.10.7.3 The MCO shall provide Confidential Data to Providers, as appropriate, that describe the

Page 291 of 397	Date
-----------------	------

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

retrospective cost and utilization patterns for Members, which shall inform the strategy and design of APMs.

- 4.15.10.7.4 For each APM entered into, the MCO shall provide timely and actionable cost, quality and utilization information to Providers participating in the APM that enables and tracks performance under the APM and notifies the Providers with clarity throughout the APM period of their progress against incentive payment formulas at least quarterly.
- 4.15.10.7.5 In addition, the MCO shall provide Member and Provider level Confidential Data (e.g., encounter and claims information) for concurrent real time utilization and care management interventions.
- 4.15.10.7.6 The APM Implementation Plan shall describe in example form to the Department the level of information that shall be given to Providers that enter into APM Agreements with the MCO, including if the level of information shall vary based on the Category and/or type of APM the Provider enters.
- 4.15.10.7.7 The information provided shall be consistent with the requirements outlined under Section 4.16.11 (Alternative Payment Model Transparency and Reporting Requirements). The MCOs shall utilize all applicable and appropriate agreements as required under State and federal law to maintain confidentiality of protected health information.

# 4.15.11 Alternative Payment Model Transparency and Reporting Requirements 4.15.12 Transparency

- 4.15.12.1 In the MCO APM Implementation Plan, the MCO shall provide to the Department for each APM, as applicable, the following information at a minimum:
  - 4.15.12.1.1 The methodology for determining Member attribution, and sharing information on Member attribution with Providers participating in the corresponding APM;
  - 4.15.12.1.2 The methodology for incentivizing Providers engage Members in Provider-Delivered Care Coordination and Primary Care and Prevention, including, but not limited to, health and wellness screenings;
  - 4.15.12.1.3 The mechanisms used to determine cost benchmarks and Provider performance, including cost target

Page 292 of 397	Date

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

- calculations, and the attachment points for cost targets, and risk adjustment methodology;
- 4.15.12.1.4 The approach to determining quality benchmarks and evaluating Provider performance, including advance communication of the specific measures that shall be used to determine quality performance, the methodology for calculating and assessing Provider performance, and any quality gating criteria that may be included in the APM design; and
- 4.15.12.1.5 The frequency at which the MCO shall regularly report Confidential Data related to APM performance to Providers on cost, quality, evaluation of progress towards incentive payments and the information that shall be included in each report.
- 4.15.12.2 Additional information may be required by the Department in supplemental guidance. All information provided to the Department shall be made available to Providers eligible to participate in or already participating in the APM unless the MCO requests and receives the Department approval for specified information not to be made available.
- 4.15.12.3 Standardized Assessment of Alternative Payment Model Usage
  - 4.15.12.3.1 Additional information may be required by the Department's Medicaid APM Strategy and supplemental guidance. All information provided to the Department shall be made available to Providers eligible to participate in or already participating in the APM unless the MCO requests and receives the Department approval for specified information not to be made available.
- 4.15.12.4 Standardized Assessment of Alternative Payment Model Usage
  - 4.15.12.4.1 The MCO shall complete, attest to the contents of, and submit to the Department the HCP-LAN APM assessment in accordance with Exhibit O: Quality and Oversight Reporting Requirements.
  - 4.15.12.4.2 Thereafter, the MCO shall complete, attest to the contents of, and submit to the Department the HCP-LAN APM assessment in accordance with Exhibit O: Quality and Oversight Reporting Requirements and/or the Department Medicaid APM Strategy.
  - 4.15.12.4.3 If the MCO reaches an agreement with the Department that its implementation of the required APM model(s) may be delayed, the MCO shall

age 293 of 397 Date	
---------------------	--

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

comply with all terms set forth by the Department for the additional and/or alternative timing of the MCO's submission of the HCP-LAN APM assessment.

- 4.15.12.5 Additional Reporting on Alternative Payment Model Outcomes
  - 4.15.12.5.1 The MCO shall provide additional information required by the Department in Exhibit O: Quality and Oversight Reporting Requirements or other Department guidance on the type, usage, effectiveness and outcomes of its APMs.

#### 4.15.13 **Development Period for MCO Implementation**

4.15.13.1 Consistent with the requirements for new MCOs, outlined in Section 4.16.9 (Qualifying Alternative Payment Models) above, the Department acknowledges that MCOs may require time to advance their MCO Implementation Plan. The Department shall provide additional detail, in its Medicaid APM Strategy, that describes how MCOs should expect to advance use of APMs over time.

# 4.15.14 Alternative Payment Model Alignment with State Priorities and Evolving Public Health Matters

- 4.15.14.1 The MCO's APM Implementation Plan shall adopt the quantitative, measurable clinical outcomes required by the Department in the Department's Medicaid APM Strategy and additional outcomes the MCO seeks to improve through its APM and QAPI initiative(s).
- 4.15.14.2 At a minimum, the MCO shall address the priorities identified in this Section 4.16.14 (Alternative Payment Model Alignment with State Priorities and Evolving Public Health Matters) and all additional priorities identified by the Department in the Department's Medicaid APM Strategy.
- 4.15.14.3 State Priorities in RSA 126-AA
  - 4.15.14.3.1 The MCO's APM Implementation Plan and/or QAPI Plan shall address the following priorities:
    - 4.15.14.3.1.1. Opportunities to decrease unnecessary service utilization, particularly as related to use of the ED, especially for Members with behavioral health needs and among low-income children;
    - 4.15.14.3.1.2. Opportunities to reduce preventable admissions and thirty (30)-day hospital readmission for all causes;
    - 4.15.14.3.1.3. Opportunities to improve the timeliness of prenatal care and other efforts that support the reduction of births of infants

Page 294 of 397	Date
-----------------	------

New Hampshire Department of Health and Human Services Medicaid Care Management Services



#### **Exhibit B - DRAFT**

born affected by exposure to substance use;

- 4.15.14.3.1.4. Opportunities to better integrate physical and behavioral health, particularly efforts to increase the timeliness of follow-up after a mental illness or Substance Use Disorder admission;
- 4.15.14.3.1.5. Opportunities to incentivize, through payments to Providers and Member incentives, Provider engagement with attributed Members in primary and preventive care, health needs assessments, Provider- Directed Care Coordination at a frequency and in a manner set forth in the Department's Medicaid APM Strategy;
- 4.15.14.3.1.6. Opportunities to better manage pharmacy utilization, including through Participating Provider incentive arrangements focused on efforts such as increasing generic prescribing and efforts aligned to the MCO's Medication Management program aimed reducing Polypharmacy, as described in Section 4.2 (Medication Management);
- 4.15.14.3.1.7. Opportunities to enhance access to and the effectiveness of medication to treat Substance Use Disorder treatment; and
- 4.15.14.3.1.8. Opportunities to address health-related social needs (further addressed in Section 4.11.7 (Provider-Delivered Care Coordination and Integration with Social Services and Community Care) of this Agreement), and in particular to address "ED boarding," in which Members that would be best treated in the community remain in the ED.

4.15.14.4 Emerging State Medicaid and Public Health Priorities

4.15.14.4.1 The MCO shall address priorities identified by the Department in the Medicaid APM Strategy or related quidance.

Page 295 of 397 Date \_\_\_\_\_

New Hampshire Department of Health and Human Services Medicaid Care Management Services



#### **Exhibit B - DRAFT**

4.15.14.4.2 If the Department adds or modifies priorities after the Program Start Date, the MCO shall incorporate plans for addressing the new or modified priorities in the next regularly-scheduled submission of it APM Implementation Plan.

## 4.15.15 Physician Incentive Plans

- 4.15.15.1 The MCO shall submit all Physician Incentive Plans to the Department for review as part of its APM Implementation Plan or upon development of Physician Incentive Plans that are separate from the MCO's APM Implementation Plan.
- 4.15.15.2 The MCO shall not implement Physician Incentive Plans until they have been reviewed and approved by the Department.
- 4.15.15.3 Any Physician Incentive Plan, including those detailed within the MCO's APM Implementation Plan, shall be in compliance with the requirements set forth in 42 CFR 422.208 and 42 CFR 422.210, in which references to "MA organization," "CMS," and "Medicare beneficiaries" should be read as references to "MCO," "DHHS," "the Department," and "Members," respectively. These include that:
  - 4.15.15.3.1 The MCO may only operate a Physician Incentive Plan if no specific payment can be made directly or indirectly under a Physician Incentive Plan to a physician or Physician Group as an incentive to reduce or limit Medically Necessary Services to a Member [Section 1903(m)(2)(A)(x) of the Social Security Act; 42 CFR 422.208(c)(1)-(2); 42 CFR 438.3(i)]; and
  - 4.15.15.3.2 If the MCO puts a physician or Physician Group at substantial financial risk for services not provided by the physician or Physician Group, the MCO shall ensure that the physician or Physician Group has adequate stop-loss protection. [Section 1903(m)(2)(A)(x) of the Social Security Act; 42 CFR 422.208(c)(2); 42 CFR 438.3(i)]
- 4.15.15.4 The MCO shall submit to the Department annually, at the time of its annual HCP-LAN assessment, a detailed written report of any implemented (and previously reviewed) Physician Incentive Plans, as described in Exhibit O: Quality and Oversight Reporting Requirements.
- 4.15.15.5 Annual Physician Incentive Plan reports shall provide assurance satisfactory to the Department that the requirements of 42 CFR 438.208 are met. The MCO shall, upon request, provide additional detail in response to any Department request to understand the terms of Provider payment arrangements.

Page 296 of 397	Date
-----------------	------

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

- 4.15.15.6 The MCO shall provide to Members upon request the following information:
  - 4.15.15.6.1 Whether the MCO uses a Physician Incentive Plan that affects the use of referral services;
  - 4.15.15.6.2 The type of incentive arrangement; and
  - 4.15.15.6.3 Whether stop-loss protection is provided. [42 CFR 438.3(i)].

#### 4.16 **Provider Payments**

#### 4.16.1 **General Requirements**

- 4.16.1.1 The MCO shall not, directly or indirectly, make payment to a physician or Physician Group or to any other Provider as an inducement to reduce or limit Medically Necessary Services furnished to a Member. [Section 1903(m)(2)(A)(x) of the Social Security Act; 42 CFR 438.3(i)]
- 4.16.1.2 The MCO shall not pay for an item or service (other than an emergency item or service, not including items or services furnished in an emergency room of a hospital) under the following circumstances: [Section 1903 of the Social Security Act]:
  - 4.16.1.2.1 When furnished under the MCO by an individual or entity during any period when the individual or entity is excluded from participation under Title V, XVIII, or XX of the Social Security Act or pursuant to sections 1128, 1128A, 1156, or 1842(j)(2) of the Social Security Act;
  - 4.16.1.2.2 When furnished at the medical direction or on the prescription of a physician, during the period when such physician is excluded from participation under Title V, XVIII, or XX pursuant to sections 1128, 1128A, 1156, or 1842(j)(2) of the Social Security Act when the person knew or had any reason to know of the exclusion (after a reasonable time period after reasonable notice has been furnished to the person);
  - 4.16.1.2.3 When furnished by an individual or entity to whom the State has suspended payments during any period when there is a pending investigation of a credible allegation of Fraud against the individual or entity, unless the State determines there is good cause not to suspend such payments or if the individual or entity has not completed their federally required enrollment revalidation with the Department;

Page 297 of 397 Date \_\_\_\_\_

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

- 4.16.1.2.4 With respect to any amount expended for roads, bridges, stadiums, or any other item or service not covered under the Medicaid State Plan; and [Section 1903(i) of the Social Security Act, final sentence; section 1903(i)(2)(A-C) of the Social Security Act; section 1903(i)(16-17) of the Social Security Act]
- 4.16.1.2.5 When furnished by a Participating Provider or entity that is not enrolled with NH Medicaid or whose Medicaid participation has been terminated by the Department.
- 4.16.1.3 No payment shall be made to a Participating Provider other than by the MCO for services covered under the Agreement between the Department and the MCO, except when these payments are specifically required to be made by the State in Title XIX of the Social Security Act, in 42 CFR Chapter IV, or when the Department makes direct payments to Participating Providers for graduate medical education costs approved under the Medicaid State Plan, or have been otherwise approved by CMS. [42 CFR 438.60]
- 4.16.1.4 The MCO shall reimburse Providers based on the Current Procedural Terminology (CPT) code's effective date. To the extent a procedure is required to be reimbursed under the Medicaid State Plan but no CPT code or other billing code has been provided by the Department, the MCO shall contact the Department and obtain a CPT code and shall retroactively reimburse claims based on the CPT effective date as a result of the CPT annual updates.
  - 4.16.1.4.1 For MCO provider contracts based on NH Medicaid fee schedules, the MCO shall reimburse providers for annual and periodic fee schedule adjustments in accordance with their effective dates.
- 4.16.1.5 The MCO shall permit Providers up to one hundred and twenty (120) calendar days to submit a timely claim. The MCO shall establish reasonable policies that allow for good cause exceptions to the one hundred and twenty (120) calendar day timeframe.
- 4.16.1.6 Good cause exceptions shall accommodate foreseeable and unforeseeable events such as:
  - 4.16.1.6.1 A Member providing the wrong Medicaid identification number;
  - 4.16.1.6.2 Natural disasters; or
  - 4.16.1.6.3 Failed information technology systems.
- 4.16.1.7 The Provider should be provided a reasonable opportunity to rectify the error, once identified, and to either file or re-file the claim.

Page 298 of 397	Date

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

- 4.16.1.8 Within the first one hundred and eighty (180) calendar days of the Program Start Date, the Department has discretion to direct MCOs to extend the one hundred and twenty (120) calendar days on case by case basis.
- 4.16.1.9 The MCO shall pay interest on any Clean Claims that are not paid within thirty (30) calendar days at the interest rate published in the Federal Register in January of each year for the Medicare program.
- 4.16.1.10 The MCO shall collect Confidential Data from Providers in standardized formats to the extent feasible and appropriate, including secure information exchanges and technologies utilized for State Medicaid quality improvement and Care Coordination efforts. [42 CFR 438.242(b)(3)(iii)]
- 4.16.1.11 The MCO shall implement and maintain arrangements or procedures for prompt reporting of all Overpayments identified or recovered, specifying the Overpayments due to potential Fraud, Waste or Abuse, to the Department. [42 CFR 438.608(a)(2)]
- 4.16.1.12 The MCO shall comply with State and federal laws requiring nonpayment to a Participating Provider for Hospital-Acquired Conditions and for Provider Preventable Conditions.
  - 4.16.1.12.1 The MCO shall not make payments to a Provider for a Provider Preventable Condition that meets the following criteria:
    - 4.16.1.12.1.1. Is identified in the Medicaid State Plan;
    - 4.16.1.12.1.2. Has been found based upon a review of medical literature by qualified professionals, to be reasonably preventable through the application of procedures supported by evidence-based guidelines;
    - 4.16.1.12.1.3. Has a negative consequence for the Member:
    - 4.16.1.12.1.4. Is auditable; and
    - 4.16.1.12.1.5. Includes, at a minimum, wrong surgical or other invasive procedure performed on a patient, surgical or other invasive procedure performed on the wrong body part, or surgical or other invasive procedure performed on the wrong patient. [42 CFR 438.3(g); 42 CFR 438.6(a)(12)(i); 42 CFR 447.26(b)]

age 299 of 397	Date
-	

New Hampshire Department of Health and Human Services Medicaid Care Management Services



#### **Exhibit B - DRAFT**

- 4.16.1.12.2 The MCO shall require all Providers to report Provider Preventable Conditions associated with claims for payment or Member treatments for which payment would otherwise be made, in accordance with Exhibit O: Quality and Oversight Reporting Requirements.
  [42 CFR 438.3(g); 42 CFR 434.6(a)(12)(ii); 42 CFR 447.26(d)]
- 4.16.1.12.3 Any directed payments proposed to CMS shall be described in the program's actuarial certification for the rating period.
- 4.16.1.12.4 The MCO shall not impose an administrative fee, cost or any other charge upon any form of payment (e.g., electronic or paper checks) to Providers rendering Covered Services to Members.
- 4.16.1.12.5 The term "minimum fee schedule" in this Section 4.17 (Provider Payments), shall infer the minimum Provider reimbursement amount(s) permissible under the terms of this Agreement.

## 4.16.2 Provider Payment Requirements<sup>19</sup>

- 4.16.2.1 Ambulance, Stretcher, and Wheelchair Van Providers
  - 4.16.2.1.1 The MCO shall reimburse ambulance, stretcher, and wheelchair van Providers for Covered Services, as follows:
    - 4.16.2.1.1.1. For the rating period ending August 31, 2024 (subject to future rating period extension(s)), the MCO shall reimburse Participating Providers for all ambulance, stretcher, and wheelchair van Covered Services at no less than NH Medicaid fee schedule rates.

## 4.16.2.2 Birthing Centers

- 4.16.2.2.1 For the rating period ending August 31, 2024 (subject to future rating period extension(s)), the MCO shall reimburse Participating Provider hospital-based and free-standing birthing centers for Covered Services at no less than NH Medicaid fee schedule rates.
- 4.16.2.3 Community Mental Health Centers (CMHCs)

Page 300 of 397

Date \_\_\_\_\_

<sup>&</sup>lt;sup>19</sup> Provider payment requirements described in this section and throughout the Agreement are generally based on reimbursement terms that were either effective at the time the Agreement draft was prepared or as proposed under the procured contract; therefore, they are subject to change.

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

4.16.2.3.1 The MCO shall enter into an agreement with Community Mental Health Centers effective September 1, 2024. The agreement shall be defined by the Department and requires a monthly per member rate payment to the Community Mental Health Centers consistent with the directed payment and incentives (subject to CMS approval, as appropriate) for the treatment of Members with Severe/Persistent Mental Illness, Severe Mental Illness, Low Utilizers, Serious Emotionally Disturbed Children (SED and SED-I) as directed by the Department and detailed in the Department's Medicaid APM Strategy. This directed payment shall include an incentive pool to pay CMHCs for achieving quality outcomes established by the Department consistent with the statewide mental health improvement goals and objectives.

## 4.16.2.4 Critical Access Hospitals (CAHs)

- 4.16.2.4.1 The MCO shall remit directed payment(s) to CAHs in accordance with separate guidance, as follows:
- 4.16.2.4.2 For the current rating period ending August 31, 2024 (subject to future rating period extension(s)), directed payment amounts determined by DHHS shall comprise a uniform rate increase for all inpatient discharges and outpatient encounters as approved by CMS, including any alternate CMS-approved directed payment methodology. Qualified directed payments are tied to actual hospital services, including the number of inpatient discharges and outpatient visits reported by qualifying Providers.

## 4.16.2.5 DME Providers

- 4.16.2.5.1 The MCO shall reimburse DME Providers for DME and DME related supplies and services, as follows:
- 4.16.2.5.2 For the current rating period ending August 31, 2024 (subject to future rating period extension(s)), MCO provider reimbursement shall comprise payments at a minimum 80% of the DHHS FFS State Plan fee schedule as approved by CMS, including any alternate CMS-approved directed payment methodology.

### 4.16.2.6 Hospice Payment Rates

4.16.2.6.1 The Medicaid hospice payment rates shall be calculated based on the annual hospice rates

Page 301 of 397	Date	
-----------------	------	--

New Hampshire Department of Health and Human Services Medicaid Care Management Services



## **Exhibit B - DRAFT**

established under Medicare. These rates are authorized by 1814(i)(1)(ii) of the Social Security Act which also provides for an annual increase in payment rates for hospice care services.

#### 4.16.2.7 Indian Health Care Providers

- 4.16.2.7.1 The MCO shall pay IHCPs, whether Participating Providers or not, for Covered Services provided to American Indian Members who are eligible to receive services at a negotiated rate between the MCO and the IHCP or, in the absence of a negotiated rate, at a rate not less than the level and amount of payment the MCO would make for the services to a Participating Provider that is not an IHCP. [42 CFR 438.14(b)(2)(i-ii)]
- 4.16.2.7.2 For contracts involving IHCPs, the MCO shall meet the requirements of FFS timely payment for all I/T/U Providers in its network, including the paying of ninety-five percent (95%) of all Clean Claims within thirty (30) calendar days of the date of receipt; and paying ninety-nine percent (99%) of all Clean Claims within ninety (90) calendar days of the date of receipt. [42 CFR 438.14(b)(2)(iii); ARRA 5006(d); 42 CFR 447.45; 42 CFR 447.46; SMDL 10-001)]
- 4.16.2.7.3 IHCPs enrolled in Medicaid as FQHCs but not Participating Providers of the MCO shall be paid an amount equal to the amount the MCO would pay an FQHC that is a Participating Provider but is not an IHCP, including any supplemental payment from the Department to make up the difference between the amount the MCO pays and what the IIHCPs FQHC would have received under FFS. [42 CFR 438.14(c)(1)]
- 4.16.2.7.4 When an IHCP is not enrolled in Medicaid as a FQHC, regardless of whether it participates in the network of an MCO, it has the right to receive its applicable encounter rate published annually in the Federal Register by the IHS, or in the absence of a published encounter rate, the amount it would receive if the services were provided under the Medicaid State Plan's FFS payment methodology. [42 CFR 438.14(c)(2)]
- 4.16.2.7.5 When the amount the IHCP receives from the MCO is less than the amount the IHCP would have received under FFS or the applicable encounter rate

Page 302 of 397 Date \_\_\_\_\_

New Hampshire Department of Health and Human Services Medicaid Care Management Services



## **Exhibit B - DRAFT**

published annually in the Federal Register by the IHS, the Department shall make a supplemental payment to the IHCP to make up the difference between the amount the MCO pays and the amount the IHCP would have received under FFS or the applicable encounter rate. [42 CFR 438.14(c)(3)]

- 4.16.2.8 Private Duty Nursing Services
  - 4.16.2.8.1 The MCO shall reimburse private duty nursing agencies for private duty nursing services at least at the FFS rates established by the Department.
- 4.16.2.9 Substance Use Disorder Providers
  - 4.16.2.9.1 The MCO shall reimburse Substance Use Disorder Providers in accordance with rates that are no less than the equivalent DHHS FFS rates on the applicable Substance Use Disorder Provider fee schedule.
- 4.16.2.10 Transition Housing Program
  - 4.16.2.10.1 The MCO shall reimburse eligible Transition Housing Program services at least at the FFS rates established by the Department.
- 4.16.2.11 Designated Receiving Facility (DRF)
  - 4.16.2.11.1 The MCO shall reimburse eligible Medicaid enrolled DRFs as designated by the Commissioner, as follows:
    - 4.16.2.11.1.1. For the current rating period ending August 31, 2024 (subject to future rating period extension(s)), the MCO directed payment remittance to the Peer Group 06 providers shall comprise the minimum Peer Group 06 NH Medicaid State Plan DRG fee schedule payment amounts.
  - 4.16.2.11.2 For administrative days and post stabilization care services delivered under the inpatient and outpatient service categories, the MCO shall pay State-Owned Hospitals and other State determined IMDs for mental illness at rates no less than those paid by the NH Medicaid FFS program, inclusive of both State and federal share of the payment, if a Member cannot be discharged due to failure to provide appropriate community-based care and services. Administrative days and post stabilization care services are inpatient

Page 303 of 397 Date \_\_\_\_\_

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

hospital days associated with Members who no longer require acute care but are left in State-Owned Hospitals and other State determined IMDs for mental illness.

## 4.16.2.12 Neuropsychological Testing Services

- 4.16.2.12.1 The MCO shall reimburse eligible Medicaid-enrolled Providers for covered neuropsychological testing services, as follows:
  - 4.16.2.12.1.1. For the current rating period ending August 31, 2024 (subject to future rating period extension(s)), directed payment remittance shall comprise NH Medicaid minimum fee schedule amounts as approved by CMS, including any alternate CMS-approved directed payment methodology.
- 4.16.3 State-Owned Inpatient Psychiatric Hospitals
  - 4.16.3.1 The MCO shall reimburse state-owned New Hampshire Hospital and Hampstead Hospital as described below:
    - 4.16.3.1.1 For inpatient psychiatric services, the MCO shall reimburse state-owned New Hampshire Hospital and Hampstead Hospital, as follows:
      - 4.16.3.1.1.1. For the current rating period ending August 31, 2024 (subject to future rating period extension(s)), the state-owned facilities shall be reimbursed for inpatient psychiatric services at no less than the NH Medicaid uniform daily rate established and periodically adjusted by the Department of Health and Human Services Commissioner.
    - 4.16.3.1.2 For psychiatric professional services, the MCO shall reimburse psychiatric professional services delivered at the state-owned New Hampshire Hospital and Hampstead Hospital, as follows:
  - 4.16.3.2 For the current rating period ending August 31, 2024 (subject to future rating period extension(s)), directed payment amounts shall comprise minimum fee schedule payments at no less than the Medicare rates for eligible psychiatric professional services delivered in the state-owned facilities established and periodically adjusted by CMS.
  - 4.16.3.3 Qualifying Children's Hospitals

Page 304 of 397	Date
-----------------	------

New Hampshire Department of Health and Human Services Medicaid Care Management Services



## **Exhibit B - DRAFT**

- 4.16.3.3.1 The MCO shall remit directed payments to qualifying Children's Hospitals substantively serving NH Medicaid Members, in accordance with separate guidance, as follows:
  - 4.16.3.3.1.1. For the rating period ending August 31, 2024 (subject to future rating period extension(s)), directed payment amounts determined by DHHS shall comprise a uniform rate increase for all inpatient discharges and outpatient encounters for all qualifying children's hospitals.

## 4.17 Readiness Requirements Prior to Operations

#### 4.17.1 **General Requirements**

- 4.17.1.1 Prior to the Program Start Date, the MCO shall demonstrate to the Department's satisfaction its operational readiness and its ability to provide Covered Services to Members at the start of this Agreement in accordance with 42 CFR 438.66(d)(2), (d)(3), and (d)(4). [42 CFR 437.66(d)(1)(i).
- 4.17.1.2 The Readiness Review requirements shall apply to all MCOs regardless of whether they have previously contracted with the Department. [42 CFR 438.66(d)(1)(ii)]
- 4.17.1.3 The MCO shall accommodate Readiness desk and site Reviews, including documentation review and system demonstrations as defined by the Department.
- 4.17.1.4 The Readiness Review requirements shall apply to all MCOs, including those who have previously covered benefits to all eligibility groups covered under this Agreement. [42 CFR 438.66(d)(2), (d)(3) and (d)(4)]
- 4.17.1.5 In order to demonstrate its readiness, the MCO shall cooperate in the Readiness Review conducted by the Department.
- 4.17.1.6 If the MCO is unable to demonstrate its ability to meet the requirements of this Agreement, as determined solely by the Department, within the timeframes determined solely by the Department, then the Department shall have the right to terminate this Agreement in accordance with Section 7.1 (Termination for Cause).
- 4.17.1.7 The MCO shall participate in all the Department trainings in preparation for implementation of the Agreement.

#### 4.17.2 Emergency Response Plan/Disaster Recovery Plan

Page 305 of 397	Date
-----------------	------

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

- 4.17.2.1 The MCO shall submit an Emergency Response Plan to the Department for review prior to the Program Start Date, in compliance with the Exhibit Q IT Requirements Workbook.
- 4.17.2.2 The Emergency Response Plan shall address, at a minimum, the following aspects of pandemic preparedness and natural disaster response and recovery:
  - 4.17.2.2.1 Staff and Provider training;
  - 4.17.2.2.2 Essential business functions and key employees within the organization necessary to carry them out;
  - 4.17.2.2.3 Contingency plans for covering essential business functions in the event key employees are incapacitated or the primary workplace is unavailable;
  - 4.17.2.2.4 Communication with staff, Members, Providers, Subcontractors and suppliers when normal systems are unavailable:
  - 4.17.2.2.5 Plans to ensure continuity of services to Providers and Members:
  - 4.17.2.2.6 How the MCO shall coordinate with and support the Department and the other MCOs; and
  - 4.17.2.2.7 How the plan shall be tested, updated and maintained.
- 4.17.2.3 On an annual basis, or as otherwise specified in Exhibit O: Quality and Oversight Reporting Requirements, the MCO shall submit a certification of "no change" to the Emergency Response Plan or submit a revised Emergency Response Plan together with a redline reflecting the changes made since the last submission.

#### 4.18 Managed Care Information System

#### 4.18.1 **System Functionality**

- 4.18.1.1 The MCO shall have a comprehensive, automated, and integrated MCIS that:
  - 4.18.1.1.1 Complies with the Exhibit Q: IT Requirements Workbook;
  - 4.18.1.1.2 Collects, analyzes, integrates, and reports Confidential Data; [42 CFR 438.242(a)];
  - 4.18.1.1.3 Provides information on areas, including but not limited to utilization, claims, grievances and appeals [42 CFR 438.242(a)];
  - 4.18.1.1.4 Collects and maintains Confidential Data on Members and Providers, as specified in this

Page 306 of 397	Date

Medicaid Care Management Services Contract
New Hampshire Department of Health and Human Services
Medicaid Care Management Services



## **Exhibit B - DRAFT**

		Agreement and on all services furnished to Members,
		through an Encounter Confidential Data system [42 CFR 438.242(b)(2)];
	4.18.1.1.5	Is capable of meeting the requirements listed throughout this Agreement; and
	4.18.1.1.6	Is capable of providing all of the Confidential Data and information necessary for the Department to meet State and federal Medicaid reporting and information regulations.
4.18.1.2	as detailed i	MCIS shall be capable of submitting Encounter Data, n Section 5.1.3 (Encounter Data) of this Agreement. all provide for:
	4.18.1.2.1	Collection and maintenance of sufficient Member Encounter Confidential Data to identify the Provider who delivers any item(s) or service(s) to Members;
	4.18.1.2.2	Submission of Member Encounter Confidential Data to the Department at the frequency and level of detail specified by CMS and by the Department;
	4.18.1.2.3	Submission of all Member Encounter Confidential Data that NH is required to report to CMS; and
	4.18.1.2.4	Submission of Member Encounter Confidential Data to the Department in standardized ASC X12N 837 format, and other proprietary file layouts as defined by the Department. [42 CFR 438.242(c)(1-4); 42 CFR 438.818]
4.18.1.3	this Section Agreement, a errors or r	actors shall meet the same standards, as described in 4.18 (Managed Care Information System) of the as the MCO. The MCO shall be held responsible for noncompliance resulting from the action of a or with respect to its provided functions.
4.18.1.4	The MCO MC	CIS shall include, but not be limited to:
	4.18.1.4.1	Management of Recipient Demographic Eligibility and Enrollment and History;
	4.18.1.4.2	Management of Provider Enrollment and Credentialing;
	4.18.1.4.3	Benefit Plan Coverage Management, History, and Reporting;
	4.18.1.4.4	Eligibility Verification;
	4.18.1.4.5	Encounter Data;

Page 307 of 397 Date \_\_\_\_\_

Medicaid Care Management Services Contract
New Hampshire Department of Health and Human Services
Medicaid Care Management Services

4.18.1.5



## **Exhibit B - DRAFT**

4.18.1.4.6	Reference File Updates;
4.18.1.4.7	Service Authorization Tracking, Support and Management;
4.18.1.4.8	Third Party Coverage and Cost Avoidance Management;
4.18.1.4.9	Financial Transactions Management and Reporting;
4.18.1.4.10	Payment Management (Checks, electronic funds transfer (EFT), Remittance Advices, Banking);
4.18.1.4.11	Reporting (Ah hoc and Pre-Defined/Scheduled and On-Demand);
4.18.1.4.12	Call Center Management;
4.18.1.4.13	Claims Adjudication;
4.18.1.4.14	Claims Payments; and
4.18.1.4.15	QOS metrics.
Specific functions limited to, the	tionality related to the above shall include, but is not following:
4.18.1.5.1	The MCIS Membership management system shall have the capability to receive, update, and maintain NH's Membership files consistent with information provided by the Department;
4.18.1.5.2	The MCIS shall have the capability to provide daily updates of Membership information to subcontractors or Providers with responsibility for processing claims or authorizing services based on Membership information;
4.18.1.5.3	The MCIS's Provider file shall be maintained with detailed information on each Provider sufficient to support Provider enrollment and payment and also meet the Department's reporting and Encounter Confidential Data requirements;
4.18.1.5.4	The MCIS's claims processing system shall have the capability to process claims consistent with timeliness and accuracy requirements of a federal MMIS system;
4.18.1.5.5	The MCIS's Services Authorization system shall be integrated with the claims processing system;
4.18.1.5.6	The MCIS shall be able to maintain its claims history with sufficient detail to meet all Department reporting and encounter requirements;

Date \_\_\_\_\_ Page 308 of 397

New Hampshire Department of Health and Human Services Medicaid Care Management Services



#### **Exhibit B - DRAFT**

- 4.18.1.5.7 The MCIS's credentialing system shall have the capability to store and report on Provider specific Confidential Data sufficient to meet the Provider credentialing requirements, Quality Management, and Utilization Management Program Requirements;
- 4.18.1.5.8 The MCIS shall be bi-directionally linked to the other operational systems maintained by the Department, in order to ensure that Confidential Data captured in encounter records accurately matches Confidential Data in Member, Provider, claims and authorization files, and in order to enable Encounter Confidential Data to be utilized for Member profiling, Provider profiling, claims validation, Fraud, Waste and Abuse monitoring activities, quality improvement, and any other research and reporting purposes defined by the Department; and
- 4.18.1.5.9 The Encounter Confidential Data system shall have a mechanism in place to receive, process, and store the required data.
- 4.18.1.6 The MCO system shall be compliant with the requirements NPI, and transaction processing, including being able to process electronic Confidential Data interchange (EDI) transactions in the ASC 5010 format.
- 4.18.1.7 The MCO system shall be compliant with Section 6504(a) of the Affordable Care Act, which requires that State claims processing and retrieval systems are able to collect Confidential Data elements necessary to enable the mechanized claims processing and information retrieval systems in operation by the State to meet the requirements of Section 1903(r)(1)(F) of the Social Security Act. [42 CFR 438.242(b)(1)]
- 4.18.1.8 MCIS capability shall include, but not be limited to the following:
  - 4.18.1.8.1 Provider network connectivity to EDI and Provider portal systems;
  - 4.18.1.8.2 Documented scheduled down time and maintenance windows, as agreed upon by DHHS, for externally accessible systems, including telephony, web, Interactive Voice Response (IVR), EDI, and online reporting;
  - 4.18.1.8.3 The Department on-line web access to applications and Confidential Data required by the State to utilize agreed upon workflows, processes, and procedures (reviewed by the Department) to access, analyze, or utilize Confidential Data captured in the MCO

Page 309 of 397 Date \_\_\_\_\_

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

system(s)	and	to	perform	appropriate	reporting	and
operationa	al act	ivit	ies;			

- 4.18.1.8.4 The Department access to user acceptance testing (UAT) environment for externally accessible systems including websites and secure portals; and
- 4.18.1.8.5 Documented instructions and user manuals for each component.

## 4.18.1.9 Managed Care Information System Up-Time

- 4.18.1.9.1 Externally accessible systems, including telephone, web, IVR, EDI, and online reporting shall be available twenty-four (24) hours a day, seven (7) days a week, three-hundred-sixty-five (365) days a year, except for scheduled maintenance upon notification of and preapproval by the Department. The maintenance period shall not exceed four (4) consecutive hours without prior the Department approval.
- 4.18.1.9.2 MCO shall provide redundant telecommunication backups and ensure that interrupted transmissions shall result in immediate failover to redundant communications path as well as guarantee Confidential Data transmission is complete, accurate and fully synchronized with operational systems.

## 4.18.2 Information System Confidential Data Transfer

- 4.18.2.1 Effective communication between the MCO and the Department requires secure, accurate, complete, and auditable transfer of Confidential Data to/from the MCO and the Department Confidential Data management information systems. Elements of Confidential Data transfer requirements between the MCO and the Department management information systems shall include, but not be limited to:
  - 4.18.2.1.1 Department read access to all MCM Confidential Data in reporting databases where Confidential Data is stored, which includes all tools required to access the Confidential Data at no additional cost to the Department;
  - 4.18.2.1.2 Exchanges of Confidential Data between the MCO and the Department in a format and schedule as prescribed by the State, including detailed mapping specifications identifying the Confidential Data source and target;
  - 4.18.2.1.3 Secure (encrypted) communication protocols to provide timely notification of any Confidential Data file

	Page 310 of 397	Date
--	-----------------	------

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

retrieval, i	receipt, load, or send transmittal issues and
provide th	ne requisite analysis and support to identify
and resolv	ve issues according to the timelines set forth
by the Sta	ate;

- 4.18.2.1.4 Collaborative relationships with the Department, its MMIS fiscal agent, and other interfacing entities to effectively implement the requisite exchanges of Confidential Data necessary to support the requirements of this Agreement;
- 4.18.2.1.5 MCO implementation of the necessary telecommunication infrastructure and tools/utilities to support secure connectivity and access to the system and to support the secure, effective transfer of data;
- 4.18.2.1.6 Utilization of Confidential Data extract, transformation, and load (ETL) or similar methods for Confidential Data conversion and Confidential Data interface handling that, to the maximum extent possible, automate the ETL processes, and provide for source to target or source to specification mappings;
- 4.18.2.1.7 Mechanisms to support the electronic reconciliation of all Confidential Data extracts to source tables to validate the integrity of Confidential Data extracts; and
- 4.18.2.1.8 A given day's Confidential Data transmissions, as specified in this Section 4.19.2 (Information System Confidential Data Transfer) of the Agreement, are to be downloaded to the Department according to the schedule prescribed by the State. If errors are encountered in batch transmissions, reconciliation of transactions shall be included in the next batch transmission.
- 4.18.2.2 The MCO shall designate a single point of contact to coordinate Confidential Data transfer issues with the Department.
- 4.18.2.3 The Department shall provide for a Centralized Electronic Repository, providing for secure access to authorized MCO and the Department staff for project plans documentation, issues tracking, deliverables, and other project-related artifacts.
- 4.18.2.4 Confidential Data transmissions from the Department to the MCO shall include, but not be limited to the following:
  - 4.18.2.4.1 Provider Extract (Daily);
  - 4.18.2.4.2 Recipient Eligibility Extract (Daily);

Page 311 of 397	Date	
-----------------	------	--

Medicaid Care Management Services Contract
New Hampshire Department of Health and Human Services
Medicaid Care Management Services



## **Exhibit B - DRAFT**

	4.18.2.4.3	Recipient Eligibility Audit/Roster (Monthly);
	4.18.2.4.4	Medical and Pharmacy Service Authorizations (Daily);
	4.18.2.4.5	Medicare and Commercial Third Party Coverage (Daily);
	4.18.2.4.6	Claims History (Bi-Weekly); and
	4.18.2.4.7	Capitation Payment Confidential Data (Monthly).
4.18.2.5		Data transmissions from the MCO to the Department but not be limited to the following:
	4.18.2.5.1	Member Demographic changes (Daily);
	4.18.2.5.2	Member Primary Care Physician Selection (Daily);
	4.18.2.5.3	MCO Provider Network Confidential Data (Daily);
	4.18.2.5.4	Medical and Pharmacy Service Authorizations (Daily);
	4.18.2.5.5	Member Encounter Confidential Data including paid, denied, adjustment transactions by pay period (Weekly);
	4.18.2.5.6	Financial Transaction Confidential Data (Weekly); and
	4.18.2.5.7	Updates to Third Party Coverage Confidential Data (Weekly).
	4.18.2.5.8	Behavioral Health Certification Data (Monthly).
4.18.2.6		all provide Department staff with access to timely and onfidential Data and shall meet the following ::
	4.18.2.6.1	All exchanges of Confidential Data between the MCO and the Department shall be in a format, file record layout, and scheduled as prescribed by the Department;
	4.18.2.6.2	The MCO shall work collaboratively with the Department, the Department's MMIS fiscal agent, the NH Department of Information Technology, and other interfacing entities to implement effectively the requisite exchanges of Confidential Data necessary to support the requirements of this Agreement;
	4.18.2.6.3	The MCO shall implement the necessary telecommunication infrastructure to support the MCIS and shall provide the Department with a network

Page 312 of 397 Date \_\_\_\_\_

4.18.3 Systems Operation

New Hampshire Department of Health and Human Services Medicaid Care Management Services



## **Exhibit B - DRAFT**

		diagram depicting the MCO's communications infrastructure, including but not limited to connectivity between the Department and the MCO, including any MCO/Subcontractor locations supporting the NH program;
	4.18.2.6.4	The MCO shall provide support to the Department and its fiscal agent to prove the validity, integrity and reconciliation of its data, including Encounter Data; and
	4.18.2.6.5	The MCO shall be responsible for correcting Confidential Data extract errors in a timeline set forth by the Department as outlined within this Agreement.
System	s Operation a	and Support
4.18.3.1	Systems ope	erations and support shall include, but not be limited to:
	4.18.3.1.1	On-call procedures and contacts;
	4.18.3.1.2	Job scheduling and failure notification documentation;
	4.18.3.1.3	Secure (encrypted) Confidential Data transmission and storage methodology;
	4.18.3.1.4	Interface acknowledgements and error reporting;
	4.18.3.1.5	Technical issue escalation procedures;
	4.18.3.1.6	Business and Member notification;
	4.18.3.1.7	Change control management;
	4.18.3.1.8	Assistance with UAT and implementation coordination;
	4.18.3.1.9	Documented Confidential Data interface specifications – Confidential Data imported and extracts exported including database mapping specifications;
	4.18.3.1.10	Journaling and internal backup procedures, for which facility for storage shall be class 3 compliant; and
	4.18.3.1.11	Communication and Escalation Plan that fully outlines the steps necessary to perform notification and monitoring of events including all appropriate contacts and timeframes for resolution by severity of

4.18.3.2 The MCO shall be responsible for implementing and maintaining necessary telecommunications and network infrastructure to support the MCIS and shall provide:

the event.

Date \_\_\_\_\_ Page 313 of 397

New Hampshire Department of Health and Human Services Medicaid Care Management Services



## **Exhibit B - DRAFT**

recipients.

- 4.18.3.2.1 Network diagram that fully defines the topology of the MCO's network;
  4.18.3.2.2 DHHS/MCO connectivity;
  4.18.3.2.3 Any MCO/Subcontractor locations requiring MCIS access/support; and
  4.18.3.2.4 Web access for the Department staff, Providers and
- 4.18.3.3 The MCO shall utilize either its own or the State's open model Electronic Visit (EVV) system as prescribed by the Department in separate guidance for all Medicaid personal care services and home health Covered Services that require an in-home visit by a Provider in accordance with Section 12006(a) of the 21st Century Cures Act. This applies to personal care services provided under sections 1905(a)(24), 1915(c), 1915(i), 1915(j), 1915(k), and Section 1115; and HHCS provided under 1905(a)(7) of the Social Security Act or a waiver, as applicable.

#### 4.18.4 Ownership and Access to Systems and Data

- 4.18.4.1 The MCO shall make available to the Department and, upon request, to CMS all collected data. [42 CFR 438.242(b)(4)]
- 4.18.4.2 Confidential Data accumulated, as part of the MCM program shall remain the property of the State.
- 4.18.4.3 The MCO shall provide the Department with system reporting capabilities that shall include access to pre-designed and agreed-upon scheduled reports, as well as the ability to respond promptly to ad-hoc requests to support the Department Confidential Data and information needs.
- 4.18.4.4 The Department acknowledges the MCO's obligations to appropriately protect Confidential Data and system performance, and the parties agree to work together to ensure the Department information needs can be met while minimizing risk and impact to the MCO's systems.

## 4.18.5 Records Retention

- 4.18.5.1 The MCO shall retain, preserve, and make available upon request all records relating to the performance of its obligations under the Agreement, including paper and electronic claim forms, for a period of not less than ten (10) years from the date of termination of this Agreement.
- 4.18.5.2 Records involving matters that are the subject of litigation shall be retained for a period of not less than ten (10) years following the termination of litigation.

Page 314 of 397	Date

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

- 4.18.5.3 Certified protected electronic copies of the documents contemplated herein may be substituted for the originals with the prior written consent of the Department, if the Department approves the electronic imaging procedures as reliable and supported by an effective retrieval system.
- 4.18.5.4 Upon expiration of the ten (10) year retention period and upon request, the subject records shall be transferred to the Department's possession, refer to the End of Contract Transition Services section for additional requirements.
- 4.18.5.5 No records shall be destroyed or otherwise disposed of without the prior written consent of the Department.

## 4.18.6 Web Access and Use by Providers and Members

- 4.18.6.1 The MCIS shall include web access for use by and support to Participating Providers and Members.
- 4.18.6.2 The services shall be provided at no cost to the Participating Provider or Members.
- 4.18.6.3 All costs associated with the development, security, and maintenance of these websites shall be the responsibility of the MCO.
- 4.18.6.4 The MCO shall create secure web access for Medicaid Providers and Members and authorized the Department staff to access case-specific information; this web access shall fulfill the following requirements, and shall be available no later than the Program Start Date:
  - 4.18.6.4.1 Providers shall have the ability to electronically submit service authorization requests and access and utilize other Utilization Management tools;
  - 4.18.6.4.2 Providers and Members shall have the ability to download and print any needed Medicaid MCO program forms and other information:
  - 4.18.6.4.3 Providers shall have an option to e-prescribe without electronic medical records or hand held devices;
  - 4.18.6.4.4 The MCO shall support Provider requests and receive general program information with contact information for phone numbers, mailing, and e-mail address(es);
- 4.18.6.5 Providers shall have access to drug information;
  - 4.18.6.5.1 The website shall provide an encrypted e-mail link to the MCO to permit Providers and Members or other interested parties to e-mail inquiries or comments.

age 315 of 397	Date

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

	4.18.6.5.2	The website shall provide a link to the State's Medicaid website;
	4.18.6.5.3	Audit logs shall be maintained reflecting access to the system and random audits shall be conducted; and
	4.18.6.5.4	Access shall be limited to verified users.
4.18.6.6	system, and	hall manage Provider and Member access to the operational services necessary to assist Providers and h gaining access and utilizing the web portal.
4.18.6.7	System Supp	oort Performance Standards shall include:
	4.18.6.7.1	Email inquiries-one (1) business day response;
	4.18.6.7.2	New information posted within one (1) business day of receipt, and up to two (2) business days of receipt for materials that shall be made ADA compliant with Section 508 of the Rehabilitation Act;
	4.18.6.7.3	Routine maintenance;
	4.18.6.7.4	Standard reports regarding portal usage such as hits per month by Providers/Members, number, and types of inquiries and requests, and email response statistics as well as maintenance reports; and
	4.18.6.7.5	Website user interfaces shall be ADA compliant with Section 508 of the Rehabilitation Act and support all major browsers (i.e. Chrome, MS Edge, Firefox, Safari, etc.). If user does not have compliant browser, MCO shall redirect user to site to install appropriate browser.

## 4.18.7 Contingency Plans and Quality Assurance

- 4.18.7.1 Critical systems within the MCIS support the delivery of critical medical services to Members and reimbursement to Providers. As such, contingency plans shall be developed and tested to ensure continuous operation of the MCIS.
- 4.18.7.2 The MCO shall host the MCIS at the MCO's data center, and provide for adequate redundancy, disaster recovery, and business continuity such that in the event of any catastrophic incident, system availability is restored to NH within twenty-four (24) hours of incident onset.
- 4.18.7.3 Archiving processes shall not modify the Confidential Data composition of the Department's records, and archived Confidential Data shall be retrievable at the request of the Department. Archiving shall be conducted at intervals agreed upon between the MCO and the Department.

age 316 of 397	Date
----------------	------

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

- 4.18.7.4 The MCIS shall be able to accept, process, and generate HIPAA compliant electronic transactions as requested, transmitted between Providers, Provider billing agents/clearing houses, or the Department and the MCO.
- 4.18.7.5 Audit logs of activities shall be maintained and periodically reviewed to ensure compliance with Exhibit G: IT Requirements Workbook and security and access rights granted to users.
- 4.18.7.6 In accordance with Exhibit O: Quality and Oversight Reporting Requirements, the MCO shall submit the following documents and corresponding checklists for the Departments Information Security review:
  - 4.18.7.6.1 Disaster Recovery Plan;
  - 4.18.7.6.2 Business Continuity Plan;
  - 4.18.7.6.3 Security Plan:
  - 4.18.7.6.4 The following documents which, if after the original documents are submitted the MCO makes modifications to them, the revised redlined documents and any corresponding checklists shall be submitted for Department review:
    - 4.18.7.6.4.1. Risk Management Plan;
    - 4.18.7.6.4.2. Systems Quality Assurance Plan; and
    - 4.18.7.6.4.3. Confirmation of 5010 compliance and Companion Guides.
- 4.18.7.7 Management of changes to the MCIS is critical to ensure uninterrupted functioning of the MCIS. The following elements, at a minimum, shall be part of the MCO's change management process:
  - 4.18.7.7.1 The complete system shall have proper configuration management/change management in place (to be reviewed by the Department).
  - 4.18.7.7.2 The MCO system shall be configurable to support timely changes to benefit enrollment and benefit coverage or other such changes.
  - 4.18.7.7.3 The MCO shall provide the Department with written notice of major systems changes and implementations no later than ninety (90) calendar days prior to the planned change or implementation, including any changes relating to Subcontractors, and specifically identifying any change impact to the Confidential Data interfaces or transaction

Page 317 of 397 Date \_\_\_\_\_

4.18.7.7.4

4.18.7.7.5

New Hampshire Department of Health and Human Services **Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

and/or the fiscal agent.

nature of the request from the MCO.

		at least one (1) business day.
	4.18.7.7.6	The MCO shall provide the Department with official points of contact for MCIS issues on an ongoing basis.
	4.18.7.7.7	The MCO shall ensure appropriate testing is done for all system changes. MCO shall also provide a test system for the Department to monitor changes in externally facing applications (i.e. NH websites). This test site shall contain no actual PHI Confidential Data of any Member.
	4.18.7.7.8	The MCO shall make timely changes or defect fixes to Confidential Data interfaces and execute testing with the Department and other applicable entities to validate the integrity of the interface changes.
4.18.7.8	The Department, or its agent, may conduct a Systems readiness review to validate the MCO's ability to meet the MCIS requirements.	
4.18.7.9	The System readiness review may include a desk review and/or an onsite review. If the Department determines that it is necessary to conduct an onsite review, the MCO shall be responsible for all reasonable travel costs associated with such onsite reviews for at least two (2) staff from the Department.	
4.18.7.10	For purposes of this Section of the Agreement, "reasonable travel costs" include airfare, lodging, meals, car rental and fuel, taxi, mileage, parking, and other incidental travel expenses incurred by the Department or its authorized agent in connection with the onsite reviews.	
4.18.7.11	If for any reason the MCO does not fully meet the MCIS requirements, the MCO shall, upon request by the Department, either correct such deficiency or submit to the Department a CAP and Risk Mitigation Plan to address such deficiency. Immediately	

exchanges between the MCO and the Department

The Department retains the right to modify or waive the notification requirement contingent upon the

The MCO shall provide the Department with updates to the MCIS organizational chart and the description of MCIS responsibilities at least thirty (30) calendar days prior to the effective date of the change, except where personnel changes were not foreseeable in such period, in which case notice shall be given within

Page 318 of 397 Date \_\_\_\_

upon identifying a deficiency, the Department may impose

New Hampshire Department of Health and Human Services Medicaid Care Management Services



## **Exhibit B - DRAFT**

contractual remedies according to the severity of the deficiency as described in Section 5.5 (Remedies) of this Agreement.

#### 4.18.7.12 QOS metrics shall include:

- 4.18.7.12.1 The security of the Care Management processing system shall minimally provide the following three types of controls to maintain Confidential Data integrity that directly impacts QOS. These controls shall be in place at all appropriate points of processing:
  - 4.18.7.12.1.1. Preventive Controls: controls designed to prevent errors and unauthorized events from occurring;
  - 4.18.7.12.1.2. Detective Controls: controls designed to identify errors and unauthorized transactions that have occurred in the system; and
  - 4.18.7.12.1.3. Corrective Controls: controls to ensure that the problems identified by the detective controls are corrected.
- 4.18.7.12.2 System Administration: Ability to comply with HIPAA, ADA, and other State and federal regulations, and perform in accordance with Agreement terms and conditions, ability to provide a flexible solution to effectively meet the requirements of upcoming HIPAA regulations and other national standards development.
- 4.18.7.12.3 The system shall accommodate changes with global impacts (e.g., implementation of electronic health record, e-Prescribe) as well as new transactions at no additional cost.

## 4.18.8 Interoperability and Patient Access

4.18.8.1 The MCO shall comply with the Centers for Medicare & Medicaid Services published final rule, "Interoperability and Patient Access for Medicare Advantage Organization and Medicaid Managed Care Plans, State Medicaid Agencies, CHIP Agencies and CHIP Managed Care Entities, Issuers of Qualified Health Plans on the Federally-Facilitated Exchanges, and Health Care Providers," (referred to as the "CMS Interoperability and Patient Access final rule") to further advance interoperability for Medicaid and Children's Health Insurance Program (CHIP) providers and improve beneficiaries' access to their data.

Page 319 of 397 Date \_\_\_\_\_

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

4.18.8.2	The MCO shall implement this final rule in a manner consistent witl			
	existing guidance and the published "21st Century Cures Act:			
	Interoperability, Information Blocking, and the ONC Health IT			
	Certification Program" final rule (referred to as the ONC 21st			
	Century Cures Act final rule), including:			

- 4.18.8.2.1 Patient Access Application Program Interfaces (API). [42 CFR 438.242(b)(5); 42 CFR 457.1233(d); 85 Fed. Reg. 25,510-25, 640 (May 1, 2020); 85 Fed. Reg. 25,642-25, 961 (May 1, 2020)];
- 4.18.8.2.2 Provider Directory Application Program Interfaces (API). [42 CFR 438.242(b)(6); 85 Fed. Reg. 25,510-25, 640 (May 1, 2020); 85 Fed. Reg. 25,642-25, 961 (May 1, 2020)]; and
- 4.18.8.2.3 Implement and maintain a Payer-to-Payer Confidential Data Exchange. [42 CFR 438.62(b)(1)(vi-vii); 85 Fed. Reg. 25,510-25, 640 (May 1, 2020); 85 Fed. Reg. 25,642-25, 961 (May 1, 2020)].
- 4.18.8.3 The MCO shall implement an Application Programming Interface (API) that meets the criteria specified at 42 CFR 431.60, and include(s):
  - 4.18.8.3.1 Confidential Data concerning adjudicated claims, including claims Confidential Data for payment decisions that may be appealed, were appealed, or are in the process of appeal, and provider remittances and beneficiary cost-sharing pertaining to such claims, no later than one (1) business day after a claim is processed;
  - 4.18.8.3.2 Encounter data, including encounter Confidential Data from any network providers the MCO is compensating on the basis of capitation payments and adjudicated claims and encounter Confidential Data from any Subcontractors no later than one (1) business day after receiving the Confidential Data from providers; and
  - 4.18.8.3.3 Clinical data, including laboratory results, if the MCO maintains any such data, no later than one (1) business day after the Confidential Data is received by the State.
  - 4.18.8.3.4 Information about covered outpatient drugs and updates to such information, including, where applicable, preferred drug list information, no later than one (1) business day after the effective date of

than one (1) business da	y after the effective date t
Page 320 of 397	Date
_	

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

any such information or updates to such information. [42 CFR 438.242(b)(5); 42 CFR 457.1233(d)(2)]

4.18.8.4 The MCO shall implement and maintain a publicly accessible standards-based API as described in 42 CFR 431.70, which must include all of the provider directory information specified in 42 CFR 438.10(h)(1) and (2). [42 CFR 438.242(b)(6); 42 CFR 457.1233(d)]

## 4.19 Claims Quality Assurance Standards

#### 4.19.1 Claims Payment Standards

- 4.19.1.1 For purposes of this Section 4.20 (Claims Quality Assurance Standards), the Department has adopted the claims definitions established by CMS. [42 CFR 447.25(b)]
  - 4.19.1.1.1 "Clean Claim" as defined in Section 2.1 (Definitions); and
  - 4.19.1.1.2 "Incomplete Claim" means a claim that is rejected for the purpose of obtaining additional information from the Provider.
  - 4.19.1.1.3 Claims payment timeliness shall be measured from the received date, which is the date a paper claim is received in the MCO's mailroom by its date stamp or the date an electronic claim is submitted.
- 4.19.1.2 The paid date is the date a payment check or EFT is issued to the service Provider [42 CFR 447.45(d)(5-6); 42 CFR 447.46; sections 1932(f) and 1902(a)(37)(A) of the Act]
- 4.19.1.3 The denied date is the date at which the MCO determines that the submitted claim is not eligible for payment.
- 4.19.1.4 The MCO shall pay or deny ninety-five percent (95%) of Clean Claims within thirty (30) calendar days of receipt, or receipt of additional information.
- 4.19.1.5 The MCO shall pay ninety-nine percent (99%) of Clean Claims within ninety (90) calendar days of receipt. [42 CFR 447.46; 42 CFR 447.45(d)(2-3) and (d)(5-6); Sections 1902(a)(37)(A) and 1932(f) of the Social Security Act].
- 4.19.1.6 The MCO shall request all additional information necessary to process Incomplete Claims from the Provider within thirty (30) calendar days from the date of original claim receipt.

#### 4.19.2 Claims Quality Assurance Program

4.19.2.1 The MCO shall verify the accuracy and timeliness of Confidential Data reported by Providers, including Confidential Data from

age 321 of 397	Date

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

- Participating Providers the MCO is compensating through a capitated payment arrangement.
- 4.19.2.2 The MCO shall screen the Confidential Data received from Providers for completeness, logic, and consistency [42 CFR 438.242(b)(3)(i)-(ii)].
- 4.19.2.3 The MCO shall maintain an internal program to routinely measure the accuracy of claims processing for MCIS and report results to the Department, in accordance with Exhibit O: Quality and Oversight Reporting Requirements.
- 4.19.2.4 As indicated in Exhibit O: Quality and Oversight Reporting Requirements, reporting to the Department shall be based on a review of a statistically valid sample of paid and denied claims determined with a ninety-five percent (95%) confidence level, +/-three percent (3%), assuming an error rate of three percent (3%) in the population of managed care claims.
- 4.19.2.5 The MCO shall implement CAPs to identify any issues and/or errors identified during claim reviews and report resolution to the Department.

## 4.19.3 Claims Financial Accuracy

- 4.19.3.1 Claims financial accuracy measures the accuracy of dollars paid to Providers. It is measured by evaluating dollars overpaid and underpaid in relation to total paid amounts taking into account the dollar stratification of claims.
- 4.19.3.2 The MCO shall pay ninety-nine percent (99%) of dollars accurately.

#### 4.19.4 Claims Payment Accuracy

- 4.19.4.1 Claims payment accuracy measures the percentage of claims paid or denied correctly. It is measured by dividing the number of claims paid/denied correctly by the total claims reviewed.
- 4.19.4.2 The MCO shall pay ninety-seven percent (97%) of claims accurately.

#### 4.19.5 Claims Processing Accuracy

- 4.19.5.1 Claims processing accuracy measures the percentage of claims that are accurately processed in their entirety from both a financial and non-financial perspective; i.e., claim was paid/denied correctly and all coding was correct, business procedures were followed, etc. It is measured by dividing the total number of claims processed correctly by the total number of claims reviewed.
- 4.19.5.2 The MCO shall process ninety-five percent (95%) of all claims correctly.

age 322 of 397	Date

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



## **Exhibit B - DRAFT**

#### 5 OVERSIGHT AND ACCOUNTABILITY

#### 5.1 Reporting

#### 5.1.1 **General Provisions**

- 5.1.1.1 As indicated throughout this Agreement, the Department shall document ongoing MCO reporting requirements through Exhibit O: Quality and Oversight Reporting Requirements and additional specifications provided by the Department.
- 5.1.1.2 The MCO shall provide data, reports, and plans in accordance with Exhibit O: Quality and Oversight Reporting Requirements, this Agreement, and any additional specifications provided by the Department.
- 5.1.1.3 The MCO shall comply with all NHID rules for Confidential Data reporting, including those related to the NH CHIS, NH code of Administrative Rule, Chapter Ins 4000. Failure to submit timely, accurate, and/or complete files to the NH CHIS shall be subject to liquidated damages as described in Section 5.5.2 (Liquidated Damages).
- 5.1.1.4 For all historical files submitted under NH Code of Administrative Rule, Chapter Ins 4000 after the submission start date, if DHHS or NHID notifies the MCO of not meeting compliance, the MCO shall remediate all related files within forty-five (45) calendar days after such notice.
- 5.1.1.5 If the MCO fails to comply with either error resolution timeline, DHHS shall require a CAP and assess liquidated damages as described in Section 5.5.2 (Liquidated Damages).
- 5.1.1.6 The MCO shall make all collected Confidential Data available to the Department upon request and upon the request of CMS. [42 CFR 438.242(b)(4)]
- 5.1.1.7 The MCO shall collect Confidential Data on Member and Provider characteristics as specified by the Department and on services furnished to Members through a MCIS system or other methods as may be specified by the Department. [42 CFR 438.242(b)(2)]
- 5.1.1.8 The MCO shall ensure that Confidential Data received from Providers are accurate and complete by:
  - 5.1.1.8.1 Verifying the accuracy and timeliness of reported data;
  - 5.1.1.8.2 Screening the Confidential Data for completeness, logic, and consistency; and

age 323 of 397	Date

Medicaid Care Management Services Contract
New Hampshire Department of Health and Human Services
Medicaid Care Management Services



## **Exhibit B - DRAFT**

	5.1.1.8.3	Collecting service information in standardized formats to the extent feasible and appropriate. [42 CFR 438.242(b)(3)]
5.1.1.9	provide, the throughout of Oversight	nent shall at a minimum collect, and the MCO shall following information, and the information specified the Agreement and within Exhibit O: Quality and Reporting Requirements, in order to improve the of the MCM program [42 CFR 438.66(c)(1)-(2) and
	5.1.1.9.1	Enrollment and disenrollment data;
	5.1.1.9.2	Member grievance and appeal logs;
	5.1.1.9.3	Medical management committee reports and minutes;
	5.1.1.9.4	Audited financial and encounter data;
	5.1.1.9.5	The MLR summary reports;
	5.1.1.9.6	Customer service performance data;
	5.1.1.9.7	Performance on required quality measures; and
	5.1.1.9.8	The MCO's QAPI Plan.
5.1.1.10	presenting to report that	shall be responsible for preparing, submitting, and to the Governor, Legislature, and the Department a includes the following information, or information dicated by the State:
	5.1.1.10.1	A description of how the MCO has addressed State priorities for the MCM Program, including those specified in RSA 126-AA, throughout this Agreement, and in other State statute, policies, and guidelines;
	5.1.1.10.2	A description of the innovative programs the MCO has developed and the outcomes associated with those programs;
	5.1.1.10.3	A description of how the MCO is addressing health- related social needs and the outcomes associated with MCO-implemented interventions;
	5.1.1.10.4	A description of how the MCO is improving health outcomes in the State; and
	5.1.1.10.5	Any other information indicated by the State for inclusion in the annual report.
5.1.1.11	Prior to Pro	ogram Start Date and at any other time upon the

Page 324 of 397 Date \_\_\_\_\_

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

Department shall conduct a review of MCO policies and procedures and/or other administrative documentation.

- 5.1.1.11.1 The Department shall deem materials as pass or fail following the Department review.
- 5.1.1.11.2 The MCO shall complete and submit a Department-developed attestation that attests that the policy, procedure or other documentation satisfies all applicable State and federal authorities.
- 5.1.1.11.3 The Department may require modifications to MCO policies and procedures or other documentation at any time as determined by the Department.
- 5.1.1.12 The MCO shall submit all required data to meet CMS standards for submission to the Transformed Medicaid Statistical Information System.

### 5.1.2 Requirements for Waiver Programs

- 5.1.2.1 The MCO shall provide to the Department the Confidential Data and information required for its current CMS waiver programs and any waiver programs it enters during the Term of this Agreement that require Confidential Data for Members covered by the MCO. These include but are not limited to:
  - 5.1.2.1.1 Substance Use Disorder and Severe Mental Illness Institute for Mental Disease 1115 waiver;
  - 5.1.2.1.2 Mandatory managed care 1915b waiver; and
  - 5.1.2.1.3 Granite Advantage 1115 waiver.

#### 5.1.3 Encounter Data

- 5.1.3.1 The MCO shall submit Encounter Confidential Data in the format and content, timeliness, completeness, and accuracy as specified by the Department and in accordance with timeliness, completeness, and accuracy standards as established by the Department. [42 CFR 438.604(a)(1); 42 CFR 438.606; 42 CFR 438.818]
- 5.1.3.2 All MCO encounter requirements apply to all Subcontractors. The MCO shall ensure that all contracts with Participating Providers and Subcontractors contain provisions that require all encounter records are reported or submitted in an accurate and timely fashion such that the MCO meets all Department reporting requirements.
- 5.1.3.3 The MCO shall submit to the Department for review, during the Readiness Review process, its policies and procedures that detail the MCO's encounter process. The MCO-submitted policies and

age 325 of 397	Date

Medicaid Care Management Services Contract
New Hampshire Department of Health and Human Services
Medicaid Care Management Services

5.1.3.4

5.1.3.5

5.1.3.6

5.1.3.7

5.1.3.8

5.1.3.9



# **Exhibit B - DRAFT**

procedures satisfaction:	shall at minimum include to the Department's
5.1.3.3.1	An end-to-end description of the MCO's encounter process;
5.1.3.3.2	Encounter specific source to target mapping detail that traces the inbound provider claim, in the applicable format, to the MCO's encounter system data storage location. The MCO shall provide the level of detail for each transmission of the source data that is used to create the encounter files that are submitted to DHHS;
5.1.3.3.3	A detailed overview of the encounter process with all Providers and Subcontractors; and
5.1.3.3.4	A detailed description of the internal reconciliation process followed by the MCO, and all Subcontractors that process claims on the MCO's behalf.
to and revise	nall, as requested by the Department, submit updates e upon request its policies and procedures that detail encounter process.
Department	er Confidential Data shall remain the property of the and the Department retains the right to use it for any eems necessary.
and the D Department's (Encounter	nall submit Encounter Confidential Data to the EQRO Department or its designated vendor upon the s request in accordance with this Section 5.1.3 Data) of the Agreement and to the Department's requested, according to the format and specification ries.
not eliminate Administrativ	of Encounter Confidential Data to the Department does the MCO's responsibility to comply with N.H. Code of Performing Reporting System are Claims Confidential Data Sets.
	nall ensure that encounter records are consistent with ent requirements and all applicable State and federal
	inters shall include all adjudicated claims, including , and adjusted claims.
5.1.3.9.1	The MCO shall submit claim and claim line denial reason codes in the level of detail and format as specified by the Department.

Page 326 of 397 Date \_\_\_\_\_

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



### **Exhibit B - DRAFT**

- 5.1.3.10 The level of detail associated with encounters from Providers with whom the MCO has a capitated payment arrangement shall be the equivalent to the level of detail associated with encounters for which the MCO received and settled a FFS claim.
- 5.1.3.11 The MCO shall maintain a record of all information submitted by Providers on claims. All Provider-submitted claim information shall be submitted in the MCO's encounter records.
- 5.1.3.12 The MCO shall have a computer and Confidential Data processing system, and staff, sufficient to accurately produce the data, reports, and encounter record set in formats and timelines as defined in this Agreement.
- 5.1.3.13 The System shall be capable of following or tracing an encounter within its system using a unique encounter record identification number for each encounter.
- 5.1.3.14 The MCO shall collect service information in the federally mandated HIPAA transaction formats and code sets, and submit these Confidential Data in a standardized format approved by the Department.
- 5.1.3.15 The MCO shall make all collected Confidential Data available to the Department after it is tested for compliance, accuracy, completeness, logic, and consistency.
- 5.1.3.16 The MCO's systems that are required to use or otherwise contain the applicable Confidential Data type shall conform to current and future HIPAA-based standard code sets; the processes through which the Confidential Data are generated shall conform to the same standards, including application of:
  - 5.1.3.16.1 Health Care Common Procedure Coding System (HCPCS);
  - 5.1.3.16.2 CPT codes;
  - 5.1.3.16.3 International Classification of Diseases, 10th revision, Clinical Modification ICD-10-CM and International Classification of Diseases, 10th revision, Procedure Coding System ICD-10-PCS;
  - 5.1.3.16.4 National Drug Codes which is a code set that identifies the vendor (manufacturer), product and package size of all drugs and biologics recognized by the FDA. It is maintained and distributed by HHS, in collaboration with drug manufacturers;
  - 5.1.3.16.5 Code on Dental Procedures and Nomenclature (CDT) which is the code set for dental services. It is

Page 327 of 397	Date

5.1.3.16.6

said methods.

New Hampshire Department of Health and Human Services Medicaid Care Management Services



### **Exhibit B - DRAFT**

Association (ADA);

		codes used throughout the health care industry;
	5.1.3.16.7	Claim Adjustment Reason Codes (CARC) which explain why a claim payment is reduced. Each CARC is paired with a dollar amount, to reflect the amount of the specific reduction, and a Group Code, to specify whether the reduction is the responsibility of the Provider or the patient when other insurance is involved; and
	5.1.3.16.8	Reason and Remark Codes (RARC) which are used when other insurance denial information is submitted to the MMIS using standard codes defined and maintained by CMS and the NCPDP.
5.1.3.17	Department transaction of formats (P – the Department formats (P – services, in the transaction)	or the State's fiscal agent in the standard HIPAA formats, namely the ANSI X12N 837 transaction Professional and I - Institutional) or at the discretion of ent the ANSI X12N 837 post adjudicated transaction Professional and I - Institutional) and, for pharmacy ne NH file format, and other proprietary file layouts as e Department.
5.1.3.18	FFS equivale	ounters shall be submitted with MCO paid amount, the ent, and, as applicable, the Medicare paid amount, ce paid amount and/or expected Member Copayment
5.1.3.19	Confidential amount paid	nount (or FFS equivalent) submitted with Encounter Data shall be the amount paid to Providers, not the I to MCO Subcontractors or Providers of shared in the MCO's organization, third party administrators, entities.
5.1.3.20	MCO paid ar and exclude	nent means that, for example for pharmacy claims, the mount shall include the amount paid to the pharmacy any and all fees paid by the MCO to the Pharmacy ager. The amount paid to the MCO's PBM is not
5.1.3.21		nall continually provide up to date documentation of

maintained and distributed by the American Dental

POS Codes which are two-digit codes placed on

health care professional claims to indicate the setting in which a service was provided. CMS maintains POS

Page 328 of 397 Date \_\_\_\_\_

payment methods used for all types of services by date of use of

New Hampshire Department of Health and Human Services Medicaid Care Management Services



### **Exhibit B - DRAFT**

- 5.1.3.22 The MCO shall continually provide up to date documentation of claim adjustment methods used for all types of claims by date of use of said methods.
- 5.1.3.23 The MCO shall collect, and submit to the State's fiscal agent, Member service level Encounter Confidential Data for all Covered Services.
- 5.1.3.24 The MCO shall be held responsible for errors or non-compliance resulting from its own actions or the actions of an agent authorized to act on its behalf.
- 5.1.3.25 The MCO shall conform to all current and future HIPAA-compliant standards for information exchange, including but not limited to the following requirements:
  - 5.1.3.25.1 Batch and Online Transaction Types are as follows:
    - 5.1.3.25.1.1. ASC X12N 820 Premium Payment Transaction;
    - 5.1.3.25.1.2. ASC X12N 834 Enrollment and Audit Transaction:
    - 5.1.3.25.1.3. ASC X12N 835 Claims Payment Remittance Advice Transaction;
    - 5.1.3.25.1.4. ASC X12N 837I Institutional Claim/Encounter Transaction;
    - 5.1.3.25.1.5. ASC X12N 837P Professional Claim/Encounter Transaction;
    - 5.1.3.25.1.6. ASC X12N 837D Dental Claim/Encounter Transaction; and
    - 5.1.3.25.1.7. NCPDP D.0 Pharmacy Claim/Encounter Transaction.
  - 5.1.3.25.2 Online transaction types are as follows:
    - 5.1.3.25.2.1. ASC X12N 270/271 Eligibility/Benefit Inquiry/Response;
    - 5.1.3.25.2.2. ASC X12N 276 Claims Status Inquiry;
    - 5.1.3.25.2.3. ASC X12N 277 Claims Status Response;
    - 5.1.3.25.2.4. ASC X12N 278/279 Utilization Review Inquiry/Response; and
    - 5.1.3.25.2.5. NCPDP D.0 Pharmacy Claim/Encounter Transaction.

Page 329 of 397	Date

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

- 5.1.3.26 Submitted Encounter Confidential Data shall include all elements specified by the Department, including but not limited to those specified in the Department Medicaid Encounter Submission Requirements Policy.
- 5.1.3.27 The MCO shall submit summary reporting in accordance with Exhibit O: Quality and Oversight Reporting Requirements, to be used to validate Encounter submissions.
- 5.1.3.28 The MCO shall use the procedure codes, diagnosis codes, and other codes as directed by the Department for reporting Encounters and fee- for-service claims.
- 5.1.3.29 Any exceptions shall be considered on a code-by-code basis after the Department receives written notice from the MCO requesting an exception.
- 5.1.3.30 The MCO shall use the Provider identifiers as directed by DHHS for both Encounter and FFS submissions, as applicable.
- 5.1.3.31 The MCO shall provide, as a supplement to the Encounter Confidential Data submission, a Member file on a monthly basis, which shall contain appropriate Member Medicaid identification numbers, the PCP assignment of each Member, and the group affiliation and service location address of the PCP.
- 5.1.3.32 The MCO shall submit complete Encounter Confidential Data in the appropriate HIPAA-compliant formats regardless of the claim submission method (hard copy paper, proprietary formats, EDI, DDE).
- 5.1.3.33 The MCO shall assign staff to participate in encounter technical work group meetings as directed by the Department.
- 5.1.3.34 The MCO shall provide complete and accurate encounters to the Department.
- 5.1.3.35 The MCO shall implement review procedures to validate Encounter Confidential Data submitted by Providers. The MCO shall meet the following standards:
  - 5.1.3.35.1 Completeness
    - 5.1.3.35.1.1. The MCO shall submit encounters that represent one hundred percent (100%) of the Covered Services provided by Participating Providers and Non-Participating Providers.
  - 5.1.3.35.2 Accuracy
    - 5.1.3.35.2.1. Transaction type (X12): Ninety-eight percent (98%) of the records in an

Page 330 of 397 Date \_\_\_\_\_

Medicaid Care Management Services Contract
New Hampshire Department of Health and Human Services
Medicaid Care Management Services



# **Exhibit B - DRAFT**

	MCO's encounter batch submission shall pass X12 EDI compliance edits and the MMIS threshold and repairable compliance edits. The standard shall apply to submissions of each individual batch and online transaction type.
5.1.3.35.2.2.	Transaction type (NCPDP): Ninety-eight percent (98%) of the records in an MCO's encounter batch submission shall pass NCPDP compliance edits and the pharmacy benefits system threshold and repairable compliance edits. The NCPDP compliance edits are described in the NCPDP.
5.1.3.35.2.3.	One-hundred percent (100%) of Member identification numbers shall be accurate and valid.
5.1.3.35.2.4.	Ninety-eight percent (98%) of billing Provider information shall be accurate and valid.
5.1.3.35.2.5.	Ninety-eight percent (98%) of servicing Provider information shall be accurate and valid.
5.1.3.35.2.6.	The MCO shall submit a monthly supplemental Provider file, to include Confidential Data elements as defined by the Department, for all Providers that were submitted on encounters in the prior month.
5.1.3.35.2.7.	For the first six (6) months of encounter production submissions, the MCO shall conduct a monthly end to end test of a statistically valid sample of claims to ensure Encounter Confidential Data quality.
5.1.3.35.2.8.	The end to end test shall include a review of the Provider claim to what Confidential Data is in the MCO claims processing system, and the encounter file record produced for that claim.
5.1.3.35.2.9.	The MCO shall report a pass or fail to

Page 331 of 397 Date \_\_\_\_\_

the Department. If the result is a fail, the MCO shall also submit a root cause

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

analysis that includes plans for remediation.

5.1.3.35.2.10. If the Department or the MCO identifies a Confidential Data defect, the MCO shall, for six (6) months post Confidential Data defect identification, conduct a monthly end to end test of a statistically valid sample of claims to ensure Encounter Confidential Data quality.

5.1.3.35.2.11. If two (2) or more Encounter Confidential Data defects are identified within a rolling twelve (12) month period, the Department may require the MCO to contract with an external vendor to independently assess the MCO Encounter Confidential Data process. The external vendor shall produce a report that shall be shared with the Department.

#### 5.1.3.35.3 Timeliness

- 5.1.3.35.3.1. Encounter Confidential Data shall be submitted weekly, within fourteen (14) calendar days of claim payment.
- 5.1.3.35.3.2. All encounters shall be submitted, both paid and denied claims.
- 5.1.3.35.3.3. The MCO shall be subject to liquidated damages as specified in Section 5.5.2 (Liquidated Damages) for failure to timely submit Encounter Data, in accordance with the accuracy standards established in this Agreement.

#### 5.1.3.35.4 Error Resolution

5.1.3.35.4.1. For all historical encounters submitted after the submission start date, if the Department or its fiscal agent notifies the MCO of encounters failing X12 EDI compliance edits or MMIS threshold and repairable compliance edits, the MCO shall remediate all related encounters within forty-five (45) calendar days after such notice.

Page 332 of 397 Date \_\_\_\_\_

New Hampshire Department of Health and Human Services Medicaid Care Management Services



#### **Exhibit B - DRAFT**

5.1.3.35.4.2.	For all ongoing claim encounters, if the
	Department or its fiscal agent notifies
	the MCO of encounters failing X12 EDI
	compliance edits or MMIS threshold and
	repairable compliance edits, the MCO
	shall remediate all such encounters
	within fourteen (14) calendar days after
	such notice.

5.1.3.35.4.3. If the MCO fails to comply with either error resolution timeline, the Department shall require a CAP and assess liquidated damages as described in Section 5.5.2 (Liquidated Damages).

5.1.3.35.4.4. The MCO shall not be held accountable for issues or delays directly caused by or as a direct result of the changes to MMIS by the Department.

5.1.3.35.5 Survival

5.1.3.35.5.1. All Encounter Confidential Data accumulated as part of the MCM program shall remain the property of the Department and, upon termination of the Agreement, the Confidential Data shall be electronically transmitted to the Department in a format and schedule prescribed by the Department and as is further described in Section 7.2

(Termination for Other Reasons).

5.1.3.35.6 Reporting

5.1.3.35.6.1. The MCO shall submit Confidential Data on the basis of which the State certifies the actuarial soundness of capitation rates to the MCO, including base Confidential Data that is generated by the MCO. [42 CFR 438.604(a)(2); 42 CFR 438.606; 42 CFR 438.3; 42 CFR 438.5(c)]

5.1.3.35.6.2. When requested by the Department, the MCO shall submit Encounter Data, financial data, and other Confidential Data to the Department to ensure actuarial soundness in development of the capitated rates, or any other

Date \_\_\_\_\_

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

actuarial analysis required by the Department or State or federal law.

5.1.3.35.6.3. The MCO's CFO shall submit and concurrently certify to the best of their information, knowledge, and belief that all Confidential Data and information described in 42 CFR 438.604(a), which the Department uses to determine the capitated rates, is accurate. [42 CFR 438.606]

#### 5.1.4 Confidential Data Certification

- 5.1.4.1 All Confidential Data submitted to the Department by the MCO shall be certified by one (1) of the following:
  - 5.1.4.1.1 The MCO's CEO:
  - 5.1.4.1.2 The MCO's CFO; or
  - 5.1.4.1.3 An individual who has delegated authority to sign for, and who reports directly to, the MCO's CEO or CFO. [42 CFR 438.604; 42 CFR 438.606(a)]
- 5.1.4.2 The Confidential Data that shall be certified include, but are not limited to, all documents specified by the Department, enrollment information, Encounter Data, and other information contained in this Agreement or proposals.
- 5.1.4.3 The certification shall attest to, based on best knowledge, information, and belief, the accuracy, completeness and truthfulness of the documents and data.
- 5.1.4.4 The MCO shall submit the certification concurrently with the certified Confidential Data and documents. [42 CFR 438.604; 42 CFR 438.606]
- 5.1.4.5 The MCO shall submit the MCO Confidential Data Certification process policies and procedures for the Department review during the Readiness Review process.
- 5.1.5 Confidential Data System Support for Quality Assurance & Performance Improvement
  - 5.1.5.1 The MCO shall have a Confidential Data collection, processing, and reporting system sufficient to support the QAPI program requirements described in Section 4.14.3 (Quality Assessment and Performance Improvement Program).
  - 5.1.5.2 The system shall be able to support QAPI monitoring and evaluation activities, including the monitoring and evaluation of the quality of clinical care provided, periodic evaluation of Participating

Page 334 of 397	Date
-----------------	------

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

Providers, Member feedback on QAPI activity, and maintenance and use of medical records used in QAPI activities.

### 5.2 **Contract Oversight Program**

- 5.2.1 The MCO shall have a formalized Contract Oversight Program to ensure that it complies with this Agreement, which at a minimum, should outline:
  - 5.2.1.1 The specific monitoring and auditing activities that the MCO shall undertake to ensure its and its Subcontractors' compliance with certain provisions and requirements of the Agreement;
  - 5.2.1.2 The frequency of those contract oversight activities; and
  - 5.2.1.3 The person(s) responsible for those contract oversight activities.
- 5.2.2 The Contract Oversight Program shall specifically address how the MCO shall oversee the MCO's and its Subcontractor's compliance with the following provisions and requirements of the Agreement:
  - 5.2.2.1 Section 3.10 (Subcontractors);
  - 5.2.2.2 Section 4 (Program Requirements); and
  - 5.2.2.3 All Confidential Data and reporting requirements.
- 5.2.3 The Contract Oversight Program shall set forth how the MCO's Chief Executive Officer (CEO)/Executive Director, Compliance Officer and Board of Directors shall be made aware of non-compliance identified through the Contract Oversight Program.
- 5.2.4 The MCO shall present to the Department for review as part of the Readiness Review a copy of the Contract Oversight Program and any implementing policies.
- 5.2.5 The MCO shall present to the Department for review redlined copies of proposed changes to the Contract Oversight Program and its implementing policies prior to adoption.
- 5.2.6 This Contract Oversight Program is distinct from the Program Integrity Plan and the Fraud, Waste and Abuse Compliance Plan discussed in Section 5.3 (Program Integrity).
- 5.2.7 The MCO shall promptly, but no later than thirty (30) calendar days after the date of discovery, report any material non-compliance identified through the Contract Oversight Program and submit a Corrective Action Plan to the Department to remediate such non-compliance.
- 5.2.8 The MCO shall implement any changes to the Corrective Action Plan requested by the Department.

#### 5.3 **Program Integrity**

5.3.1 **General Requirements** 

Page 335 of 397	Date	

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

- 5.3.1.1 The MCO shall present to the Department for review, as part of the Readiness Review process, a Program Integrity Plan and a Fraud, Waste and Abuse Compliance Plan and shall comply with policies and procedures that guide and require the MCO and the MCO's officers, employees, agents and Subcontractors to comply with the requirements of this Section 5.3 (Program Integrity). [42 CFR 438.608]
- 5.3.1.2 Within thirty (30) calendar days from the date of contract signing and annually thereafter, the MCO shall submit all updates and modifications to the Department for approval at least thirty (30) calendar days in advance of the effective date. The MCO shall present to the Department for review redlined copies of proposed changes to the Program Integrity Plan and the Fraud, Waste and Abuse Compliance Plan prior to adoption.
- 5.3.1.3 The MCO shall include program integrity requirements in its Subcontracts and provider application, credentialing and recredentialing processes.
- 5.3.1.4 The MCO is expected to be familiar with, comply with, and require compliance by its Subcontractors with all regulations and subregulatory guidance related to program integrity whether or not those regulations are listed below:
  - 5.3.1.4.1 Section 1902(a)(68) of the Social Security Act;
  - 5.3.1.4.2 42 CFR Section 438;
  - 5.3.1.4.3 42 CFR Section 455;
  - 5.3.1.4.4 42 CFR Section 1000 through 1008; and
  - 5.3.1.4.5 CMS Toolkits.
- 5.3.1.5 The MCO shall ensure compliance with the program integrity provisions of this Agreement, including proper payments to providers or Subcontractors, methods for detection and prevention of Fraud, Waste and Abuse, and the MCO's and its Subcontractors' compliance with all program integrity reporting requirements to the Department.
- 5.3.1.6 The MCO shall have a Program Integrity Plan and a Fraud, Waste and Abuse Compliance Plan that are designed to guard against Fraud, waste and abuse.
- 5.3.1.7 The Program Integrity Plan and the Fraud, Waste and Abuse Compliance Plan shall include, at a minimum, the establishment and implementation of internal controls, policies, and procedures to prevent and deter Fraud, Waste and Abuse.
- 5.3.1.8 The MCO shall be compliant with all applicable federal and State regulations related to Medicaid program integrity. [42 CFR 455, 42]

Page 336 of 397	Date	
-----------------	------	--

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

- CFR 456, 42 CFR 438, 42 CFR 1000 through 1008 and Section 1902(a)(68) of the Social Security Act]
- 5.3.1.9 The MCO shall work with the Department on program integrity issues, and with MFCU as directed by the Department, on Fraud, Waste or Abuse investigations. This shall include, at a minimum, the following:
  - 5.3.1.9.1 Participation in MCO program integrity meetings with the Department following the submission of the monthly allegation log submitted by the MCO in accordance with Exhibit O: Quality and Oversight Reporting Requirements:
    - 5.3.1.9.1.1. The frequency of the program integrity meetings shall be as often as monthly.
  - 5.3.1.9.2 Discussion at these meetings shall include, but not be limited to, case development and monitoring, implementation of Fraud, Waste, and Abuse Annual Plans, plan use of data analytic Fraud detection algorithms required in Section 5.3.2.2.4.4, quality control and review of Encounter Confidential Data submitted to the Department, and corrective actions from the Department Program Integrity audits.
  - 5.3.1.9.3 The MCO shall ensure Subcontractors attend monthly meetings based on relevant agenda items and ensure agenda items are removed if essential MCO or Subcontractor staff are unavailable:
  - 5.3.1.9.4 Participation in any MCO and Subcontractor forums to discuss best practices, performance metrics, provider risk assessments, analytics, and lessons learned; and
  - 5.3.1.9.5 Participation in meetings with MFCU, as determined by MFCU and the Department.

# 5.3.2 Fraud, Waste and Abuse

- 5.3.2.1 The MCO, or a Subcontractor which has been delegated responsibility for coverage of services and payment of claims under this Agreement, shall implement and maintain administrative and management arrangements or procedures designed to detect and prevent Fraud, Waste and Abuse. [42 CFR 438.608(a)]
- 5.3.2.2 The arrangements or procedures shall include the following:
  - 5.3.2.2.1 The Program Integrity Plan and the Fraud, Waste and Abuse Compliance Plan that includes, at a minimum, all of the following elements:

Page 337 of 397	Date

Medicaid Care Management Services Contract
New Hampshire Department of Health and Human Services
Medicaid Care Management Services



# **Exhibit B - DRAFT**

5.3.2.2.1.1.	Written policies, procedures, and standards of conduct that articulate the organization's commitment to comply with all applicable requirements and standards under this Agreement, and all applicable federal and State requirements;
5.3.2.2.1.2.	Designation of a Compliance Officer who is accountable for developing and implementing policies and procedures, and practices designed to ensure compliance with the requirements of the Agreement and who directly reports to the CEO and the Board of Directors;
5.3.2.2.1.3.	Establishment of a Regulatory Compliance Committee of the Board of Directors and at the senior management level charged with overseeing the MCO's compliance program and its compliance with this Agreement;
5.3.2.2.1.4.	System for training and education for the Compliance Officer, the MCO's senior management, employees, and Subcontractor on the federal and State standards and requirements under this Agreement;
5.3.2.2.1.5.	Effective lines of communication between the Compliance Officer and MCO's staff and Subcontractors;
5.3.2.2.1.6.	Enforcement of standards through well-publicized disciplinary guidelines; and
5.3.2.2.1.7.	Establishment and implementation of procedures and a system with dedicated staff of routine internal monitoring and auditing of compliance risks, prompt response to compliance issues as they are raised, investigation of potential problems as identified in the course of self-evaluation and audits, correction of such problems promptly and thoroughly (or coordination of suspected criminal acts with law enforcement agencies) to reduce the potential for recurrence, and ongoing compliance with the

Page 338 of 397 Date \_\_\_\_

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

I	requireme	ents under this	Agreeme	ent. [42
(	CFR	438.608(a);	42	CFR
•	438.608(	a)(1)(i-vii)]		

- 5.3.2.2.2 The process by which the MCO shall monitor their marketing representative activities to ensure that the MCO does not engage in inappropriate activities, such as inducements;
- 5.3.2.2.3 A requirement that the MCO shall report on staff termination for engaging in prohibited marketing conduct or Fraud, Waste and Abuse to the Department within thirty (30) business days;
- 5.3.2.2.4 The MCO shall maintain and report as requested specific controls to detect and prevent potential Fraud, Waste and Abuse including, without limitation:
  - 5.3.2.2.4.1. A list of automated pre-payment claims edits, including National Correct Coding Initiative (NCCI) edits;
  - 5.3.2.2.4.2. A list of automated post-payment claims edits;
  - 5.3.2.2.4.3. In accordance with 42 CFR 438.602(b), the MCO shall maintain edits on its claims systems to ensure in-network claims include New Hampshire Medicaid enrolled billing and rendering provider NPIs. The MCO shall amend edits on its claims systems as required by any changes in federal and State requirements for managed care billing;
  - 5.3.2.2.4.4. At least three (3) Confidential Data analytic algorithms for Fraud detection specified by the Department Program Integrity and three (3) additional Confidential Data analytic algorithms as determined by the MCO for a total of at least six (6) algorithms, which should include services provided Subcontractors. These algorithms are subject to change based on outcomes of the algorithms and Department approval;
  - 5.3.2.2.4.5. Visit verification procedures and practices, including sample sizes and targeted provider types or locations:



Page 339 of 397 Date \_\_\_\_\_

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

5.3.2.2.4.6.	A list of surveillance and/or utilization			
	management protocols used to			
	safeguard against unnecessary or			
	inappropriate use of Medicaid services			
	and a description demonstrating the			
	results of such protocols when			
	requested by the Department;			

- 5.3.2.2.4.7. A method to verify, by sampling or other method, whether services that have been represented to have been delivered by Participating Providers and were received by Members and the application of such verification processes on a regular basis. The MCO may use an explanation of benefits (EOB) for such verification only if the MCO suppresses information on EOBs that would be a violation of Member confidentiality requirements for women's health care, family planning, sexually transmitted diseases, and behavioral health services [42 CFR 455.20];
- 5.3.2.2.4.8. Provider and Member materials identifying the MCO's Fraud and Abuse reporting hotline number;
- 5.3.2.2.4.9. Work plans for conducting both announced and unannounced site visits and field audits of Participating Providers determined to be at high risk to ensure services are rendered and billed correctly;
- 5.3.2.2.4.10. The State reserves the right to direct the MCO to conduct up to ten (10) on-site investigations per year;
- 5.3.2.2.4.11. The process for putting a Participating Provider on and taking a Participating Provider off prepayment review, metrics used including. the and frequency of evaluating whether prepayment review continues to be appropriate;
- 5.3.2.2.4.12. The ability to suspend a Participating Provider's or Non-Participating Provider's payment due to credible

Page 340 of 397

Date					

New Hampshire Department of Health and Human Services Medicaid Care Management Services



#### **Exhibit B - DRAFT**

	Dopartinont'i Togram intognty, and
5.3.2.2.4.13.	The process by which the MCO shall recover inappropriately paid funds if the MCO discovers wasteful and/or abusive, incorrect billing trends with a particular Participating Provider or provider type, specific billing issue trends, or quality trends.

allegation of Fraud if directed by the Department Program Integrity; and

- 5.3.2.2.5 A provision for the prompt reporting of all Overpayments identified and recovered, specifying the Overpayments due to potential Fraud;
- 5.3.2.2.6 A provision for referral of any potential Participating Provider or Non-Participating Provider Fraud, Waste and Abuse that the MCO or Subcontractor identifies to the Department Program Integrity and any potential Fraud directly to the MFCU as required under this Agreement [42 CFR 438.608(a)(7)];
- 5.3.2.2.7 A provision for the MCO's suspension of payments to a Participating Provider for which the Department determines there is credible allegation of Fraud in accordance with this Agreement and 42 CFR 455.23; and
- 5.3.2.2.8 A provision for notification to the Department when the MCO receives information about a change in a Participating Provider's circumstances that may affect the Participating Provider's eligibility to participate in the MCM program, including the termination of the provider agreement with the MCO as detailed in Exhibit O: Quality and Oversight Reporting Requirements.
- 5.3.2.3 The MCO and Subcontractors shall implement and maintain written policies for all employees and any Subcontractor or agent of the entity, that provide detailed information about the False Claims Act (FCA) and other federal and State laws described in Section 1902(a)(68) of the Social Security Act, including information about rights of employees to be protected as whistleblowers. [Section 1902(a)(68) of the Social Security Act; 42 CFR 438.608(a)(6)]
- 5.3.2.4 The MCO, and if required by the MCO's Subcontractors, shall post and maintain the Department-approved information related to Fraud, Waste and Abuse on its website, including but not limited to, provider notices, current listing of Participating Providers, providers that have been excluded or sanctioned from the Medicaid Care

Page 341 of 397 Date \_\_\_\_\_

New Hampshire Department of Health and Human Services Medicaid Care Management Services



#### **Exhibit B - DRAFT**

Management Program, any updates, policies, provider resources, contact information and upcoming educational sessions/webinars.

### 5.3.3 Identification and Recoveries of Overpayments

- 5.3.3.1 The MCO shall maintain an effective Fraud, Waste and Abuserelated Provider overpayment identification, Recovery and tracking process.
- 5.3.3.2 The MCO shall perform ongoing analysis of its authorization, utilization, claims, Provider's billing patterns, and encounter Confidential Data to detect improper payments, and shall perform audits and investigations of Subcontractors, Providers and Provider entities.
- 5.3.3.3 This process shall include a methodology for a means of estimating overpayment, a formal process for documenting communication with Providers, and a system for managing and tracking of investigation findings, Recoveries, and underpayments related to Fraud, Waste and Abuse investigations/audit/any other overpayment recovery process as described in the Fraud, Waste and Abuse reports provided to the Department in accordance with Exhibit O: Quality and Oversight Reporting Requirements.
- 5.3.3.4 The MCO and Subcontractors shall each have internal policies and procedures for documentation, retention and recovery of all Overpayments, specifically for the recovery of Overpayments due to Fraud, Waste and Abuse, and for reporting and returning Overpayments as required by this Agreement. [42 CFR 438.608(d)(1)(i)]
- 5.3.3.5 The MCO and its subcontractors shall report to the Department within sixty (60) calendar days when it has identified Capitation Payments or other payment amounts received are in excess to the amounts specified in this Agreement. [42 CFR 438.608(c)(3)].
- 5.3.3.6 The Department may recover Overpayments that are not recovered by or returned to the MCO within sixty (60) calendar days of notification by final findings letter to the Provider by the MCO unless the MCO has a recovery agreement with the Provider, or is actively recovering through claims recoupment. The Department will notify MCO if the Department has plans to pursue recovery.
- 5.3.3.7 This section of the Agreement does not apply to any amount of a recovery to be retained under False Claim Act cases.
- 5.3.3.8 Any settlement reached by the MCO or its Subcontractors and a Provider shall not bind or preclude the State from further action.
- 5.3.3.9 The Department shall utilize the information and documentation collected under this Agreement, as well as nationally recognized information on average recovery amounts as reported by State

Page 342 of 397	Date

New Hampshire Department of Health and Human Services Medicaid Care Management Services



#### **Exhibit B - DRAFT**

MFCUs and commercial insurance plans for setting actuarially sound Capitation Payments for each MCO consistent with the requirements in 42 CFR 438.4.

5.3.3.10 If the MCO does not meet the required metrics related to expected Fraud referrals, overpayment recoupments, and other measures set forth in this Agreement and Exhibit O: Quality and Oversight Reporting Requirements, the Department shall impose liquidated damages, unless the MCO can demonstrate good cause for failure to meet such metrics.

# 5.3.4 Referrals of Credible Allegations of Fraud, Waste or Abuse and Provider and Payment Suspensions

- 5.3.4.1 General
  - 5.3.4.1.1 The MCO shall, and shall require any Subcontractor to, establish policies and procedures for referrals to the Department Program Integrity Unit and the MFCU on credible allegations of Fraud and for payment suspension when there is a credible allegation of Fraud. [42 CFR 438.608(a)(8); 42 CFR 455.23].
  - 5.3.4.1.2 The MCO shall complete a Department "Request to Open" form for any potential Fraud, waste, or abuse case, including those that lead to a credible allegation of Fraud. The Department's Program Integrity Unit shall have fifteen (15) business days to respond to the MCO's "Request to Open" form.
  - 5.3.4.1.3 When the MCO or its Subcontractor has concluded that a credible allegation of Fraud or abuse exists, the MCO shall make a referral to the Department's Program Integrity Unit and any potential Fraud directly to MFCU within five (5) business days of the determination on a template provided by the Department. [42 CFR 438.608(a)(7)]
  - 5.3.4.1.4 Unless and until prior written approval is obtained from the Department, neither the MCO nor a Subcontractor shall take any administrative action or any of the following regarding the allegations of suspected Fraud:
    - 5.3.4.1.4.1. Suspend Provider payments:
    - 5.3.4.1.4.2. Contact the subject of the investigation about any matters related to the investigation;
    - 5.3.4.1.4.3. Continue the investigation into the matter:

	matter;	
Page 343 of 397		Date _

Medicaid Care Management Services Contract
New Hampshire Department of Health and Human Services
Medicaid Care Management Services



Enter into or attempt to negotiate any settlement or agreement regarding the

# **Exhibit B - DRAFT**

5.3.4.1.4.4.

		matter; or
	5.3.4.1.4.5.	Accept any monetary or other thing of valuable consideration offered by the subject of the investigation in connection with the incident.
5.3.4.1.5		shall employ pre-payment review when the Department.
5.3.4.1.6		, the MCO may employ pre-payment the following circumstances without
	5.3.4.1.6.1.	Upon new Participating Provider enrollment;
	5.3.4.1.6.2.	For delayed payment during Provider education;
	5.3.4.1.6.3.	For existing Providers with billing inaccuracies; or
	5.3.4.1.6.4.	Fraud upon identification from Confidential Data analysis or other grounds.
5.3.4.1.7	agency acc Department Officer wit acceptance suspend pa determined	tment, MFCU or another law enforcement cepts the allegation for investigation, the t shall notify the MCO's Compliance thin two (2) business days of the notification, along with a directive to ayment to the affected Provider(s) if it is that an exception to suspension does not letermined by the Department under 42 3.
5.3.4.1.8		ment shall notify the MCO if the referral is investigation.
5.3.4.1.9	agencies de	ts contained in this section of the
5.3.4.1.10	- 1	ot of notification from the Department, the send notice of the decision to suspend

Date \_\_\_\_\_ Page 344 of 397

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

program payments to the Provider within the following	
timeframe:	

- 5.3.4.1.10.1. Within five (5) calendar days of taking such action unless requested in writing by the Department, the MFCU, or law enforcement to temporarily withhold such notice; or
- 5.3.4.1.10.2. Within thirty (30) calendar days if requested by the Department, MFCU, or law enforcement in writing to delay sending such notice.
- 5.3.4.1.10.3. The request for delay may be renewed in writing no more than twice and in no event may the delay exceed ninety (90) calendar days.
- 5.3.4.1.11 The notice shall include or address all of the following (42 CFR 455.23(b)(2)):
  - 5.3.4.1.11.1. That payments are being suspended in accordance with this provision;
  - 5.3.4.1.11.2. Set forth the general allegations as to the nature of the suspension action. The notice need not disclose any specific information concerning an ongoing investigation;
  - 5.3.4.1.11.3. That the suspension is for a temporary period and cite the circumstances under which the suspension shall be lifted;
  - 5.3.4.1.11.4. Specify, when applicable, to which type or types of claims or business units the payment suspension relates; and
  - 5.3.4.1.11.5. Where applicable and appropriate, inform the Provider of any appeal rights available to the Provider, along with the Provider's right to submit written evidence for consideration by the MCO.
- 5.3.4.1.12 All suspension of payment actions under this section of the Agreement shall be temporary and shall not continue after either of the following:
  - 5.3.4.1.12.1. The MCO is notified by the Department that there is insufficient evidence of

Page 345 of 397	Date

5.3.4.1.15

5.3.4.1.16

5.3.4.1.17

New Hampshire Department of Health and Human Services **Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

Fraud, Waste or Abuse by the Provider; 5.3.4.1.12.2. The MCO is notified by the Department that the legal proceedings related to the Provider's alleged Fraud, Waste or Abuse are completed. The MCO shall document in writing the termination of 5.3.4.1.13 a payment suspension and issue a notice of the termination to the Provider and to the Department. 5.3.4.1.14 The DHHS Program Integrity Unit may find that good cause exists not to suspend payments, in whole or in part, or not to continue a payment suspension previously imposed, to an individual or entity against which there is an investigation of a credible allegation of Fraud, Waste or Abuse as set forth in 42 CFR 455.23. Every thirty (30) calendar days that a payment suspension exists, the Department shall direct the MCO to continue, reduce, or remove the payment suspension. The MCO shall maintain for a minimum of six (6) years from the date of issuance all materials documenting: 5.3.4.1.16.1. Details of payment suspensions that were imposed in whole or in part; and 5.3.4.1.16.2. Each instance when a payment suspension was not imposed or was discontinued for good cause. If the MCO fails to suspend payments to an entity or individual for whom there is a pending investigation of a credible allegation of Fraud, Waste or Abuse

5.3.4.1.18 If any government entity, either from restitutions, recoveries, penalties or fines imposed following a criminal prosecution or guilty plea, or through a civil settlement or judgment, or any other form of civil action, receives a monetary recovery from any entity or individual, the entirety of such monetary recovery belongs exclusively to the State, and the MCO and

impose liquidated damages.

without good cause, and the Department directed the MCO to suspend payments, the Department may

Date \_\_\_\_\_

New Hampshire Department of Health and Human Services Medicaid Care Management Services



#### **Exhibit B - DRAFT**

any involved Subcontractor have no claim to any portion of such recovery.

- 5.3.4.1.19 Furthermore, the MCO is fully subrogated, and shall require its Subcontractors to agree to subrogate, to the State for all criminal, civil and administrative action recoveries undertaken by any government entity, including but not limited to all claims the MCO or its Subcontractor(s) has or may have against any entity or individual that directly or indirectly receives funds under this Agreement, including but not limited to any health care Provider, manufacturer, wholesale or retail supplier, sales representative, laboratory, or other Provider in the design, manufacture, Marketing, pricing, or quality of drugs, pharmaceuticals, medical supplies, medical devices, DME, or other health care related products or services.
- 5.3.4.1.20 Any funds recovered and retained by a government entity shall be reported to the actuary to consider in the rate-setting process.

### 5.3.5 Investigations

- 5.3.5.1 The MCO and its Subcontractors shall cooperate with all State and federal agencies that investigate Fraud, Waste and Abuse.
- 5.3.5.2 The MCO shall ensure its Subcontractors and any other contracted entities are contractually required to also participate fully with any State or federal agency or their contractors.
- 5.3.5.3 The MCO and its Subcontractors shall suspend its own investigation and all program integrity activities if notified in writing to do so by any applicable State or federal agency (e.g., MFCU, the Department, OIG, and CMS).
- 5.3.5.4 The MCO and its Subcontractors shall comply with any and all directives resulting from State or federal agency investigations.
- 5.3.5.5 The MCO and its Subcontractors shall maintain all records, documents and claim or encounter Confidential Data for Members, Providers and Subcontractors who are under investigation by any State or federal agency in accordance with retention rules or until the investigation is complete and the case is closed by the investigating State or federal agency.
- 5.3.5.6 The MCO shall provide any Confidential Data access or detail records upon written request from the Department for any potential Fraud, Waste and Abuse investigation, Provider or claim audit, or for MCO oversight review.

age 347 of 397	Date

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



### **Exhibit B - DRAFT**

- 5.3.5.7 The additional access shall be provided within three (3) business days of the request.
- 5.3.5.8 The MCO and its Subcontractors shall request a refund from a third-party payer, Provider or Subcontractor when an investigation indicates that such a refund is due.
- 5.3.5.9 These refunds shall be reported to the Department as Overpayments.
- 5.3.5.10 The Department shall conduct investigations related to suspected Provider Fraud, Waste and Abuse cases, and reserves the right to pursue and retain recoveries for all claims (regardless of paid date) to a Provider with a paid date older than four (4) months for which the MCO has not submitted a request to open and for which the MCO continued to pursue the case. The State shall notify the MCO of any investigation it intends to open prior to contacting the Provider.
- 5.3.5.11 Investigations should be concluded within nine (9) months of the approval of the request to open. The MCO must submit a justification for the investigation remaining open if it exceeds nine (9) months with an expected date for the conclusion of the investigation and receive approval from the Department to continue the investigation. The MCO may be penalized if the justification is not approved in accordance with Exhibit N: Liquidated Damages Matrix. A case shall be considered completed when a final conclusion letter is sent to the provider or a referral has been made to MFCU.
- 5.3.5.12 The MCO shall conduct a follow up investigation twelve (12) months after the final recovery letter date to ensure the same issue is not repeated.

### 5.3.6 Reporting

- 5.3.6.1 Annual Fraud Prevention Report
  - 5.3.6.1.1 The MCO shall submit an annual summary (the "Fraud Prevention Report") that shall document the outcome and scope of the activities performed under Section 5.3 (Program Integrity).
  - 5.3.6.1.2 The annual Fraud Prevention summary shall include, at a minimum, the following elements, in accordance with Exhibit O: Quality and Oversight Reporting Requirements:
    - 5.3.6.1.2.1. The name of the person and department responsible for submitting the Fraud Prevention Report;

Page 348 of 397	Date

template.

5.3.6.2

5.3.6.3

Medicaid Care Management Services Contract New Hampshire Department of Health and Human Services Medicaid Care Management Services



# **Exhibit B - DRAFT**

	5.3.6.1.2.2.	The date the report was prepared;
	5.3.6.1.2.3.	The date the report was prepared,  The date the report is submitted;
		•
	5.3.6.1.2.4.	A description of the SIU;
	5.3.6.1.2.5.	Cumulative Overpayments identified and recovered;
	5.3.6.1.2.6.	Investigations initiated, completed, and referred;
	5.3.6.1.2.7.	Analysis of the effectiveness of the activities performed; and
	5.3.6.1.2.8.	Other information in accordance with Exhibit O: Quality and Oversight Reporting Requirements.
5.3.6.1.3	Department by its CFO to of their in required b	this report, the MCO shall submit to the the Overpayments it recovered, certified that this information is accurate to the best aformation, knowledge, and belief, as by Exhibit O: Quality and Oversight Requirements. [42 CFR 438.606]
Reporting	Member Fraud	
5.3.6.2.1	which the Monday of Member	hall notify the Department of any cases in MCO believes there is a serious likelihood Fraud, Waste and Abuse by sending a ail to the Department Special Investigation
5.3.6.2.2	Fraud, Wa Fraud, Was shall provid	is responsible for investigating Member ste and Abuse and referring Member ste or Abuse to the Department. The MCO de initial allegations, investigations and of Member Fraud, Waste and Abuse to ment.
Terminatio	n Report	
5.3.6.3.1	Termination due to sar Confidentia related pro	shall submit to the Department a monthly Report including Providers terminated action, invalid licenses, services, billing, I Data mining, investigation and any ogram integrity involuntary termination; minations for convenience; and Providers minated.
5.3.6.3.2	The report	shall be completed using the Department

Date \_\_\_\_\_ Page 349 of 397

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

#### 5.3.6.4 Other Reports

- 5.3.6.4.1 The MCO shall submit to the Department demographic changes that may impact eligibility (e.g., Address, etc.).
- 5.3.6.4.2 The MCO shall report at least annually to the Department, and as otherwise required by this Agreement, on their recoveries of Overpayments. [42 CFR 438.604(a)(7); 42 CFR 438.606; 42 CFR 438.608(d)(3)]

#### 5.3.7 Access to Records, On-Site Inspections and Periodic Audits

- 5.3.7.1 As an integral part of the MCO's program integrity function, and in accordance with 42 CFR 455 and 42 CFR 438, the MCO shall provide the Department program integrity staff (or its designee), real time access to all of the MCO electronic encounter and claims Confidential Data (including the Department third-party liability) from the MCO's current claims reporting system.
- 5.3.7.2 The MCO shall provide the Department with the capability to access accurate, timely, and complete Confidential Data as specified in Section 4.20.2 Claims Quality Assurance Program).
- 5.3.7.3 Upon request, the MCO and the MCO's Providers and Subcontractors shall permit the Department, MFCU or any other authorized State or federal agency, or duly authorized representative, access to the MCO's and the MCO's Providers and Subcontractors premises during normal business hours to inspect, review, audit, investigate, monitor or otherwise evaluate the performance of the MCO and its Providers and Subcontractors.
- 5.3.7.4 The MCO and its Providers and Subcontractors shall forthwith produce all records, documents, or other Confidential Data requested as part of such inspection, review, audit, investigation, monitoring or evaluation.
- 5.3.7.5 Copies of records and documents shall be made at no cost to the requesting agency. [42 CFR 438.3(h)]; 42 CFR 455.21(a)(2); 42 CFR 431.107(b)(2)]. A record includes, but is not limited to:

		, , , , ,
	5.3.7.5.1	Medical records;
	5.3.7.5.2	Billing records;
*	5.3.7.5.3	Financial records;
	5.3.7.5.4	Any record related to services rendered, and quality, appropriateness, and timeliness of such service;
	5.3.7.5.5	Any record relevant to an administrative, civil or criminal investigation or prosecution; and

Page 350 of 397 Date \_\_\_\_\_

New Hampshire Department of Health and Human Services Medicaid Care Management Services



#### **Exhibit B - DRAFT**

- 5.3.7.5.6 Any record of an MCO-paid claim or encounter, or an MCO-denied claim or encounter.
- 5.3.7.6 Upon request, the MCO, its Provider or Subcontractor shall provide and make staff available to assist in such inspection, review, audit, investigation, monitoring or evaluation, including the provision of adequate space on the premises to reasonably accommodate the Department, MFCU or other State or federal agencies.
- 5.3.7.7 The MCO and the MCO's Providers and Subcontractors shall permit the Department, MFCU or any other authorized State or federal agency, or duly authorized representative, access to the MCO's and the MCO's Providers and Subcontractors premises at any time to inspect, review, audit, investigate, monitor or otherwise evaluate the performance of the MCO and its Providers and Subcontractors. When reasonable, such access shall be sought during normal business hours. [42 CFR 438.3(h)]
- 5.3.7.8 The MCO and its Subcontractors shall be subject to on-site or offsite reviews by the Department and shall comply within fifteen (15) business days with any and all Department documentation and records requests.
- 5.3.7.9 Documents shall be furnished by the MCO or its Subcontractors at the MCO's expense.
- 5.3.7.10 The right to inspect and audit any records or documents of the MCO or any Subcontractor shall extend for a period of ten (10) years from the final date of this Agreement's contract period or from the date of completion of any audit, whichever is later. [42 CFR 438.3(h)]
- 5.3.7.11 The Department shall conduct, or contract for the conducting of, periodic audits of the MCO no less frequently than once every three (3) years, for the accuracy, truthfulness, and completeness of the encounter and financial Confidential Data submitted by, or on behalf of, each MCO. [42 CFR 438.602(e)]
- 5.3.7.12 This shall include, but not be limited to, any records relevant to the MCO's obligation to bear the risk of financial losses or services performed or payable amounts under the Agreement.

#### 5.3.8 Transparency

- 5.3.8.1 The Department shall post on its website, as required by 42 CFR 438.10(c)(3), the following documents and reports:
  - 5.3.8.1.1 The Agreement;
  - 5.3.8.1.2 42 CFR 438.604(a)(5) where the Department certifies that the MCO has complied with the Agreement requirements for availability and accessibility of

Page 351 of 397 Date \_\_\_\_\_

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

services, including adequacy of the Participating Provider network, as set forth in 42 CFR 438.206;

- 5.3.8.1.3 Under 42 CFR 438.602(e), a quality report on the accuracy, truthfulness, and completeness of the encounter and financial Data submitted and certified by the MCO resulting from the State's periodic audit; and
- 5.3.8.1.4 Performance metrics and outcomes.

#### 5.4 MCM Withhold and Incentive Program

- 5.4.1 The Department shall institute a withhold arrangement through which an actuarially sound percentage of the MCO's risk adjusted Capitation Payment will be recouped from the MCO and be available for distribution in future years upon meeting specific criteria.
- 5.4.2 The Department shall issue Withhold and Incentive Program Guidance by August 1st each year and/or at other times as determined by the Department.
- 5.4.3 The Department shall institute a Withhold and Incentive Program which directs an annual actuarially sound two percent (2%) retention of the MCO's risk adjusted total Capitation for the rating period. The Withhold shall be available for distribution in future contract years upon meeting specific performance criteria as described in separate guidance.
- 5.4.4 Pursuant to 42 CFR 438.6 (b)(3), this withhold arrangement shall:
  - 5.4.4.1 Be for a fixed period of time and performance is measured during the rating period under the Agreement in which the withhold arrangement is applied;
  - 5.4.4.2 Not be renewed automatically;
  - 5.4.4.3 Be made available to both public and private contractors under the same terms of performance;
  - 5.4.4.4 Not condition MCO participation in the withhold arrangement on the MCO entering into or adhering to intergovernmental transfer agreements; and
  - 5.4.4.5 Is necessary for the specified activities, targets, performance measures, or quality-based outcomes that support program initiatives as specified in the NH MCM Quality Strategy.
- 5.4.5 The MCO shall not receive incentive payments in excess of five percent (5%) of the approved Capitation Payments attributable to the Members or services covered by the incentive arrangements.
  - 5.4.5.1 Pursuant to 42 CFR 438.6(b)(2), this incentive arrangement shall:
    - 5.4.5.1.1 Be for a fixed period of time and performance is measured during the rating period under the

Page 352 of 397 Date \_\_\_\_\_

New Hampshire Department of Health and Human Services Medicaid Care Management Services



#### **Exhibit B - DRAFT**

	applied;
5.4.5.1.2	Not be renewed automatically;
5.4.5.1.3	Be made available to both public and private contractors under the same terms of performance;
5.4.5.1.4	Not condition MCO participation in the incentive arrangement on the MCO entering into or adhering to intergovernmental transfer Agreements; and
5.4.5.1.5	Is necessary for the specified activities, targets, performance measures, or quality-based outcomes that support program initiatives as specified in the NH

Agreement in which the withhold arrangement is

5.4.6 Any differences in performance and rating periods shall be described in the program's actuarial certification for the rating period.

MCM Quality Strategy.

5.4.7 Insofar as the withhold incentive is capped at one hundred five percent (105%) of approved Capitation Payments, and the design of the Withhold and Incentive Program is to maintain withhold funds in the program for actuarial soundness, should there be a remaining amount in withheld funds within the program, additional incentives shall be available through performance metrics determined by the State so that all funds will be disbursed before the end of the contract term in accordance with separate guidance.

#### 5.5 Remedies

# 5.5.1 Reservation of Rights and Remedies

- 5.5.1.1 The MCO acknowledges that failure to comply with provisions of this Agreement may, at the Department's sole discretion, result in the assessment of liquidated damages, termination of the Agreement in whole or in part, and/or imposition of other sanctions as set forth in this Agreement and as otherwise available under State and federal law.
- 5.5.1.2 In the event of any claim for default or breach of this Agreement, no provision of this Agreement shall be construed, expressly or by implication, as a waiver by the State to any existing or future right or remedy available by law.
- 5.5.1.3 Failure of the State to insist upon the strict performance of any term or condition of this Agreement or to exercise or delay the exercise of any right or remedy provided in the Agreement or by law, or the acceptance of (or payment for) materials, equipment or services, shall not release the MCO from any responsibilities or obligations imposed by this Agreement or by law, and shall not be deemed a

Page 353 of 397 Date \_\_\_\_\_

New Hampshire Department of Health and Human Services Medicaid Care Management Services



#### **Exhibit B - DRAFT**

- waiver of any right of the State to insist upon the strict performance of this Agreement.
- 5.5.1.4 In addition to any other remedies that may be available for default or breach of the Agreement, in equity or otherwise, the State may seek injunctive relief against any threatened or actual breach of this Agreement without the necessity of proving actual damages.
- 5.5.1.5 The State reserves the right to recover any or all administrative costs incurred in the performance of this Agreement during or as a result of any threatened or actual breach.
- 5.5.1.6 The remedies specified in this section of the Agreement shall apply until the failure is cured or a resulting dispute is resolved in the MCO's favor.

#### 5.5.2 Liquidated Damages

- 5.5.2.1 The Department may perform an annual review to assess if the liquidated damages set forth in Exhibit N: Liquidated Damages Matrix align with actual damages and/or with the Department's strategic aims and areas of identified non-compliance, and update Exhibit N: Liquidated Damages Matrix as needed via contract amendment.
- 5.5.2.2 DHHS and the MCO agree that it shall be extremely impracticable and difficult to determine actual damages that the Department will sustain in the event the MCO fails to maintain the required performance standards within this section during this Agreement.
- 5.5.2.3 The parties agree that the liquidated damages as specified in this Agreement and set forth in Exhibit N: Liquidated Damages Matrix, and as updated by the Department, are reasonable.
- 5.5.2.4 Assessment of liquidated damages shall be in addition to, not in lieu of, such other remedies that may be available to the Department.
- 5.5.2.5 To the extent provided herein, the Department shall be entitled to recover liquidated damages for each day, incidence or occurrence, as applicable, of a violation or failure.
- 5.5.2.6 The liquidated damages shall be assessed based on the categorization of the violation or non-compliance and are set forth in Exhibit N: Liquidated Damages Matrix.
- 5.5.2.7 The MCO shall be subject to liquidated damages for failure to comply in a timely manner with all reporting requirements in accordance with Exhibit O: Quality and Oversight Reporting Requirements.
- 5.5.2.8 At its sole discretion, the Department may temporarily provide the MCO partial relief or exemption from one or more Liquidated Damages.

Page 354 of 397 Date	Page 354 of 397	Date
----------------------	-----------------	------

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



### **Exhibit B - DRAFT**

#### 5.5.3 **Suspension of Payment**

- 5.5.3.1 Payment of Capitation Payments may be suspended at the Department's sole discretion when the MCO fails:
  - 5.5.3.1.1 To cure a default under this Agreement to the Department's satisfaction within thirty (30) calendar days of notification:
  - 5.5.3.1.2 To implement a CAP addressing violations or non-compliance; and
  - 5.5.3.1.3 To implement an approved Program Management Plan.
- 5.5.3.2 Upon correction of the deficiency or omission, Capitation Payments shall be reinstated.

#### 5.5.4 Intermediate Sanctions

- 5.5.4.1 The Department shall have the right to impose intermediate sanctions as set forth in 42 CFR Section 438.702(a), which include:
  - 5.5.4.1.1 Civil monetary penalties (the Department shall not impose any civil monetary penalty against the MCO in excess of the amounts set forth in 42 CFR 438.704(c), as adjusted);
  - 5.5.4.1.2 Temporary management of the MCO;
  - 5.5.4.1.3 Permitting Members to terminate enrollment without cause;
  - 5.5.4.1.4 Suspending all new enrollment;
  - 5.5.4.1.5 Suspending payments for new enrollment; and
  - 5.5.4.1.6 Agreement termination.
- 5.5.4.2 The Department shall impose intermediate sanctions if the Department finds that the MCO acts or fails to act as follows:
  - 5.5.4.2.1 Fails to substantially provide Medically Necessary services to a Member that the MCO is required to provide services to by law and/or under its Agreement with the Department.
  - 5.5.4.2.2 The Department may impose a civil monetary penalty of up to \$25,000 for each failure to provide medically necessary services, and may also:
    - 5.5.4.2.2.1. Appoint temporary management for the MCO,
    - 5.5.4.2.2.2. In the event of multiple MCOs, the Department may:

Page 355 of 397	Date	
-----------------	------	--

Medicaid Care Management Services Contract
New Hampshire Department of Health and Human Services
Medicaid Care Management Services



# **Exhibit B - DRAFT**

	5.5.4.2.2.3.	Grant Members the right to disenroll without cause;				
	5.5.4.2.2.4.	Suspend all new enrollments to the MCO after the date the HHS Secretary or the Department notifies the MCO of a determination of a violation of any requirement under sections 1903(m) or 1932 of the Social Security Act; and/or				
	5.5.4.2.2.5.	Suspend payments for new enrollments to the MCO until CMS or the Department is satisfied that the reason for imposition of the sanction no longer exists and is not likely to recur. [42 CFR 438.700(b)(1); 42 CFR 438.702(a); 42 CFR 438.704(b)(1); sections 1903(m)(5)(A)(i); 1903(m)(5)(B); 1932(e)(1)(A)(i); 1932(e)(2)(A)(i) of the Social Security Act]				
5.5.4.2.3	Imposes premiums or charges on Members that are in excess of those permitted in the Medicaid program, in which case, the State may impose a civil monetary of up to \$25,000 or double the amount of the excess charges (whichever is greater). The State may also:					
	5.5.4.2.3.1.	Appoint temporary management to the MCO;				
	5.5.4.2.3.2.	Grant Members the right to disenroll without cause;				
	5.5.4.2.3.3.	Suspend all new enrollments to the MCO after the date the HHS Secretary or the Department notifies the MCO of a determination of a violation of any requirement under sections 1903(m) or 1932 of the Social Security Act; and/or				
	5.5.4.2.3.4.	Suspend payments for new enrollments to the MCO until CMS or the Department is satisfied that the reason for imposition of the sanction no longer exists and is not likely to recur. [42 CFR 438.700(b)(2); 42 CFR 438.702(a); 42 CFR 438.704(c); sections 1903(m)(5)(A)(ii); 1903(m)(5)(B); 1932(e)(1)(A)(ii); 1932(e)(2)(A)(iii) of the Social Security Act]				

Page 356 of 397 Date \_\_\_\_\_

New Hampshire Department of Health and Human Services **Medicaid Care Management Services** 



### **Exhibit B - DRAFT**

- 5.5.4.2.4 Discriminates among Members on the basis of their health status or need for health services, in which case, the Department may impose a civil monetary penalty of up to one hundred thousand dollars (\$100,000) for each determination by the Department of discrimination. The Department may impose a civil monetary penalty of up to fifteen thousand dollars (\$15,000) for each individual the MCO did not enroll because of a discriminatory practice, up to the one hundred thousand dollar (\$100,000) maximum. The Department may also:
  - 5.5.4.2.4.1. Appoint temporary management to the MCO:
  - 5.5.4.2.4.2. Grant Members the right to disenroll without cause;
  - Suspend all new enrollments to the 5.5.4.2.4.3. MCO after the date the HHS Secretary or the Department notifies the MCO of a determination of a violation of any requirement under sections 1903(m) or 1932 of the Social Security Act; and/or
  - 5.5.4.2.4.4. Suspend payments for new enrollments to the MCO until CMS or the Department is satisfied that the reason for imposition of the sanction no longer exists and is not likelv to recur. [42 CFR 438.700(b)(3); 42 CFR 438.702(a); 42 CFR 438.704(b)(2) and (3); sections 1903(m)(5)(A)(iii); 1903(m)(5)(B); 1932(e)(1)(A)(iii); 1932(e)(2)(A)(ii) & (iv) of the Social Security Act]
- 5.5.4.2.5 Misrepresents or falsifies information that it furnishes to a Member, potential Member, or health care Provider, in which case, the Department may impose a civil monetary penalty of up to \$25,000 for each instance of misrepresentation. The Department may
  - Appoint temporary management to the MCO:
  - 5.5.4.2.5.2. Grant Members the right to disenroll without case:
  - 5.5.4.2.5.3. Suspend all new enrollments to the MCO after the date the HHS Secretary

also: 5.5.4.2.5.1.

Page 357 of 397

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

or the Department notifies the MCO of a determination of a violation of any requirement under sections 1903(m) or 1932 of the Social Security Act; and/or

- 5.5.4.2.5.4. Suspend payments for new enrollments to the MCO until CMS or the Department is satisfied that the reason for imposition of the sanction no longer exists and is not likely to recur. [42 CFR 438.702(a); 42 CFR 438.700(b)(5); 42 CFR 438.704(b)(1); sections 1903(m)(5)(A)(iv)(II); 1903(m)(5)(B); 1932(e)(1)(A)(iv)(II); 1932(e)(2)(A)(i) of the Social Security Act]
- 5.5.4.2.6 Misrepresents or falsifies information that it furnishes to CMS or to the Department, in which case, the Department may impose a civil monetary penalty of up to one hundred thousand dollars (\$100,000) for each instance of misrepresentation. The Department may also:
  - 5.5.4.2.6.1. Appoint temporary management to the MCO;
  - 5.5.4.2.6.2. Grant Members the right to disenroll without cause;
  - 5.5.4.2.6.3. Suspend all new enrollments to the MCO after the date the HHS Secretary or the Department notifies the MCO of a determination of a violation of any requirement under sections 1903(m) or 1932 of the Social Security Act; and/or
  - 5.5.4.2.6.4. Suspend payments for new enrollments to the MCO until CMS or the Department is satisfied that the reason for imposition of the sanction no longer exists and is not likely to recur. [42 CFR 438.702(a); 42 CFR 438.700(b)(5); 42 CFR 438.704(b)(1); sections 1903(m)(5)(A)(iv)(II); 1903(m)(5)(B); 1932(e)(1)(A)(iv)(II); 1932(e)(2)(A)(i) of the Social Security Act]
- 5.5.4.2.7 Fails to comply with the Medicare Physician Incentive Plan requirements, in which case, DHHS may impose

Page 358 of 397 Date \_\_\_\_\_

New Hampshire Department of Health and Human Services Medicaid Care Management Services



#### **Exhibit B - DRAFT**

a civil monetary	penalty	of	up	to	\$25,000	for	each
failure to comply.	DHHS I	ma	y al	so:			

- 5.5.4.2.7.1. Appoint temporary management to the MCO;
- 5.5.4.2.7.2. Grant Members the right to disenroll without cause;
- 5.5.4.2.7.3. Suspend all new enrollments to the MCO after the date the HHS Secretary or DHHS notifies the MCO of a determination of a violation of any requirement under sections 1903(m) or 1932 of the Social Security Act, and/or
- 5.5.4.2.7.4. Suspend payments for new enrollments to the MCO until CMS or DHHS is satisfied that the reason for imposition of the sanction no longer exists and is not likely to recur. [42 CFR 438.702(a); 42 CFR 438.700(b)(5); CFR 42 438.704(b)(1); sections 1903(m)(5)(A)(iv)(II);1903(m)(5)(B); 1932(e)(1)(A)(iv)(II); 1932(e)(2)(A)(i) of the Social Security Act]
- 5.5.4.3 The Department shall have the right to impose civil monetary penalty of up to \$25,000 for each distribution if the Department determines that the MCO has distributed directly, or indirectly through any agent or independent contractor, Marketing Materials that have not been approved by the Department or that contain false or materially misleading information. [42 CFR 438.700(c); 42 CFR 438.704(b)(1); sections 1932(e)(1)(A); 1932(e)(2)(A)(i) of the Social Security Act]
- 5.5.4.4 The Department shall have the right to terminate this Agreement and enroll the MCO's Members in other MCOs if the Department determines that the MCO has failed to either carry out the terms of this Agreement or meet applicable requirements in Sections 1905(t), 1903(m), and 1932 of the Social Security Act. [42 CFR 438.708(a); 42 CFR 438.708(b); sections 1903(m); 1905(t); 1932 of the Social Security Act]
- 5.5.4.5 The Department shall grant Members the right to terminate MCO enrollment without cause when an MCO repeatedly fails to meet substantive requirements in sections 1903(m) or 1932 of the Social Security Act or 42 CFR 438. [42 CFR 438.706(b-d); section 1932(e)(2)(B)(ii) of the Social Security Act]

Page 359 of 397	Date	

Medicaid Care Management Services Contract New Hampshire Department of Health and Human Services Medicaid Care Management Services

5.5.5



# **Exhibit B - DRAFT**

5.5.4.6	intermediate MCO violate	ment shall only have the right to impose the following sanctions when the Department determines that the dany of the other requirements of Sections 1903(m) or Social Security Act, or any implementing regulations:			
	5.5.4.6.1	Grant Members the right to terminate enrollment without cause and notifying the affected Members of their right to disenroll immediately;			
	5.5.4.6.2	Provide notice to Members of the Department's intent to terminate the Agreement;			
	5.5.4.6.3	Suspend all new enrollment, including default enrollment, after the date the HHS Secretary or the Department notifies the MCO of a determination of a violation of any requirement under Sections 1903(m) or 1932 of the Social Security Act; and			
	5.5.4.6.4	Suspend payment for Members enrolled after the effective date of the sanction and until CMS or the Department is satisfied that the reason for imposition of the sanction no longer exists and is not likely to recur. [42 CFR 438.700; 42 CFR 438.702(a); 42 CFR 438.704; 42 CFR 438.706(b); 42 CFR 438.722(a)-(b); Sections 1903(m)(5); 1932(e) of the Social Security Act]			
Administrative and Other Remedies					
5.5.5.1	At its sole discretion, the Department may, in addition to the other Remedies described within this Section 5.5 (Remedies), also impose the following remedies:				
	5.5.5.1.1	Requiring immediate remediation of any deficiency as determined by the Department;			
	5.5.5.1.2	Requiring the submission of a CAP;			
	5.5.5.1.3	Suspending part of or all new enrollments;			
	5.5.5.1.4	Suspending part of the Agreement;			
	5.5.5.1.5	Requiring mandated trainings; and/or			
	5.5.5.1.6	Suspending all or part of Marketing activities for varying lengths of time.			
5.5.5.2	Temporary N	Management			
	5.5.5.2.1	The Department, at its sole discretion, shall impose temporary management when the Department finds, through onsite surveys, Member or other complaints,			

Page 360 of 397 Date \_\_\_\_\_

financial status, or any other source:

Medicaid Care Management Services Contract New Hampshire Department of Health and Human Services Medicaid Care Management Services



# **Exhibit B - DRAFT**

	5.5.5.2.1.1.	There is continued egregious behavior by the MCO;
	5.5.5.2.1.2.	There is substantial risk to Members' health;
	5.5.5.2.1.3.	The sanction is necessary to ensure the health of the MCO's Members in one (1) of two (2) circumstances: while improvements are made to remedy violations that require sanctions, or until there is an orderly termination or reorganization of the MCO. [42 CFR 438.706(a); section 1932(e)(2)(B)(i) of the Social Security Act]
	5.5.5.2.1.4.	The Department shall impose mandatory temporary management when the MCO repeatedly fails to meet substantive requirements in sections 1903(m) or 1932 of the Social Security Act or 42 CFR 438; and
	5.5.5.2.1.5.	The Department shall not delay the imposition of temporary management to provide a hearing and may not terminate temporary management until it determines, in its sole discretion that the MCO can ensure the sanctioned behavior shall not reoccur. [42 CFR 438.706(b)-(d); Section 1932(e)(2)(B)(ii) of the Social Security Act]
Correcti	ve Action Plan	
5.5.6.1		ment, the MCO shall submit a CAP within f the Department's request, unless the ension to such timeframe.
5.5.6.2	The Department shall rev business days of receipt.	iew and approve the CAP within five (5)
5.5.6.3	The MCO shall implem timeframes specified in the	ent the CAP in accordance with the eCAP.
5.5.6.4	impose liquidated damage	date the implementation of the CAP and es if it determines that the MCO failed to rovision thereof as required.

5.5.7 **Publication** 

5.5.6

Page 361 of 397	Date	

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



### **Exhibit B - DRAFT**

- 5.5.7.1 The Department may publish on its website, on a quarterly basis, a list of MCOs that had remedies imposed on them by the Department during the prior quarter, the reasons for the imposition, and the type of remedy(ies) imposed.
- 5.5.7.2 MCOs that had their remedies reversed pursuant to the dispute resolution process prior to the posting shall not be listed.

#### 5.5.8 **Notice of Remedies**

- 5.5.8.1 Prior to the imposition of remedies under this Agreement, except in the instance of required temporary management, the Department shall issue written notice of remedies that shall include, as applicable, the following:
  - 5.5.8.1.1 A citation to the law, regulation or Agreement provision that has been violated;
  - 5.5.8.1.2 The remedies to be applied and the date the remedies shall be imposed;
  - 5.5.8.1.3 The basis for the Department's determination that the remedies shall be imposed;
  - 5.5.8.1.4 The appeal rights of the MCO;
  - 5.5.8.1.5 Whether a CAP is being requested; and
  - 5.5.8.1.6 The timeframe and procedure for the MCO to dispute the Department's determination.
- 5.5.8.2 An MCO's dispute of a liquidated damage or remedies shall not stay the effective date of the proposed liquidated damages or remedies; and
- 5.5.8.3 Liquidated damages may be imposed retroactively to the date of failure to perform and continue until the failure is cured or any resulting dispute is resolved in the MCO's favor. [42 CFR 438.710(a)(1)-(2)]
- 5.5.8.4 The Department shall monitor accrual of performance standards-based Liquidated Damages for a period of three (3) to nine (9) months as a means to monitor performance to allow for adjustments to start-up operations; thereafter, Liquidated Damages shall be levied and collected at the Department's discretion, as described in this Agreement and any subregulatory guidance.

# 5.6 State Audit Rights

5.6.1 The Department, CMS, NHID, NH Department of Justice, the OIG, the Comptroller General and their designees shall have the right to audit the records and/or documents of the MCO or the MCO's Subcontractors during the term of this Agreement and for ten (10) years from the final date of the

Page 362 of 397	Date
-----------------	------

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

- Agreement period or from the date of completion of any audit, whichever is later. [42 CFR 438.3(h)]
- 5.6.2 HHS, the HHS Secretary, (or any person or organization designated by either), and the Department, have the right to audit and inspect any books or records of the MCO or its Subcontractors pertaining to:
  - 5.6.2.1 The ability of the MCO to bear the risk of financial losses; and
  - 5.6.2.2 Services performed or payable amounts under the Agreement. [Section 1903(m)(2)(A)(iv) of the Social Security Act]
- In accordance with Exhibit O: Quality and Oversight Reporting Requirements, no later than forty (40) business days after the end of the State Fiscal Year, the MCO shall provide the Department a "SOC1" or a "SOC2" Type 2 report of the MCO or its corporate parent in accordance with American Institute of Certified Public Accountants, Statement on Standards for Attestation Engagements (SSAE) No. 16, Reporting on Controls at a Service Organization.
- 5.6.4 The report shall assess the design of internal controls and their operating effectiveness. The reporting period shall cover the previous twelve (12) months or the entire period since the previous reporting period.
- 5.6.5 The Department shall share the report with internal and external auditors of the State and federal oversight agencies. The SSAE 16 Type 2 report shall include:
  - 5.6.5.1 Description by the MCO's management of its system of policies and procedures for providing services to user entities (including control objectives and related controls as they relate to the services provided) throughout the twelve (12) month period or the entire period since the previous reporting period;
  - 5.6.5.2 Written assertion by the MCO's management about whether:
    - 5.6.5.2.1 The aforementioned description fairly presents the system in all material respects;
    - 5.6.5.2.2 The controls were suitably designed to achieve the control objectives stated in that description; and
    - 5.6.5.2.3 The controls operated effectively throughout the specified period to achieve those control objectives.
  - 5.6.5.3 Report of the MCO's auditor, which:
    - 5.6.5.3.1 Expresses an opinion on the matters covered in management's written assertion; and
    - 5.6.5.3.2 Includes a description of the auditor's tests of operating effectiveness of controls and the results of those tests.

Page 363 of 397 Date \_\_\_\_\_

New Hampshire Department of Health and Human Services Medicaid Care Management Services



#### **Exhibit B - DRAFT**

- 5.6.6 The MCO shall notify the Department if there are significant or material changes to the internal controls of the MCO.
- 5.6.7 If the period covered by the most recent SSAE16 report is prior to June 30, the MCO shall additionally provide a bridge letter certifying to that fact.
- 5.6.8 The MCO shall respond to and provide resolution of audit inquiries and findings relative to the MCO Managed Care activities.
- 5.6.9 The Department may require monthly plan oversight meetings to review progress on the MCO's Program Management Plan, review any ongoing CAPs and review MCO compliance with requirements and standards as specified in this Agreement.
- 5.6.10 The MCO shall use reasonable efforts to respond to the Department oral and written correspondence within one (1) business day of receipt.
- 5.6.11 The MCO shall file annual and interim financial statements in accordance with the standards set forth below.
- 5.6.12 Within one hundred and eighty (180) calendar days or other mutually agreed upon date following the end of each calendar year during this Agreement, the MCO shall file, in the form and content prescribed by the National Association of Insurance Commissioners, annual audited financial statements that have been audited by an independent Certified Public Accountant. [42 CFR 438.3(m)]
- 5.6.13 Financial statements shall be submitted in either paper format or electronic format, provided that all electronic submissions must be sent encrypted, if PHI or PII is included, and in PDF format or another read-only format that maintains the documents' security and integrity.
- 5.6.14 The MCO shall also file, within seventy-five (75) calendar days following the end of each calendar year, certified copies of the annual statement and reports as prescribed and adopted by NHID.
- 5.6.15 The MCO shall file within sixty (60) calendar days following the end of each calendar quarter, quarterly financial reports in form and content as prescribed by the National Association of Insurance Commissioners.

#### 5.7 **Dispute Resolution Process**

#### 5.7.1 Informal Dispute Process

5.7.1.1 In connection with any action taken or decision made by the Department with respect to this Agreement, within thirty (30) calendar days following the action or decision, the MCO may protest such action or decision by the delivery of a written notice of protest to the Department and by which the MCO may protest said action or decision and/or request an informal hearing with the NH Medicaid Director ("Medicaid Director").

Page 364 of 397 Date \_\_\_\_\_

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



### **Exhibit B - DRAFT**

- 5.7.1.2 The MCO shall provide the Department with a written statement of the action being protested, an explanation of its legal basis for the protest, and its position on the action or decision.
- 5.7.1.3 The Director shall determine a time that is mutually agreeable to the parties during which they may present their views on the disputed issue(s).
- 5.7.1.4 The presentation and discussion of the disputed issue(s) shall be informal in nature.
- 5.7.1.5 The Director shall provide written notice of the time, format and location of the presentations.
- 5.7.1.6 At the conclusion of the presentations, the Director shall consider all evidence and shall render a written recommendation, subject to approval by the Department Commissioner, as soon as practicable, but in no event more than thirty (30) calendar days after the conclusion of the presentation.
- 5.7.1.7 The Director may appoint a designee to hear the matter and make a recommendation.

# 5.7.2 **Hearing**

- 5.7.2.1 In the event of a termination by the Department, pursuant to 42 CFR Section 438.708, the Department shall provide the MCO with notice and a pre-termination hearing in accordance with 42 CFR Section 438.710.
- 5.7.2.2 The Department shall provide written notice of the decision from the hearing.
- 5.7.2.3 In the event of an affirming decision at the hearing, the Department shall provide the effective date of the Agreement termination.
- 5.7.2.4 In the event of an affirming decision at the hearing, the Department shall give the Members of the MCO notice of the termination, and shall inform Members of their options for receiving Medicaid services following the effective date of termination. [42 CFR 438.710(b); 42 CFR 438.710(b)(2)(i-iii); 42 CFR 438.10]

#### 5.7.3 No Waiver

5.7.3.1 The MCO's exercise of its rights under Section 5.5.1 (Reservation of Rights and Remedies) shall not limit, be deemed a waiver of, or otherwise impact the Parties' rights or remedies otherwise available under law or this Agreement, including but not limited to the MCO's right to appeal a decision of the Department under RSA chapter 541-A, if applicable, or any applicable provisions of the NH Code of Administrative Rules, including but not limited to Chapter He-C 200 Rules of Practice and Procedure.

Page 365 of 397 Date \_\_\_\_\_

New Hampshire Department of Health and Human Services Medicaid Care Management Services



### **Exhibit B - DRAFT**

#### **6 FINANCIAL MANAGEMENT**

#### 6.1 Financial Standards

- 6.1.1 In compliance with 42 CFR 438.116, the MCO shall maintain a minimum level of capital as determined in accordance with NHID regulations, to include RSA Chapter 404-F, and any other relevant laws and regulations.
- 6.1.2 The MCO shall maintain a risk-based capital ratio to meet or exceed the NHID regulations, and any other relevant laws and regulations.
- 6.1.3 With the exception of payment of a claim for a medical product or service that was provided to a Member, and that is in accordance with a written agreement with the Provider, the MCO may not pay money or transfer any assets for any reason to an affiliate without prior approval from the Department, if any of the following criteria apply:
  - 6.1.3.1 Risk-based capital ratio was less than two (2) for the most recent year filing, per RSA 404-F:14 (III); and
  - 6.1.3.2 The MCO was not in compliance with the NHID solvency requirement.
- 6.1.4 The MCO shall notify the Department within ten (10) calendar days when its agreement with an independent auditor or actuary has ended and seek approval of, and the name of the replacement auditor or actuary, if any from the Department.
- 6.1.5 The MCO shall maintain current assets, plus long-term investments that can be converted to cash within seven (7) calendar days without incurring a penalty of more than twenty percent (20%) that equal or exceed current liabilities.
- 6.1.6 The MCO shall submit Confidential Data on the basis of which the Department has the ability to determine that the MCO has made adequate provisions against the risk of insolvency.
- 6.1.7 The MCO shall inform the Department and NHID staff by phone and by email within five (5) business days of when any key personnel learn of any actual or threatened litigation, investigation, complaint, claim, or transaction that may reasonably be considered to have a material financial impact on and/or materially impact or impair the ability of the MCO to perform under this Agreement.
- 6.1.8 The MCO shall prohibit clawback business arrangements whereby Pharmacy Benefit Managers (PBM) reimburse network pharmacies an initial drug reimbursement amount and dispensing fee, and subsequently the PBM receives remuneration for a portion of that fee that is unreported to the Department and its actuary.

Page 366 of 397	Date	

New Hampshire Department of Health and Human Services Medicaid Care Management Services



### **Exhibit B - DRAFT**

# 6.2 **Capitation Payments**

- 6.2.1 Capitation payments made by the Department and retained by the MCO shall be for Medicaid-eligible Members. [42 CFR 438.3(c)(2)]
  - 6.2.1.1 The per member per month (PMPM) capitation rates for the current contract period are shown in Exhibit C: Payment Terms.
  - 6.2.1.2 For each of the subsequent years of the Agreement, actuarially sound per Member, per month capitated rates shall be paid as calculated and certified by the Department's actuary, subject to approval by CMS and Governor and Executive Council.
  - 6.2.1.3 Any rate adjustments shall be subject to the availability of State appropriations.
  - 6.2.1.4 Capitation rates shall be based on generally accepted actuarial principles and practices that are applied to determine aggregate utilization patterns, are appropriate for the population and services to be covered, and have been certified by actuaries who meet the qualification standards established by the Actuarial Standards Board. [42 CFR 457.10]
- 6.2.2 In the event the MCO incurs costs in the performance of this Agreement that exceed the capitation payments, the State and its agencies are not responsible for those costs and shall not provide additional payments to cover such costs.
- 6.2.3 Capitation rates shall use an actuarially sound prospective risk adjustment model to adjust the rates for each participating MCO.
  - 6.2.3.1 The risk adjustment process shall use the most recent version of the CDPS+Rx model to assign scored individuals to a demographic category and disease categories based on their medical claims and drug utilization during the study period. The methodology shall also incorporate a custom risk weight related to the cost of opioid addiction services, as deemed necessary by the Department and its actuary. Scored individuals are those with at least six months of eligibility and claims experience in the base data. The methodology shall exclude diagnosis codes related to radiology and laboratory services to avoid including false positive diagnostic indicators for tests run on an individual. Additionally, each scored member with less than 12 months of experience in the base data period shall also be assigned a durational adjustment to compensate for missing diagnoses due to shorter enrollment durations, similar to a missing data adjustment.
  - 6.2.3.2 Each unscored member shall be assigned a demographic-only risk weight instead of receiving the average risk score for each MCO's scored members in the same rate cell. The risk adjustment methodology shall also incorporate a specific adjustment to

Page 367 of 397 Date \_\_\_\_\_

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

address cost and acuity differences between the scored and unscored populations, which shall be documented by a thorough review of historical data for those populations based on generally accepted actuarial techniques.

- 6.2.3.3 Members shall be assigned to MCOs and rate cells using the actual enrollment by MCO in each quarter to calculate risk scores in order to capture actual membership growth for each MCO.
- 6.2.3.4 The Department and its actuary reserve the right to modify the risk adjustment methodology.
- 6.2.4 The MCO shall report to the Department within sixty (60) calendar days upon identifying any capitation or other payments in excess of amounts provided in this Agreement. [42 CFR 438.608(c)(3)]
- 6.2.5 The MCO and the Department agree the Capitation Rates may be adjusted periodically (at least annually) to maintain actuarial soundness as determined by the Department's actuary, subject to approval by CMS and Governor and Executive Council.
- 6.2.6 The MCO shall submit Confidential Data on the basis of which the State certifies the actuarial soundness of capitation rates to an MCO, including base Confidential Data that is generated by the MCO. [42 CFR 438.604(a)(2); 42 CFR 438.606; 42 CFR 438.3; 42 CFR 438.5(c)]
- 6.2.7 When requested by the Department, the MCO shall submit Encounter Data, financial data, and other Confidential Data to the Department to ensure actuarial soundness in development of the capitation rates, or any other actuarial analysis required by the Department or State or federal law.
- 6.2.8 The MCO's CFO shall submit and concurrently certify to the best of their information, knowledge, and belief that all Confidential Data and information described in 42 CFR 438.604(a), which the Department uses to determine the capitation rates, is complete and accurate. [42 CFR 438.606]
- 6.2.9 The MCO has responsibility for implementing systems and protocols to maximize the collection of TPL recoveries and subrogation activities. The MCO may retain such recoveries, subject to the parameters in the Agreement, since the capitation rates are calculated net of expected MCO recoveries.
- 6.2.10 The Department shall make a monthly payment to the MCO for each Member enrolled in the MCO's plan as the Department currently structures its capitation payments.
  - 6.2.10.1 Capitation Payments for all standard Medicaid Members shall be made retrospectively with a one month plus five (5) business day lag.
  - 6.2.10.2 Capitation Payments for all Granite Advantage Members shall be made before the end of each month of coverage.

Page 368 of 397	Date	

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



# **Exhibit B - DRAFT**

- 6.2.11 The capitation rate cell is determined based on the Member characteristics as of the earliest date of Member plan enrollment span(s) within the month.
- 6.2.12 The capitation rate does not change during the month, regardless of Member changes (e.g., age), unless the Member's plan enrollment is terminated and the Member is re-enrolled resulting in multiple spans within the month.
- 6.2.13 Capitation adjustments are processed systematically each month by the Department's MMIS.
- 6.2.14 The Department shall make systematic adjustments based on factors that affect rate cell assignment or plan enrollment.
- 6.2.15 If a Member is deceased, the Department shall recoup any and all capitation payments after the Member's date of death including any prorated share of a capitation payment intended to cover dates of services after the Member's date of death.

#### 6.2.16 Capitation Settlement

- 6.2.16.1 The Department has sole discretion over the capitation settlement process.
- 6.2.16.2 The MCO shall follow policies and procedures for the settlement process as developed by the Department.
- 6.2.16.3 Based on the provisions herein, the Department shall not make any further retroactive adjustments other than those described herein or elsewhere in this Agreement.
- 6.2.16.4 The Department and the MCO agree there is a nine (9) month limitation from the date of the Capitation Payment and is applicable only to retroactive Capitation Payments described herein, and shall in no way be construed to limit the effective date of enrollment in the MCO.
- 6.2.16.5 The Department shall have the discretion to recoup payments retroactively up to twenty-four (24) months for Members whom the Department later determines were not eligible for Medicaid during the enrollment month for which Capitation Payment was made.
- 6.2.16.6 For each live birth, the Department shall:
  - 6.2.16.6.1 Make a one-time maternity kick payment to the MCO with whom the mother is enrolled on the DOB.
  - 6.2.16.6.2 This payment is a global fee to cover all delivery care.
  - 6.2.16.6.3 In the event of a multiple birth DHHS shall make only one (1) maternity kick payment.
  - 6.2.16.6.4 A live birth is defined in accordance with NH Vital Records reporting requirements for live births as specified in RSA 5-C.

Page 369 of 397 Date \_\_\_\_\_

New Hampshire Department of Health and Human Services Medicaid Care Management Services



# **Exhibit B - DRAFT**

0.2.10.7		olled on the DOB.
	6.2.16.7.1	This payment is a global fee to cover all newborn expenses incurred in the first two (2) full or partial calendar months of life, including all hospital, professional, pharmacy, and other services.
	6.2.16.7.2	Enrolled babies shall be covered under the MCO capitated rates thereafter.
	6.2.16.7.3	Different rates of newborn kick payments may be employed by the Department, in its sole discretion, to increase actuarial soundness.
	6.2.16.7.4	Two (2) newborn kick payments shall be employed, one (1) for newborns with NAS and one (1) for all other newborns.
	6.2.16.7.5	Each type of payment is distinct and only one payment is made per newborn.
	6.2.16.7.6	The MCO shall submit information on maternity and newborn events to DHHS, and shall follow written policies and procedures, as developed by DHHS, for receiving, processing and reconciling maternity and newborn payments.
6.2.16.8	period extens the MCO for	ed ending August 31, 2024 (subject to future rating sion(s)), DHHS shall make a one-time kick payment to each Member psychiatric admission stay with DRG 87, except as described in Section 6.2.17.3 below.
6.2.16.9	Groups estab	yment shall be specific to the corresponding Peer blished by DHHS. Separate kick payments exist for 01 and 07, Peer Group 02, Peer Group 06, and Peer
6.2.16.10		dmissions for dually eligible Members are not subject yment and shall be paid out of the capitation rates.
6.2.16.11	and Hampste	dmissions for Members at New Hampshire Hospital and Hospital are not subject to the kick payment and out of the MCO's capitation rates.
6.2.16.12	Payment for b	behavioral health rate cells shall be determined based

6.2.16.7 Make a one-time newborn kick payment to the MCO with whom the

Page 370 of 397 Date \_\_\_\_\_

on a Member's CMH Program or CMH Provider behavioral certification level as supplied in an interface to the Department's

based on a Member having had an encounter in the last six (6)

6.2.16.13 The CMH Program or CMH Provider behavioral certification level is

MMIS by the MCO.

months.

New Hampshire Department of Health and Human Services Medicaid Care Management Services



#### **Exhibit B - DRAFT**

6.2.16.14 Changes in the certification level for a Member shall be reflected as of the first of each month and does not change during the month.

#### 6.2.17 Capitation Adjustments

- 6.2.17.1 After the completion of each Agreement year, an actuarially sound withhold percentage of each MCO's risk adjusted Capitation Payment net of directed payments to the MCO shall be calculated as having been withheld by the Department. On the basis of the MCO's performance, as determined under DHHS's MCM Withhold and Incentive Guidance.
  - 6.2.17.1.1 Details of the MCM Withhold and Incentive Program are described in MCM Withhold and Incentive Program Guidance provided by the Department as indicated in Section 5.4 (Withhold and Incentive Payment Program).
  - 6.2.17.1.2 The Department shall inform the MCO of any required program revisions or additions in a timely manner.
  - 6.2.17.1.3 The Department may adjust the rates to reflect these changes as necessary to maintain actuarial soundness.
- 6.2.17.2 The Department shall only make a monthly capitation payment to the MCO for a Member aged 21–64 receiving inpatient treatment in an IMD, as defined in 42 CFR 435.1010, so long as the facility is a hospital providing psychiatric or substance use disorder inpatient care or a sub-acute facility providing psychiatric or substance use disorder crisis residential services permitted by CMS through a waiver obtained from CMS. [42 CFR 438.6(e)]
- 6.2.17.3 In the event an enrolled Medicaid Member was previously admitted as a hospital inpatient and is receiving continued inpatient hospital services on the first day of coverage with the MCO, the MCO shall receive the applicable capitation payment for that Member.
- 6.2.17.4 The entity responsible for coverage of the Member at the time of admission as an inpatient (either DHHS or another MCO) shall be fully responsible for all inpatient care services and all related services authorized while the Member was an inpatient until the day of discharge from the hospital.
- 6.2.17.5 Should any part of the scope of work under this Agreement relate to a State program that is no longer authorized by law (e.g., which has been vacated by a court of law, or for which CMS has withdrawn federal authority, or which is the subject of a legislative repeal), the MCO must do no work on that part after the effective date of the loss of program authority.

Page 371 of 397	Date
-----------------	------

New Hampshire Department of Health and Human Services Medicaid Care Management Services



### **Exhibit B - DRAFT**

- 6.2.17.6 The State must adjust capitation rates to remove costs that are specific to any program or activity that is no longer authorized by law.
- 6.2.17.7 If the MCO works on a program or activity no longer authorized by law after the date the legal authority for the work ends, the MCO will not be paid for that work.
- 6.2.17.8 If the State paid the MCO in advance to work on a no-longerauthorized program or activity and under the terms of this contract the work was to be performed after the date the legal authority ended, the payment for that work should be returned to the State.
- 6.2.17.9 However, if the MCO worked on a program or activity prior to the date legal authority ended for that program or activity, and the State included the cost of performing that work in its payments to the MCO, the MCO may keep the payment for that work even if the payment was made after the date the program or activity lost legal authority.

## 6.2.18 Other Reimbursement Considerations

6.2.18.1 Unless MCOs are exempted, through legislation or otherwise, from having to make payments to the NH Insurance Administrative Fund (Fund) pursuant to RSA 400-A:39, the Department shall reimburse MCO for MCO's annual payment to the Fund on a supplemental basis within 30 days following receipt of invoice from the MCO and verification of payment by the NHID.

#### 6.3 Medical Loss Ratio (MLR) Reporting and Settlement

#### 6.3.1 Minimum MLR Performance and Rebate Requirements

- 6.3.1.1 The MCO shall meet a minimum MLR of eighty-five percent (85%) or higher.
- 6.3.1.2 In the event the MCO's MLR for any single reporting year is below the minimum of the eighty-five percent (85%) requirement, the MCO shall provide to the Department a rebate, no later than sixty (60) calendar days following the Department notification, that amounts to the difference between the total amount of Capitation Payments received by the MCO from the Department multiplied by the required MLR of eighty-five percent (85%) and the MCO's actual MLR.
- 6.3.1.3 If the MCO fails to pay any rebate owed to the Department in accordance with the time periods set forth by the Department, in addition to providing the required rebate to the Department, the MCO shall pay the Department interest at the current Federal Reserve Board lending rate or ten percent (10%) annually, whichever is higher, on the total amount of the rebate.

Page 372 of 397	Date	
-----------------	------	--

New Hampshire Department of Health and Human Services Medicaid Care Management Services



# **Exhibit B - DRAFT**

#### 6.3.2 Calculation of the MLR

- 6.3.2.1 The MCO shall calculate and report to the Department the MLR for each MLR reporting year, in accordance with 42 CFR 438.8 and the standards described within this Agreement. [42 CFR 438.8(a)]
- 6.3.2.2 The MLR calculation is the ratio of the numerator (as defined in accordance with 42 CFR 438.8(e)) to the denominator (as defined in accordance with 42 CFR 438.8(f)). [42 CFR 438.8 (d-f)]
  - 6.3.2.2.1 The calculation of the MLR will be updated to consider new provisions added or amended by CMS through published rules and guidance.
- 6.3.2.3 Each MCO expense shall be included under only one (1) type of expense, unless a portion of the expense fits under the definition of, or criteria for, one (1) type of expense and the remainder fits into a different type of expense, in which case the expense shall be prorated between the two types of expenses.
  - 6.3.2.3.1 Expenditures that benefit multiple contracts or populations, or contracts other than those being reported, shall be reported on a pro rata basis. [42 CFR 438.8(g)(1)(i)-(ii)]
- 6.3.2.4 Expense allocation shall be based on a generally accepted accounting method that is extended to yield the most accurate results.
  - 6.3.2.4.1 Shared expenses, including expenses under the terms of a management contract, shall be apportioned pro rata to the contract incurring the expense.
  - 6.3.2.4.2 Expenses that relate solely to the operation of a reporting entity, such as personnel costs associated with the adjusting and paying of claims, shall be borne solely by the reporting entity and are not to be apportioned to other entities. [42 CFR 438.8(g)(2)(i)-(iii)]
- 6.3.2.5 The MLR report must include non-claims costs, which are those expenses for administrative services that are not: incurred claims, expenditures for activities that improve health care quality or licensing and regulatory fees or federal and state taxes.
  - 6.3.2.5.1 Revenue and expenses for administrative services should exclude the Health Insurer Tax, any allocation for premium taxes and any other revenue-based assessments.

Page 373 of 397	Date
-----------------	------

New Hampshire Department of Health and Human Services Medicaid Care Management Services



#### **Exhibit B - DRAFT**

- 6.3.2.5.2 Expenses for administrative services may include amounts that exceed a third party's costs (profit margin), but these amounts must be justified and consistent with prudent management and fiscal soundness requirements to be includable when these transactions are between related parties. [42 C.F.R. § 422.516(b)].
- 6.3.2.6 Health Care Quality Improvement (HCQI) expenses are limited to the portion of salaries and benefits for employees directly performing administrative functions for inclusion in the MLR calculation. Expenses for items such as office space (including rent or depreciation, facility maintenance, janitorial, utilities, property taxes, insurance, wall art), human resources, salaries of counsel and executives, equipment, computer and telephone usage, travel and entertainment, company parties and retreats, information technology infrastructure and systems, and software licenses do not qualify as direct HCQI expenses.
- 6.3.2.7 The MCO may add a credibility adjustment in accordance with 42 CFR 438.8(h) to a calculated MLR if the MLR reporting year experience is partially credible.
  - 6.3.2.7.1 The credibility adjustment, if included, shall be added to the reported MLR calculation prior to calculating any remittances.
  - 6.3.2.7.2 The MCO may not add a credibility adjustment to a calculated MLR if the MLR reporting year experience is fully credible.
  - 6.3.2.7.3 If the MCO's experience is non-credible, it is presumed to meet or exceed the MLR calculation standards. [42 CFR 438.8(h)(1)-(3)]

# 6.3.3 MLR Reporting

- 6.3.3.1 The MCO shall submit MLR summary reports quarterly to the Department in accordance with Exhibit O: Quality and Oversight Reporting Requirements. [42 CFR 438.8(k)(2); 42 CFR 438.8(k)(1)].
- 6.3.3.2 The MLR summary reports shall include all information required by 42 CFR 438.8(k) within nine (9) months of the end of the MLR reporting year, including:
  - 6.3.3.2.1 Total incurred claims:
  - 6.3.3.2.2 Expenditures on quality improvement activities;
  - 6.3.3.2.3 Expenditures related to activities compliant with the program integrity requirements;

	Page 374 of 397	Date
--	-----------------	------

6.3.3.2.4

Medicaid Care Management Services Contract New Hampshire Department of Health and Human Services Medicaid Care Management Services

6.3.3.3

6.3.3.4

6.3.3.5

6.3.3.6

6.3.3.7



# **Exhibit B - DRAFT**

Non-claims costs;

0.0.0.2.1	Tion dame dode,
6.3.3.2.5	Premium revenue;
6.3.3.2.6	Taxes;
6.3.3.2.7	Licensing fees;
6.3.3.2.8	Regulatory fees;
6.3.3.2.9	Methodology(ies) for allocation of expenditures;
6.3.3.2.10	Any credibility adjustment applied;
6.3.3.2.11	The calculated MLR;
6.3.3.2.12	Any remittance owed to the State, if applicable;
6.3.3.2.13	A comparison of the information reported with the audited financial report;
6.3.3.2.14	A description of the aggregate method used to calculate total incurred claims; and
6.3.3.2.15	The number of Member months. [42 CFR $438.8(k)(1)(i-xiii)$ ; 42 CFR $438.608(a)(1-5)$ ; 42 CFR $438.608(a)(7-8)$ ; 42 CFR $438.608(b)$ ; 42 CFR $438.8(i)$ ]
calculation of	all attest to the accuracy of the summary reports and the MLR when submitting its MLR summary reports ment. [42 CFR 438.8(n); 42 CFR 438.8(k)]
	ary reports shall be based on a template provided by ent within sixty (60) calendar days of the Program Start R 438.8(a)]
Data for all	all in its MLR summary reports aggregate Confidential Medicaid eligibility groups covered under this nless otherwise required by the Department.
adjudication associated w eighty (180) of within thirty (3 comes soon	shall require any Subcontractor providing claims activities to provide all underlying Confidential Data ith MLR reporting to the MCO within one hundred and calendar days or the end of the MLR reporting year or 30) calendar days of a request by the MCO, whichever iter, regardless of current contract limitations, to divalidate the accuracy of MLR reporting. [42 CFR]
change to the	nce in which the Department makes a retroactive e Capitation Payments for a MLR reporting year and ort has already been submitted to the Department, the

Page 375 of 397 Date \_\_\_\_\_

New Hampshire Department of Health and Human Services Medicaid Care Management Services



#### **Exhibit B - DRAFT**

- 6.3.3.7.1 Re-calculate the MLR for all MLR reporting years affected by the change; and
- 6.3.3.7.2 Submit a new MLR report meeting the applicable requirements. [42 CFR 438.8(m); 42 CFR 438.8(k)]
- 6.3.3.8 The MCO and its Subcontractors (as applicable) shall retain MLR reports for a period of no less than ten (10) years.

# 6.3.4 **Risk Mitigation**

# 6.3.4.1 Stop-Loss Protections

- 6.3.4.1.1 The Department will provide an actuarially sound High-Cost Pharmacy Risk Pool (HCPRP) funded though the MCO capitation rates that will allocate HCPRP funding to each MCO based on the qualifying pharmacy claim payments for Members with annual pharmacy claim payments over a specified threshold. The HCPRP will provide MCO protection for Members with pharmacy claims in access of the attachment point. Detailed program features and parameters will be established on an annual basis in quidance.
- 6.3.4.2 The Department shall implement a budget neutral-risk pool for services provided at Boston Children's Hospital in order to better allocate funds based on MCO-specific spending for these services. Inpatient and outpatient facility services provided at Boston Children's Hospital qualify for risk pool calculation.

#### 6.3.5 Minimum and Maximum MLR

- 6.3.5.1 The Department reserves the right to modify its risk mitigation strategies in accordance with actuarially sound practices.
- 6.3.5.2 For each year under this Agreement, the Department and its actuary will determine if a minimum and maximum MLR should be implemented due to unforeseen events that could materially impact the level of uncertainty associated with the financial soundness of the MCM program. For example, for the period July 1, 2023 through August 31, 2024, the target MLR for at-risk services is 90.8 for Standard Medicaid and 89.9 for GAHCP based on the SFY 2023 projected enrollment distribution. Please note, each program's target MLR may change in future rate amendments as a result of changes to underlying assumptions, such as enrollment projections, emerging utilization experience, and the retroactive acuity adjustments, if applicable, as described in the State's capitation rate letter, exhibits, and certification filed with the Centers for Medicare and Medicaid Services for the period based on the target MLRs below:

Page 376 of 397 Date \_\_\_\_\_

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

- 6.3.5.2.1 The minimum MLR is set on a program-wide basis for each major population, such that the maximum profit achievable is 4.0%, which is equal to the 1.5% target margin plus the amount between the target MLR and the minimum MLR 2.5%. The minimum MLR shall be 88.3% for the Standard Medicaid population and 87.4% for the GAHCP population.
- 6.3.5.2.2 The maximum MLR is also set on a program-wide basis for each major population 3.5% above the target MLR, such that the MCOs will have a maximum loss of 2.0%. Based on the target MLRs, the maximum MLR shall be 94.3% for the Standard Medicaid population and 92.4% for the GAHCP population.
- 6.3.5.2.3 The final maximum and minimum MLRs for Standard Medicaid and GAHCP shall be updated for the final retroactive acuity factor and any other changes implemented in the contract amendment, as applicable.
- 6.3.5.2.4 The settlement shall be done separately for the Standard Medicaid and GAHCP populations.
- 6.3.5.3 Other MCM program risk mitigation provisions shall apply prior to the minimum and maximum MLR calculation (i.e., High-Cost Drug Risk Pool, Boston Children's Hospital risk pool, high-cost patient stop-loss arrangement, prospective risk adjustment, and retrospective acuity adjustment), if applicable, as described in the State's capitation rate letter, exhibits, and certification filed with the Centers for Medicare and Medicaid Services for the period.
- 6.3.5.4 Minimum MLR settlement operational requirements include:
  - 6.3.5.4.1 The numerator for the actual MLR shall include all payments made to providers, such as fee-for-service payments, sub-capitation payments, incentive payments, and settlement payments. The numerator for the actual MLR shall not include costs related to quality improvement activities or Fraud, Waste and Abuse prevention.
  - 6.3.5.4.2 The denominator for the actual MLR shall equal the risk adjusted capitation revenue including risk mitigation settlement amounts as described in Section 6.3.5 (Risk Mitigation).
  - 6.3.5.4.3 Payments and revenue related to directed payments and premium taxes shall be excluded from the numerator and denominator for the actual MLR.

Page 377 of 397 Date \_\_\_\_\_

New Hampshire Department of Health and Human Services **Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

6.3.5.4.4	Any incentive payments made to higher-performing MCOs as part of the Withhold Program shall not impact the minimum or maximum MLR provision of the Agreement.
6.3.5.4.5	The timing of the minimum and maximum MLR settlement shall occur after the contract year is closed and substantial paid claims runout is available.
6.3.5.4.6	Payments or recoupments related to the Withhold and Incentive Program shall be excluded from the MLR settlement. The Withhold and Incentive

6.3.5.4.7 The MLR settlement shall occur after the contract year is closed and sufficient paid claims runout is available.

Program settlement shall occur after the MLR

#### 6.4 **Financial Responsibility for Dual-Eligible Members**

6.4.1 For Medicare Part A crossover claims and Medicare Part B crossover claims billed on the UB-04, the MCO shall pay the patient responsibility amount (deductible and coinsurance) for Covered Services.

settlement is complete.

- 6.4.2 For Part B crossover claims billed on the CMS-1500, the MCO shall pay the lesser of:
  - The patient responsibility amount (deductible and coinsurance) for 6.4.2.1 Covered Services, or
  - 6.4.2.2 The difference between the amount paid by the primary payer and the Medicaid allowed amount.
  - 6.4.2.3 For both Medicare Part A and Part B claims, if the Member responsibility amount is "0" then the MCO shall make no payment.

#### 6.5 **Medical Cost Accruals**

6.5.1 The MCO shall establish and maintain an actuarially sound process to estimate Incurred But Not Reported (IBNR) claims, services rendered for which claims have not been received.

#### 6.6 Audits

- The MCO shall permit the Department or its designee(s) and/or the NHID to 6.6.1 inspect and audit any of the financial records of the MCO and its Subcontractors.
- 6.6.2 There shall be no restrictions on the right of the State or federal government to conduct whatever inspections and audits are necessary to assure quality, appropriateness or timeliness of services and reasonableness of their costs. [SMM 2087.7; 42 CFR 434.6(a)(5)]

Page 378 of 397 Date \_\_\_\_\_

New Hampshire Department of Health and Human Services Medicaid Care Management Services



### **Exhibit B - DRAFT**

- 6.6.3 The MCO shall file annual and interim financial statements in accordance with the standards set forth in this Section 6 (Financial Management) of this Agreement.
- 6.6.4 Within one hundred and eighty (180) calendar days or other mutually agreed upon date following the end of each calendar year during this Agreement, the MCO shall file, in the form and content prescribed by the NAIC, annual audited financial statements that have been audited by an independent Certified Public Accountant.
- 6.6.5 Financial statements shall be submitted in either paper format or electronic format, provided that all electronic submissions shall be in be sent encrypted, if PHI or PII is included, and PDF format or another read-only format that maintains the documents' security and integrity.
- 6.6.6 The MCO shall also file, within seventy-five (75) calendar days following the end of each calendar year, certified copies of the annual statement and reports as prescribed and adopted by the NHID.
- 6.6.7 The MCO shall file within sixty (60) calendar days following the end of each calendar quarter, quarterly financial reports in form and content as prescribed by the NAIC.

## 6.7 **Member Liability**

- 6.7.1 The MCO shall not hold MCM Members liable for:
  - 6.7.1.1 The MCO's debts, in the event of the MCO's insolvency;
  - 6.7.1.2 The Covered Services provided to the Member, for which the State does not pay the MCO;
  - 6.7.1.3 The Covered Services provided to the Member, for which the State, or the MCO does not pay the individual or health care Provider that furnishes the services under a contractual, referral, or other arrangement; or
  - 6.7.1.4 Payments for Covered Services furnished under an agreement, referral, or other arrangement, to the extent that those payments are in excess of the amount that the Member would owe if the MCO provided those services directly. [42 CFR 438.106(a)-(c); section 1932(b)(6) of the Social Security Act; 42 CFR 438.3(k); 42 CFR 438.230]
- 6.7.2 The MCO shall provide assurances satisfactory to the Department that its provision against the risk of insolvency is adequate to ensure that Medicaid Members shall not be liable for the MCO's debt if the MCO becomes insolvent. [42 CFR 438.116(a)]
- 6.7.3 Subcontractors and Referral Providers may not bill Members any amount greater than would be owed if the entity provided the services directly

Page 379 of 397	Date
. age c. c c. cc.	24.0

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



### **Exhibit B - DRAFT**

- [Section 1932(b)(6) of the SSA; 42 CFR 438.106(c); 42 CFR 438.3(k); 42 CFR 438.230; SMDL 12/30/97].
- 6.7.4 The MCO shall cover services to Members for the period for which payment has been made, as well as for inpatient admissions up until discharge during insolvency. [SMM 2086.6B]
- 6.7.5 The MCO shall meet the Department's solvency standards for private health maintenance organizations, or be licensed or certified by the Department as a risk-bearing entity. [Section 1903(m)(1) of the Social Security Act; 42 CFR 438.116(b)]

#### 6.8 **Denial of Payment**

- 6.8.1 Payments provided for under the Agreement shall be denied for new Members when, and for so long as, payment for those Members is denied by CMS.
- 6.8.2 CMS may deny payment to the State for new Members if its determination is not timely contested by the MCO. [42 CFR 438.726(b); 42 CFR 438.730(e)(1)(ii)]

## 6.9 Federal Matching Funds

- 6.9.1 Federal matching funds are not available for amounts expended for Providers excluded by Medicare, Medicaid, or CHIP, except for Emergency Services. [42 CFR 431.55(h) and 42 CFR 438.808; 1128(b)(8) and Section1903(i)(2) of the SSA; SMDL 12/30/97]
- 6.9.2 Payments made to such Providers are subject to recoupment from the MCO by the Department.

#### 6.10 Third Party Liability

- 6.10.1 NH Medicaid shall be the payer of last resort for all Covered Services in accordance with federal regulations.
- 6.10.2 The MCO shall develop and implement policies and procedures to meet its obligations regarding TPL. [42 CFR 433 Sub D; 42 CFR 447.20]
- 6.10.3 The Department and the MCO shall cooperate in implementing cost avoidance and cost recovery activities.
- 6.10.4 The MCO shall be responsible for making every reasonable effort to determine the liable third party to pay for services rendered and cost avoid and/or recover any such liabilities from the third party.
- 6.10.5 The Department shall conduct two (2) policy and procedure audits of the MCO and its Subcontractors.
  - 6.10.5.1 Noncompliance with CAPs issued due to deficiencies may result in liquidated damages as outlined in Exhibit N.

Page 380 of 397 Date \_\_\_\_\_

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



# **Exhibit B - DRAFT**

- 6.10.6 The MCO shall have one (1) dedicated contact person for the Department for TPL.
- 6.10.7 The Department and/or its actuary shall identify a market-expected median TPL percentage amount and deduct an appropriate amount from the gross medical costs included in the Department Capitation Payment rate setting process.
- 6.10.8 All cost recovery amounts shall be retained by the MCO, except overpayments by other insurance. For recoveries over the Provider paid amount see 6.10.12.5.3.
- 6.10.9 The MCO and its Subcontractors shall comply with all regulations and State laws related to TPL, including but not limited to:
  - 6.10.9.1 42 CFR 433.138;
  - 6.10.9.2 42 CFR 433.139; and
  - 6.10.9.3 RSA 167:14-a.

#### 6.10.10 Cost Avoidance

- 6.10.10.1 The MCO and its Subcontractors performing claims processing duties shall be responsible for cost avoidance through the Coordination of Benefits (COB) relating to federal and private health insurance resources, including but not limited to Medicare, Medicare Advantage plans, private health insurance, Employees Retirement Income Security Act of 1974 (ERISA), 42 U.S.C. 1396a(a)(25) plans and workers compensation.
- 6.10.10.2 The MCO shall establish claims edits and deny payment of claims when active Medicare, Medicare Advantage Plans, or active private insurance exist at the time the claim is adjudicated and the claim does not reflect payment from the other payer.
- 6.10.10.3 The MCO shall deny payment on a claim that has been denied by Medicare Advantage Plan or private insurance when the reason for denial is the Provider or Member's failure to follow prescribed procedures including, but not limited to, failure to obtain Prior Authorization or timely claim filing.
- 6.10.10.4 The MCO shall establish claim edits to ensure claims with Medicare, Medicare Advantage plan, or private insurance denials are properly adjudicated based on the denial reason. The MCO is required to determine which specific Medicare, Medicare Advantage plan, and private insurance denials should be processed for payment or denial by the MCO.
- 6.10.10.5 The MCO shall make its own independent decisions about approving claims for payment that have been denied by the private insurance, Medicare, or Medicare Advantage plans if either:

Page 381 of 397	Date

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

- 6.10.10.5.1 The primary payer does not cover the services and the MCO does; or
- 6.10.10.5.2 The service was denied as not Medically Necessary and the Provider followed the dispute resolution and/or appeal process of the private insurance or Medicare and the denial was upheld.
- 6.10.10.6 If a claim is denied by the MCO based on active Medicare, Medicare Advantage Plan, or private insurance, the MCO shall provide the Medicare, Medicare Advantage Plan, or private insurance information to the Provider.
- 6.10.10.7 To ensure the MCO is cost avoiding, the MCO shall implement a file transfer protocol between the Department MMIS and the MCO's MCIS to send new, terminated, and changed Medicare or private insurance information and other information as required pursuant to 42 CFR 433.138.
- 6.10.10.8 The MCO shall implement a nightly file transfer protocol with its Subcontractors to ensure Medicare, private health insurance, ERISA, 42 U.S.C. 1396a(a)(25) plans, and workers compensation policy information is updated and utilized to ensure claims are properly denied for Medicare or private insurance.
- 6.10.10.9 The MCO shall perform monthly electronic confidential data matches with private insurance companies (medical and pharmacy) unless the Department performs these functions.
  - 6.10.10.9.1 Should the Department establish data matching and provide to the MCO individual member private insurance data, then the MCO will not be required to perform direct data matching.
  - 6.10.10.9.2 The date of the Department transmission of the data will be considered the date of discovery for the plan regarding member private insurance. The MCO will be required to meet cost avoidance requirements outlined in this section of this Agreement within two (2) business days of the date of discovery and four (4) business days for any subcontractors.
  - 6.10.10.9.3 The Department shall provide the MCO with the Member name, Medicaid ID, private insurance company name, the Department's private insurance ID, private insurance policy number, type of coverage, policy begin date, policy end date (if open, end date will be 12/31/9999), and policy holder information, if available.

Page 382 of 397 Date \_\_\_\_\_

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



# **Exhibit B - DRAFT**

- 6.10.10.10 If the Department is not performing the data matching with other insurances, then it will be the responsibility of the MCO to establish, and shall ensure the MCO and its Subcontractors utilize, monthly electronic Confidential Data matches with private insurance companies (Medical and pharmacy), and Medicare Advantage plans that sell insurance in the State to obtain current and accurate private insurance information for their Members in accordance with this Agreement. This provision may be satisfied by a contract with a third-party vendor to the MCO or its Subcontractors.
- 6.10.10.11 Upon audit, the MCO shall demonstrate with written documentation that good faith efforts were made to establish Confidential Data matching agreements with insurers selling in the State who have refused to participate in Confidential Data matching agreements with the MCO. All communication with the insurer relating to and including the Confidential data matching agreements shall be in writing and in accordance with this Agreement
- 6.10.10.12 The MCO shall maintain the following private insurance Confidential Data within their system for all insurance policies that a Member may have and include for each policy:
  - 6.10.10.12.1 Member's first and last name;
  - 6.10.10.12.2 Member's policy number;
  - 6.10.10.12.3 Member's group number, if available;
  - 6.10.10.12.4 Policyholder's first and last name, if available;
  - 6.10.10.12.5 Policy coverage type to include at a minimum:
    - 6.10.10.12.5.1. Medical coverage (including, mental health, DME, Chiropractic, skilled nursing, home health, or other health coverage not listed below);
    - 6.10.10.12.5.2. Hospital coverage;
    - 6.10.10.12.5.3. Pharmacy coverage;
    - 6.10.10.12.5.4. Dental coverage; and
    - 6.10.10.12.5.5. Vision Coverage.
  - 6.10.10.12.6 Begin date of insurance; and
  - 6.10.10.12.7 End date of insurance (when terminated).
- 6.10.10.13 The MCO shall submit any new, changed, or terminated private insurance Confidential Data to the Department through file transfer on a monthly basis.

Page 383 of 397	Date	
-----------------	------	--

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



### **Exhibit B - DRAFT**

- 6.10.10.14 The MCO shall not cost avoid claims for preventive pediatric services (including EPSDT), that is covered under the Medicaid State Plan per 42 CFR 433.139(b)(3).
- 6.10.10.15 The MCO shall pay all preventive pediatric services and collect reimbursement from private insurance after the claim adjudicates.
- 6.10.10.16 The MCO shall pay the Provider for the Member's private insurance cost sharing (Copays and deductibles) up to the MCO Provider contract allowable or any other agreement to payment in the MCO/Provider contract.
- 6.10.10.17 The MCO shall disregard the TPL lesser of logic payment methodology for claims that require Medicaid or Medicare minimum fee schedule rates under this Agreement.
  - 6.10.10.17.1 The MCO shall pay the difference between the TPL amount and the minimum Medicaid or Medicare fee schedule amount required.
  - 6.10.10.17.2 If the TPL payment is more than the Medicaid or Medicare minimum fee schedule amount requirement, then the MCO pays nothing.
- 6.10.10.18 On a quarterly basis, the MCO shall submit a cost avoidance summary, as described in Exhibit O: Quality and Oversight Reporting Requirements.
- 6.10.10.19 This report shall reflect the number of claims and billed dollar amount avoided by private insurance, including Medicare and Medicare Advantage plans for all types of coverage as follows:
  - 6.10.10.19.1 Medical coverage (including, mental health, DME, Chiropractic, skilled nursing, home health, or other health coverage not listed below);
  - 6.10.10.19.2 Hospital coverage;
  - 6.10.10.19.3 Pharmacy coverage:
  - 6.10.10.19.4 Dental coverage; and
  - 6.10.10.19.5 Vision coverage.

# 6.10.11 Pay and Chase Private Insurance

6.10.11.1 If private insurance exists for services provided and paid by the MCO, but was not known by the MCO at time the claim was adjudicated, then the MCO shall pursue recovery of funds expended from the private insurance company, including Medicare Advantage plans.

age 384 of 397	Date

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

- 6.10.11.2 The MCO shall submit quarterly TPL billed and recovery reports, in accordance with Exhibit O: Quality and Oversight Reporting Requirements.
- 6.10.11.3 These reports shall reflect detail and summary information of the MCO's billing, collection efforts, and recovery from Standards Medicare, Medicare Advantage Plans, and private insurance for all types of coverage as follows:
  - 6.10.11.3.1 Medical coverage (including, mental health, DME, Chiropractic, skilled nursing, home health, or another other health coverage not listed below);
  - 6.10.11.3.2 Hospital coverage;
  - 6.10.11.3.3 Pharmacy coverage;
  - 6.10.11.3.4 Dental coverage; and
  - 6.10.11.3.5 Vision Coverage.
- 6.10.11.4 The MCO shall have eight (8) months from the original paid date to initiate recovery of funds from private insurance.
  - 6.10.11.4.1 The Department may, beginning one year from the date the MCO paid the claim, directly bill and recover the private insurance amount paid by the MCO but not collected. The Department shall inform the MCO in writing any claims in which the Department plans to pursue Pay and Chase recovery, and the Department shall retain any recovered funds.
  - 6.10.11.4.2 If a recovery is closed on the Exhibit O: Quality and Oversight Reporting Requirements TPLCOB.02 or TPLCOB.03 report for any reason, the Department has the right to initiate collections from private insurance, after the closure, and retain any funds recovered.
- 6.10.11.5 The MCO shall treat funds recovered from private insurance and Medicare Advantage plans as offsets to claims payments by posting within the claim system.
  - 6.10.11.5.1 The MCO shall post all payments to claim level detail by Member.
  - 6.10.11.5.2 Any Overpayment by private insurance can be applied to other claims not paid or covered by private insurance for the same Member.
  - 6.10.11.5.3 The MCO shall submit amounts beyond a Member's outstanding MCO payment to the Department semi-annually to determine if the Department has any

Page 385 of 397	Date	

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



### **Exhibit B - DRAFT**

claims to apply the funds. If there no claims in which to apply the funds, the MCO must return any remaining over payments to the Member annually.

- 6.10.11.6 The MCO and its Subcontractors shall not deny or delay approval of otherwise covered treatment or services based on TPL considerations, nor bill or pursue collection from a Member for services.
- 6.10.11.7 The MCO may neither unreasonably delay payment nor deny payment of claims unless the probable existence of TPL is established at the time the claim is adjudicated. [42 CFR 433 Sub D; 42 CFR 447.20]
- 6.10.11.8 The MCO or its Subcontractor shall follow up on any billed TPL that is not collected or properly denied by the other insurance once a \$1,500 cumulative minimum threshold for medical claims is reached per member or single claim of fifty dollars (\$50) and one hundred dollars (\$100) per cumulative prescription is reached per member.
- 6.10.11.9 Subrogation Recoveries
  - 6.10.11.9.1 The MCO shall be responsible for pursuing recoveries of claims paid when there is an accident or trauma in which there is a third party liable, such as automobile insurance, malpractice, lawsuit, including class action lawsuits.
  - 6.10.11.9.2 The MCO is responsible for class action lawsuits when the member is enrolled in an MCO on the date of injury and only includes MCO claims related to the class action. If the class action has fee for service and MCO claims, the Department is responsible for the case and will settle for both MCO and fee for service claims and will retain all funds.
  - 6.10.11.9.3 The MCO shall act upon any information from insurance carriers or attorneys regarding potential subrogation cases. The MCO shall be required to seek Subrogation amounts regardless of the amount believed to be available as required by federal Medicaid guidelines.
  - 6.10.11.9.4 The MCO shall establish detailed policies and procedures for determining, processing, and recovering funds based on accident and trauma Subrogation cases.
  - 6.10.11.9.5 The MCO shall submit its policies and procedures, including those related to their case tracking system

Page 386 of 397	Date	

New Hampshire Department of Health and Human Services **Medicaid Care Management Services** 



# **Exhibit B - DRAFT**

as	describ	ed i	n Sec	tion	6.10	0.1	1.9.7	of	this
Agr	eement,	to the	Depart	ment f	or a	рр	roval d	luring	the
Rea	adiness F	Revie	w proce	ess. Th	ne N	ЛC	O shal	I hav	e in
its	policies	and	proced	lures,	at	а	minim	num,	the
follo	wing:								

- 6.10.11.9.5.1. The MCO shall establish a paid claims review process based on diagnosis and trauma codes to identify claims that may constitute an accident or trauma in which there may be a liable third party.
- 6.10.11.9.5.2. The claims required to be identified, at a minimum, should include ICD-10 diagnosis codes related to accident or injury and claims with an accident trauma indicator of "Y".
- 6.10.11.9.5.3. The MCO shall present a list of ICD-10 diagnostic codes to the Department for approval in identifying claims for review.
- 6.10.11.9.5.4. The Department reserves the right to require specific codes be reviewed by MCO.
- 6.10.11.9.5.5. The MCO shall establish a monthly process request additional to information from Members to determine if there is a liable third party for any accident or trauma related claims by establishing a questionnaire to be sent to Members.
- 6.10.11.9.5.6. The MCO shall submit a report of questionnaires generated and sent as described in Exhibit O: Quality and Oversight Reporting Requirements.
- 6.10.11.9.5.7. The MCO shall establish timeframes and claim logic for determining when additional letters to Members should be sent relating to specific accident

	diagnosis codes and indictors.
6.10.11.9.5.8.	The MCO shall respond to accident referrals and lien request within twenty-one (21) calendar days of the notice per RSA 167:14-a.

Page 387 of 397 Date \_\_\_\_\_

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



### **Exhibit B - DRAFT**

6.10.11.9.6	The MCO shall establish a case tracking system to
	monitor and manage Subrogation cases.

- 6.10.11.9.7 This system shall allow for reporting of case status at the request of DHHS, OIG, CMS, and any of their designees. The tracking system shall, at a minimum, maintain the following record:
  - 6.10.11.9.7.1. Date inquiry letter sent to Member, if applicable;
  - 6.10.11.9.7.2. Date inquiry letter received back from Member, if applicable;
  - 6.10.11.9.7.3. Date of contact with insurance company, attorney, or Member informing the MCO of an accident;
  - 6.10.11.9.7.4. Date case is established;
  - 6.10.11.9.7.5. Date of incident;
  - 6.10.11.9.7.6. Reason for incident;
  - 6.10.11.9.7.7. Claims associated with incident;
  - 6.10.11.9.7.8. All correspondence and dates;
  - 6.10.11.9.7.9. Case comments by date;
  - 6.10.11.9.7.10. Lien amount and date updated;
  - 6.10.11.9.7.11. Settlement amount:
  - 6.10.11.9.7.12. Date settlement funds received; and
  - 6.10.11.9.7.13. Date case closed.
- 6.10.11.9.8 The MCO shall submit Subrogation reports in accordance with Exhibit O: Quality and Oversight Reporting Requirements. [42 CFR 433 Sub D; 42 CFR 447.20]
- 6.10.11.9.9 DHHS shall inform the MCO of any claims related to an MCO Subrogation cases. The MCO shall pursue the Department's claim recovery as part of their case.
- 6.10.11.9.10 The MCO shall submit to the Department any and all information regarding the case upon request if the Department also has a Subrogation lien.
- 6.10.11.9.11 The MCO shall coordinate with the Department on any dual Subrogation settlement recoveries identified in writing by the Department.
  - 6.10.11.9.11.1. The MCO shall pay the Department claims first in the event of any settlement

Page 388 of 397	Date	
-----------------	------	--

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



#### **Exhibit B - DRAFT**

less than the combined total MCO and Department lien amount.

- 6.10.11.9.11.2. The MCO shall be liable for repayment to the Department for the total Department lien amount in situations when the Department informed the MCO of the State's lien in advance of the settlement, regardless of whether the Department lien amount exceeds the total settlement amount recovered when the MCO settles a subrogation case and accepts a settlement amount without written authorization from the Department.
- 6.10.11.9.12 If the MCO notifies the Department that they have closed a case prior to the case settling, the Department reserves the right to pursue and retain payment of any remaining paid MCO and FFS claims related to the case.
- 6.10.11.9.13 The MCO shall submit to the Department for approval any Subrogation proposed settlement agreement that is less than eighty percent (80%) of the total lien in which the MCO intends to accept prior to acceptance of the settlement.
- 6.10.11.9.14 The Department shall have twenty (20) business days to review the case once the MCO provides all relevant information as determined by the Department to approve the settlement from date received from the MCO.
- 6.10.11.9.15 If the Department does not respond within twenty (20) business days, the MCO may proceed with settlement.
- 6.10.11.9.16 If the Department does not approve of the settlement agreement, then the Department may work with the MCO and other parties on the settlement.
- 6.10.11.9.17 The MCO must notify the Department TPL unit within ten (10) calendar days of a Subrogation case in which the Member was not eligible under the MCO for the date of incident. The MCO cannot close these cases with no lien letter until the Department responds to the notification.
- 6.10.11.9.18 DHHS shall have exclusive rights to pursue subrogations in which the MCO does not have an

Page 389 of 397	Date	

New Hampshire Department of Health and Human Services Medicaid Care Management Services



### **Exhibit B - DRAFT**

active subrogation case within ninety (90) calendar days of receiving a referral, of sending the first questionnaire as referenced in 6.10.12.9.4.5 of this section, or of claim paid date if no action was taken since claims paid date, or if the MCO closes the case, as noted on the MCO Subrogation.01 report which indicates the MCO is no longer pursuing the case.

- 6.10.11.9.19 In the event that there are outstanding Subrogation settlements at the time of Agreement termination, the MCO shall assign the Department all rights to such cases to complete and collect on those Subrogation settlements.
- 6.10.11.9.20 The Department shall retain all recoveries after Agreement termination.
- 6.10.11.9.21 The MCOs shall report on all subrogation recoveries in a manner prescribed by the Department.

#### 6.10.11.10 Medicare

- 6.10.11.10.1 The MCO shall be responsible for coordinating benefits for dually eligible Members, if applicable.
- 6.10.11.10.2 The MCO shall enter into a Coordination of Benefits Agreement (COBA) for NH with Medicare and participate in the automated crossover process. [42 CFR 438.3(t)]
- 6.10.11.10.3 A newly contracted MCO shall have ninety (90) calendar days from the start of this Agreement to establish and start file transfers with COBA.
- 6.10.11.10.4 The MCO and its Subcontractors shall establish claims edits to ensure that:
  - 6.10.11.10.4.1. Claims covered by Medicare part D are denied when a Member has an active Medicare part A or Medicare part B;
  - 6.10.11.10.4.2. Claims covered by Medicare part B are denied when a Member has an active Medicare part B; and
  - 6.10.11.10.4.3. The MCO treats Members with Medicare part C as if they had Medicare part A and Medicare part B and shall establish claims edits and deny part D for those part C Members.

Page 390 of 397	Date
1 ago 000 01 001	

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



### **Exhibit B - DRAFT**

- 6.10.11.10.4.4. The MCO shall pursue collection for Medicare Part D from the Medicare Part D plan.
- 6.10.11.10.5 If Medicare was not known or active at the time a claim was submitted by a Provider to the MCO, but was determined active or retroactive subsequent to the MCO's payment of the claim, the MCO shall recoup funds from the Provider and the Provider may pursue Medicare payment, except for Medicare Part D, for all claim types, provided the claims remain within the Medicare timely filing requirements.
  - 6.10.11.10.5.1. The MCO shall pursue collection for Medicare Part D from the Medicare Part D plan.
- 6.10.11.10.6 The MCO shall contact DHHS if Members' claims were denied due to the lack of active Medicare part D or Medicare part B.
- 6.10.11.10.7 The MCO shall pay applicable Medicare coinsurance and deductible amounts as outlined in Section 6.4 (Financial Responsibility for Dual-Eligible Members). These payments are included in the calculated Capitation Payment. The MCO shall not pay any member liability for Medicare Part D claims.
- 6.10.11.11 The MCO shall pay any wrap around services not covered by Medicare that are Covered Services under the Medicaid State Plan Amendment and this Agreement.

# 6.10.12 Estate Recoveries

6.10.12.1 The Department shall be solely responsible for estate recovery activities and shall retain all funds recovered through these activities.

#### 7 TERMINATION OF AGREEMENT

#### 7.1 Termination for Cause

- 7.1.1 The Department shall have the right to terminate this Agreement, in whole or in part, without liability to the State, if the MCO:
  - 7.1.1.1 Takes any action or fails to prevent an action that threatens the health, safety or welfare of any Member, including significant Marketing abuses:
  - 7.1.1.2 Takes any action that threatens the fiscal integrity of the Medicaid program;

age 391 of 397	Date
ago oo i oi ooi	

New Hampshire Department of Health and Human Services Medicaid Care Management Services



### **Exhibit B - DRAFT**

- 7.1.1.3 Has its certification suspended or revoked by any federal agency and/or is federally debarred or excluded from federal procurement and/or non-procurement agreement;
- 7.1.1.4 Materially breaches this Agreement or fails to comply with any term or condition of this Agreement that is not cured within twenty (20) business days of the Department's notice and written request for compliance;
- 7.1.1.5 Violates State or federal law or regulation;
- 7.1.1.6 Fails to carry out a substantive term or terms of this Agreement that is not cured within twenty (20) business days of the Department's notice and written request for compliance;
- 7.1.1.7 Becomes insolvent;
- 7.1.1.8 Fails to meet applicable requirements in Sections 1932, 1903 (m) and 1905(t) of the Social Security Act.; [42 CFR 438.708(a); 42 CFR 438.708(b); sections 1903(m); 1905(t); 1932 of the Social Security Act]
- 7.1.1.9 Receives a "going concern" finding in an annual financial report or indications that creditors are unwilling or unable to continue to provide goods, services or financing or any other indication of insolvency; or
- 7.1.1.10 Brings a proceeding voluntarily, or has a proceeding brought against it involuntarily under Title 11 of the U.S. Code.

# 7.2 Termination for Other Reasons

- 7.2.1 The MCO shall have the right to terminate this Agreement if the Department fails to make agreed-upon payments in a timely manner or fails to comply with any material term or condition of this Agreement, provided that, the Department has not cured such deficiency within sixty (60) business days of its receipt of written notice of such deficiency.
- 7.2.2 This Agreement may be terminated immediately by the Department if federal financial participation in the costs hereof becomes unavailable or if State funds sufficient to fulfill its obligations of the Department hereunder are not appropriated by the Legislature. In either event, the Department shall give MCO prompt written notice of such termination.
- 7.2.3 Notwithstanding the above, the MCO shall not be relieved of liability to the Department or damages sustained by virtue of any breach of this Agreement by the MCO.
- 7.2.4 Upon termination, all documents, data, and reports prepared by the MCO under this Agreement shall become the property of and be delivered to the Department immediately on demand.

Page 392 of 397 Date \_\_\_\_\_

**New Hampshire Department of Health and Human Services Medicaid Care Management Services** 



### **Exhibit B - DRAFT**

- 7.2.5 The Department may terminate this Agreement, in whole or in part, and place Members into a different MCO or provide Medicaid benefits through other Medicaid State Plan Authority, if the Department determines that the MCO has failed to carry out the substantive terms of this Agreement or meet the applicable requirements of Sections 1932, 1903(m) or 1905(t) of the Social Security Act. [42 CFR 438.708(a); 42 CFR 438.708(b); sections 1903(m); 1905(t); 1932 of the Social Security Act].
- 7.2.6 In such event, Section 4.7.9 (Access to Providers During Transitions of Care) shall apply.

#### 7.3 Claims Responsibilities

- 7.3.1 The MCO shall be fully responsible for all inpatient care services and all related services authorized while the Member was an inpatient until the day of discharge from the hospital.
- 7.3.2 The MCO shall be financially responsible for all other authorized services when the service is provided on or before the last day of the Closeout Period (defined in Section 7.5.2 (Service Authorization/Continuity of Care) of this Agreement, or if the service is provided through the date of discharge.

# 7.4 Final Obligations

- 7.4.1 The Department may withhold payments to the MCO, to the reasonable extent it deems necessary, to ensure that all final financial obligations of the MCO have been satisfied. Such withheld payments may be used as a set-off and/or applied to the MCO's outstanding final financial obligations.
- 7.4.2 If all financial obligations of the MCO have been satisfied, amounts due to the MCO for unpaid premiums, risk settlement, high-dollar stop-loss, shall be paid to the MCO within one (1) year of date of termination of the Agreement.

# 7.5 Survival of Terms

- 7.5.1 Termination or expiration of this Agreement for any reason shall not release either the MCO or the Department from any liabilities or obligations set forth in this Agreement that:
  - 7.5.1.1 The parties have expressly agreed shall survive any such termination or expiration; or
  - 7.5.1.2 Arose prior to the effective date of termination and remain to be performed or by their nature would be intended to be applicable following any such termination or expiration, or obliges either party by law or regulation.

## 7.5.2 Service Authorization/Continuity of Care

7.5.2.1 Effective fourteen (14) calendar days prior to the last day of the closeout period, the MCO shall work cooperatively with the

1 age 393 of 391	Page 393 of 397	Date
------------------	-----------------	------

New Hampshire Department of Health and Human Services Medicaid Care Management Services



### **Exhibit B - DRAFT**

Department and/or its designee to process service authorization requests received.

- 7.5.2.1.1 Disputes between the MCO and the Department and/or its designee regarding service authorizations shall be resolved by the Department in its sole discretion.
- 7.5.2.2 The MCO shall give written notice to the Department of all service authorizations that are not decided upon by the MCO within fourteen (14) calendar days prior to the last day of the closeout period.
  - 7.5.2.2.1 Untimely service authorizations constitute a denial and are thus adverse actions [42 CFR 438.404(c)(5)].
- 7.5.2.3 The Member has access to services consistent with the access they previously had, and is permitted to retain their current Provider for the period referenced in Section 4.7.9 (Access to Providers During Transitions of Care) for the transition timeframes if that Provider is not in the new MCO's network of Participating Providers.
- 7.5.2.4 The Member shall be referred to appropriate Participating Providers.
- 7.5.2.5 The MCO that was previously serving the Member, fully and timely complies with requests for historical utilization Confidential Data from the new MCO in compliance with State and federal law.
- 7.5.2.6 Consistent with State and federal law, the Member's new Provider(s) are able to obtain copies of the Member's medical records, as appropriate.
- 7.5.2.7 Any other necessary procedures as specified by the HHS Secretary to ensure continued access to services to prevent serious detriment to the Member's health or reduce the risk of hospitalization or institutionalization.
- 7.5.2.8 The Department shall make any other transition of care requirements publically available.

# 7.6 State Owned Devices, Systems and Network Usage

- 7.6.1 If Contractor End Users, as defined in Exhibit K: DHHS Information Security Requirements are authorized by the Department's Information Security Office to use a State issued device (e.g. computer, tablet, mobile telephone) and/or access the State' network or system in the fulfilment of this Agreement, each individual being granted access must:
  - 7.6.1.1 Sign and abide by applicable Department and NH Department of Information Technology (DOIT) use agreements, policies,

age 394 of 397	Date

New Hampshire Department of Health and Human Services Medicaid Care Management Services



### **Exhibit B - DRAFT**

- standards, procedures and/or guidelines, and complete applicable trainings as required;
- 7.6.1.2 Use the information that they have permission to access solely for conducting official Department or State business. All other use or access is strictly forbidden including, but not limited, to personal or other private and non-State use, and that at no time must they access or attempt to access information without having the express authority of the Department to do so;
- 7.6.1.3 Not access or attempt to access information in a manner inconsistent with the approved policies, standards, procedures, and/or agreement relating to system entry/access;
- 7.6.1.4 Not copy, share, distribute, sub-license, modify, reverse engineer, rent, or sell software licensed, developed, or being evaluated by the Department, and at all times must use utmost care to protect and keep such software strictly confidential in accordance with the license or any other agreement executed by the Department or State:
- 7.6.1.5 Only use equipment, software or subscription(s) authorized by the Department's Information Security Officer or designee;.
- 7.6.1.6 Not install non-standard software on any equipment unless authorized by the Department's Information Security Officer or designee;
- 7.6.1.7 Agree that email and other electronic communication messages created, sent, and received on a State-issued email system are the property of the State of New Hampshire and to be used for business purposes only. Email is defined as "internal email systems" or "state-funded email systems."
- 7.6.1.8 Agree that use of email must follow Department and NH DoIT policies, standards, and procedures and:
- 7.6.1.9 When utilizing the State's email system, the MCO must;
  - 7.6.1.9.1 Only use a State email address assigned to them with a "@ affiliate.DHHS.NH.Gov".
  - 7.6.1.9.2 Include in the signature lines information identifying the End User as a non-state workforce member; and
  - 7.6.1.9.3 Ensure the following confidentiality notice is embedded underneath the signature line:

CONFIDENTIALITY NOTICE: "This message may contain information that is privileged and confidential and is intended only for the use of the individual(s) to whom it is addressed. If you receive this message in error, please notify the sender immediately and delete this electronic

Page 395 of 397 Date \_\_\_\_\_

New Hampshire Department of Health and Human Services Medicaid Care Management Services



#### **Exhibit B - DRAFT**

message and any attachments from your system. Thank you for your cooperation."

- 7.6.2 If applicable in 7.6.1, Contractor End Users with a State issued email, access or potential access to Confidential Information, as defined in Exhibit K: DHHS Information Security Requirements, and/or workspace in a Department building/ facility must:
  - 7.6.2.1 Complete the Department's online Annual Information Security & Compliance Awareness Training prior to accessing, viewing, handling, hearing or transmitting State Data or Confidential Information.
  - 7.6.2.2 Sign the Department's Business Use and Confidentiality Agreement and Asset Use Agreement, and the NH DoIT Statewide Computer Use Policy upon execution of the Agreement and annually throughout the Term.
  - 7.6.2.3 Agree End User's will only access the State's intranet to view the Department's Policies and Procedures and Information Security webpages.
  - 7.6.2.4 If any End User is found to be in violation of any of the above-stated terms and conditions of the Agreement, said End User may face removal from the Agreement, and/or criminal or civil prosecution, if the act constitutes a violation of law.

#### 7.7 Website And Social Media

- 7.7.1 The Contractor must agree, if performance of services on behalf of the Department involve using social media or a website for marketing or to solicit information of individuals, or Confidential Information, the Contractor shall work with the Department's Communications Bureau to ensure that any social media or website designed, created, or managed on behalf of the State meets all of the Department's and NH Department of Information Technology's website and social media requirements and policies as prioritized and approved by the New HEIGHTS Project Manager.
- 7.7.2 The Contractor must agree protected health information (PHI), personally identifiable information (PII), or other Confidential Information solicited either by social media or the website maintained, stored or captured shall not be further disclosed unless expressly provided in the Agreement. The solicitation or disclosure of PHI, PII, or other Confidential Information shall be subject to the Department's Exhibit K: Information Security Requirements, Exhibit I: Health Insurance Portability and Accountability Act Business Associate Agreement, the IT Requirements Workbook, and all applicable State rules and State and federal law. Unless specifically required by this Agreement and unless clear notice is provided to users of the website or social media, the Contractor agrees that site visitation will not be tracked, disclosed or used for website or social media analytics or marketing.

Page 396 of 397 Date \_\_\_\_\_

New Hampshire Department of Health and Human Services Medicaid Care Management Services



### **Exhibit B - DRAFT**

#### 7.8 Privacy Impact Assessment

- 7.8.1 Upon request, the Contractor and its End Users must allow and assist the Department to conduct a Privacy Impact Assessment (PIA) of the Contractor's Applications/Systems/Websites/Web Portals or as applicable, Department applications/systems/websites/web portals hosted by the Contractor if Personally Identifiable Information (PII) is collected, used, accessed, shared, or stored. To conduct the PIA the Contractor must provide the State access to the aforementioned applicable systems and documentation sufficient to allow the State to assess, at minimum, the following:
  - 7.8.1.1 How PII is gathered and stored;
  - 7.8.1.2 Who will have access to PII;
  - 7.8.1.3 How PII will be used in the system;
  - 7.8.1.4 If federal PII is being gathered and stored:
  - 7.8.1.5 How individual consent will be achieved and revoked; and
  - 7.8.1.6 Privacy practices.
- 7.8.2 The Department may conduct follow-up PIA's in the event there are either significant process changes or new technologies impacting the collection, processing or storage of PII.

Page 397 of 397 Date \_\_\_\_\_